THIS DOCUMENT IS IMPORTANT AND REQUIRES YOUR IMMEDIATE ATTENTION. PART TWO OF THIS DOCUMENT COMPRISES AN EXPLANATORY STATEMENT IN COMPLIANCE WITH ARTICLE 126 OF THE JERSEY COMPANIES LAW. THIS DOCUMENT CONTAINS DETAILS OF A PROPOSED TRANSACTION WHICH, IF IMPLEMENTED, WILL RESULT IN THE CANCELLATION OF THE LISTING AND OF THE ADMISSION TO TRADING OF KWE SHARES ON THE MAIN MARKET OF THE LONDON STOCK EXCHANGE.

If you are in any doubt as to the action you should take, you are recommended to seek your own independent advice as soon as possible from your stockbroker, bank, solicitor, accountant, fund manager or other appropriate independent professional adviser who, if you are taking advice in the United Kingdom, must be appropriately authorised to provide such advice under the United Kingdom Financial Services and Markets Act 2000, or another appropriately authorised independent financial adviser if you are in a territory outside the United Kingdom.

Where legally permissible, Scheme Shareholders, and other KWE Shareholders, in all jurisdictions will receive this document and the accompanying Forms of Proxy to notify them of, and permit them to attend and/or vote for or against the Scheme at, the meetings at which they are entitled to vote. However, there may be restrictions on the extent to which certain shareholders of KWE in jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands and the Cayman Islands can receive either this document or an Election Form, make Elections and/or receive New KW Shares pursuant to the Scheme. Any persons who are resident in, or citizens or nationals of, or otherwise subject to the law of, any jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands should inform themselves about, and observe, any applicable legal or regulatory requirements. Any failure to comply with the applicable requirements may constitute a violation of the securities laws of any such jurisdiction.

If you have sold or otherwise transferred all of your KWE Shares, please send this document and the accompanying documents, but not the personalised Forms of Proxy or Election Form, at once to the purchaser or transferee or to the stockbroker, bank or other agent through whom the sale or transfer was effected for delivery to the purchaser or transferee. Neither this document nor any Election Form should be forwarded or transmitted in, into or from any jurisdiction in which to do so would give rise to a violation of the relevant laws in such jurisdiction. If you have sold or otherwise transferred part of your holding of KWE Shares, please retain these documents and consult the stockbroker, bank or other agent through whom the sale or transfer was effected.

The release, publication or distribution of this document and/or the Election Form in or into jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands may be restricted or prohibited by the laws of those jurisdictions and therefore persons into whose possession any Election Form comes should inform themselves about, and observe, any such restrictions or applicable requirements. Any failure to comply with such restrictions or applicable requirements may constitute a violation of the securities laws of any such jurisdiction.

The New KW Shares are expected to be issued in the United States in reliance upon the exemption from the registration requirements of the US Securities Act provided by Section 3(a)(10) thereof.

RECOMMENDED MERGER OF KENNEDY WILSON EUROPE REAL ESTATE PLC AND KENNEDY-WILSON HOLDINGS, INC.

to be effected by means of a Scheme of Arrangement under Article 125 of the Companies (Jersey) Law 1991, as amended

You should read this document and the documents incorporated by reference into this document in their entirety, together with the accompanying Forms of Proxy and Election Form. Your attention is drawn to Part One of this document, which contains the unanimous recommendation of the KWE Independent Directors that Scheme Shareholders vote in favour of the Scheme at the Scheme Court Meeting, elect to receive the consideration and special distribution available under the New Offer and vote in favour of the Special Resolution at the KWE General Meeting. A letter from Rothschild and J.P. Morgan Cazenove explaining the Merger in greater detail and the action to be taken by you appears in Part Two of this document and constitutes an explanatory statement in compliance with Article 126 of the Jersey Companies Law.

Notices of the Scheme Court Meeting and the KWE General Meeting, both of which are to be held at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN on 10 October 2017, are set out in Part Twelve and Part Thirteen of this document. The Scheme Court Meeting will start at 10.00 a.m. on that date and the KWE General Meeting will start at 10.15 a.m. on that date (or as soon thereafter as the Scheme Court Meeting is concluded or adjourned). Please also refer to pages 10 to 11 of this document, which contain an indicative timetable of certain principal events in relation to the Scheme.

Your attention is drawn to pages 14 to 22 of this document, which explain the actions you may take in relation to the Scheme. The BLUE Form of Proxy is to be used in connection with the KWE General Meeting. Whether or not you intend to attend both or either of these KWE Meetings, please complete and sign the Form(s) of Proxy that have been sent to you and return them in accordance with the accompanying instructions, by post or (during normal business hours only) by hand to the KWE Registrar as soon as possible but in any event so as to be received by no later than 10.00 a.m. on 8 October 2017 (in the case of the BLUE Form of Proxy for the Scheme Court Meeting) or 10.15 a.m. on 8 October 2017 (in the case of the GREEN Form of Proxy for the KWE General Meeting) or, in the case of any adjournment, not later than 48 hours before the time fixed for the holding of the adjourned meeting. Alternatively, you can log on to the website www.investorcentre.co.uk/je to submit your proxy or proxies electronically so as to be received by not later than 48 hours before the relevant KWE Meeting (or any adjournment thereof). You will need your unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on your Form of Proxy) in order to submit your

proxies in this manner. If you hold Scheme Shares or KWE Shares through CREST and wish to appoint a proxy or proxies, you can do so by using the CREST electronic proxy appointment service. Proxies submitted via CREST must be received by the KWE Registrar (ID is 3RA50) not later than 10.00 a.m. on 8 October 2017 in the case of the Scheme Court Meeting and 10.15 a.m. on 8 October 2017 in the case of the KWE General Meeting or, in the case of any adjournment, not later than 48 hours before the time fixed for the holding of the adjourned meeting. A proxy, however appointed, need not be a KWE Shareholder and may be the Chairman of the relevant meeting.

Alternatively, the BLUE Form of Proxy relating to the Scheme Court Meeting will be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or any adjournment thereof). However, in the case of the KWE General Meeting, the GREEN Form of Proxy will be valid only if it is returned by 10.15 a.m. on 8 October 2017 or, in the case of an adjourned KWE General Meeting, by not later than 48 hours before the time fixed for the holding of the adjourned KWE General Meeting.

Rothschild, which is authorised and regulated by the FCA in the UK, is acting exclusively for KWE and the KWE Independent Directors and no one else in connection with the Merger and any other matter referred to in this document and will not be responsible to anyone other than KWE and the KWE Independent Directors for providing the protections afforded to clients of Rothschild, or for providing advice in relation to the Merger or any other matters referred to in this document.

J.P. Morgan Limited, which conducts its UK investment banking business as J.P. Morgan Cazenove ("J.P. Morgan Cazenove") and is authorised and regulated in the UK by the FCA, is acting exclusively for KWE and the KWE Independent Directors and no one else in connection with the Merger and any other matter referred to in this document and will not regard any other person as its client in relation to the matters in this document and will not be responsible to anyone other than the KWE Independent Directors for providing the protections afforded to clients of J.P. Morgan Cazenove, or for providing advice in relation to the Merger or any other matters referred to in this document.

Goldman Sachs, which is authorised by the Prudential Regulation Authority and regulated by the FCA and the Prudential Regulation Authority in the UK, is acting exclusively for KW and no one else in connection with the Merger and any other matter referred to in this document and will not be responsible to anyone other than KW for providing the protections afforded to clients of Goldman Sachs, or for providing advice in relation to the Merger or any other matters referred to in this document.

Merrill Lynch International, which is authorised by the Prudential Regulation Authority and regulated by the FCA and the Prudential Regulation Authority in the UK, and Merrill Lynch, Pierce, Fenner & Smith Incorporated are acting exclusively for KW and no one else in connection with the Merger or any other matter referred to in this document and will not be responsible to anyone other than KW for providing the protections afforded to clients of BofA Merrill Lynch, or for providing advice in relation to the Merger or any other matters referred to in this document.

Deutsche Bank AG is authorised under German Banking Law (competent authority: European Central Bank) and, in the UK, by the Prudential Regulation Authority. It is subject to supervision by the European Central Bank and by BaFin, Germany's Federal Financial Supervisory Authority, and is subject to limited regulation in the UK by the Prudential Regulation Authority and the Financial Conduct Authority. Details about the extent of its authorisation and regulation by the Prudential Regulation Authority, and regulation by the Financial Conduct Authority, are available on request or from www.db.com/en/content/ eu_disclosures.htm.

Deutsche Bank AG acting through its London Branch ("DB London") and Deutsche Bank Securities Inc. (("DBSI") and together with DB London, "DB") are acting as financial advisers to KW and no other person in connection with this Merger. DB will not be responsible to any person other than KW for providing any of the protections afforded to clients of DB, nor for providing any advice, in relation to any matter referred to herein. Without limiting a person's liability for fraud, neither DB nor any of their affiliates nor any of their or their affiliates' respective directors, officers, representatives, employees, advisers or agents shall have any liability to any other person in connection with this document and its contents.

Certain words and terms used in this document are defined in Part Eleven of this document. All times referred to are London times unless otherwise stated.

TABLE OF CONTENTS

EXPECTED TIMETABLE OF PRINCIPAL EVENTS	10
IMPORTANT NOTICE ABOUT VOTING ON THE MERGER	12
ACTION TO BE TAKEN	14
PART ONE LETTER FROM THE INDEPENDENT DIRECTORS OF KENNEDY WILSON EUROPE REAL ESTATE PLC	23
PART TWO EXPLANATORY STATEMENT	41
PART THREE CONDITIONS AND CERTAIN FURTHER TERMS OF THE SCHEME	81
PART FOUR THE SCHEME OF ARRANGEMENT	91
PART FIVE NOTES FOR MAKING ELECTIONS AND SUBMITTING IRS FORMS W-8 OR W-9	111
PART SIX FINANCIAL AND OTHER INFORMATION ON KWE AND KW	121
PART SEVEN TAXATION	123
PART EIGHT ADDITIONAL INFORMATION	144
PART NINE BASES AND SOURCES	174
PART TEN DESCRIPTION OF KW SHARES	177
PART ELEVEN DEFINITIONS	202
PART TWELVE NOTICE OF SCHEME COURT MEETING	213
PART THIRTEEN NOTICE OF KWE GENERAL MEETING	215
PART FOURTEEN PROFIT FORECASTS	221
ANNEX I VALUATION REPORTS	237

IMPORTANT NOTICE

This document and the accompanying Forms of Proxy and Election Form have been prepared for the purposes of complying with Jersey law and will be subject to the requirements of the Code, the Panel, the London Stock Exchange, the Listing Rules, the FCA and the Jersey Financial Services Commission. Information disclosed may not be the same as that which would have been disclosed in accordance with the laws of jurisdictions other than Jersey or the United Kingdom or if requirements other than those imposed by the Code, the Panel, the London Stock Exchange, the Listing Rules, the FCA and the Jersey Financial Services Commission had applied. This document, the Conditions and the further terms of the Scheme, as set out in this document, are governed by Jersey law and the Code and are subject to the jurisdiction of the Jersey courts and the Panel and, in respect of issues relating to compliance with the Code, the Courts of England and Wales.

This document relates to a scheme of arrangement among KWE and Scheme Shareholders by which KW will undertake to the Court to be bound. It is not intended to and does not constitute, or form part of, any offer to sell or issue or an invitation to purchase or subscribe for any securities in any jurisdiction, nor is it a prospectus for the purposes of the Companies (General Provisions) (Jersey) Order 2002. Further, this document does not constitute, or form part of, any solicitation of any vote or approval in any jurisdiction in which such solicitation is unlawful, nor shall there be any sale, issuance or transfer of securities of KWE or KW in any jurisdiction in contravention of applicable law.

This document does not constitute an offer of securities or financial product advice or investment advice. It is general information only and has been prepared without reference to the investment objectives, financial situation, tax position or particular needs of any KWE Shareholder or any other person. This document should not be relied upon as the primary basis for any investment decision in relation to KWE Shares or any other securities, and each KWE Shareholder should seek independent financial, legal, tax and all other relevant professional advice before making any such investment decision.

The KW Prospectus, which has been approved by the UKLA, may be accessed free of charge at KW's website at <u>ir.kennedywilson.com</u> (subject to any restrictions relating to persons with a registered address in or who are a citizen, resident or national of certain jurisdictions). The KW Prospectus has been issued solely by KW in relation to the New KW Shares and contains a description of KW, its business, strategy, financial condition and results of operations. KWE Shareholders' attention is drawn to the risk factors described in Part 2 "Risk Factors" of the KW Prospectus. The materialisation of one or more of the risks described in the KW Prospectus may have a material adverse effect on KW's activities, assets, financial position, results or prospects as well as the market price of KW Shares.

The publication and content of the KW Prospectus are the responsibility solely of the KW Directors.

KW and the KW Directors have accepted sole responsibility for all information, opinions, estimates, valuations, projections and commentary contained in certain communications that KW has issued in connection with the Merger (such communications to include, without limitation, the KW Prospectus and the KW Proxy Statement, which is further described in paragraph 12 of Part One (Letter from the Independent Directors of Kennedy Wilson Europe Real Estate PLC) of this document and all the information contained in Part Fourteen (Profit Forecasts) of this document), including, without limitation, all financial and commercial information, opinions, estimates, valuations, projections and commentary relating directly or indirectly to KWE or its assets or liabilities or derived from or reflecting any information, opinion, estimate, valuation, projection, or commentary relating to KWE or its assets or liabilities included or incorporated by reference in any such communications and all information in this document relating to KW or persons acting in concert with it ("Covered Information"). No representation has been made, is made or will be made, at any time by KWE or any of the KWE Independent Directors, and none of KWE or any of the KWE Independent Directors have authorised anyone to represent that KWE or any KWE Independent Director has adopted or verified the accuracy, completeness, reasonableness or achievability of any Covered Information, that all or part of it is not misleading or that all or part of it complies with all or any applicable legal, regulatory or other requirements. Accordingly, none of KWE nor any of the KWE Independent Directors assumes any duty of care or other duty to KW, any holder of KW Shares or other securities of KW or any other person or regulatory body in relation to the content of, or any omission from, any Covered Information and KWE and each of the KWE Independent Directors expressly disclaims all and any responsibility for the accuracy or completeness of any Covered

Information or for the Covered Information not being misleading in the context in which it is used and/or complying with applicable legal, regulatory, accounting or other requirements. None of KWE nor any of the KWE Independent Directors owes any obligation, whether to KW, any holder of KW Shares or other securities of KW or any other person or regulatory body, to correct or update any Covered Information or accepts any liability, whether arising in tort, contract or otherwise, to any person or regulatory body in connection with any reliance or expectation placed on any Covered Information or in connection with any Covered Information, or any matter derived from it, including, without limitation, in relation to any decision to vote or not to vote in relation to any matter or to acquire or not to acquire any securities or to dispose or not to dispose of any securities.

The New KW Shares will be issued in reliance upon the exemption from the registration requirements of the US Securities Act provided by Section 3(a)(10) thereof.

Notice to US Investors

The New KW Shares have not been and will not be registered under the US Securities Act or under the securities laws of any state or other jurisdiction of the United States. Accordingly, the New KW Shares may not be offered, sold, resold, delivered, distributed or otherwise transferred, directly or indirectly, in or into the United States without an exemption from registration under the US Securities Act. This document does not constitute an offer to sell or the solicitation of an offer to buy, nor shall there be any sale of, the New KW Shares in any state of the United States in which such offer, solicitation or sale would be unlawful prior to qualification under the securities laws of any such state. The New KW Shares are expected to be issued in the United States in reliance upon the exemption from the registration requirements of the US Securities Act provided by Section 3(a)(10) thereof. Scheme Shareholders (whether or not US persons) who are or will be affiliates (within the meaning of the US Securities Act) of KW prior to or after the Effective Date will be subject to certain restrictions on transfers of the New KW Shares received pursuant to the Scheme. For a description of these and certain further restrictions on offers, sales and transfers of the New KW Shares and the distribution of this document, and additional information applicable to US shareholders, see paragraph 33 of Part Two (Explanatory Statement) of this document.

For the purposes of qualifying for the exemption from the registration requirements of the US Securities Act afforded by Section 3(a)(10), the Court will be advised that KW will rely on the Section 3(a)(10) exemption based on the Court's sanctioning of the Scheme.

The Merger relates to the shares of a Jersey company and is being effected by means of a scheme of arrangement pursuant to the Jersey Companies Law. A transaction effected by means of a scheme of arrangement under Jersey law is not subject to the tender offer rules under the US Exchange Act and is exempt from the registration requirements under the US Securities Act. Accordingly, the Merger will only be subject to disclosure and other requirements and practices applicable under Jersey law and the Code to schemes of arrangement, which are different from the disclosure and other requirements of the US tender offer and securities laws.

It may be difficult for US holders of KWE Shares to enforce their rights and/or any claims they may have arising under US federal securities laws in connection with the Merger, since KWE is organised under the laws of a country other than the United States, and some of its officers and directors may be residents of countries other than the United States and most of the assets of KWE are located outside the United States. US holders of KWE Shares may not be able to sue a non-US company or its officers or directors in a non-US court for violations of US securities laws. Further, it may be difficult to compel a non-US company and its affiliates to subject themselves to a US court's jurisdiction or judgment.

The Merger may, in certain circumstances and subject to the consent of the Panel, be carried out by way of a Takeover Offer under the Jersey Companies Law rather than using a scheme of arrangement. If KW elects to implement the Merger by way of a Takeover Offer, such Takeover Offer will be subject to applicable US tender offer and securities laws and regulations.

In accordance with normal practice under the Code and Jersey law, KW or its nominees, or its brokers (acting as agents), may from time to time make certain purchases of, or arrangements to purchase, KWE Shares outside the United States, other than pursuant to the Scheme, until the date on which the Merger becomes effective, lapses or is withdrawn. These purchases may occur either in the open market at prevailing prices or in private transactions at negotiated prices.

In addition, the Manager is entitled to an annual management fee (payable quarterly in arrear) equal to 1% of KWE's adjusted net asset value, half of which may be paid in certain circumstances in

KWE Shares (for further information please see paragraph 15 of Part Two (*Explanatory Statement*) of this document). Any information about such purchases and/or fees will be disclosed, as required in the UK, will be reported to a Regulatory Information Service and will be available on the London Stock Exchange website at www.londonstockexchange.com.

None of the securities referred to in this document have been approved or disapproved by the SEC, any state securities commission in the United States or any other US regulatory authority, nor have such authorities passed upon or determined the fairness or merits of such securities or upon the adequacy or accuracy of the information contained in this document. Any representation to the contrary is a criminal offence in the United States.

Notice to Canadian Investors

The New KW Shares are expected to be issued in Canada pursuant to an exemption from the requirement that KW prepare and file a prospectus in the relevant jurisdictions of Canada. Any resale of the New KW Shares if made through the facilities of the NYSE will generally be exempt from the prospectus requirement under Canadian securities legislation. However, such securities legislation may require resales of KW Shares to be made under other statutory exemptions or a discretionary exemption granted by the applicable Canadian securities regulatory authority. Scheme Shareholders resident in Canada are advised to seek legal advice prior to any resale of the New KW Shares issued in connection with the Merger.

Notice to Cayman Islands Investors

There is no registration required nor will any registration be made under the Securities Investment Business Law in the Cayman Islands or with the Cayman Islands Monetary Authority in relation to this document. This document is distributed to Scheme Shareholders on the basis that it does not represent an offer to the public in the Cayman Islands under any law in the Cayman Islands.

Notice to French Investors

This document does not constitute a prospectus within the meaning of Directive 2003/71/EC of the European Parliament and Council dated 4 November 2003, as amended, or otherwise. This document has not been prepared and is not being distributed in the context of an offer to the public of financial securities (offre au public de titres financiers) in France within the meaning of Article L.411-1 of the French Monetary and Financial Code (Code monétaire et financier) and Title 1 of Book II of the General Regulation of the French Autorité des marchés financiers (AMF), and has therefore not been submitted to the AMF for prior approval and clearance procedure.

Notice to Irish Investors

This document is not a prospectus within the meaning of the Companies Act 2014 of Ireland, the Prospectus Directive (2003/71/EC) Regulations 2005 of Ireland (as amended) or the Prospectus Rules issued by the Central Bank of Ireland. This document has not been approved or reviewed by or registered with the Central Bank of Ireland. This document does not constitute investment advice or the provision of investment services within the meaning of the European Communities (Markets in Financial Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Directive (2004/39/EC). Neither KWE nor KW is an authorised investment firm within the meaning of the European Communities (Markets in Financial Instruments) Regulations 2007 of Ireland (as amended) or the Markets in Financial Instruments Directive (2004/39/EC) and the recipients of this document should seek independent legal and financial advice in determining their actions in respect of or pursuant to this document.

Overseas Shareholders

Where legally permissible, Scheme Shareholders, and other KWE Shareholders, in all jurisdictions will receive this document and the accompanying Forms of Proxy to notify them of, and permit them to attend and/or vote for or against the Scheme at, the meetings at which they are entitled to vote. However, there may be restrictions on the extent to which certain shareholders of KWE in jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands and the Cayman Islands can receive either this document or an Election Form, make Elections and/or receive New KW Shares pursuant to the Scheme. The arrangements for such shareholders are summarised below and set out more fully in Clause 9 of the Scheme. Any persons who are resident in, or citizens or nationals of, or otherwise subject to the law of, any jurisdiction other than Jersey, the United

Kingdom, the United States, the Netherlands or the Cayman Islands should inform themselves about, and observe, any applicable legal or regulatory requirements. Any failure to comply with the applicable requirements may constitute a violation of the securities laws of any such jurisdiction. To the fullest extent permitted by applicable law, the companies and persons involved in the Merger disclaim any responsibility or liability for the violation of such requirements by any person.

This document and the accompanying documents have been prepared for the purpose of complying with Jersey law, the Listing Rules and the Code and the information disclosed may not be the same as that which would have been disclosed if this document and the accompanying documents had been prepared in accordance with the laws of any jurisdiction outside of England and Wales and Jersey. Nothing in this document or in the accompanying documents should be relied upon on any other basis. Unless otherwise determined by KW and KWE or required by the Code, and permitted by applicable law and regulation, the Merger will not be made available, directly or indirectly, in, into or from a jurisdiction to the extent that to do so would violate the laws in that jurisdiction.

The release, publication or distribution of this document and/or the Election Form in or into jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands may be restricted or prohibited by the laws of those jurisdictions and therefore persons into whose possession any Election Form comes should inform themselves about, and observe, any such restrictions or applicable requirements. Any failure to comply with such restrictions or applicable requirements may constitute a violation of the securities laws of any such jurisdiction. Neither this document nor any Election Form should be forwarded or transmitted in, into or from any jurisdiction in which to do so would give rise to a violation of the relevant laws in such jurisdiction.

The availability of the Scheme and the Merger to certain Overseas Shareholders (including the right to make an Election under the Mix and Match Facility or for the Original Offer) may be restricted or prohibited by the laws of the relevant jurisdiction. Overseas Shareholders should inform themselves about, and observe, any applicable legal or regulatory requirements. KWE Shareholders who are in any doubt regarding such matters should consult an appropriate independent professional adviser in the relevant jurisdiction without delay.

Elections with respect to the Mix and Match Facility or the Original Offer may not be available to certain Overseas Shareholders. Notwithstanding the submission of an Election, KW or KWE may, in its discretion, determine that the local laws applicable to the jurisdiction of any Scheme Shareholder making an Election who is an Overseas Shareholder prohibit or restrict (i) the extension or availability of the Merger or the Election in, into or from such jurisdiction; (ii) any Election or the granting of a right to make an Election pursuant to the Merger; or (iii) the issue of New KW Shares or KW CDIs pursuant to the Merger, and in each case to act in accordance with the terms of Clause 9 of the Scheme in this respect. Neither KW nor KWE will be liable to any KWE Shareholder for making any such determination or acting in accordance with this paragraph.

Forward-Looking Statements

This document, including information incorporated by reference into this document, oral statements made regarding the Merger, and other information published by KW and/or KWE may contain statements which are, or may be deemed to be, "forward-looking statements" with respect to the financial condition, results of operations and businesses of KWE and/or KW and/or the KWE Group and/or the KW Group, and certain plans and objectives of KWE and/or KW with respect to the Combined Group. All statements, other than statements of historical facts, or of opinion related to historical fact, are forward-looking statements. A forward-looking statement is prospective in nature and is not based on historical facts, but rather on current expectations and/or projections of the persons making the statement about future events, and is therefore subject to risks and uncertainties which could cause actual facts and/or circumstances to differ materially from the future facts and/or circumstances expressed or implied by the forward-looking statement. The forward-looking statements contained in this document include statements relating to the expected effects of the Merger on KW and KWE, the expected timing and scope of the Merger and other statements other than historical facts. Often, but not always, forward-looking statements can be identified by the use of forwardlooking words such as "plans", "expects", "does not expect", "is expected", "is subject to", "budget", "scheduled", "estimates", "forecasts", "intends", "anticipates", "does not anticipate", "believes" or variations of such words and phrases or statements that certain actions, events or circumstances "may", "could", "should", "would", "might" or "will" be taken, occur or be achieved. Although the persons responsible for any forward-looking statements in this document believe that

the expectations reflected in such forward-looking statements are reasonable, neither KW nor KWE, nor any of their respective officers, directors, associates or advisers, gives any assurance that any such expectation will prove to be correct. By their nature, forward-looking statements involve risk and uncertainty because they relate to events and depend on circumstances that will occur in the future. There are a number of factors that could cause actual facts and/or circumstances to differ materially from those expressed or implied by such forward-looking statements. These factors include the nonsatisfaction of any of the Conditions, as well as additional factors, such as: risks relating to the KW Group's and/or the KWE Group's credit rating; local and global political and economic conditions, including Brexit; the KW Group's and the KWE Group's economic model and liquidity risks; financial services risk; the risks associated with KW's and KWE's brand, reputation and trust; environmental risks; safety, technology, data security and data privacy risks; the ability to realise the anticipated benefits and synergies of the Merger, including as a result of a delay in completing the Merger or difficulty in integrating the businesses of the companies involved; legal or regulatory developments and changes; the outcome of any litigation; the impact of any acquisitions or similar transactions; competition and market risks; the impact of foreign exchange rates; pricing pressures; and business continuity and crisis management. Other unknown or unpredictable factors could cause actual results to differ materially from those in the forward-looking statements. Such forward-looking statements should therefore be construed in the light of such factors. Neither KW nor KWE, nor any of their respective associates, directors, officers or advisers, gives any representation, assurance or guarantee that the occurrence of the events expressed or implied in any forward-looking statements in this document will actually occur. You are cautioned not to place undue reliance on these forwardlooking statements. Each forward-looking statement speaks only as of the date of this document. None of KWE, the KWE Group, KW or the KW Group nor any of their officers, directors, associates or advisers accepts any obligation to update or revise any forward-looking statements, whether as a result of new information, future events or otherwise.

Publication on Website and Availability of Hard Copies

This document and the documents required to be published pursuant to Rule 26.1 of the Code will (subject to any applicable restrictions with respect to certain Overseas Shareholders) be made available free of charge on KW's website at ir.kennedywilson.com and on KWE's website at www.kennedywilson.eu by no later than 12 noon (London time) on the Business Day following the date of posting of this document to KWE Shareholders. For the avoidance of doubt, the contents of those websites are not incorporated into, and do not form part of, this document.

KWE Shareholders may request a hard copy of this document, and any future documents, announcements and information, by contacting the Corporate Services team at Crestbridge during business hours on +44 (0) 1534 835600 or by submitting a request in writing to the Corporate Services Team, Crestbridge, 47 Esplanade, St Helier, Jersey JE1 0BD.

Dealing Disclosure Requirements

Under Rule 8.3(a) of the Code, any person who is interested in 1% or more of any class of relevant securities of an offeree company or of any securities exchange offeror (being any offeror other than an offeror in respect of which it has been announced that its offer is, or is likely to be, solely in cash) must make an Opening Position Disclosure following the commencement of the offer period and, if later, following the announcement in which any securities exchange offeror is first identified. An Opening Position Disclosure must contain details of the person's interests and short positions in, and rights to subscribe for, any relevant securities of each of (i) the offeree company and (ii) any securities exchange offeror(s). An Opening Position Disclosure by a person to whom Rule 8.3(a) applies must be made by no later than 3.30 p.m. (London time) on the 10th business day following the commencement of the offer period and, if appropriate, by no later than 3.30 p.m. (London time) on the 10th business day following the announcement in which any securities exchange offeror is first identified. Relevant persons who deal in the relevant securities of the offeree company or of a securities exchange offeror prior to the deadline for making an Opening Position Disclosure must instead make a Dealing Disclosure.

Under Rule 8.3(b) of the Code, any person who is, or becomes, interested in 1% or more of any class of relevant securities of the offeree company or of any securities exchange offeror must make a Dealing Disclosure if the person deals in any relevant securities of the offeree company or of any securities exchange offeror. A Dealing Disclosure must contain details of the dealing concerned and of the person's interests and short positions in, and rights to subscribe for, any relevant securities of

each of (i) the offeree company and (ii) any securities exchange offeror(s), save to the extent that these details have previously been disclosed under Rule 8. A Dealing Disclosure by a person to whom Rule 8.3(b) applies must be made by no later than 3.30 p.m. (London time) on the business day following the date of the relevant dealing.

If two or more persons act together pursuant to an agreement or understanding, whether formal or informal, to acquire or control an interest in relevant securities of an offeree company or a securities exchange offeror, they will be deemed to be a single person for the purpose of Rule 8.3.

Opening Position Disclosures must also be made by the offeree company and by any offeror and Dealing Disclosures must also be made by the offeree company, by any offeror and by any persons acting in concert with any of them (see Rules 8.1, 8.2 and 8.4).

Details of the offeree and offeror companies in respect of whose relevant securities Opening Position Disclosures and Dealing Disclosures must be made can be found in the Disclosure Table on the Panel's website at www.thetakeoverpanel.org.uk, including details of the number of relevant securities in issue, when the offer period commenced and when any offeror was first identified. You should contact the Panel's Market Surveillance Unit on +44 (0)20 7638 0129 if you are in any doubt as to whether you are required to make an Opening Position Disclosure or a Dealing Disclosure.

Rounding

Certain figures included in this document have been subjected to rounding adjustments. Accordingly, figures shown for the same category presented in different tables may vary slightly and figures shown as totals in certain tables may not be an arithmetic aggregation of the figures that precede them.

Date of publication

This document is published on 13 September 2017.

EXPECTED TIMETABLE OF PRINCIPAL EVENTS

The following timetable sets out the expected dates for implementation of the Merger (some of which are indicative):

Time and/or date Voting Record Time⁽¹⁾ 6.00 p.m. on 6 October 2017 Latest time for lodging the BLUE Form of Proxy (or appointing a 10.00 a.m. on 8 October 2017 proxy electronically or submitting a proxy via CREST) for the Scheme Court Meeting⁽²⁾⁽⁴⁾ Latest time for lodging the GREEN Form of Proxy (or appointing a 10.15 a.m. on 8 October 2017 proxy electronically or submitting a proxy via CREST) for the KWE General Meeting⁽³⁾⁽⁴⁾ Scheme Court Meeting⁽¹⁾ 10.00 a.m. on 10 October 2017 **KWE General Meeting**⁽¹⁾ 10.15 a.m. on 10 October 2017 or as soon thereafter as the Scheme Court Meeting is concluded or adjourned 9.00 a.m. (Pacific Daylight Time) on **KW Special Meeting**

The following dates are provided by way of indicative guidance only, are subject to change and will depend on, among other things, the date on which the Conditions are satisfied or, to the extent capable of waiver, waived, the date on which the Court sanctions the Scheme and the date on which the Court Order sanctioning the Scheme is delivered to the Registrar of Companies. KWE will give notice of all of these dates, when known, by issuing an announcement through a Regulatory Information Service. Further updates or changes to other times and/or dates indicated below will be notified in the same way.

12 October 2017

Scheme Court Sanction Hearing ⁽⁵⁾	10.00 a.m. on 18 October 2017
Last day of dealings in, and for registration of transfers of, and disablement in CREST of, KWE Shares	19 October 2017
Election Return Time (being the latest time for lodging an Election Form or making an Electronic Election either under the Mix and Match Facility or for the Original Offer and for returning IRS Forms W-9 or applicable IRS Forms W-8) ⁽⁶⁾	1.00 p.m. on 19 October 2017
Scheme Record Time (including the record time with respect to the KWE Special Distribution and the Closing Dividend Record Time)	6.00 p.m. on 19 October 2017
Suspension of listing of, and dealings, settlement and transfers in, KWE Shares	7:30 a.m. on 20 October 2017
Effective Date	20 October 2017
Cancellation of listing and admission to trading on the London Stock Exchange of KWE Shares	On or after 23 October 2017
Latest date for despatch of the KW Cash Component and the KWE Special Distribution	Within 14 days after the Effective Date
New KW Shares issued and listed on NYSE ⁽⁷⁾	Within 14 days after the Effective Date
Crediting of KW CDIs in CREST accounts	Within 14 days after the Effective Date
Longstop Date ⁽⁸⁾	30 November 2017
Payment of Closing Dividend	By 10 January 2018

Unless otherwise stated all references to time in this document are to London time.

Notes:

⁽¹⁾ The Scheme Court Meeting and the KWE General Meeting will both be held at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN on 10 October 2017. The Scheme Court Meeting will start at 10.00 a.m. and the KWE General Meeting will start at 10.15 a.m. (or as soon thereafter as the Scheme Court Meeting has been concluded or adjourned). If

- the Scheme Court Meeting and/or the KWE General Meeting are or is adjourned, the Voting Record Time for the adjourned meeting will be postponed to 6.00 p.m. on the last date to fall two Jersey Working Days before the date of the adjourned meeting.
- (2) A BLUE Form of Proxy for the Scheme Court Meeting not lodged by this time (or, in the case of any adjournment of the Scheme Court Meeting, not later than 48 hours before the time fixed for the holding of the adjourned Scheme Court Meeting) will be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or, if relevant, any adjournment thereof).
- (3) A GREEN Form of Proxy for the KWE General Meeting will be invalid if not lodged by this time (or, in the case of any adjournment of the KWE General Meeting, by not later than 48 hours before the time fixed for the holding of the adjourned KWE General Meeting).
- (4) The completion and return of a Form of Proxy, the electronic appointment of a proxy or the submission of a proxy via CREST will not prevent an eligible Scheme Shareholder from attending and voting in person at the Scheme Court Meeting, or any adjournment thereof, or an eligible KWE Shareholder from attending and voting in person at the KWE General Meeting, or any adjournment thereof. Attending and voting in person will result in the invalidation of any previously lodged proxy. At both of the KWE Meetings, voting on the resolutions set out in the notices at Part Twelve (*Notice of Scheme Court Meeting*) and Part Thirteen (*Notice of KWE General Meeting*) of this document will be conducted on a poll rather than on a show of hands.
- (5) All Scheme Shareholders will have the right to attend and be heard at the Scheme Court Sanction Hearing in person or through counsel to support or oppose the sanctioning of the Scheme by the Court.
- (6) If the Scheme Court Sanction Hearing is adjourned, the Election Return Time will be extended to a date and time to be announced.
- (7) The precise date will be the subject of an announcement by KWE through a Regulatory Information Service nearer the relevant time.
- (8) This is the latest date by which the Scheme must become effective or lapse unless (a) KWE and KW otherwise agree to a later date and (b) the Panel and, if required, the Court permit such later date.

IMPORTANT NOTICE ABOUT VOTING ON THE MERGER

This page should be read in conjunction with the section entitled "Action to be Taken", commencing on page 14 of this document, the rest of this document, the accompanying Forms of Proxy, any document incorporated by reference and the KW Prospectus (the publication and content of which are the responsibility solely of the KW Directors).

Please check you have received the following with this document:

KWE Shareholders holding uncertificated shares (i.e. through CREST) and/or certificated shares

- a **BLUE Form of Proxy** for use in respect of the Scheme Court Meeting by Scheme Shareholders entitled to attend and vote at such meeting;
- a GREEN Form of Proxy for use in respect of the KWE General Meeting by KWE Shareholders entitled to attend and vote at such meeting;

KWE Shareholders holding KWE Shares in certificated form

- a WHITE Election Form for use by Scheme Shareholders to make an Election under the Mix and Match Facility or for the Original Offer; and
- a reply-paid envelope for use in the UK only for the return of the WHITE Election Form.

If you have not received all of these documents, please contact the Shareholder Helpline on the numbers indicated below.

Whether or not you plan to attend either or both of the KWE Meetings, please complete and sign both the BLUE Form of Proxy (for the Scheme Court Meeting) and the GREEN Form of Proxy (for the KWE General Meeting) and return them in accordance with the instructions set out on these forms as soon as possible but in any event so as to be received by the KWE Registrar:

- not later than 10.00 a.m. on 8 October 2017 in the case of the BLUE Form of Proxy for the Scheme Court Meeting; and
- not later than 10.15 a.m. on 8 October 2017 in the case of the GREEN Form of Proxy for the KWE General Meeting

(or, in the case of an adjournment of the relevant meeting, not later than 48 hours before the time fixed for the holding of the adjourned meeting).

If the BLUE Form of Proxy for use at the Scheme Court Meeting is not lodged by 10.00 a.m. on 8 October 2017, it will still be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in either case before the start of the Scheme Court Meeting (or, if relevant, any adjournment thereof).

In the case of the KWE General Meeting, the GREEN Form of Proxy will be valid only if it is returned by the time specified above or, in the case of an adjourned KWE General Meeting, by not later than 48 hours before the time fixed for the holding of the adjourned KWE General Meeting.

The person appointed as a proxy in relation to the Scheme Court Meeting will vote for or against the Scheme as per your instructions in the BLUE Form of Proxy. For any other business arising at the Scheme Court Meeting (including any proper procedural resolution not listed in the notice of the Scheme Court Meeting), the person appointed as a proxy will vote at his or her discretion.

The person appointed as a proxy in relation to the KWE General Meeting will vote for or against, or abstain from voting in relation to, the Special Resolution as per your instructions in the GREEN Form of Proxy. In the absence of instructions and for any business arising at the KWE General Meeting (including any proper procedural resolution not listed in the notice of the KWE General Meeting), the person appointed may vote, or abstain from voting, as he or she thinks fit on any business which may properly come before the KWE General Meeting.

The Chairman of the KWE Meetings shall be at liberty to accept a Form of Proxy (and to accept as having voted the number of Scheme Shares or, where relevant, KWE Shares, in respect of which the relevant holder seeks to vote) notwithstanding that the relevant Form of Proxy has not been completed in accordance with the instructions contained in the Form of Proxy, provided that the Chairman considers that the information contained in the Form of Proxy is sufficient to establish the entitlement of the holder of the relevant Scheme Shares or, where relevant, the KWE Shares, to vote and is satisfied as to the authority of the person signing it to do so.

Proxies can also be appointed electronically by Scheme Shareholders and other KWE Shareholders, irrespective of whether they hold their shares in CREST or in certificated form, by logging on to www.investorcentre.co.uk/je to submit their proxies not later than 48 hours before the relevant KWE Meeting (or any adjournment thereof).

Scheme Shareholders and other KWE Shareholders who hold their shares through CREST may submit their proxies by using the CREST electronic proxy appointment service. Proxies submitted via CREST must be received by the KWE Registrar (ID is 3RA50) not later than 10.00 a.m. on 8 October 2017 in relation to the Scheme Court Meeting and 10.15 a.m. on 8 October 2017 in relation to the KWE General Meeting or, in relation to any adjournment, not later than 48 hours before the time fixed for the holding of the adjourned meeting.

The completion and return of a Form of Proxy, the electronic appointment of a proxy or the submission of a proxy via CREST will not prevent an eligible Scheme Shareholder from attending and voting in person at the KWE Meetings, or any adjournment thereof, or an eligible KWE Shareholder from attending and voting in person at the KWE General Meeting, or any adjournment thereof. Attending and voting in person will result in the invalidation of any previously lodged proxy. At both of the KWE Meetings, voting on the resolutions set out in the notices at Part Twelve (*Notice of Scheme Court Meeting*) and Part Thirteen (*Notice of KWE General Meeting*) of this document will be conducted on a poll rather than on a show of hands.

FOR THE SCHEME COURT MEETING, IT IS IMPORTANT THAT AS MANY VOTES AS POSSIBLE ARE CAST SO THAT THE COURT MAY BE SATISFIED THAT THERE IS FAIR AND REASONABLE REPRESENTATION OF SCHEME SHAREHOLDERS AT THE SCHEME COURT MEETING. YOU ARE THEREFORE STRONGLY URGED TO COMPLETE, SIGN AND RETURN YOUR BLUE FORM OF PROXY, APPOINT A PROXY ELECTRONICALLY OR SUBMIT YOUR PROXY VIA CREST AS SOON AS POSSIBLE.

THE SCHEME COURT SANCTION HEARING IS SCHEDULED TO BE HELD AT 10.00 A.M. ON 18 OCTOBER 2017 AT THE ROYAL COURT OF JERSEY, ROYAL COURT BUILDING, ST. HELIER, JE1 1BA, JERSEY, CHANNEL ISLANDS. SCHEME SHAREHOLDERS MAY, IF THEY WISH, ATTEND AND BE HEARD AT THE SCHEME COURT SANCTION HEARING, IN PERSON OR THROUGH COUNSEL, TO SUPPORT OR OPPOSE THE COURT'S SANCTIONING OF THE SCHEME.

Shareholder Helpline

If you require assistance, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

ACTION TO BE TAKEN

The KWE Independent Directors, who have been so advised by Rothschild and J.P. Morgan Cazenove as to the financial terms of the Original Offer and the New Offer, consider the terms of the New Offer to be fair and reasonable. In providing advice to the KWE Independent Directors, Rothschild and J.P. Morgan Cazenove have taken into account the commercial assessments of the KWE Independent Directors.

Accordingly, the KWE Independent Directors unanimously recommend Scheme Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and to choose to receive the New Offer Consideration and Distribution in the manner explained in this document and also recommend all KWE Shareholders to vote in favour of the Special Resolution at the KWE General Meeting but do not recommend them to elect to receive the Original Offer Consideration and also make no recommendation in relation to the Mix and Match Facility.

As further explained in this section ("Action to be Taken"), a Scheme Shareholder may cast a vote for or against the Scheme in accordance with the instructions set out herein. A Scheme Shareholder may vote for or against the Scheme in respect of all Scheme Shares held, or may vote for the Scheme in respect of some Scheme Shares held and against the Scheme in respect of others (this may be of particular relevance to Scheme Shareholders who are nominees or trustees for more than one person).

If the Scheme becomes effective, it will be binding on all holders of Scheme Shares, irrespective of whether or not they attended or voted in favour of the Scheme at the Scheme Court Meeting or in favour of the Special Resolution at the KWE General Meeting. It will also be binding in law.

Whether or not you attend or vote in favour of the Scheme at the Scheme Court Meeting or in favour of the Special Resolution at the KWE General Meeting, the choices available to you with respect to your Scheme Shares are as follows:

- You may choose to receive the New Offer Consideration and Distribution, in which case you should make no Election for the Original Offer.
- If you are entitled to receive the New Offer Consideration and Distribution, you may elect to vary the proportions in which you receive the New KW Share Component and the KW Cash Component by making a valid Mix and Match Election.

The Mix and Match Facility is available only in respect of the consideration available from KW under the New Offer. It is not available in relation to the KWE Special Distribution or the Closing Dividend. Furthermore, it is available only to the extent permitted by valid countervailing elections from other Scheme Shareholders who are entitled to receive the consideration available from KW under the New Offer.

A Scheme Shareholder who is entitled to receive the consideration available under the New Offer may make a Mix and Match Election to receive a greater proportion of the KW Cash Component than would otherwise have been payable under the terms of the New Offer (the "Additional Cash Election") or to receive a greater proportion of the New KW Share Component than would otherwise have been required to be issued under the terms of the New Offer (the "Additional Share Election").

To the extent that you are entitled to receive the consideration available under the New Offer in respect of all or some of your Scheme Shares, you can make an Additional Cash Election or an Additional Share Election in respect of all or only some of those Scheme Shares.

You should note that making a Mix and Match Election does not guarantee that you will receive either the number of New KW Shares under the Additional Share Election you elected to receive or the amount of cash under the Additional Cash Election you elected to receive. To the extent that an Additional Share Election or an Additional Cash Election cannot be satisfied in full due to an insufficiency of valid countervailing elections, it will be scaled down on a *pro rata* basis in accordance with Clause 6.7 and Clause 6.8 of the Scheme.

• You may elect to receive the consideration available under the Original Offer, which consists of 0.667 KW Shares for each Scheme Share in respect of which such an Election is validly made (the "Original Offer Consideration"), rather than the New Offer Consideration and Distribution. You may elect to receive the Original Offer Consideration in respect of all of your Scheme Shares or in respect of certain of your Scheme Shares only. The Mix and Match Facility is not available in relation to Scheme Shares in respect of which an Election for the Original Offer has been made.

• In all cases, you will be entitled to receive the Closing Dividend from KWE. As explained in paragraph 5 of Part Two (*Explanatory Statement*) of this document, the amount of the Closing Dividend will depend on, among other matters, the number of New KW Shares you acquire in the Merger and thus will be affected by any Election you may make as described above.

If you wish to receive the New Offer Consideration and Distribution, consisting of (i) 0.3854 New KW Shares; (ii) 300 pence in cash payable by KW; and (iii) 250 pence in cash payable by KWE in the form of a special distribution in respect of each Scheme Share held, without variation pursuant to the Mix and Match Facility, DO NOT MAKE ANY ELECTION, i.e. do not return the Election Form or send an Electronic Election via CREST. If the Scheme becomes effective and you do not submit any Election, you will be entitled to receive the New Offer Consideration and Distribution without variation pursuant to the Mix and Match Facility. You will, in addition, be entitled to receive the Closing Dividend from KWE as explained in paragraph 5 of Part Two (Explanatory Statement) of this document.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT BELOW IN "US TAX DEDUCTIONS".

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A <u>SEPARATE</u> APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT BELOW IN "US TAX DEDUCTIONS".

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 OR TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 OR IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 OR IRS FORM W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 OR IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT BELOW IN "US TAX DEDUCTIONS" BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM

OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

Voting at the Scheme Court Meeting and the KWE General Meeting

For the Scheme to become effective it will require the approval at the Scheme Court Meeting (and at any separate class meeting which may be required), or any adjournment thereof, by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the Register at the Voting Record Time, and present and voting (either in person or by proxy) at the Scheme Court Meeting, representing three-quarters or more of the votes attached to Scheme Shares which are cast at the Scheme Court Meeting (or at the meeting of each class of Scheme Shareholders, as the case may be). The Scheme Court Meeting has been convened by order of the Court to be held at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN at 10.00 a.m. on 10 October 2017.

Implementation of the Scheme will also require the approval of the Special Resolution as set out in the notice of the KWE General Meeting by a majority of not less than two-thirds of votes cast by KWE Shareholders present and voting (and entitled to vote) in accordance with the KWE Articles, either in person or by proxy (including holders of Excluded Shares), at the KWE General Meeting to be held at the same venue as the Scheme Court Meeting at 10.15 a.m. on 10 October 2017 (or, if later, as soon as the Scheme Court Meeting has concluded or been adjourned).

At both of the KWE Meetings, voting on the resolutions set out in the notices at Part Twelve (*Notice of Scheme Court Meeting*) and Part Thirteen (*Notice of KWE General Meeting*) of this document will be conducted on a poll rather than on a show of hands.

As at the Latest Practicable Date, KW and its subsidiaries held approximately 23.8% of the issued ordinary shares of KWE. Any KWE Shares held by or on behalf of any member of the KW Group (including, for the avoidance of doubt, any KWE Shares held by or on behalf of the Manager) will be Excluded Shares (i.e. not subject to the Scheme) and no member of the KW Group (including the Manager), or its nominee, is entitled to vote at the Scheme Court Meeting in respect of the Excluded Shares held by it.

The 23.8% of KWE Shares held by KW and/or its subsidiary undertakings as at the Latest Practicable Date included 532,230 KWE Shares held for the purpose of satisfying RSUs which will vest in connection with the Merger. It is intended that these RSUs will vest in consequence of the Merger after the KWE General Meeting but before the Scheme Record Time and that the 532,230 KWE Shares underlying these RSUs (less a certain number of KWE Shares withheld by KW to satisfy tax withholding obligations in accordance with the terms of the underlying award agreements) will be transferred to RSU holders (in accordance with the terms of the underlying award agreements) before the Scheme Record Time, i.e. in time for the relevant KWE Shares to become Scheme Shares and to be acquired by KW pursuant to the Scheme. However, irrespective of whether RSU holders or KW or any of its subsidiaries are holders of record of any KWE Shares to which these arrangements apply at any time when votes may be cast or proxies appointed for the Scheme Court Meeting, KW has undertaken to KWE to take all reasonable steps to procure that no proxies are appointed and no votes are cast at the Scheme Court Meeting in respect of any such underlying KWE Shares. However, votes attached to these underlying KWE Shares may be freely cast at the KWE General Meeting and Elections in respect of the underlying KWE Shares may be made prior to the Election Return Time.

As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.

Under the Confidentiality Agreement, KW has also agreed that, other than KWE Shares acquired pursuant to the IMA, neither it nor any person acting in concert with it may acquire any interest in any KWE Shares without the prior written consent of the KWE Independent Directors. Please see paragraph 30(a) of Part Two (Explanatory Statement) for more details.

Further information on the procedures to vote at the Scheme Court Meeting and the KWE General Meeting are set out in "Important Notice About Voting on the Merger" and paragraph 24 of Part Two (Explanatory Statement).

The BLUE Form of Proxy is to be used in connection with the Scheme Court Meeting and the GREEN Form of Proxy is to be used in connection with the KWE General Meeting. Whether or not you intend to attend both or either of these KWE Meetings, please complete and sign the Form(s) of Proxy that have been sent to you and return them in accordance with the accompanying instructions, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, as soon as possible but in any event so as to be received by no later than 10.00 a.m. on 8 October 2017 (in the case of the BLUE Form of Proxy for the Scheme Court Meeting) or 10.15 a.m. on 8 October 2017 (in the case of the GREEN Form of Proxy for the time fixed for the holding of the adjourned meeting. Alternatively, the BLUE Form of Proxy relating to the Scheme Court Meeting will be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or any adjournment thereof).

Electronic appointment of proxies

All Scheme Shareholders and KWE Shareholders, whether holding uncertificated or certificated shares, can appoint proxies electronically, by logging on to the website www.investorcentre.co.uk/je to submit their proxies not later than 48 hours before the relevant KWE Meeting (or any adjournment thereof). You will need your unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on your Form of Proxy) to submit your proxies in this manner.

Shareholders holding shares through CREST

Scheme Shareholders who hold Scheme Shares through CREST and who wish to appoint a proxy or proxies for the Scheme Court Meeting, or any adjournment(s) thereof, and KWE Shareholders who hold KWE Shares through CREST and who wish to appoint a proxy or proxies for the KWE General Meeting, or any adjournment(s) thereof, may do so by using the CREST electronic proxy appointment service and following the procedures described in the CREST Manual. CREST personal members or other CREST sponsored members, and those CREST members who have appointed (a) voting service provider(s), should refer to their CREST sponsor or voting service provider(s), and request them to take the appropriate action on their behalf.

In order for a proxy appointment or instruction made by means of CREST to be valid, the appropriate CREST message (a "CREST Proxy Instruction") must be properly authenticated in accordance with Euroclear's specifications and must contain the information required for such instructions, as described in the CREST Manual. In order to be valid, this message, regardless of whether it constitutes the appointment of a proxy or an amendment to the instructions given to a previously appointed proxy, must be transmitted so as to be received by the KWE Registrar (ID is 3RA50) not later than 10.00 a.m. on 8 October 2017 in the case of the Scheme Court Meeting and not later than 10.15 a.m. on 8 October 2017 in the case of the KWE General Meeting (or, in the case of any adjournment, not less than 48 hours prior to the time fixed for the holding of the adjourned meeting). For this purpose, the time of receipt will be taken to be the time (as determined by the time stamp applied to the message by the CREST Applications Host) from which the KWE Registrar is able to retrieve the message by enquiry to CREST in the manner prescribed by CREST.

CREST members and, where applicable, their CREST sponsors or voting service provider(s), should note that Euroclear does not make available special procedures in CREST for any particular messages. Normal system timings and limitations will therefore apply in relation to the input of CREST Proxy Instructions. It is the responsibility of the CREST member concerned to take (or, if the CREST member is a CREST personal member or sponsored member or has appointed (a) voting service provider(s), to procure that his CREST sponsor or voting service provider(s) take(s)) such action as shall be necessary to ensure that a message is transmitted by means of the CREST system by any particular time. CREST members and, where applicable, their CREST sponsor or voting

service provider(s) are referred, in particular, to those sections of the CREST Manual concerning practical limitations of the CREST system and timings.

KWE may treat as invalid a CREST Proxy Instruction in the circumstances set out in Article 34 of the Uncertificated Securities Order.

A proxy, however appointed, need not be a KWE Shareholder and may be the Chairman of the relevant meeting.

Notes for making an Election

In order to make an Election with respect to the Mix and Match Facility or the Original Offer, a Scheme Shareholder must complete the Election Form or make an Electronic Election. For more information regarding Original Offer Elections, see paragraph 4 of Part Two (*Explanatory Statement*) and Part Five (*Notes for Making Elections and Submitting IRS Forms W-8 or W-9*) of this document. For more information regarding the Mix and Match Facility, see paragraph 3 of Part Two (*Explanatory Statement*) and Part Five (*Notes for Making Elections and Submitting IRS Forms W-8 or W-9*) of this document.

If you hold Scheme Shares in certificated form (i.e. not in CREST) and you wish to make an Election for the Original Offer or under the Mix and Match Facility, please complete and return the relevant enclosed **WHITE Election Form** by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom. A pre-paid envelope is provided for use only in the UK. Instructions on how to complete the Election Form are printed thereon.

If you hold Scheme Shares in uncertificated form (i.e. in CREST) and you wish to make an Election in respect of the Mix and Match Facility or the Original Offer, you must submit your election electronically by taking (or procuring to be taken) the actions set out in Part Five (*Notes for Making Elections and Submitting IRS Forms W-8 or W-9*) of this document to transfer the Scheme Shares in respect of which you wish to make that Election to an escrow balance, using an Electronic Election specifying the CREST Receiving Agent (in its capacity as a CREST participant under the ID 8RA25) as the escrow agent. If you wish to make your Election by completing an Election Form (rather than making an Electronic Election), you must first materialise your Scheme Shares (i.e. withdraw them from CREST) by completing a CREST stock withdrawal form, and you may request an Election Form by contacting the Shareholder Helpline on the number set out on page 22.

If you hold Scheme Shares in both certificated and uncertificated form and/or if you hold Scheme Shares in two or more designated accounts and you wish to make an Election in respect of all such holdings, you must make a separate Election in respect of each holding.

The Election Return Time (being the latest time for lodging an Election Form or making an Electronic Election either under the Mix and Match Facility or for the Original Offer) is 1.00 p.m. on 19 October 2017, as set out in the expected timetable of principal events on pages 10 to 11 of this document (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service).

Scheme Shareholders who wish to receive the New Offer Consideration and Distribution (without amendment pursuant to the Mix and Match Facility) should not make an Election and should therefore not return the Election Form or make an Electronic Election.

By submitting an Election, you undertake not to sell or encumber any KWE Shares to which such Election relates unless and until that Election has been validly withdrawn.

US Tax Deductions

The payment of the KW Cash Component, the KWE Special Distribution and/or any other amount (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer to Scheme Shareholders entitled to the New Offer may be subject to US federal backup withholding (currently, at a rate of 28%). An explanation of this backup withholding is given in paragraphs 2.1(6) and 2.2(5) of Part Seven (*Taxation*) of this document.

In accordance with Clause 8.10 of the Scheme, KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free

and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-8 or IRS Forms W-9 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-9 or, as applicable, IRS Forms W-8 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed IRS Forms W-9 or applicable IRS Forms W-8, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of IRS Form W-9 and applicable IRS Form W-8 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A <u>SEPARATE</u> APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 OR TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL

DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 OR IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 OR IRS FORM W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 OR IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

<u>US federal backup withholding is not an additional tax.</u> Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability) or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner.

FAILURE TO COMPLETE AND RETURN IRS FORMS W-9 OR, IF APPLICABLE, IRS FORMS W-8 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

The various versions of IRS Forms W-8 and W-9 for use by different categories of taxpayer are generally as follows:

- W-9 for use by beneficial owners of Scheme Shares which are "United States persons";
- W-8BEN for use by individual beneficial owners of Scheme Shares that are not "United States persons";
- W-8BEN-E for use by corporate and certain other non-individual beneficial owners of Scheme Shares that are not "United States persons"; and

• W-8IMY for use by non-US intermediaries and non-US partnerships (or entities treated as partnerships for US federal income tax purposes).

Beneficial owners of Scheme Shares and Scheme Shareholders should also review the discussion of US federal backup withholding in Part Seven (*Taxation*) of this document and consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e. Form W-9 or applicable Form W-8) and any supporting documentation to be provided to avoid potential US federal backup withholding.

In addition to the backup withholding regime, there are special rules regarding US taxation in relation to the KW Cash Component of the New Offer. For a more complete explanation of these, beneficial owners of KWE Shares and KWE Shareholders should carefully review Part Seven (*Taxation*) of this document and seek advice from their tax advisers.

In overview, these rules could, in certain circumstances, cause non-US Holders of Scheme Shares to be subject to US federal income tax in respect of the KW Cash Component despite the fact that KW pays the KW Cash Component without deduction or withholding of US federal income tax. In general, these rules could only apply if beneficial owners of Scheme Shares that own at least 50% of KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme come to own at least 50% of KW immediately after such time, in each case taking into account constructive ownership rules.

In that case, all or a portion of the cash received by a non-US Holder in respect of the KW Cash Component could be subject to US tax on deemed dividends by KW or treated as a gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme.

- Any such deemed dividend would be treated as paid by KW to the extent of its 2017 "earnings and profits" for US federal income tax purposes. This tax would not exceed 30% of the lesser of (a) the KW Cash Component received by the relevant non-US Holder and (b) the non-US Holder's attributable proportion of such earnings and profits (and may, in appropriate cases, be eligible for a reduced rate of US income taxation under an applicable tax treaty). Any earnings and profits of KW would be allocated ratably among actual and deemed dividends paid by KW in 2017. KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profits (for US federal income tax purposes) for the financial year ending 31 December 2017.
- In addition, even if, contrary to KW's stated expectation, KW were to have a material amount of earnings and profits for US federal income tax purposes for the financial year ending 31 December 2017, any non-US Holder whose percentage interest (taking into account the constructive ownership rules described below) in all of the outstanding KW Shares immediately after the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme (including, but not limited to, New KW Shares acquired pursuant to the Scheme, as well as any KW Shares otherwise acquired in connection therewith) is less than 80% of such non-US Holder's relative interest in KWE Shares immediately prior to such time (the "Less Than 80% Test") will not be treated as receiving a deemed dividend (and will thus not be subject to the 30% deemed dividend tax referred to above on any portion of the KW Cash Component received pursuant to the Scheme)

For example, a non-US Holder that held a 5% interest in KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme could not be subject to the 30% deemed dividend tax unless such non-US Holder holds a 4% or greater interest in KW Shares immediately after such time. Accordingly, the 30% deemed dividend tax would not apply to any non-US Holder who does not actually or constructively hold any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme, but who only acquires KW Shares pursuant to the Scheme. Under constructive ownership rules, a non-US Holder generally will be treated as owning stock owned by certain related persons (such as an individual's spouse, children, grandchildren and parents), as well as, in certain circumstances, such holder's allocable share of stock held by any corporation (such as KWE shares owned by KW immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the

Scheme) or any partnership, estate or trust, in each case, in which the non-US Holder or its relatives hold an interest. In addition, a non-US Holder which actually or constructively holds an option to purchase stock, is treated as owning the underlying stock.

A non-US Holder will not be subject to this form of 30% deemed dividend tax as a result of choosing to receive the consideration available under the New Offer or as a result of making any Additional Share Elections under the Mix and Match Facility if that non-US Holder does not own, actually or constructively, any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme. Accordingly, non-US Holders of Scheme Shares may wish to consider whether to avoid (or limit) any further acquisitions of KW Shares prior to such time. Those non-US Holders who already own KW Shares may also wish to avoid (or limit) any Additional Share Elections under the Mix and Match Facility, which could jeopardise their ability to satisfy the Less Than 80% Test, or to consider disposing of some or all of their KW Shares, or to reduce the KW Shares they will receive under the Scheme by reducing their holdings of KWE Shares in time, in either case insofar as that would remove or mitigate their potential liability to this form of deemed dividend taxation.

Assuming that a substantial number of beneficial owners of Scheme Shares satisfy the Less Than 80% Test and thus are not treated as if they had received deemed dividends, the remaining recipients of the KW Cash Component payable to non-US Holders (i.e. those which do **not** satisfy the Less Than 80% Test) would be allocated a significantly larger portion of KW's earnings and profits. However, a non-US holder will never be allocated earnings and profits in excess of the KW Cash Component it receives and would thus never be subject to the 30% deemed dividend tax in respect of an amount greater than the KW Cash Component it receives.

To the extent that the KW Cash Component is not treated as a deemed dividend on the basis described above, it may nonetheless be treated as gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme and subjected to US federal income tax if the non-US Holder owned (or was treated as owning) more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the deemed redemption and (ii) the non-US Holder's holding period in such shares (a "5% Holder"). This tax would be payable on the graduating scale of US federal income tax rates (including, if applicable, reduced tax rates on long-term capital gains derived by certain non-corporate holders, including individuals) on the amounts deemed to be received after allowing for certain reductions in respect of tax basis. The Less than 80% Test will not avoid or mitigate this tax.

Any beneficial holders of Scheme Shares who are also holders of KW Shares and who wish to avoid the potential application of these rules should consult their tax advisers, including as to the desirability and tax consequences of disposing of some or all their Scheme Shares or KW Shares in the market prior to the Merger. Any person (whether or not a holder of shares in KW or KWE) who is considering acquiring shares in KW or KWE prior to the Effective Date should consider whether to refrain from doing so if, as a result, it may become liable to additional US tax as a result of the requirements described above.

This summary is, of necessity, a simplification of numerous complex tax considerations. Beneficial owners of Scheme Shares and Scheme Shareholders should not rely on it as a comprehensive statement of the potential tax implications of their receipt of the KW Cash Component under the New Offer or the First KW Dividend and should in addition read and consider carefully the explanations in Part Seven (*Taxation*) of this document in full and seek independent tax advice on them.

Assistance

If you require assistance, including in connection with IRS Forms W-8 or W-9, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

PART ONE

LETTER FROM THE INDEPENDENT DIRECTORS OF KENNEDY WILSON EUROPE REAL ESTATE PLC



KWE Independent Directors: Charlotte Valeur Mark McNicholas Simon Radford Registered office:
47 Esplanade
St Helier
Jersey
JE1 0BD

Registered number: 114680

13 September 2017

To KWE Shareholders

Dear Shareholder,

RECOMMENDED MERGER OF KENNEDY WILSON EUROPE REAL ESTATE PLC AND KENNEDY-WILSON HOLDINGS, INC.

1. Introduction

On 24 April 2017, KWE and KW published an announcement setting out the terms and conditions of a recommended offer by KW to acquire all of the outstanding shares of KWE (other than the Excluded Shares) in an all-share merger (the "Original Offer").

On 13 June 2017, KWE and KW published a further announcement setting out the terms and conditions of an additional offer by KW to acquire all of the outstanding shares of KWE (other than the Excluded Shares) (the "New Offer"). As the New Offer comprises fewer New KW Shares for each Scheme Share than are available under the Original Offer, it is necessary, under the requirements of the Code, for KW to continue to make the Original Offer available in parallel to the New Offer.

It is intended that the Merger will be effected by means of a Court-sanctioned scheme of arrangement of KWE under Article 125 of the Jersey Companies Law (or by a Takeover Offer under certain circumstances described in this document and subject to the consent of the Panel), and KW will undertake to the Court to be bound by and comply with the Scheme.

This letter sets out the terms of the Merger and the Conditions which must be satisfied before the Merger can become effective, and explains the background to the Merger and the reasons for the recommendation of the Scheme and the New Offer by the KWE Independent Directors.

The vote by Scheme Shareholders to approve the Merger is separate from any Election the Scheme Shareholder may make under the Mix and Match Facility or to receive the Original Offer. A Scheme Shareholder will be able to receive different forms of consideration on the basis explained in this document, irrespective of how the Scheme Shareholder votes at the Scheme Court Meeting to approve the Scheme.

Details of the action which may be taken by KWE Shareholders are set out on pages 14 to 22 of this document, and the recommendations of the KWE Independent Directors appear in paragraph 24 of this Part One.

Although the planned date for the Scheme to become effective is 20 October 2017, KW and KWE have agreed to extend the Longstop Date to 30 November 2017 to allow time to deal with any unforeseen delay.

Further details of the terms of the Merger are set out in Part Two (Explanatory Statement) of this document.

2. Terms of the New Offer

Under the terms of the Merger, which is subject to the Conditions and the other terms set out in Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document, if the Scheme becomes effective, a Scheme Shareholder which does not submit a valid Election in the manner set out in this document will be entitled to receive:

In respect of each Scheme Share:

0.3854 New KW Shares (the "New KW Share Component"); and

300 pence in cash, paid by KW (the "KW Cash Component"); and

250 pence in cash, paid by KWE in the form of a special distribution pursuant to the Scheme (the "KWE Special Distribution") (collectively, the "New Offer Consideration and Distribution").

The KWE Special Distribution will be payable by KWE shortly after the Effective Date to Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution. It will be paid in addition to all other dividends payable by KWE, including the Closing Dividend, and dividends paid by KW referred to in this document.

KWE seeks to record its real estate assets in its accounts at their fair value as derived from half yearly valuations. As at 30 June 2017, the accounting value of KWE's net assets, as recorded in its interim accounts for the six months ended 30 June 2017, was £1,567.9 million. If the maximum amount of £241.6 million potentially payable by KWE in respect of the KWE Special Distribution had been paid on 30 June 2017, KWE's net asset value would have been approximately £1,326.3 million as at 30 June 2017.

The attention of beneficial owners of Scheme Shares and Scheme Shareholders is drawn to Part Seven (*Taxation*) of this document. The tax implications of the Merger will depend on the individual circumstances of each beneficial owner of Scheme Shares and Scheme Shareholders. Beneficial owners of Scheme Shares and Scheme Shareholders should consult their own tax advisers, including as to the tax consequences of disposing of their shares pursuant to the Scheme or in the market. Beneficial owners of Scheme Shares and Scheme Shareholders who believe that the tax consequences of the Scheme may be unattractive to them should seek independent professional advice on the courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme).

Based on the Closing Price of a KW Share of US\$19.15 and the US\$ / £ exchange rate of US\$1.3209:£1, in each case at the Latest Practicable Date, the New Offer Consideration and Distribution valued each Scheme Share at approximately 1,109 pence and KWE's entire issued share capital at approximately £1.40 billion. This represents:

- a premium of approximately 13.3% to the Closing Price of 979 pence of one KWE Share on 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- a premium of approximately 15.5% to the volume weighted average price of 960 pence of one KWE Share for the three month period that ended on 21 April 2017 (the last Business Day before the commencement of the Offer Period); and
- a discount of approximately 10.7% to KWE's last reported Adjusted NAV of 1,241 pence of one KWE Share as at 30 June 2017.

Following the completion of the Merger, if all Scheme Shareholders received the New Offer Consideration and Distribution and no Scheme Shareholders received the Original Offer Consideration, former KWE Shareholders (other than KW and other holders of Excluded Shares) would own approximately 25% of the Combined Group and existing KW Shareholders would own approximately 75% of the Combined Group.

3. Mix and Match Facility

Under the Mix and Match Facility, Scheme Shareholders (other than certain Overseas Shareholders) entitled to receive the New Offer Consideration and Distribution may elect to vary the proportions in which they receive the New KW Share Component and the KW Cash Component in respect of each Scheme Share held, subject to the availability of valid countervailing elections under the Mix and Match Facility by other Scheme Shareholders. The Mix and Match Facility will remain available until the Election Return Time, i.e. 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service). The Mix and Match Facility is not available in relation to the Original Offer.

The Mix and Match Facility will not change the aggregate number of New KW Shares issued by KW or the aggregate amount of the KW Cash Component paid by KW pursuant to the New Offer. To the extent that Mix and Match Elections for the New KW Share Component or the KW Cash Component cannot be satisfied in full due to an insufficiency of valid countervailing elections, they will be scaled down on a *pro rata* basis in accordance with Clause 6.7 and Clause 6.8 of the Scheme. As a result, Scheme Shareholders who make an Additional Share Election and/or an Additional Cash Election will not know the exact amount of the KW Cash Component or number of additional New KW Shares they are entitled to receive until settlement of the consideration under the Merger.

The Mix and Match Facility will not enable any Scheme Shareholder to vary the amount of the KWE Special Distribution receivable under the New Offer. It will also not enable any Scheme Shareholder to vary the amount of cash receivable in respect of the Closing Dividend, although a Mix and Match Election which results in a change in the number of New KW Shares received by a Scheme Shareholder will, as a consequence, alter the amount of dividends receivable from KW after the Effective Date and consequently alter the amount receivable in respect of the Closing Dividend, as explained more fully in paragraph 5 of Part 2 (Explanatory Statement).

A Mix and Match Election will be of no effect to the extent that it is made in respect of a Scheme Share for which a valid Original Offer Election has been received and not validly withdrawn.

Elections under the Mix and Match Facility may be made on the following basis:

for every 300 pence of the KW Cash Component 0.20

0.2069 of a New KW Share

OR

for every 0.3854 of a New KW Share

559 pence of the KW Cash Component.

SCHEME SHAREHOLDERS SHOULD NOTE THAT:

- THE MIX AND MATCH RATIOS SPECIFIED ABOVE ARE BASED ON THE CLOSING PRICE OF A KW SHARE OF US\$19.15 AND THE EXCHANGE RATE OF US\$1.3209:£1, IN EACH CASE AS AT THE LATEST PRACTICABLE DATE.
- THE CLOSING PRICES OF KW SHARES AND EXCHANGE RATES MAY FLUCTUATE BEFORE AND AFTER THE ELECTION RETURN TIME.

Further information about the Mix and Match Facility is provided in paragraph 3 of Part Two (Explanatory Statement) and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

The availability of the Mix and Match Facility to certain Overseas Shareholders may be restricted or prohibited by law. Further details are set out in paragraph 33 of Part Two (Explanatory Statement) and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

4. Terms of the Original Offer

The Original Offer will be made available to Scheme Shareholders (other than certain Overseas Shareholders) on the terms and subject to the Conditions set out in Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document. Each Scheme Shareholder validly electing for the Original Offer will be entitled to receive:

In respect of each Scheme Share:

0.667 New KW Shares.

Scheme Shareholders who elect to receive the Original Offer in respect of their entire holdings of Scheme Shares will not be entitled to receive the New KW Share Component, the KW Cash Component or the KWE Special Distribution available under the New Offer and will not be entitled to make any Election under the Mix and Match Facility. However, the dividend arrangements described in paragraph 5 of this Part One below, including the Closing Dividend, will still apply.

Based on the Closing Price of a KW Share of US\$19.15 and the US\$ / £ exchange rate of US\$1.3209:£1, in each case at the Latest Practicable Date, the terms of the Original Offer valued each Scheme Share at approximately 967 pence and KWE's entire issued share capital at approximately £1.22 billion. This represents:

- a discount of approximately 1.2% to the Closing Price of 979 pence of one KWE Share on 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- a premium of approximately 0.7% to the volume weighted average price of 960 pence of one KWE Share for the three month period that ended on 21 April 2017 (the last Business Day before the commencement of the Offer Period); and
- a discount of approximately 22.1% to KWE's last reported Adjusted NAV of 1,241 pence of one KWE Share as at 30 June 2017.

If all Scheme Shareholders received the Original Offer Consideration and no Scheme Shareholders received the New Offer Consideration and Distribution, former KWE Shareholders (other than KW and other holders of Excluded Shares) would own approximately 36% of the Combined Group and existing KW Shareholders would own approximately 64% of the Combined Group.

The availability of the Original Offer Election to certain Overseas Shareholders may be restricted or prohibited by law. Further details are set out in paragraph 33 of Part Two (Explanatory Statement) and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

As at the Latest Practicable Date, the Original Offer valued each Scheme Share at 967 pence compared with 1,109 pence under the New Offer on the basis of the values, prices and exchange rates referred to above. As at the Latest Practicable Date, the value of the Original Offer on a pre-tax basis was therefore materially less than the value of the New Offer on a pre-tax basis. The KW Share price would need to rise from US\$19.15 as at the Latest Practicable Date to US\$25.82 (i.e. an increase of over approximately 34%) in order for the Original Offer to represent greater value than the New Offer on a pre-tax basis (assuming no Mix and Match Election and a US\$ / £ exchange rate of US\$1.3209:£1).

5. Dividends

KW and KWE have the same quarterly dividend periods. KW and KWE have agreed arrangements under which each KWE Shareholder (including those who are not Scheme Shareholders) will receive the usual KWE quarterly dividends payable on their KWE Shares, subject to certain adjustments designed to avoid duplicated payments which would otherwise result from the First KW Dividend being paid in respect of a period which includes part of the Final KWE Dividend Period. These arrangements apply in the event that the Scheme becomes effective and the KWE Special Distribution is paid if, when and to the extent required under the Scheme with the result that the remaining arrangements under the Scheme also have effect.

The Final KWE Dividend will accrue in the usual way up to, but excluding, the Effective Date, on the basis that the full amount of the Final KWE Dividend payable to KWE Shareholders will be reduced to be the Pre-Effective Date Proportion of the full amount of the Final KWE Dividend (i.e. the amount of that dividend that accrues from and including the first day of the Final KWE Dividend Period to, but excluding, the Effective Date). As a result of this adjustment, the amount of the Final KWE Dividend of 12 pence per KWE Share will be reduced *pro rata* based on the number of days between (and including) the first day of the Final KWE Dividend Period up to (but excluding) the Effective Date (as a proportion of the full number of days in the Final KWE Dividend Period).

The amount of the Final KWE Dividend, as so adjusted, will be further reduced, in respect of each KWE Share which is a Scheme Share, by an amount equal to the pound sterling equivalent of the Pre-Effective Date Proportion of the First KW Dividend receivable on New

KW Shares issued in exchange for such Scheme Share (i.e. the amount of the First KW Dividend that accrues from and including the first day of the Final KWE Dividend Period up to (but excluding) the Effective Date).

For some beneficial owners of KWE Shares and KWE Shareholders who are not US taxpayers, the First KW Dividend (as well as subsequent KW dividends) may be subject to US federal withholding tax at a rate of up to 30%, as explained below under "Taxation Information". As a result, for those beneficial owners of KWE Shares and KWE Shareholders, the deduction made in respect of the gross amount of the First KW Dividend will reduce the net sum they receive in respect of the Closing Dividend to less than would have been received if those beneficial owners of KWE Shares and KWE Shareholders had received dividends only from KWE.

KWE dividends are payable in pounds sterling while KW dividends are payable in US Dollars. Accordingly, the extent to which the Pre-Effective Date Proportion of the First KW Dividend will reduce the Pre-Effective Date Proportion of the Final KWE Dividend will be determined by translating the US Dollar amount of the First KW Dividend into pounds sterling using the Applicable Translation Rate. The portion of the Final KWE Dividend payable after these adjustments is referred to as the "Adjusted Final KWE Dividend". The Adjusted Final KWE Dividend will be rounded to the second decimal place.

Both the Adjusted Final KWE Dividend and the First KW Dividend are expected to be paid on the same day, being the First KW Dividend Payment Date (i.e. not later than 10 days after the end of the First KW Dividend Period).

Where, pursuant to the Scheme, a Scheme Shareholder reduces the number of New KW Shares to be received in respect of that Scheme Shareholder's Scheme Shares (e.g. by choosing to receive the New Offer Consideration and Distribution in preference to the Original Offer or by electing for more cash under the Mix and Match Facility), the aggregate amount of the First KW Dividend receivable by that Scheme Shareholder for each Scheme Share will also correspondingly reduce (because the Scheme Shareholder will be receiving fewer New KW Shares). As a result, the amount by which the Adjusted Final KWE Dividend payable in respect of such Scheme Shares is reduced to take account of receipt of the First KW Dividend will also be correspondingly less (i.e. the amount of the Scheme Shareholder's Adjusted Final KWE Dividend will be correspondingly larger).

The "Closing Dividend" will comprise the Adjusted Final KWE Dividend and the Penultimate KWE Dividend. The Closing Dividend will be payable by KWE on the First KW Dividend Payment Date (i.e. it will be payable on the same date as the First KW Dividend), which shall be not later than 10 days after the end of the First KW Dividend Period (i.e. not later than 10 January 2018 if, as expected, the Effective Date falls in 2017). The Closing Dividend will be paid to all KWE Shareholders who are on the Register at the Closing Dividend Record Time.

If the value of the US Dollar were to strengthen very considerably against the pound sterling and such strengthening, either alone or in combination with an increase in the number of New KW Shares received as a result of a valid Election in respect of the Mix and Match Facility or the Original Offer, were to result in the Pre-Effective Date Proportion of the First KW Dividend receivable by a Scheme Shareholder reducing the Adjusted Final KWE Dividend below zero, the deficit would be applied to reduce the Penultimate KWE Dividend, on the basis that the Pre-Effective Date Proportion of the Final KWE Dividend is added to the full amount of the Penultimate KWE Dividend to establish the full amount of the Closing Dividend. A similar outcome would result if the First KW Dividend were unexpectedly substantially greater than US\$0.19 per KW Share. However, assuming dividends paid by KW in respect of the First KW Dividend Period do not exceed US\$0.19 per KW Share, for a Scheme Shareholder who becomes entitled to receive the New Offer Consideration and Distribution but makes no Mix and Match Election, the Applicable Translation Rate would have to change from US\$1.3209:£1 (the Applicable Translation Rate as at the Latest Practicable Date) to approximately US\$0.60:£1 in order for the Adjusted Final KWE Dividend to be reduced to below zero and for the Penultimate KWE Dividend to be reduced in this way.

The amount of the Closing Dividend (i.e. the total of the Adjusted Final KWE Dividend and the Penultimate KWE Dividend) can never be reduced in this way to less than zero.

KWE Quarterly Dividends before the Penultimate KWE Dividend

In addition to the Closing Dividend, KW has agreed that KWE Shareholders will be entitled to receive any quarterly dividend declared and paid by KWE in amounts of up to £0.12 per KWE Share in aggregate in respect of any KWE Dividend Period expiring before the start of the Penultimate KWE Dividend Period. Accordingly, as announced on 27 June 2017 by KWE, on 31 August 2017, KWE paid KWE Shareholders who were on the Register on 18 August 2017 its usual quarterly dividend of £0.12 per KWE Share in respect of the KWE Dividend Period ended on 30 June 2017.

Taxation information relevant to dividends

The attention of beneficial owners of KWE Shares and KWE Shareholders is drawn to Part Seven (*Taxation*) of this document. Beneficial owners of KWE Shares and KWE Shareholders should note that distributions paid on KW Shares to a beneficial owner of KW Shares or a KW Shareholder (including a former beneficial owner of KWE Shares or KWE Shareholder who receives New KW Shares pursuant to the Merger) that is not a "United States person" for US federal income tax purposes generally will be subject to US withholding tax at a rate of 30% of the gross amount of the distribution (or a reduced rate prescribed by an applicable double tax treaty if the beneficial owner of KW Shares or KW Shareholder is entitled to claim and makes a valid claim under such treaty). Accordingly, the net amount received in aggregate by a beneficial owner of KWE Shares or KWE Shareholder as a result of the combination of (i) the Closing Dividend calculated in the manner set out in this document and (ii) the First KW Dividend may not be the same as if such beneficial owner of KWE Shares or KWE Shareholder had only received dividends from KWE in respect of the same aggregate period.

Beneficial owners of KWE Shares and KWE Shareholders who believe that they may prefer alternative tax treatments to those described in Part Seven (*Taxation*) should consider seeking independent professional advice on alternative courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme) in the event that KWE were to be treated as a partnership for US tax purposes.

6. Fractional Entitlements

Fractions of New KW Shares will not be issued to Scheme Shareholders. Instead, Scheme Shareholders who would otherwise have received a fraction of a New KW Share will receive an amount in cash rounded to the nearest cent, based on the amount obtained by multiplying such fraction by the average of the highest and lowest sales prices of KW Shares on the NYSE on each of the five consecutive trading days ending on the trading day that is two trading days prior to the Effective Date, except that individual entitlements of less than US\$5.00 will not be paid but will be retained for the benefit of the Combined Group. For the purposes of determining fractional entitlements, each portion of a Scheme Shareholder's holding recorded in the Register by reference to a separate designation at the Scheme Record Time, whether in certificated or uncertificated form, shall be treated as a separate holding. Fractional entitlements will be determined after allowing for the outcome of Mix and Match Elections and Original Offer Elections.

7. Irrevocable Undertakings and Statements of Intent

7.1 KWE Shareholders

KW has received irrevocable undertakings to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept or procure acceptance of the Takeover Offer) in respect of a total of 27,529,150 Scheme Shares, representing, in aggregate, approximately 28.6% of the KWE Shares whose holders are entitled to vote at the Scheme Court Meeting and approximately 21.8% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date.

KW has also received a letter of intent from the Värde Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept the Takeover Offer) in respect of a total of 5,452,722 Scheme Shares, representing, in

aggregate, approximately 5.7% of the KWE Shares whose holders are entitled to vote at the Scheme Court Meeting and approximately 4.3% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date.

Details of these irrevocable undertakings and the letter of intent obtained by KW (including details of the circumstances in which the irrevocable undertakings will cease to be binding) are set out in paragraph 16 of Part Two (*Explanatory Statement*) and paragraph 4 of Part Eight (*Additional Information*) of this document.

7.2 KW Shareholders

Fairfax Financial Holdings Limited and Elkhorn Partners LP have indicated to KW that they intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 14.3% of KW's outstanding shares of common stock as at the Latest Practicable Date.

The KW Directors, as well as Mary L. Ricks, Matt Windisch and Justin Enbody (being all of the executive officers of KW who hold KW Shares) and In Ku Lee, also intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 17.1% of KW's outstanding shares of common stock as at the Latest Practicable Date.

Therefore, support from KW Shareholders to vote in favour of the KW Shareholder Resolution at the KW Special Meeting has been received by KW in respect of approximately 31.4% of KW's outstanding shares of common stock as at the Latest Practicable Date.

8. Independent Valuation of KWE's assets

For information regarding the independent valuation of KWE's real estate assets by CBRE and Colliers in accordance with Rule 29 of the Code, please see Annex I (Valuation Reports) to this document.

9. Background to and reasons for the recommendation of the Scheme and the New Offer by the KWE Independent Directors

Since its IPO in 2014, KWE's strategy of unlocking value in under-resourced real estate through active portfolio management, asset re-positioning and rigorous capital management has yielded strong returns, generating growth in net asset value and a growing dividend.

The KWE Independent Directors believe that KWE's balanced portfolio, strong cash flows and ample liquidity place the business on a strong foundation and, accordingly, the KWE Independent Directors continue to believe in the strong standalone prospects of KWE in the event that the Merger does not become effective.

The KWE Independent Directors note that based on the Closing Price of a KW Share of US\$19.15 and the spot rate for conversion of US Dollar into pounds sterling of US\$1.3209:£1, in each case as at the Latest Practicable Date, the terms of the New Offer implied a value as at the Latest Practicable Date representing 89.3% of KWE's reported Adjusted NAV per KWE Share as at 30 June 2017 and a premium of 13.3% to the Closing Price of a KWE Share on 21 April 2017 (being the last Business Day prior to the Original Announcement).

The KWE Independent Directors have also taken note of the financial effects of the Merger on the capital and income position of Scheme Shareholders on a pre-tax basis, as summarised in the table in paragraph 21 of Part Two (Explanatory Statement).

The KW Share price would need to rise from US\$19.15 as at the Latest Practicable Date to US\$25.82 (i.e. an increase of over approximately 34%) in order for the Original Offer to represent greater value than the New Offer on a pre-tax basis (assuming no Mix and Match Election and a US\$ / £ exchange rate of US\$1.3209:£1).

The KWE Independent Directors also recognise that the Merger has the potential to deliver to those KWE Shareholders who receive New KW Shares pursuant to the Merger the following other advantages:

- access to a diversified international platform with strong growth prospects;
- the potential for future upside from investment in the Combined Group based on opportunistic investment policies of management across multiple geographies and asset classes;

- greater diversification of the underlying asset base with exposure to KW's existing asset portfolio; and
- access to the same management expertise as currently in place at KWE with a proven track record of generating returns on investment.

In formulating their views in relation to the Merger and the terms of the New Offer and the Original Offer, the KWE Independent Directors have taken into account a number of other considerations, including the existing externally managed structure of KWE, the terms of the IMA, KWE's shareholder base (including KW's position as a 23.8% shareholder of KWE) and KWE's existing capital structure, business plan and prospects, as well as the irrevocable undertakings referred to in paragraph 7 of this Part One.

Having regard to these and other relevant considerations (including the information published by KW in Part Fourteen (*Profit Forecasts*) of this document), the KWE Independent Directors are unanimously recommending Scheme Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and to choose to receive the New Offer Consideration and Distribution and are recommending all KWE Shareholders to vote in favour of the Special Resolution at the KWE General Meeting. They do not make any recommendation in relation to the Mix and Match Facility and are not recommending Scheme Shareholders to elect for the Original Offer as, at the Latest Practicable Date, it had a materially lower value in pound sterling terms than the New Offer based on the values, the Closing Price of a KW Share and the US\$ / £ exchange rate applicable at the Latest Practicable Date.

10. Strategic Plans, Management, Employees and Pension Schemes

10.1 KW Statement

The KW Directors believe that the Merger will create a leading global real estate investment and asset management company with enhanced scale, diversification, liquidity and access to capital.

The KW Board recognises that, in order to achieve the expected benefits of the Merger, operational and administrative restructuring may be required following completion of the Merger. The detailed steps for such restructuring (if any) are not yet known, but KW will aim to retain the best talent across the Combined Group and has confirmed that the existing contractual and statutory employment rights of all KW and KWE employees will be fully safeguarded. No decisions have yet been made by KW in relation to specific actions that may be taken, though KW does intend to terminate the IMA in due course following completion of the Merger.

KW continues its detailed review of the combined business to identify realisable cost synergies, but that review is on-going and will not be completed until after completion of the Merger. Until such review is complete, KW cannot be certain what impact there will be on the Combined Group's business, KWE's management, location of business or any redeployment of KWE's fixed assets. However, given the close existing relationship between KW and KWE, the KW Board's expectation is that the effects of any restructuring will not be material. KWE is an externally managed investment company with no pension schemes.

Following completion of the Merger, the Combined Group will continue to invest both in the US and globally and make investment decisions on the basis of risk-adjusted returns on capital. KW has confirmed that the Merger will not have any effect on its management, on the location of its business or (save as disclosed in this paragraph 10.1) on its employees.

10.2 Response from the KWE Independent Directors

The KWE Independent Directors note the statement made by KW that, in order to achieve the expected benefits of the Merger, operational and administrative changes may be required but that the detailed steps for any restructuring, other than the expected termination of the IMA in due course, are not yet known.

The KWE Independent Directors note in particular that KWE is an externally managed investment company and that, having regard to the close relationship between KW and KWE, the KW Board does not currently expect that the effects of any restructuring will be material. KWE does not have employees (other than hotel staff) and, as an externally managed company, has no offices of its own.

The KWE Independent Directors welcome the statements made by KW that the existing contractual and statutory employment rights of all KWE employees in its operating hotel assets will be fully safeguarded.

In view of the on-going review to be completed following the successful completion of the Merger, as described above, the KWE Independent Directors are unable to express a more detailed opinion on the impact of the Merger on KWE's business.

11. Conditions

- 11.1 The Merger will be subject to the Conditions set out in Part Three (Conditions and Certain Further Terms of the Scheme) of this document, including, among other things:
- (1) (i) approval of the Scheme at the Scheme Court Meeting (and at any separate class meeting which may be required), or any adjournment thereof, by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the Register at the Scheme Voting Record Time, and present and voting (either in person or by proxy) at the Scheme Court Meeting, representing three-quarters or more of the votes attached to Scheme Shares which are cast at the Scheme Court Meeting (or at the meeting of each class of Scheme Shareholders, as the case may be); and (ii) such Scheme Court Meeting being held on or before the later of (1) the 22nd day after the expected date of the Scheme Court Meeting set out in this document; and (2) the fifth Business Day following the date on which the KW Shareholder Resolution has been duly passed or, in each case, such later date (if any) as KW and KWE may agree (any such agreement would require the consent of the Court and the Panel);
- (2) (i) all resolutions required to approve and implement the Scheme as set out in the notice of the KWE General Meeting (including, without limitation, the Special Resolution) being duly passed by the requisite majority of KWE Shareholders at the KWE General Meeting or at any adjournment thereof; and (ii) the KWE General Meeting being held on or before the later of (1) the 22nd day after the expected date of the KWE General Meeting set out in this document; and (2) the fifth Business Day following the date on which the KW Shareholder Resolution has been duly passed or, in each case, such later date (if any) as KW and KWE may agree (any such agreement would require the consent of the Court and the Panel);
- (3) the sanction of the Scheme by the Court without modification, or with modification on terms acceptable to KW and KWE, and the delivery of the Court Order to the Registrar of Companies for registration;
- (4) the KW Shareholder Resolution being duly passed at the KW Special Meeting;
- (5) the NYSE having authorised the Listing upon official notice of issuance and not having withdrawn such authorisation; and
- (6) to the extent that a mandatory notification to the CCPC pursuant to section 18(1)(a) of the Irish Competition Act is required in respect of the Merger, the Merger having been notified to the CCPC and one of the following having occurred: (a) the CCPC having informed KW of its determination pursuant to either section 21(2)(a) or section 22(3)(a) of the Irish Competition Act that the Merger may be put into effect; (b) the CCPC having informed KW of its determination pursuant to section 22(3)(c) of the Irish Competition Act that the Merger may be put into effect subject to condition(s) specified by the CCPC being complied with; (c) the period specified in section 21(2) of the Irish Competition Act having lapsed without the CCPC having informed KW of its determination (if any) under section 21(2)(a) or (b) of the Irish Competition Act; or (d) where applicable, if 120 working days after the appropriate date (as defined by section 19(6) of the Irish Competition Act) have elapsed without the CCPC having informed KW of its determination (if any) under section 22 of the Irish Competition Act in relation to the Merger.
- 11.2 The New Offer was, when originally announced, subject to the additional Conditions set out in Section 3 of Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document and below:
 - (1) the KW Prospectus being (a) approved by the UK Listing Authority; and (b) made available to the public in accordance with the prospectus rules made by the Financial Conduct Authority under section 73A of the Financial Services and Markets Act 2000 relating to the offer of securities to the public and admission of securities to trading on a regulated market; and

(2) to the extent required, the Jersey Financial Services Commission having consented to the circulation in Jersey of the offer of the New KW Shares pursuant to Article 8(2) of the Control of Borrowing (Jersey) Order 1958, as amended.

For the full wording of all of the Conditions, please see Part Three (Conditions and Certain Further Terms of the Scheme) of this document.

The conditions referred to in paragraphs 11.2(1), 11.2(2) and 11.1(6) above have been satisfied in accordance with their terms as at the date of this document.

11.3 Payment of the KWE Special Distribution is subject to such payment being permissible under the Jersey Companies Law at the time of its payment. Although this is not a condition of the Scheme becoming effective, the terms of the Scheme provide that if unanticipated circumstances arose which prevented payment of the KWE Special Distribution as, when and to the extent provided for under the Scheme, the transfer of Scheme Shares to KW, the delivery of the Original Offer Consideration and/or the delivery of the KW Share Component and the payment of the KW Cash Component under the New Offer, as well as the arrangements relating to the Closing Dividend, would not be implemented.

12. KW Shareholder Approval

KW is required to obtain the approval of its shareholders for the issue of the New KW Shares in connection with the Merger at the KW Special Meeting.

The Merger is conditional on, among other matters described in paragraph 11 of this Part One and in Part Three (*Conditions and Certain Further Terms of the Scheme*), the KW Shareholder Resolution being passed by a majority of the votes cast at the KW Special Meeting (or an adjournment thereof).

KW will make the KW Proxy Statement available to KW Shareholders on or about the date of this document, which will include a notice convening the KW Special Meeting at 9.00 a.m. (Pacific Daylight Time) on 12 October 2017.

The dissemination and content of the KW Proxy Statement are the responsibility solely of the KW Directors. See also "Important Notice" on pages 4 to 9 of this document.

For the avoidance of doubt, none of the KWE Independent Directors, and neither Rothschild nor J.P. Morgan Cazenove expresses any views as to the advantages or disadvantages of any aspects of the Merger as far as KW Shareholders are concerned.

13. New KW Shares

The New KW Shares to be issued in connection with the Merger will be issued credited as fully paid and free from all Encumbrances, will rank *pari passu* in all respects with the existing issued KW Shares and will carry the right to receive all dividends and other distributions declared, paid or made by KW on or after the Effective Date or by reference to a record date on or after the Effective Date (in each case whether or not wholly or partly in respect of a period which precedes the Effective Date).

Application will be made to the NYSE for the New KW Shares to be admitted to listing and trading on the NYSE. Authorisation of the Listing by the NYSE is a Condition to the Scheme becoming effective and if such authorisation is not provided or is withdrawn, the Scheme shall not become effective. It is anticipated that the listing of the New KW Shares will become effective and that trading in New KW Shares will commence on the Effective Date. The New KW Shares will be denominated in US Dollars and, when admitted to trading on the NYSE, will be registered with the ISIN US 4893981070 and ticker symbol "KW".

The price at which KW Shares are publicly traded on the NYSE is subject to fluctuations and may be influenced by a large number of factors. These factors could be specific to KW and its operations or may affect the industry generally. The price at which New KW Shares are publicly traded on the NYSE at the Effective Date and the price which KW Shareholders may subsequently realise for their New KW Shares cannot be predicted.

The KW Prospectus summarises certain risk factors relating to the Merger in respect of the receipt of the New KW Shares. Copies of the KW Prospectus and this document will be available to view online, as set out in paragraph 17.2 of Part Eight (*Additional Information*) of this document. The publication and content of the KW Prospectus are the responsibility solely of the KW Directors. See also "Important Notice" on pages 4 to 9 of this document.

Further details of the rights attaching to the New KW Shares are set out in Part Ten (Description of KW Shares) of this document.

14. The KWE Independent Directors and the effect of the Scheme on their interests

Except as referred to in paragraph 14 of Part Two (*Explanatory Statement*) of this document, the effect of the Scheme on the interests of each KWE Independent Director does not differ from its effect on the like interests of any other Scheme Shareholder. Your attention is drawn to paragraph 14 of Part Two (*Explanatory Statement*) of this document, which sets out details of the KWE Independent Directors and the effect of the Scheme on their interests.

15. Confidentiality Agreement

KW, the Manager and KWE entered into the Confidentiality Agreement on 23 March 2017 pursuant to which the parties have undertaken, among other things, to (a) keep confidential all confidential information relating to the Merger and the other parties and not to disclose it to third parties (other than to certain permitted recipients) unless such disclosure is permitted pursuant to the Confidentiality Agreement; and (b) use the confidential information for the sole purpose of considering, facilitating or advising on, or seeking advice in relation to, the Merger.

Under the Confidentiality Agreement, KW and the Manager undertook to KWE to instruct the Ring Fenced Employees promptly to supply to the KWE Independent Directors all information belonging to KWE requested by any KWE Independent Director and to instruct them to ensure that such information was, when supplied, complete, accurate and not misleading.

KW also agreed that, other than KWE Shares acquired pursuant to the IMA, neither it nor any person acting in concert with it may acquire any interest in any KWE Shares without the prior written consent of the KWE Independent Directors. Further details of the Confidentiality Agreement are contained in paragraph 30(a) of Part Two (*Explanatory Statement*) of this document. The Confidentiality Agreement can also be reviewed on KWE's website at www.kennedywilson.eu and on KW's website at ir.kennedywilson.com.

16. Separate Manager's Powers Agreement

KWE and the Manager entered into the Separate Manager's Powers Agreement on 10 April 2017 pursuant to which the Manager has agreed that during a specified period, it will make all arrangements requested by the Independent Directors of KWE to allow certain individuals to make payments out of KWE bank accounts. The Manager has also agreed that, during the specified period, it will not take certain actions without the prior written approval of the Independent Directors of KWE. Further details of the Separate Manager's Powers Agreement are contained in paragraph 30(b) of Part Two (*Explanatory Statement*) of this document. The Separate Manager's Powers Agreement can also be reviewed on KWE's website at www.kennedywilson.eu and on KW's website at ir.kennedywilson.com.

17. Transaction Agreement

KW and KWE entered into the Transaction Agreement on 24 April 2017 in relation to certain mutual commitments to regulate the basis on which they are willing to implement the Merger. Further details of the Transaction Agreement are contained in paragraph 30(c) of Part Two (*Explanatory Statement*) of this document. The Transaction Agreement can also be reviewed on KWE's website at www.kennedywilson.eu and on KW's website at ir.kennedywilson.com.

18. Indemnity Agreement and KWE Independent Directors' Additional Remuneration Agreements

KW and KWE entered into the Indemnity Agreement on 23 June 2017 pursuant to which KW and the Manager, subject to certain exceptions and limitations, will hold harmless and indemnify KWE and the KWE Independent Directors for liability arising from inaccuracies, deficiencies and omissions relating to certain KWE information communicated externally or filed with a regulatory body. The Indemnity Agreement further provides that the Manager will, when requested, promptly furnish the KWE Independent Directors with KWE information relating to KWE's financial condition and use reasonable efforts to ensure such information is accurate and not misleading.

Pursuant to the terms of the KWE Independent Directors' Additional Remuneration Agreements, KWE agreed to pay each KWE Independent Director extra remuneration for any services rendered in his or her capacity as a KWE Director in relation to the Merger.

Further details of the Indemnity Agreement and the additional remuneration agreements are contained in paragraph 30(d) of Part Two (*Explanatory Statement*) and paragraph 12 of Part Eight (*Additional Information*) of this document, respectively. The Indemnity Agreement and the additional remuneration agreements can also be reviewed on KWE's website at www.kennedywilson.eu and on KW's website at ir.kennedywilson.com.

19. Taxation

Your attention is drawn to Part Seven (*Taxation*) of this document, which contains a summary of certain aspects of the Jersey, UK and US tax treatment of the Merger. That summary relates only to the position of certain categories of beneficial owners of Scheme Shares and Scheme Shareholders (as explained further in Part Seven (*Taxation*) of this document), does not constitute tax advice and does not purport to be a complete analysis of all potential Jersey, UK or US tax consequences of the Merger.

If KWE were to elect to be treated as a partnership for US tax purposes at or before the time at which KW acquires the Scheme Shares pursuant to the Scheme, beneficial owners of Scheme Shares and Scheme Shareholders who are US Holders and who have made a "qualified electing fund" ("QEF") election (a "QEF election") with respect to their Scheme Shares may incur additional US tax liabilities in connection with the Scheme. Such beneficial owners of Scheme Shares and Scheme Shareholders may also incur other tax liabilities if this occurs after that time. KW undertakes for the benefit of all beneficial owners of Scheme Shares and Scheme Shareholders that it will not take any step, and will procure that none of its Affiliates will take any step, calculated to cause KWE to be treated as a partnership for US tax purposes effective on or prior to the date on which it acquires the Scheme Shares pursuant to the Scheme, except with the prior written consent of the KWE Independent Directors. The KWE Independent Directors will not give their consent to this unless, seven (7) days prior to the Scheme Court Meeting, they have announced their intention to do so through a Regulatory Information Service and unless KW undertakes not to take, and to prevent its Affiliates from taking, any such step until after the Court agrees to sanction the Scheme on the basis that this will occur.

Any US Holder (as defined in Part Seven (*Taxation*) of this document) who has made, or is considering making, a QEF election with respect to his or her Scheme Shares or which is a tax-exempt organisation, or a non-US Holder which is an entity controlled by a foreign government under the US federal income tax laws should note, in particular, the matters discussed in paragraphs 2.1(3), 2.1(4) and 2.2(3) of Part Seven (*Taxation*) of this document.

Beneficial owners of KWE Shares and KWE Shareholders who believe that they may prefer alternative tax treatments to those described in Part Seven (*Taxation*) should consider seeking independent professional advice on alternative courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme) in the event that KWE were to be treated as a partnership for US tax purposes.

The payment of the KW Cash Component, the KWE Special Distribution and/or any other amount (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer to Scheme Shareholders entitled to the New Offer may be subject to US federal backup withholding (currently, at a rate of 28%). An explanation of this backup withholding is given in paragraphs 2.1(6) and 2.2(5) of Part Seven (*Taxation*) of this document.

In accordance with Clause 8.10 of the Scheme, KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct IRS Form

W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017 the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-8 or IRS Forms W-9 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-9 or, as applicable, IRS Forms W-8 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed IRS Forms W-9 or applicable IRS Forms W-8, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of IRS Form W-9 and applicable IRS Form W-8 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A SEPARATE APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MATTER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 OR TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 OR IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 OR IRS FORM W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 OR IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT THE INTERMEDIARY, OR RECEIVING AGENT, AND YOU RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

US federal backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability) or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner.

FAILURE TO COMPLETE AND RETURN IRS FORMS W-9 OR, IF APPLICABLE, IRS FORMS W-8 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

The various versions of IRS Forms W-8 and W-9 for use by different categories of taxpayer are generally as follows:

- W-9 for use by beneficial owners of Scheme Shares which are "United States persons";
- W-8BEN for use by individual beneficial owners of Scheme Shares that are not "United States persons";
- W-8BEN-E for use by corporate and certain other non-individual beneficial owners of Scheme Shares that are not "United States persons"; and

• W-8IMY for use by non-US intermediaries and non-US partnerships (or entities treated as partnerships for US federal income tax purposes).

Beneficial owners of Scheme Shares and Scheme Shareholders should also review the discussion of US federal backup withholding in Part Seven (*Taxation*) of this document and consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e., Form W-9 or applicable Form W-8) and any supporting documentation to be provided to avoid potential US federal backup withholding.

In addition to the backup withholding regime, there are special rules regarding US taxation in relation to the KW Cash Component of the New Offer. For a more complete explanation of these, beneficial owners of KWE Shares and KWE Shareholders should carefully review Part Seven (*Taxation*) of this document and seek advice from their tax advisers.

In overview, these rules could, in certain circumstances, cause non-US Holders of Scheme Shares to be subject to US federal income tax in respect of the KW Cash Component despite the fact that KW pays the KW Cash Component without deduction or withholding of US federal income tax. In general, these rules could only apply if beneficial owners of Scheme Shares that own at least 50% of KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme come to own at least 50% of KW immediately after such time, in each case taking into account constructive ownership rules.

In that case, all or a portion of the cash received by a non-US Holder in respect of the KW Cash Component could be subject to US tax on deemed dividends by KW or treated as a gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme.

- Any such deemed dividend would be treated as paid by KW to the extent of its 2017 "earnings and profits" for US federal income tax purposes. This tax would not exceed 30% of the lesser of (a) the KW Cash Component received by the relevant non-US Holder and (b) the non-US Holder's attributable proportion of such earnings and profits (and may, in appropriate cases, be eligible for a reduced rate of US income taxation under an applicable tax treaty). Any earnings and profits of KW would be allocated ratably among actual and deemed dividends paid by KW in 2017. KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profits (for US federal income tax purposes) for the financial year ending 31 December 2017.
- In addition, even if, contrary to KW's stated expectation, KW were to have a material amount of earnings and profits for US federal income tax purposes for the financial year ending 31 December 2017, any non-US Holder whose percentage interest (taking into account the constructive ownership rules described below) in all of the outstanding KW Shares immediately after the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme (including, but not limited to, New KW Shares acquired pursuant to the Scheme, as well as any KW Shares otherwise acquired in connection therewith) is less than 80% of such non-US Holder's relative interest in KWE Shares immediately prior to such time (the "Less Than 80% Test") will not be treated as receiving a deemed dividend (and will thus not be subject to the 30% deemed dividend tax referred to above on any portion of the KW Cash Component received pursuant to the Scheme).

For example, a non-US Holder that held a 5% interest in KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme could not be subject to the 30% deemed dividend tax unless such non-US Holder holds a 4% or greater interest in KW Shares immediately after such time. Accordingly, the 30% deemed dividend tax would not apply to any non-US Holder who does not actually or constructively hold any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme, but who only acquires KW Shares pursuant to the Scheme. Under constructive ownership rules, a non-US Holder generally will be treated as owning stock owned by

certain related persons (such as an individual's spouse, children, grandchildren and parents), as well as, in certain circumstances, such holder's allocable share of stock held by any corporation (such as KWE shares owned by KW immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme) or any partnership, estate or trust, in each case, in which the non-US Holder or its relatives hold an interest. In addition, a non-US Holder which actually or constructively holds an option to purchase stock, is treated as owning the underlying stock.

A non-US Holder will not be subject to this form of 30% deemed dividend tax as a result of choosing to receive the consideration available under the New Offer or as a result of making any Additional Share Elections under the Mix and Match Facility if that non-US Holder does not own, actually or constructively, any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme. Accordingly, non-US Holders of Scheme Shares may wish to consider whether to avoid (or limit) any further acquisitions of KW Shares prior to such time. Those non-US Holders who already own KW Shares may also wish to avoid (or limit) any Additional Share Elections under the Mix and Match Facility, which could jeopardise their ability to satisfy the Less Than 80% Test, or to consider disposing of some or all of their KW Shares, or to reduce the KW Shares they will receive under the Scheme by reducing their holdings of KWE Shares in time, in either case insofar as that would remove or mitigate their potential liability to this form of deemed dividend taxation.

Assuming that a substantial number of beneficial owners of Scheme Shares satisfy the Less Than 80% Test and thus are not treated as if they had received deemed dividends, the remaining recipients of the KW Cash Component payable to non-US Holders (i.e. those which do not satisfy the Less Than 80% Test) would be allocated a significantly larger portion of KW's earnings and profits. However, a non-US holder will never be allocated earnings and profits in excess of the KW Cash Component it receives and would thus never be subject to the 30% deemed dividend tax in respect of an amount greater than the KW Cash Component it receives.

• To the extent that the KW Cash Component is not treated as a deemed dividend on the basis described above, it may nonetheless be treated as gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme and subjected to US federal income tax if the non-US Holder owned (or was treated as owning) more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the deemed redemption and (ii) the non-US Holder's holding period in such shares (a "5% Holder"). This tax would be payable on the graduating scale of US federal income tax rates (including, if applicable, reduced tax rates on long-term capital gains derived by certain non-corporate holders, including individuals) on the amounts deemed to be received after allowing for certain reductions in respect of tax basis. The Less than 80% Test will not avoid or mitigate this tax.

Any beneficial owners of Scheme Shares who are also holders of KW Shares and who wish to avoid the potential application of these rules should consult their tax advisers, including as to the desirability and tax consequences of disposing of some or all their Scheme Shares or KW Shares in the market prior to the Merger.

Any person (whether or not a holder of shares in KW or KWE) who is considering acquiring shares in KW or KWE prior to the Effective Date should consider whether to refrain from doing so if, as a result, it may become liable to additional US tax as a result of the requirements described above.

This summary is, of necessity, a simplification of numerous complex tax considerations. Beneficial owners of Scheme Shares and Scheme Shareholders should not rely on it as a comprehensive statement of the potential tax implications of their receipt of the KW Cash Component under the New Offer or the First KW Dividend and should in addition read and consider carefully the explanations in Part Seven (*Taxation*) of this document in full and seek independent tax advice on them.

20. Overseas Shareholders

Persons resident in, or citizens or nationals of, or otherwise subject to the law of, jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands and the Cayman Islands should refer to paragraph 33 of Part Two (*Explanatory Statement*) of this document.

21. Action to be taken

Your attention is drawn to the sections entitled "Important Notice about Voting on the Merger" commencing on page 12 of this document and "Action to be Taken" commencing on page 14 of this document.

The Scheme Court Sanction Hearing is scheduled to be held at 10.00 a.m. on 18 October 2017 at the Royal Court of Jersey, Royal Court Building, Royal Square, St Helier, JE1 1BA, Jersey, Channel Islands. Scheme Shareholders are entitled to attend and be heard at the Scheme Court Sanction Hearing.

22. Further information

You should ensure that you read the remainder of this document, including the explanatory statement from Rothschild and J.P. Morgan Cazenove contained in Part Two (*Explanatory Statement*) of this document. Please note that the information contained in the explanatory statement is in summary form only and reading the explanatory statement is not a substitute for reading the remainder of this document.

The KW Prospectus contains a description of KW, its business, strategy, financial condition and results of operations. KW draws the attention of KWE Shareholders to the risk factors described in Part 2 "Risk Factors" of the KW Prospectus. The materialisation of one or more of the risks described in the KW Prospectus may have a material adverse effect on KW's activities, assets, financial position, results or prospects, as well as on the market price of KW Shares

The KW Prospectus (the publication and content of which are the responsibility solely of the KW Directors) may be accessed free of charge at KW's website at <u>ir.kennedywilson.com</u> (subject to any restrictions relating to persons with a registered address in or who are a citizen, resident or national of certain jurisdictions). See also the "Important Notice about Voting on the Merger" on pages 12 to 13 of this document.

23. KW Officers and Employees

As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.

24. Recommendation

The KWE Independent Directors, who have been so advised by Rothschild and J.P. Morgan Cazenove as to the financial terms of the Original Offer and the New Offer, consider the terms of the New Offer to be fair and reasonable. In providing advice to the KWE Independent Directors, Rothschild and J.P. Morgan Cazenove have taken into account the commercial assessments of the KWE Independent Directors.

Accordingly, the KWE Independent Directors unanimously recommend Scheme Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and to choose to receive the New Offer Consideration and Distribution in the manner explained in this document and also recommend all KWE Shareholders to vote in favour of the Special Resolution at the KWE General Meeting but do not recommend them to elect to receive the Original Offer Consideration and also make no recommendation in relation to the Mix and Match Facility.

Simon Radford, the only KWE Independent Director who holds any KWE Shares, has irrevocably undertaken to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting, and in favour of the Special Resolution at the KWE General Meeting, in respect of his own beneficial holding of 12,500 Scheme Shares representing, in aggregate, approximately 0.01% of the existing issued ordinary share capital of KWE as at the Latest Practicable Date (further details of this irrevocable undertaking are contained in paragraph 16 of Part Two (Explanatory Statement) of this document).

For the avoidance of doubt, none of the KWE Independent Directors and neither Rothschild nor J.P. Morgan Cazenove expresses any view as to the advantages or disadvantages of any aspects of the Scheme as far as KW Shareholders are concerned.

Yours faithfully,

Charlotte Valeur
Chair and KWE Independent Non-Executive Director
Mark McNicholas
KWE Independent Non-Executive Director
Simon Radford
KWE Independent Non-Executive Director

PART TWO

EXPLANATORY STATEMENT

(in compliance with Article 126 of the Jersey Companies Law)



J.P.Morgan CAZENOVE

N.M. Rothschild and Sons Limited
New Court
St Swithin's Lane, London, EC4N 8AL
and
J.P. Morgan Limited
25 Bank Street
Canary Wharf
London
E14 5JP
13 September 2017

To KWE Shareholders

Dear Shareholder.

RECOMMENDED MERGER OF KENNEDY WILSON EUROPE REAL ESTATE PLC AND KENNEDY-WILSON HOLDINGS, INC.

1. Introduction

On 24 April 2017, KWE and KW published an announcement setting out the terms and conditions of a recommended offer by KW to acquire all of the outstanding shares of KWE (other than the Excluded Shares) in an all-share merger (the "Original Offer").

On 13 June 2017, KWE and KW published a further announcement setting out the terms and conditions of an additional offer by KW to acquire all of the outstanding shares of KWE (other than the Excluded Shares) (the "New Offer"). As the New Offer comprises fewer New KW Shares for each Scheme Share than are available under the Original Offer, it is necessary, under the requirements of the Code, for KW to continue to make the Original Offer available in parallel to the New Offer.

It is intended that the Merger will be effected by means of a Court-sanctioned scheme of arrangement of KWE under Article 125 of the Jersey Companies Law (or by a Takeover Offer under certain circumstances described in this document and subject to the consent of the Panel), and KW will undertake to the Court to be bound by and comply with the Scheme.

Your attention is drawn to the letter from the KWE Independent Directors set out in Part One (Letter from the Independent Directors of Kennedy Wilson Europe Real Estate Plc) of this document. That letter contains, among other things, information on the background to and reasons for the recommendation by the KWE Independent Directors to Scheme Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and to choose to receive the New Offer Consideration and Distribution and their recommendation to KWE Shareholders to vote in favour of the Special Resolution at the KWE General Meeting.

For the Scheme to become effective it will require the approval at the Scheme Court Meeting (and at any separate class meeting which may be required), or any adjournment thereof, by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the Register at the Voting Record Time, and present and voting (either in person or by proxy) at the Scheme Court Meeting, representing three-quarters or more of the votes attached to Scheme Shares which are cast at the Scheme Court Meeting (or at the meeting of each class of Scheme Shareholders, as the case may be).

Implementation of the Scheme will also require the approval of the Special Resolution as set out in the notice of the KWE General Meeting in Part Thirteen (*Notice of KWE General Meeting*) of this document by a majority of not less than two-thirds of votes cast by KWE Shareholders present and voting (and entitled to vote) in accordance with the KWE Articles, either in person or by proxy (including holders of Excluded Shares), at the KWE General Meeting.

As at the Latest Practicable Date, KW and its subsidiaries held approximately 23.8% of the issued ordinary shares of KWE. Any KWE Shares held by or on behalf of any member of the KW Group (including, for the avoidance of doubt, any KWE Shares held by or on behalf of the Manager) will be Excluded Shares (i.e. not subject to the Scheme) and no member of the KW Group (including the Manager), or its nominee, is entitled to vote at the Scheme Court Meeting in respect of the Excluded Shares held by it.

The 23.8% of KWE Shares held by KW and/or its subsidiary undertakings as at the Latest Practicable Date included 532,230 KWE Shares held for the purpose of satisfying RSUs which will vest in connection with the Merger. It is intended that these RSUs will vest in consequence of the Merger after the KWE General Meeting but before the Scheme Record Time and that the 532,230 KWE Shares underlying these RSUs (less a certain number of KWE Shares withheld by KW to satisfy tax withholding obligations in accordance with the terms of the underlying award agreements) will be transferred to RSU holders (in accordance with the terms of the underlying award agreements) before the Scheme Record Time, i.e. in time for the relevant KWE Shares to become Scheme Shares and to be acquired by KW pursuant to the Scheme. However, irrespective of whether RSU holders or KW or any of its subsidiaries are holders of record of any KWE Shares to which these arrangements apply at any time when votes may be cast or proxies appointed for the Scheme Court Meeting, KW has undertaken to KWE to take all reasonable steps to procure that no proxies are appointed and no votes are cast at the Scheme Court Meeting in respect of any such underlying KWE Shares. However, votes attached to these underlying KWE Shares may be freely cast at the KWE General Meeting and Elections in respect of the underlying KWE Shares may be made prior to the Election Return Time.

As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.

Under the Confidentiality Agreement, KW has also agreed that, other than KWE Shares acquired pursuant to the IMA, neither it nor any person acting in concert with it may acquire any interest in any KWE Shares without the prior written consent of the KWE Independent Directors. Please see paragraph 30(a) of this Part Two for more details.

We have been authorised by the KWE Independent Directors to write to you to explain the terms of the Merger and to provide you with other relevant information.

The terms of the Scheme are set out in full in Part Four (*The Scheme of Arrangement*) of this document. Your attention is also drawn to the further information contained in this document which forms part of this explanatory statement.

The Merger is subject to the Conditions being satisfied or (to the extent capable of waiver) waived and to the further terms set out in Part Three (Conditions and Certain Further Terms of the Scheme) of this document.

Although the planned date for the Scheme to become effective is 20 October 2017, KW and KWE have agreed to extend the Longstop Date to 30 November 2017 to allow time to deal with any unforeseen delay.

Scheme Shareholders should read the whole of this document before deciding whether or not to vote in favour of the Scheme at the Scheme Court Meeting and whether or not to vote in favour of the Special Resolution at the KWE General Meeting.

2. Terms of the New Offer

Under the terms of the Merger, which is subject to the Conditions and the other terms set out in Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document, if the Scheme becomes effective, a Scheme Shareholder which does not submit a valid Election in the manner set out in this document will be entitled to receive:

In respect of each Scheme Share:

0.3854 New KW Shares (the "New KW Share Component"); and

300 pence in cash, paid by KW (the "KW Cash Component"); and

250 pence in cash, paid by KWE in the form of a special distribution pursuant to the Scheme (the "KWE Special Distribution") (collectively, the "New Offer Consideration and Distribution").

The KWE Special Distribution will be payable by KWE shortly after the Effective Date to Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution. It will be paid in addition to all other dividends payable by KWE, including the Closing Dividend, and dividends paid by KW referred to in this document.

KWE seeks to record its real estate assets in its accounts at their fair value as derived from half yearly valuations. As at 30 June 2017, the accounting value of KWE's net assets, as recorded in its interim accounts for the six months ended 30 June 2017, was £1,567.9 million. If the maximum amount of £241.6 million potentially payable by KWE in respect of the KWE Special Distribution had been paid on 30 June 2017, KWE's net asset value would have been approximately £1,326.3 million as at 30 June 2017.

The attention of beneficial owners of Scheme Shares and Scheme Shareholders is drawn to Part Seven (*Taxation*) of this document. The tax implications of the Merger will depend on the individual circumstances of each beneficial owner of Scheme Shares and Scheme Shareholders. Beneficial owners of Scheme Shares and Scheme Shareholders should consult their own tax advisers, including as to the tax consequences of disposing of their shares pursuant to the Scheme or in the market. Beneficial owners of Scheme Shares and Scheme Shareholders who believe that the tax consequences of the Scheme may be unattractive to them should seek independent professional advice on the courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme).

Based on the Closing Price of a KW Share of US\$19.15 and the US\$ / £ exchange rate of US\$1.3209:£1, in each case at the Latest Practicable Date, the New Offer Consideration and Distribution valued each Scheme Share at approximately 1,109 pence and KWE's entire issued share capital at approximately £1.40 billion. This represents:

- a premium of approximately 13.3% to the Closing Price of 979 pence of one KWE Share on 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- a premium of approximately 15.5% to the volume weighted average price of 960 pence of one KWE Share for the three month period that ended on 21 April 2017 (the last Business Day before the commencement of the Offer Period); and
- a discount of approximately 10.7% to KWE's last reported Adjusted NAV of 1,241 pence of one KWE Share as at 30 June 2017.

Following the completion of the Merger, if all Scheme Shareholders received the New Offer Consideration and Distribution and no Scheme Shareholders received the Original Offer Consideration, former KWE Shareholders (other than KW and other holders of Excluded Shares) would own approximately 25% of the Combined Group and existing KW Shareholders would own approximately 75% of the Combined Group.

3. Mix and Match Facility

Under the Mix and Match Facility, Scheme Shareholders (other than certain Overseas Shareholders) entitled to receive the New Offer Consideration and Distribution may elect to vary the proportions in which they receive the New KW Share Component and the KW Cash Component in respect of each Scheme Share held, subject to the availability of valid countervailing elections under the Mix and Match Facility by other Scheme Shareholders. The

Mix and Match Facility will remain available until the Election Return Time, i.e. 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service). The Mix and Match Facility is not available in relation to the Original Offer.

The Mix and Match Facility will not change the aggregate number of New KW Shares issued by KW or the aggregate amount of the KW Cash Component paid by KW pursuant to the New Offer. To the extent that Mix and Match Elections for the New KW Share Component or the KW Cash Component cannot be satisfied in full due to an insufficiency of valid countervailing elections, they will be scaled down on a *pro rata* basis in accordance with Clause 6.7 and Clause 6.8 of the Scheme. As a result, Scheme Shareholders who make an Additional Share Election and/or an Additional Cash Election will not know the exact amount of the KW Cash Component or additional number of New KW Shares they are entitled to receive until settlement of the consideration under the Merger.

The Mix and Match Facility will not enable any Scheme Shareholder to vary the amount of the KWE Special Distribution receivable under the New Offer. It will also not enable any Scheme Shareholder to vary the amount of cash receivable in respect of the Closing Dividend, although a Mix and Match Election which results in a change in the number of New KW Shares received by a Scheme Shareholder will, as a consequence, alter the amount of dividends receivable from KW after the Effective Date and consequently alter the amount receivable in respect of the Closing Dividend, as explained more fully in paragraph 5 of this Part Two.

A Mix and Match Election will be of no effect to the extent that it is made in respect of a Scheme Share for which a valid Original Offer Election has been received and not validly withdrawn.

A Scheme Shareholder's entitlement to receive additional cash or shares under the Mix and Match Facility will depend on valid countervailing Mix and Match Elections being made by other Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution.

A Mix and Match Election may only be made in respect of a whole number of Scheme Shares. Elections under the Mix and Match Facility may be made on the following basis:

OR

for every 300 pence of the KW Cash Component

0.2069 of a New KW Share

for every 0.3854 of a New KW Share

559 pence of the KW Cash Component.

SCHEME SHAREHOLDERS SHOULD NOTE THAT:

- THE MIX AND MATCH RATIOS SPECIFIED ABOVE ARE BASED ON THE CLOSING PRICE OF A KW SHARE OF US\$19.15 AND THE EXCHANGE RATE OF US\$1.3209:£1, IN EACH CASE AS AT THE LATEST PRACTICABLE DATE.
- THE CLOSING PRICES OF KW SHARES AND EXCHANGE RATES MAY FLUCTUATE BEFORE AND AFTER THE ELECTION RETURN TIME.

If following a Mix and Match Election a Scheme Shareholder would be due to receive a fraction of a New KW Share, paragraph 8 of this Part Two will apply and such Scheme Shareholder will be entitled to receive an amount in cash rounded to the nearest cent, based on the amount obtained by multiplying such fraction by the average of the highest and lowest sales prices of KW Shares on the NYSE on each of the five consecutive trading days ending on the trading day that is two trading days prior to the Effective Date, except that individual entitlements of less than US\$5.00 will not be paid but will be retained for the benefit of the Combined Group. Fractional entitlements will be determined after allowing for the outcome of Mix and Match Elections and Original Offer Elections.

The table below shows, for illustrative purposes only, the possible outcomes for a Scheme Shareholder who holds one Scheme Share and, pursuant to the Mix and Match Facility, makes: (i) a valid Additional Cash Election; (ii) a valid Additional Share Election; or (iii) does not make (or is deemed not to have made) a valid Mix and Match Election. The table also assumes that the Closing Price of a KW Share is US\$19.15 and a US\$ / £ exchange rate of US\$1.3209:£1 (which equate to those values as at the Latest Practicable Date):

Election	KW Cash Component (pence) ⁽¹⁾	New KW Shares ⁽¹⁾	KWE Special Distribution (pence)
Additional Cash Election	859	_	250
Additional Share Election	_	0.5923	250
No Election	300	0.3854	250

Note:

An announcement will be made as soon as reasonably practicable after the Effective Date concerning the extent to which Mix and Match Elections will be satisfied.

A Scheme Shareholder who is entitled to receive the New Offer Consideration and Distribution and who makes a valid Additional Cash Election under the Mix and Match Facility in respect of all that Scheme Shareholder's Scheme Shares may nonetheless receive some New KW Shares as consideration under the Merger if the Additional Cash Elections exceed valid countervailing Additional Share Elections. A Scheme Shareholder who is entitled to receive the New Offer Consideration and Distribution and who makes a valid Additional Share Election in respect of all of the Scheme Shareholder's Scheme Shares may still receive some of the KW Cash Component as consideration under the Merger if the Additional Share Elections exceed valid countervailing Additional Cash Elections.

The Mix and Match Facility will not affect the entitlements of those Scheme Shareholders who do not make a valid Mix and Match Election under the New Offer or who elect to receive the Original Offer Consideration in respect of their entire holdings of Scheme Shares.

Further details on how and when a Scheme Shareholder can make a Mix and Match Election are set out in Part Five (*Notes for Making Elections and Submitting IRS Forms W-8 or W-9*) of this document.

A Mix and Match Election will be of no effect to the extent that it is submitted in respect of a Scheme Share for which a valid Original Offer Election has been received and not validly withdrawn.

The availability of the Mix and Match Facility to certain Overseas Shareholders may be restricted or prohibited by law. Overseas Shareholders should read paragraph 33 of this Part Two and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document in relation to their ability to make a Mix and Match Election.

4. Terms of the Original Offer

The Original Offer will be made available to Scheme Shareholders (other than certain Overseas Shareholders) on the terms and subject to the Conditions set out in Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document. Each Scheme Shareholder validly electing for the Original Offer will be entitled to receive:

In respect of each Scheme Share: 0.667 New KW Shares.

Scheme Shareholders who elect to receive the Original Offer in respect of their entire holdings of Scheme Shares will not be entitled to receive the New KW Share Component, the KW Cash Component or the KWE Special Distribution available under the New Offer and will not be entitled to make any Election under the Mix and Match Facility. However, the dividend arrangements described in paragraph 5 of this Part Two below, including the Closing Dividend, will still apply.

⁽¹⁾ Figures shown in this table are rounded for illustration purposes, and do not take into account the effect of the provisions regarding fractional entitlements set out in this Part Two. Actual amounts may vary.

Based on the Closing Price of a KW Share of US\$19.15 and the US\$ / £ exchange rate of US\$1.3209:£1, in each case at the Latest Practicable Date, the terms of the Original Offer valued each Scheme Share at approximately 967 pence and KWE's entire issued share capital at approximately £1.22 billion. This represents:

- a discount of approximately 1.2% to the Closing Price of 979 pence of one KWE Share on 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- a premium of approximately 0.7% to the volume weighted average price of 960 pence of one KWE Share for the three month period that ended on 21 April 2017 (the last Business Day before the commencement of the Offer Period); and
- a discount of approximately 22.1% to KWE's last reported Adjusted NAV of 1,241 pence of one KWE Share as at 30 June 2017.

If all Scheme Shareholders received the Original Offer Consideration and no Scheme Shareholders received the New Offer Consideration and Distribution, former KWE Shareholders (other than KW and other holders of Excluded Shares) would own approximately 36% of the Combined Group and existing KW Shareholders would own approximately 64% of the Combined Group.

The availability of the Original Offer to certain Overseas Shareholders may be restricted or prohibited by law. Further details are set out in paragraph 33 of Part Two (Explanatory Statement) and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

As at the Latest Practicable Date, the Original Offer valued each Scheme Share at 967 pence compared with 1,109 pence under the New Offer on the basis of the values, prices and exchange rates referred to above. As at the Latest Practicable Date, the value of the Original Offer on a pre-tax basis was therefore materially less than the value of the New Offer on a pre-tax basis. The KW Share price would need to rise from US\$19.15 as at the Latest Practicable Date to US\$25.82 (i.e. an increase of over approximately 34%) in order for the Original Offer to represent greater value than the New Offer on a pre-tax basis (assuming no Mix and Match Election and a US\$ / £ exchange rate of US\$1.3209:£1).

5. Dividends

KW and KWE have the same quarterly dividend periods. KW and KWE have agreed arrangements under which each KWE Shareholder (including those who are not Scheme Shareholders) will receive the usual KWE quarterly dividends payable on their KWE Shares, subject to certain adjustments designed to avoid duplicated payments which would otherwise result from the First KW Dividend being paid in respect of a period which includes part of the Final KWE Dividend Period. These arrangements apply in the event that the Scheme becomes effective and the KWE Special Distribution is paid if, when and to the extent required under the Scheme with the result that the remaining arrangements under the Scheme also have effect.

The Final KWE Dividend will accrue in the usual way up to, but excluding, the Effective Date, on the basis that the full amount of the Final KWE Dividend payable to KWE Shareholders will be reduced to be the Pre-Effective Date Proportion of the full amount of the Final KWE Dividend (i.e. the amount of that dividend that accrues from and including the first day of the Final KWE Dividend Period to, but excluding, the Effective Date). As a result of this adjustment, the amount of the Final KWE Dividend of 12 pence per KWE Share will be reduced *pro rata* based on the number of days between (and including) the first day of the Final KWE Dividend Period up to (but excluding) the Effective Date (as a proportion of the full number of days in the Final KWE Dividend Period).

The amount of the Final KWE Dividend, as so adjusted, will be further reduced, in respect of each KWE Share which is a Scheme Share, by an amount equal to the pound sterling equivalent of the Pre-Effective Date Proportion of the First KW Dividend receivable on New KW Shares issued in exchange for such Scheme Share (i.e. the amount of the First KW Dividend that accrues from and including the first day of the Final KWE Dividend Period up to (but excluding) the Effective Date).

For some beneficial owners of KWE Shares and KWE Shareholders who are not US taxpayers, the First KW Dividend (as well as subsequent KW dividends) may be subject to US federal withholding tax at a rate of up to 30%, as explained below under "Taxation Information". As a

result, for those beneficial owners of KWE Shares and KWE Shareholders, the deduction made in respect of the gross amount of the First KW Dividend will reduce the net sum they receive in respect of the Closing Dividend to less than would have been received if those beneficial owners of KWE Shares and KWE Shareholders had received dividends only from KWE.

KWE dividends are payable in pounds sterling while KW dividends are payable in US Dollars. Accordingly, the extent to which the Pre-Effective Date Proportion of the First KW Dividend will reduce the Pre-Effective Date Proportion of the Final KWE Dividend will be determined by translating the US Dollar amount of the First KW Dividend into pounds sterling using the Applicable Translation Rate. The portion of the Final KWE Dividend payable after these adjustments is referred to as the "Adjusted Final KWE Dividend". The Adjusted Final KWE Dividend will be rounded to the second decimal place.

Both the Adjusted Final KWE Dividend and the First KW Dividend are expected to be paid on the same day, being the First KW Dividend Payment Date (i.e. not later than 10 days after the end of the First KW Dividend Period).

Illustration for the New Offer

The following illustration assumes that a Scheme Shareholder is entitled to receive the New Offer Consideration and Distribution and makes no Mix and Match Election, and is therefore entitled to receive 0.3854 New KW Shares for each Scheme Share (in addition to the KW Cash Component and the KWE Special Distribution). It also assumes that the First KW Dividend is US\$0.19 per KW Share (which KW has stated its intent to pay); that the Effective Date is exactly half way through the Final KWE Dividend Period (and thus exactly half way through the First KW Dividend Period); that the Applicable Translation Rate is US\$1.3209:£1 (i.e. the Applicable Translation Rate as at the Latest Practicable Date); and that the Final KWE Dividend for the full Final KWE Dividend Period would be £0.12 if the Effective Date were the final date of that period. These assumptions permit the following illustration of the effects of the New Offer:

- (i) Final KWE Dividend: £0.12.
- (ii) Pre-Effective Date Proportion of the Final KWE Dividend (i.e. 50% of £0.12): £0.06 per KWE Share.
- (iii) First KW Dividend: US\$0.19 per New KW Share (before reduction for up to 30% US federal withholding tax, if applicable).
- (iv) Pre-Effective Date Proportion of the First KW Dividend (i.e. 50% of US\$0.19): US\$0.095 per New KW Share.
- (v) £ equivalent of (iv) above i.e. $\frac{0.095}{1.3209}$ = £0.0719 per New KW Share.
- (vi) Proportion of (v) above which is payable per KWE share on the basis of 0.3854 New KW shares per KWE Share (i.e. £0.0719 x 0.3854): £0.0277 per KWE Share.
- (vii) Adjusted Final KWE Dividend (i.e. (ii) above less (vi) above): £0.0323 per KWE Share, rounded to £0.03 per KWE Share.
- (viii) Closing Dividend (i.e. Adjusted Final KWE Dividend per KWE Share (£0.03) plus Penultimate KWE Dividend per KWE Share (£0.12)): £0.15.

Illustration for the Original Offer

The following illustration assumes that a Scheme Shareholder elects to receive the Original Offer Consideration, and is therefore entitled to receive 0.667 New KW Shares for each Scheme Share (and will not receive the KW Cash Component or the KWE Special Distribution in respect of that Scheme Share). It also assumes that the First KW Dividend is US\$0.19 per KW Share (which KW has stated its intent to pay); that the Effective Date is exactly half way through the Final KWE Dividend Period (and thus exactly half way through the First KW Dividend Period); that the Applicable Translation Rate is US\$1.3209:£1 (i.e. the Applicable Translation Rate as at the Latest Practicable Date); and that the Final KWE Dividend for the full Final KWE Dividend Period would be £0.12 if the Effective Date were the final date of that period. These assumptions permit the following illustration of the effects of the Original Offer:

- (i) Final KWE Dividend: £0.12.
- (ii) Pre-Effective Date Proportion of the Final KWE Dividend (i.e. 50% of £0.12): £0.06 per KWE Share.

- (iii) First KW Dividend: US\$0.19 per New KW Share (before reduction for up to 30% US federal withholding tax, if applicable).
- (iv) Pre-Effective Date Proportion of the First KW Dividend (i.e. 50% of US\$0.19): US\$0.095 per New KW Share.
- (v) £ equivalent of (iv) above i.e. $\frac{0.095}{1.3209}$ = £0.0719 per New KW Share (i.e. a greater amount than under the New Offer).
- (vi) Proportion of (v) above which is payable on each KWE share on the basis of 0.667 New KW shares per Scheme Share (i.e. £0.0719 x 0.667): £0.0480 per KWE Share.
- (vii) Adjusted Final KWE Dividend (i.e. (ii) above less (vi) above): £0.0120 per KWE Share, rounded to £0.01 per KWE Share (i.e. a smaller amount than under the New Offer).
- (viii) Closing Dividend (i.e. Adjusted Final KWE Dividend per KWE Share (£0.01) plus Penultimate KWE Dividend per KWE Share (£0.12)): £0.13.

Where, pursuant to the Scheme, a Scheme Shareholder reduces the number of New KW Shares to be received in respect of that Scheme Shareholder's Scheme Shares (e.g. by choosing to receive the New Offer Consideration and Distribution in preference to the Original Offer or by electing for more cash under the Mix and Match Facility), the aggregate amount of the First KW Dividend receivable by that Scheme Shareholder for each Scheme Share will also correspondingly reduce (because the Scheme Shareholder will be receiving fewer New KW Shares). As a result, the amount by which the Adjusted Final KWE Dividend payable in respect of such Scheme Shares is reduced to take account of receipt of the First KW Dividend will also be correspondingly less (i.e. the amount of the Scheme Shareholder's Adjusted Final KWE Dividend will be correspondingly larger).

Accordingly, if a Scheme Shareholder entitled to the New Offer Consideration and Distribution were to become entitled to receive only cash in respect of one Scheme Share as a result of a Mix and Match Election, neither the Adjusted Final KWE Dividend nor the Penultimate KWE Dividend would be reduced to take account of the payment of the First KW Dividend because that Scheme Shareholder would receive no New KW Shares, and accordingly, no First KW Dividend would be payable to that Scheme Shareholder in consequence of the transfer to KW of the relevant Scheme Share pursuant to the Scheme.

Similarly, any KWE Shareholder holding KWE Shares which are not Scheme Shares will not be entitled to any New KW Shares under the Scheme or under an Election, and will therefore not receive a dividend from KW and such a KWE Shareholder will therefore receive the Penultimate KWE Dividend and the Pre-Effective Date Proportion of the Final KWE Dividend on the First KW Dividend Payment Date without any reduction referable to any dividend paid by KW.

Conversely, where an election for the Original Offer or under the Mix and Match Facility increases the New KW Shares received for a Scheme Share, the amount of the First KW Dividend receivable in respect of that Scheme Share will also correspondingly increase. As a result, the amount by which the Adjusted Final KWE Dividend payable in respect of such Scheme Share is reduced to take account of receipt of the First KW Dividend will also be correspondingly greater (i.e. the amount which that Scheme Shareholder will receive from KWE in respect of the Adjusted Final KWE Dividend will be correspondingly smaller).

The "Closing Dividend" will comprise the Adjusted Final KWE Dividend and the Penultimate KWE Dividend. The Closing Dividend will be payable by KWE on the First KW Dividend Payment Date (i.e. it will be payable on the same date as the First KW Dividend), which shall be not later than 10 days after the end of the First KW Dividend Period (i.e. not later than 10 January 2018 if, as expected, the Effective Date falls in 2017). The Closing Dividend will be paid to all KWE Shareholders who are on the Register at the Closing Dividend Record Time.

If the value of the US Dollar were to strengthen very considerably against the pound sterling and such strengthening, either alone or in combination with an increase in the number of New KW Shares received as a result of a valid Election in respect of the Mix and Match Facility or the Original Offer, were to result in the Pre-Effective Date Proportion of the First KW Dividend receivable by a Scheme Shareholder reducing the Adjusted Final KWE Dividend below zero, the deficit would be applied to reduce the Penultimate KWE Dividend, on the basis that the Pre-Effective Date Proportion of the Final KWE Dividend is added to the full amount of the Penultimate KWE Dividend to establish the full amount of the Closing Dividend. A similar outcome would result if the First KW Dividend were unexpectedly substantially greater than

US\$0.19 per KW Share. However, assuming dividends paid by KW in respect of the First KW Dividend Period do not exceed US\$0.19 per KW Share, for a Scheme Shareholder who becomes entitled to receive the New Offer Consideration and Distribution but makes no Mix and Match Election, the Applicable Translation Rate would have to change from US\$1.3209:£1 (the Applicable Translation Rate as at the Latest Practicable Date) to approximately US\$0.60:£1 in order for the Adjusted Final KWE Dividend to be reduced to below zero and for the Penultimate KWE Dividend to be reduced in this way.

The amount of the Closing Dividend (i.e. the total of the Adjusted Final KWE Dividend and the Penultimate KWE Dividend) can never be reduced in this way to less than zero.

Illustration for New Offer and Mix and Match Election

The following illustration assumes that a Scheme Shareholder becomes entitled to the New Offer Consideration and Distribution and makes an Additional Share Election which is satisfied in full, entitling it to receive 0.5923 New KW Shares, but no KW Cash Component, for each Scheme Share held (in addition to the KWE Special Distribution). It also assumes that the First KW Dividend is US\$0.19 per KW Share (which KW has stated its intent to pay); that the Effective Date is exactly half way through the Final KWE Dividend Period (and thus exactly half way through the First KW Dividend Period); that the Applicable Translation Rate is US\$1.3209:£1 (i.e. the Applicable Translation Rate as at the Latest Practicable Date); and that the Final KWE Dividend for the full Final KWE Dividend Period would be £0.12 if the Effective Date were the final date of that period. These assumptions permit the following illustration of the effects of the New Offer and the Mix and Match Election described above:

- (i) Final KWE Dividend: £0.12.
- (ii) Pre-Effective Date Proportion of the Final KWE Dividend (i.e. 50% of £0.12): £0.06 per KWE Share.
- (iii) First KW Dividend: US\$0.19 per New KW Share (before reduction for up to 30% US federal withholding tax, if applicable).
- (iv) Pre-Effective Date Proportion of the First KW Dividend (i.e. 50% of US\$0.19): US\$0.095 per New KW Share.
- (v) £ equivalent of (iv) above i.e. $\frac{0.095}{1.3209}$ = £0.0719 per New KW Share.
- (vi) Proportion of (v) above which is payable per KWE Share on the basis of 0.5923 New KW Shares per KWE Share (i.e. £0.0719 x 0.5923): £0.0426 per KWE Share.
- (vii) Adjusted Final KWE Dividend (i.e. (ii) above less (vi) above): £0.0174 per KWE Share, rounded to £0.02 per KWE Share.
- (viii) Closing Dividend (i.e. Adjusted Final KWE Dividend per KWE Share (£0.02) plus Penultimate KWE Dividend per KWE Share (£0.12)): £0.14.

The above illustration shows that, at a constant US\$ / £ translation rate, a full Additional Share Election under the Mix and Match Facility will not, on its own, cause the Pre-Effective Date Proportion of the First KW Dividend receivable by the Scheme Shareholder to reduce the Adjusted Final KWE Dividend below zero (and, in doing so, to reduce the amount of the Penultimate KWE Dividend).

KWE Quarterly Dividends before the Penultimate KWE Dividend

In addition to the Closing Dividend, KW has agreed that KWE Shareholders will be entitled to receive any quarterly dividend declared and paid by KWE in amounts of up to £0.12 per KWE Share in aggregate in respect of any KWE Dividend Period expiring before the start of the Penultimate KWE Dividend Period. Accordingly, as announced on 27 June 2017 by KWE, on 31 August 2017, KWE paid KWE Shareholders who were on the Register on 18 August 2017 its usual quarterly dividend of £0.12 per KWE Share in respect of the KWE Dividend Period ended on 30 June 2017.

Taxation information relevant to dividends

The attention of beneficial owners of KWE Shares and KWE Shareholders is drawn to Part Seven (*Taxation*) of this document. Beneficial owners of KWE Shares and KWE Shareholders should note that distributions paid on KW Shares to a beneficial owner of KW Shares or a KW Shareholder (including a former beneficial owner of KWE Shares or KWE Shareholder

who receives New KW Shares pursuant to the Merger) that is not a "United States person" for US federal income tax purposes generally will be subject to US withholding tax at a rate of 30% of the gross amount of the distribution (or a reduced rate prescribed by an applicable double tax treaty if the beneficial owner of KW Shares or KW Shareholder is entitled to claim and makes a valid claim under such treaty). Accordingly, the net amount received in aggregate by a beneficial owner of KWE Shares or KWE Shareholder as a result of the combination of (i) the Closing Dividend calculated in the manner set out in this document and (ii) the First KW Dividend may not be the same as if such beneficial owner of KWE Shares or KWE Shareholder had only received dividends from KWE in respect of the same aggregate period.

Beneficial owners of KWE Shares and KWE Shareholders who believe that they may prefer alternative tax treatments to those described in Part Seven (*Taxation*) should consider seeking independent professional advice on alternative courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme) in the event that KWE were to be treated as a partnership for US tax purposes.

6. Permitted KWE Distributions

- 6.1 Under the terms of the Scheme, the term "Permitted KWE Distributions" refers to Ordinary Quarterly Dividends, the KWE Special Distribution and the Closing Dividend. Subject to paragraph 6.3 below, if, on or after 24 April 2017, any dividend, other distribution or other return of capital is declared, made or paid or becomes payable in respect of KWE Shares by reference to a record date falling prior to the Effective Date in excess of the Permitted KWE Distributions, KW shall have the right to require the reduction of:
 - (1) in the case of the New Offer:
 - (a) the KW Cash Component and, to the extent necessary, the KWE Special Distribution, by a cash amount equal to the gross amount of such dividend, other distribution or other return of capital; and/or
 - (b) the New KW Share Component by a number of New KW Shares having a value equal to not more than the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital);
 - (2) in the case of the Original Offer, the Original Offer Consideration by a number of New KW Shares having a value equal to not more than the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital),
 - in which case (i) any reference to the New Offer Consideration and Distribution and the Original Offer Consideration in this document shall be deemed to be a reference to such New Offer Consideration and Distribution or Original Offer Consideration as so reduced and (ii) KWE Shareholders will be entitled to receive and retain the amount by reference to which the consideration or distribution has been reduced.
- 6.2 Any adjustment to the New Offer Consideration and Distribution and the Original Offer Consideration by KW in accordance with the above paragraph shall be the subject of an announcement made by KW via a Regulatory Information Service.
- 6.3 To the extent that any dividend, distribution or other return of capital is declared, made or paid or is payable in respect of KWE Shares by reference to a record date falling prior to the Effective Date in excess of the Permitted KWE Distributions and it is: (i) transferred pursuant to the Merger on a basis which entitles KW to receive the dividend or distribution and to retain it; or (ii) cancelled prior to payment, KW shall not be entitled to reduce the New Offer Consideration and Distribution or the Original Offer Consideration, notwithstanding paragraph 6.1.

7. Permitted KW Dividends

7.1 Under the terms of the Scheme, the term "**Permitted KW Dividends**" refers to quarterly dividends in amounts not exceeding, in aggregate, US\$0.17 per KW Share with respect to any KW Dividend Period preceding the First KW Dividend Period and US\$0.19 per KW Share in

respect of the First KW Dividend Period which, in any such case, are payable by reference to a record date preceding the Effective Date, and any dividend paid by KW on or after the Effective Date to which Scheme Shareholders are entitled pursuant to the Scheme. Other than a Permitted KW Dividend or an on-market share buyback of KW Shares at a price no greater than the then prevailing market price, KW will not, from the date of this document until after the Effective Date, effect or announce any intention to effect any distribution, share redemption or capital reduction, whether in or in exchange for cash or other assets.

- 7.2 Without prejudice to paragraph 7.1, if, on or after 24 April 2017, any dividend, other distribution or other return of capital (other than a Permitted KW Dividend or an on-market share buyback of KW Shares at a price no greater than the then prevailing market price) is declared, made or paid or becomes payable in respect of KW Shares, KW shall be required to:
 - (1) in the case of the New Offer, increase the New KW Share Component; or
 - (2) in the case of the Original Offer, increase the Original Offer Consideration,

in each case by a number of New KW Shares to the extent necessary to ensure that the additional value from such increase is equal to the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital, less the gross amount of such dividend, other distribution or other return of capital) and in which case any reference to the New KW Share Component and the Original Offer Consideration in this document shall be deemed to be a reference to such consideration as so increased.

7.3 Any increase in the consideration available under the Original Offer and the New Offer by KW in accordance with the above paragraphs shall be the subject of an announcement made by KW via a Regulatory Information Service. An increase in the consideration available under the New Offer will not alter the terms (in particular the ratio) of the Mix and Match Facility.

8. Fractional Entitlements

Fractions of New KW Shares will not be issued to Scheme Shareholders. Instead, Scheme Shareholders who would otherwise have received a fraction of a New KW Share will receive an amount in cash rounded to the nearest cent, based on the amount obtained by multiplying such fraction by the average of the highest and lowest sales prices of KW Shares on the NYSE on each of the five consecutive trading days ending on the trading day that is two trading days prior to the Effective Date, except that individual entitlements of less than US\$5.00 will not be paid but will be retained for the benefit of the Combined Group. For the purposes of determining fractional entitlements, each portion of a Scheme Shareholder's holding recorded in the Register by reference to a separate designation at the Scheme Record Time, whether in certificated or uncertificated form, shall be treated as a separate holding. Fractional entitlements will be determined after allowing for the outcome of Mix and Match Elections and Original Offer Elections.

9. Background to and reasons for the Merger

The KW Board has stated that it believes that a combination of KW and KWE will create a leading global real estate investment and asset management company with enhanced scale, diversification, liquidity and access to capital, as follows:

The Merger creates a leading real estate investment and asset management platform

 Based on the terms of the New Offer, the Combined Group would have a market capitalisation of approximately US\$3.0 billion and an enterprise value of approximately US\$8.0 billion and a portfolio of almost 400 properties.

Enhanced portfolio diversification

• As at 30 June 2017, the Combined Group would have a global portfolio located across the Western United States (approximately 42%), the United Kingdom (approximately 31%), Ireland (approximately 20%), Spain (approximately 5%), Italy (approximately 3%) and Japan (approximately 0.2%). Additionally, the Combined Group would have broad diversification across real estate sectors including: office (approximately 36%), multifamily (approximately 25%), retail (approximately 21%), hotels (approximately 7%), industrial

(approximately 3%) and loans, residential & other (approximately 8%). All percentages in this paragraph are percentages of pro forma carrying equity value of property as at 30 June 2017.

Flexibility to allocate capital across asset classes and markets that offer attractive risk-adjusted returns

• KW's approximately 475 employees in the US and approximately 97 employees in Europe provide the knowledge base to continue to make investment decisions that offer attractive risk-adjusted returns on capital. Additionally, the enhanced scale and profile of the Combined Group may result in enhanced access to capital and an expanded set of acquisition and development opportunities. KW's capital allocation flexibility is further supplemented by its ability to use its balance sheet for attractive long-term investment opportunities and take advantage of shorter-term opportunities via funds and partnerships with third parties.

Strong pro forma capital structure to support growth

Based on the New Offer, the Combined Group would have a strong pro forma capital structure, with US\$658 million of available cash and liquidity (as at 30 June 2017). After completion of the Merger, the Combined Group will look to generate capital and liquidity to fund growth through a variety of sources, including asset sales (with approximately US\$500 million of additional sales targeted by July 2018), development activities (with approximately US\$20 million in estimated incremental net operating income to be generated from development projects by the end of 2018) and additional cashflow arising from full ownership of KWE. The Combined Group would also benefit from well-staggered debt maturities, with approximately 30% of debt maturing over the next five years, and access to diverse, global equity and debt capital sources. As a result of predominantly fixed-rate debt, the Combined Group would be well-positioned in the event of a rise in interest rates.

Earnings per share accretion, enhanced cash flows and growing dividend

- The KW Board believes that the Merger will be accretive to adjusted net income per share immediately following completion of the Merger.¹
- The KW Board expects the Merger to generate certain synergies resulting from the elimination of duplicative public company costs in relation to KWE, elimination of KWE's revolving credit facility, potential cost of capital improvements related to the differential between US and European interest rates, and potential additional income arising from the Combined Group's ability to manage capital more efficiently as one group, which is expected to result in expanded capacity for investment to drive growth.
- On a pro forma basis, the weighted average ownership of real estate assets held as at 30 June 2017 would increase from 39% to 62% following completion of the Merger, demonstrating KW's continued focus on growing recurring cash flows from owned real estate assets. Based on results from the first half of 2017, full ownership of KWE would increase operating cash flows (post incremental debt service) by approximately US\$98 million per annum. Additionally, as KW already (indirectly) manages KWE, the majority of the additional cash flow translates directly to bottom line profitability. The Combined Group's strengthened cash flow profile would enable KW to enhance potential future capital returns to shareholders. Earlier this year, KW announced a 21% increase in its common dividend per share to US\$0.17 per quarter or US\$0.68 on an annualised basis and, in connection with the Merger, KW has announced that the KW Board intends to increase the first quarterly dividend payable following the Effective Date from US\$0.17 per KW Share to US\$0.19 per KW Share. Thereafter, the KW Board plans to continue to evaluate the Combined Group's dividend policy on a quarterly basis, with the overall aim of maintaining strong dividend coverage.

¹ Unless otherwise specifically stated, no statement in this document is intended as a profit forecast or estimate for any period and no statement in this document should be interpreted to mean that adjusted earnings per share for KW, KWE or the Combined Group, as appropriate, for the current or future financial years would necessarily match or exceed the historical published adjusted net income per share for KW, KWE or the Combined Group, as appropriate.

 Additionally, KW intends to evaluate the most efficient tax structure for the Combined Group over time, which may enhance cash flows to shareholders in the future. This evaluation will take into account near-term expected changes to US tax laws and may result in, among other things, maintaining C-corporation status or a potential conversion into a real estate investment trust.

Continuity of leadership team with a strong track record

- Since its initial public offering, KWE has been managed by KW's leadership team, using KW's platform and systems. There will be no change to systems as a result of the Merger, which should minimise integration risk and disruption to the business and ensure KW's high performance culture is maintained.
- KW's leadership team has a strong track record spanning several decades. Since becoming a public company on the NYSE in November 2009 until 30 June 2017, KW has raised US\$14 billion of public and private equity and acquired over US\$19 billion of assets globally. During that same period, KW has generated a 27% internal rate of return or a 1.9x equity multiple on its realised real estate investments.

Following completion of the Merger, if all Scheme Shareholders received the New Offer Consideration and Distribution, KW's Board and management would be expected to own approximately 14.5% of the shares in the Combined Group, which the KW Board believes creates a strong alignment with shareholders.

10. KW's intentions regarding the business, employees and pension schemes of KWE and KW

10.1 KW Statement

The KW Directors believe that the Merger will create a leading global real estate investment and asset management company with enhanced scale, diversification, liquidity and access to capital.

The KW Board recognises that, in order to achieve the expected benefits of the Merger, operational and administrative restructuring may be required following completion of the Merger. The detailed steps for such restructuring (if any) are not yet known, but KW will aim to retain the best talent across the Combined Group and has confirmed that the existing contractual and statutory employment rights of all KW and KWE employees will be fully safeguarded. No decisions have yet been made by KW in relation to specific actions that may be taken, though KW does intend to terminate the IMA in due course following completion of the Merger.

KW continues its detailed review of the combined business to identify realisable cost synergies, but that review is on-going and will not be completed until after completion of the Merger. Until such review is complete, KW cannot be certain what impact there will be on the Combined Group's business, KWE's management, location of business or any redeployment of KWE's fixed assets. However, given the close existing relationship between KW and KWE, the KW Board's expectation is that the effects of any restructuring will not be material. KWE is an externally managed investment company with no pension schemes.

Following completion of the Merger, the Combined Group will continue to invest both in the US and globally and make investment decisions on the basis of risk-adjusted returns on capital. KW has confirmed that the Merger will not have any effect on its management, on the location of its business or (save as disclosed in this paragraph 10.1) on its employees.

10.2 Response from the KWE Independent Directors

The KWE Independent Directors note the statement made by KW that, in order to achieve the expected benefits of the Merger, operational and administrative changes may be required but that the detailed steps for any restructuring, other than the expected termination of the IMA in due course, are not yet known.

The KWE Independent Directors note in particular that KWE is an externally managed investment company and that, having regard to the close relationship between KW and KWE, the KW Board does not currently expect that the effects of any restructuring will be material. KWE does not have employees (other than hotel staff) and, as an externally managed company, has no offices of its own.

The KWE Independent Directors welcome the statements made by KW that the existing contractual and statutory employment rights of all KWE employees in its operating hotel assets will be fully safeguarded.

In view of the on-going review to be completed following the successful completion of the Merger, as described above, the KWE Independent Directors are unable to express a more detailed opinion on the impact of the Merger on KWE's business.

11. Information relating to KWE

KWE invests in real estate across the United Kingdom, Ireland, Spain and Italy. KWE aims to generate superior shareholder returns by unlocking value of under-resourced real estate across its target geographies. KWE's existing portfolio is primarily invested across offices and retail in the United Kingdom and Ireland, weighted towards London, the South East of the United Kingdom and Dublin.

KWE is externally managed by KW through the Manager, KW's wholly owned Jersey-incorporated subsidiary.

KWE's registered office is located at 47 Esplanade, St Helier, Jersey JE1 0BD, Channel Islands, and the telephone number for its registered office is +44 (0) 1534 835 722. KWE's website is www.kennedywilson.eu. The information contained in, or that can be accessed through, its website is not part of or incorporated by reference into this document. KWE Shares trade on the London Stock Exchange under the ticker symbol "KWE".

12. Information relating to KW

KW owns, operates and invests in real estate, both on its own and through its investment management platform. KW focuses on multifamily and commercial properties located in the Western United States, the United Kingdom, Ireland and to a lesser extent Spain, Italy and Japan. KW also provides real estate services, primarily to financial services clients. As at 30 June 2017, KW had an average ownership interest across all investments of approximately 39%.

KW's value is primarily derived from its ownership of income-producing real estate assets and its real estate asset management and related services businesses. As at 30 June 2017, KW had an ownership interest in approximately 39 million square feet of property globally, including 23,956 multifamily rental units and 18.9 million square feet of commercial property, with approximately US\$13.2 billion of investment management and real estate assets under management (excluding KWE) and over US\$1.5 billion of new initiatives under development (with approximately US\$200 million of additional investments expected in the next two-three years). In addition to its core income-producing real estate, KW engages in development, redevelopment and value-add initiatives through which it seeks to enhance cash flows or reposition assets to increase disposal value. As at 30 June 2017, KW had over 570 employees in offices throughout the United States, the United Kingdom, Ireland, Jersey, Spain and Japan, and managed and worked with over 5,000 operating associates.

KW's principal executive offices are located at 151 S. El Camino Drive, Beverly Hills, CA 90212, United States of America, and the telephone number for its principal executive offices is +1 310 887 6400. KW's website is www.kennedywilson.com. Except as expressly stated in this document, the information contained in, or that can be accessed through, KW's website is not part of or incorporated by reference into this document. KW Shares trade on the NYSE under the ticker symbol "KW".

13. Current Trading and Prospects and KW 2017 Profit Forecast

13.1 *KWE*

KWE benefits from a diversified portfolio of approximately 470 tenants, low capital commitments and ample liquidity. With £646.7 million of disposals completed and exchanged since 2015 across 140 assets, delivering attractive returns and a further pipeline of non-core disposals, the quality of the portfolio continues to improve and provide stable underlying cash flows with opportunities for further income growth.

Since 30 June 2017, KWE completed the sales of the following assets:

Description	Portfolio	Disposal Date	Sale Price (£m)
Postigo de San Martín	Postigo	20 July 2017	11.9
Staines Road Retail Park	Gatsby	11 August 2017	10

Since 30 June 2017, KWE also entered into a contract to sell Gardner House, an office building in Dublin. The sale is expected to close in November 2017 for a sale price of €63.25 million.

13.2 *KW*

On 4 August 2017, KW issued an earnings release including its results for the quarter ended 30 June 2017 and subsequently filed with the SEC its quarterly report on Form 10-Q in respect of the same period.

KW reported GAAP net income to common shareholders of US\$0.08 per share for the second quarter of 2017, compared with a loss of US\$0.02 per share for the second quarter of 2016. Second quarter adjusted EBITDA and adjusted net income grew by 39% and 18% year-on-year, respectively, to US\$102.2 million and US\$51.0 million, respectively. KW's share of recurring property net operating income grew by 5% in the second quarter, while KW's share of same property net operating income grew by 6%. KW's share of acquisitions and disposals completed during the second quarter of 2017 was US\$193.5 million and US\$258.4 million, respectively.

KW is also publishing the KW 2017 Profit Forecast and certain other forward-looking information set out in Part Fourteen (*Profit Forecasts*) of this document.

14. The KWE Independent Directors and the effect of the Scheme on their interests

The KWE Independent Directors and details of their interests in the share capital of KWE are set out in paragraph 6 of Part Eight (*Additional Information*) of this document. Particulars of the letters of appointment and service agreements (including their fees) for each KWE Independent Director are set out in paragraph 12 of Part Eight (*Additional Information*) of this document.

Simon Radford, the sole KWE Independent Director who holds KWE Shares, has irrevocably undertaken to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting in respect of his own beneficial holding of 12,500 Scheme Shares, representing, in aggregate, approximately 0.01% of the existing issued ordinary share capital of KWE as at the Latest Practicable Date (as well as any further Scheme Shares he may acquire).

Save as referred to in this paragraph 14 of this Part Two, the effect of the Scheme on the interests of each KWE Independent Director does not differ from its effect on the like interests of any other Scheme Shareholder.

15. Investment Management Agreement, RSUs and RSGs

Pursuant to the IMA, the Manager is entitled to receive a management fee for the services it provides as investment manager to KWE. In the normal course, half of the management fee is settled in cash and the other half by KWE issuing new KWE Shares to the Manager or by making market purchases of KWE Shares for delivery to the Manager. The number of KWE Shares to be issued or delivered to the Manager depends on the latest reported net asset value per KWE Share at the relevant time. The share element of the management fee in respect of a quarter is settled approximately two months after the end of that quarter. The management fee

in respect of the first quarter of 2017 was £3.9 million and was paid on 8 May 2017, half of which was settled in KWE Shares purchased on the open market at market prices. The management fee in respect of the second quarter of 2017 was £3.9 million and was paid on 5 August 2017 entirely in cash. The management fee in respect of the third quarter of 2017 will be calculated and paid in accordance with the terms of the IMA, and if the management fee is paid before the Effective Date, the amount and form of payment of the management fee will be publicly announced. For more information about the IMA please see paragraph 13.1(e) of Part Eight (Additional Information) of this document.

The 23.8% of KWE Shares held by KW and/or its subsidiary undertakings as at the Latest Practicable Date included 532,230 KWE Shares held for the purpose of satisfying RSUs which will vest in connection with the Merger. It is intended that these RSUs will vest in consequence of the Merger after the KWE General Meeting but before the Scheme Record Time and that the 532,230 KWE Shares underlying these RSUs (less a certain number of KWE Shares withheld by KW to satisfy tax withholding obligations in accordance with the terms of the underlying award agreements) will be transferred to RSU holders (in accordance with the terms of the underlying award agreements) before the Scheme Record Time, i.e. in time for the relevant KWE Shares to become Scheme Shares and to be acquired by KW pursuant to the Scheme. However, irrespective of whether RSU holders or KW or any of its subsidiaries are holders of record of any KWE Shares to which these arrangements apply at any time when votes may be cast or proxies appointed for the Scheme Court Meeting, KW has undertaken to KWE to take all reasonable steps to procure that no proxies are appointed and no votes are cast at the Scheme Court Meeting in respect of any such underlying KWE Shares. However, votes attached to these underlying KWE Shares may be freely cast at the KWE General Meeting and Elections in respect of the underlying KWE Shares may be made prior to the Election Return Time.

As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.

KW also intends to continue to grant RSGs to certain of its own directors and employees in respect of KW Shares and will continue to do so in the ordinary course and consistent with past practice. These will not apply to KWE Shares.

16. Irrevocable Undertakings and Letters of Intent

Simon Radford, the sole KWE Independent Director who holds KWE Shares, has given an irrevocable undertaking to KW to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting in respect of those Scheme Shares owned and/or controlled by him, as well as any further Scheme Shares he may acquire. Such Scheme Shares amount to, in aggregate, 12,500 Scheme Shares, representing approximately 0.01% of the share capital of KWE in issue as at the Latest Practicable Date. The irrevocable undertaking does not restrict Simon Radford's ability to receive the New Offer Consideration and Distribution, to make any election under the Mix and Match Facility or to elect for the Original Offer Consideration at his discretion. As at the date of this document, Simon Radford intends to elect to receive the New Offer Consideration and Distribution in respect of the 12,500 Scheme Shares owned and/or controlled by him. For information on the intention of the other KWE Directors holding KWE Shares to elect to receive the New Offer Consideration and Distribution, see paragraph 4.1 of Part Eight (Additional Information) of this document.

KW has also received irrevocable undertakings from Quantum Strategic Partners Ltd. and Franklin Templeton Institutional, LLC (in its capacity as investment manager on behalf of certain funds and accounts managed by it) in respect of a total of 27,516,650 Scheme Shares, representing, in aggregate, approximately 28.6% of the KWE Shares whose holders are entitled

to vote at the Scheme Court Meeting and approximately 21.8% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date.

These undertakings are for Quantum Strategic Partners Ltd. and Franklin Templeton Institutional, LLC to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting, but allow Quantum Strategic Partners Ltd. and Franklin Templeton Institutional, LLC to make or refrain from making, at their discretion, any election under the Mix and Match Facility or for the Original Offer Consideration.

As at the Latest Practicable Date all of the other KWE Directors who hold KWE Shares (being William J. McMorrow and Mary L. Ricks) intend to elect to receive the New Offer Consideration and Distribution in respect of all Scheme Shares held by them or on their behalf, but for the avoidance of doubt, they are not bound to do so.

KW has also received a letter of intent from the Värde Shareholders to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting in respect of a total of 5,452,722 Scheme Shares, representing, in aggregate, approximately 5.7% of the KWE Shares whose holders are entitled to vote at the Scheme Court Meeting and approximately 4.3% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date.

Therefore, KW has received irrevocable undertakings to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, accept the Takeover Offer) in respect of a total of 27,529,150 Scheme Shares representing, in aggregate, approximately 28.6% of the KWE Shares whose holders are entitled to vote at the Scheme Court Meeting and approximately 21.8% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date, and a letter of intent to vote in the same way in relation to a further 5,452,722 Scheme Shares representing, approximately, an additional 5.7% of the KWE Shares whose holders are entitled to vote at the Scheme Court Meeting and approximately 4.3% of the KWE Shares whose holders are entitled to vote at the KWE General Meeting, in each case as at the Latest Practicable Date.

The irrevocable undertaking from Quantum Strategic Partners Ltd. described above will cease to apply if the Scheme does not become effective by midday Eastern time on 31 October 2017.

Further details of these irrevocable undertakings and the letter of intent (including details of the circumstances in which the irrevocable undertakings will cease to be binding) are set out in paragraph 4 of Part Eight (*Additional Information*) of this document.

17. Conditions to the Merger

- 17.1 The Merger will be subject to the Conditions set out in Part Three (Conditions and Certain Further Terms of the Scheme) of this document, including, among other things:
 - (1) (i) approval of the Scheme at the Scheme Court Meeting (and at any separate class meeting which may be required), or any adjournment thereof, by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the Register at the Scheme Voting Record Time, and present and voting (either in person or by proxy) at the Scheme Court Meeting, representing three-quarters or more of the votes attached to Scheme Shares which are cast at the Scheme Court Meeting (or at the meeting of each class of Scheme Shareholders, as the case may be); and (ii) such Scheme Court Meeting being held on or before the later of (1) the 22nd day after the expected date of the Scheme Court Meeting set out in this document; and (2) the fifth Business Day following the date on which the KW Shareholder Resolution has been duly passed or, in each case, such later date (if any) as KW and KWE may agree (any such agreement would require the consent of the Court and the Panel);
 - (2) (i) all resolutions required to approve and implement the Scheme as set out in the notice of the KWE General Meeting (including, without limitation, the Special Resolution) being duly passed by the requisite majority of KWE Shareholders at the KWE General Meeting or at any adjournment thereof; and (ii) the KWE General Meeting being held on or before the later of (1) the 22nd day after the expected date of the KWE General Meeting set out in this document; and (2) the fifth Business Day following the date on which the KW

- Shareholder Resolution has been duly passed or, in each case, such later date (if any) as KW and KWE may agree (any such agreement would require the consent of the Court and the Panel);
- (3) the sanction of the Scheme by the Court without modification, or with modification on terms acceptable to KW and KWE, and the delivery of the Court Order to the Registrar of Companies for registration;
- (4) the KW Shareholder Resolution being duly passed at the KW Special Meeting;
- (5) the NYSE having authorised the Listing upon official notice of issuance and not having withdrawn such authorisation; and
- (6) to the extent that a mandatory notification to the CCPC pursuant to section 18(1)(a) of the Irish Competition Act is required in respect of the Merger, the Merger having been notified to the CCPC and one of the following having occurred: (a) the CCPC having informed KW of its determination pursuant to either section 21(2)(a) or section 22(3)(a) of the Irish Competition Act that the Merger may be put into effect; (b) the CCPC having informed KW of its determination pursuant to section 22(3)(c) of the Irish Competition Act that the Merger may be put into effect subject to condition(s) specified by the CCPC being complied with; (c) the period specified in section 21(2) of the Irish Competition Act having lapsed without the CCPC having informed KW of its determination (if any) under section 21(2)(a) or (b) of the Irish Competition Act; or (d) where applicable, if 120 working days after the appropriate date (as defined by section 19(6) of the Irish Competition Act) have elapsed without the CCPC having informed KW of its determination (if any) under section 22 of the Irish Competition Act in relation to the Merger.
- 17.2 The New Offer was, when originally announced, subject to the additional Conditions set out in Section 3 of Part Three (*Conditions and Certain Further Terms of the Scheme*) of this document and below:
 - (1) the KW Prospectus being (a) approved by the UK Listing Authority; and (b) made available to the public in accordance with the prospectus rules made by the Financial Conduct Authority under section 73A of the Financial Services and Markets Act 2000 relating to the offer of securities to the public and admission of securities to trading on a regulated market; and
 - (2) to the extent required, the Jersey Financial Services Commission having consented to the circulation in Jersey of the offer of the New KW Shares pursuant to Article 8(2) of the Control of Borrowing (Jersey) Order 1958, as amended.

The conditions referred to in paragraphs 17.2(1), 17.2(2) and 17.1(6) have been satisfied in accordance with their terms as at the date of this document.

Under Rule 13.5(a) of the Code, KW may not invoke a Condition to the Merger so as to cause the Merger not to proceed, to lapse or to be withdrawn, unless the circumstances which give rise to the right to invoke the Condition are of material significance to KW in the context of the Merger. The Conditions contained in paragraphs 1, 2(a) and 2(b) of Part A of Part Three (Conditions and Certain Further Terms of the Scheme) of this document (and any acceptance condition if the Merger is implemented by means of a Takeover Offer) are not subject to this provision of the Code.

For the full wording of all of the Conditions, please see Part Three (Conditions and Certain Further Terms of the Scheme) of this document.

17.3 Payment of the KWE Special Distribution is subject to such payment being permissible under the Jersey Companies Law at the time of its payment. Although this is not a condition of the Scheme becoming effective, the terms of the Scheme provide that if unanticipated circumstances arose which prevented payment of the KWE Special Distribution as, when and to the extent provided for under the Scheme, the transfer of Scheme Shares to KW, the delivery of the Original Offer Consideration and/or the delivery of the KW Share Component and the payment of the KW Cash Component under the New Offer, as well as the arrangements relating to the Closing Dividend, would not be implemented.

18. KW Shareholder Approval

KW is required to obtain the approval of its shareholders for the issue of the New KW Shares in connection with the Merger at the KW Special Meeting.

The Merger is conditional on, among other matters described in paragraph 17 of this Part Two and in Part Three (*Conditions and Certain Further Terms of the Scheme*), the KW Shareholder Resolution being passed by a majority of the votes cast at the KW Special Meeting.

Fairfax Financial Holdings Limited and Elkhorn Partners LP have indicated to KW that they intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 14.3% of KW's outstanding shares of common stock as at the Latest Practicable Date.

The KW Directors, as well as Mary L. Ricks, Matt Windisch and Justin Enbody (being all of the executive officers of KW who hold KW Shares) and In Ku Lee, also intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 17.1% of KW's outstanding shares of common stock as at the Latest Practicable Date.

Therefore, support from KW Shareholders to vote in favour of the KW Shareholder Resolution at the KW Special Meeting has been received by KW in respect of approximately 31.4% of KW's outstanding shares of common stock as at the Latest Practicable Date.

The KW Directors have unanimously recommended that KW Shareholders vote in favour of the KW Shareholder Resolution at the KW Special Meeting.

KW will make the KW Proxy Statement available to KW Shareholders on or about the date of this document, which will include a notice convening the KW Special Meeting at 9.00 a.m. (Pacific Daylight Time) on 12 October 2017.

The dissemination and content of the KW Proxy Statement are the responsibility solely of the KW Directors. See also the "Important Notice" on pages 4 to 9 of this document.

For the avoidance of doubt, none of the KWE Independent Directors and neither Rothschild nor J.P. Morgan Cazenove expresses any views as to the advantages or disadvantages of the Merger as far as KW Shareholders are concerned.

19. New KW Shares

The New KW Shares to be issued in connection with the Merger will be issued credited as fully paid and free from all Encumbrances, will rank *pari passu* in all respects with the existing issued KW Shares and will carry the right to receive all dividends and other distributions declared, paid or made by KW on or after the Effective Date or by reference to a record date on or after the Effective Date (in each case whether or not wholly or partly in respect of a period which precedes the Effective Date).

Application will be made to the NYSE for the New KW Shares to be admitted to listing and trading on the NYSE. Authorisation of the Listing by the NYSE is a Condition to the Scheme becoming effective and if such authorisation is not provided or is withdrawn, the Scheme shall not become effective. It is anticipated that the listing of the New KW Shares will become effective and that trading in New KW Shares will commence on the Effective Date. The New KW Shares will be denominated in US Dollars and, when admitted to trading on the NYSE, will be registered with the ISIN US 4893981070 and ticker symbol "KW".

The price at which KW Shares are publicly traded on the NYSE is subject to fluctuations and may be influenced by a large number of factors. These factors could be specific to KW and its operations or may affect the industry generally. The price at which New KW Shares are publicly traded on the NYSE at the Effective Date and the price which KW Shareholders may subsequently realise for their New KW Shares cannot be predicted.

The KW Prospectus summarises certain risk factors relating to the Merger in respect of the receipt of the New KW Shares. Copies of the KW Prospectus and this document will be available to view online, as set out in paragraph 17.2 of Part Eight (*Additional Information*) of this document.

The publication and content of the KW Prospectus are the responsibility solely of the KW Directors. Further details of the rights attaching to the New KW Shares are set out in Part Ten (*Description of KW Shares*) of this document. See also the "Important Notice" on pages 4 to 9 of this document.

20. Financing of the New Offer

The maximum aggregate value of the KW Cash Component of approximately £290 million payable by KW under the New Offer will be funded from cash resources of KW.

Goldman Sachs International, as lead financial adviser to KW, is satisfied that sufficient resources are available to KW to satisfy, in full, the KW Cash Component payable by KW pursuant to the New Offer.

21. Financial effect of the Merger on KWE Shareholders

The following table shows, for illustrative purposes only, and on the bases and assumptions set out in the notes below, the financial effects of the Merger on the capital and income position of the Scheme Shareholders on a pre-tax basis.

The table below assumes that no Election is made under the Mix and Match Facility. In assessing the financial effects on the capital and income position of the Scheme Shareholders, no account has been taken of any potential liability to taxation of a Scheme Shareholder, or a beneficial owner of Scheme Shares, or of any timing differences in the payment of dividends. All figures in the table are in pence unless otherwise stated. The attention of beneficial owners of KWE Shares and KWE Shareholders is drawn to Part Seven (*Taxation*) of this document. The tax implications of the financial effects of the Merger will depend on the individual circumstances of each beneficial owner of Scheme Shares and Scheme Shareholders. Beneficial owners of Scheme Shares and Scheme Shares.

_	Note	New Offer ⁽¹⁾	Original Offer ⁽²⁾
Illustrative effect on capital position ⁽³⁾			
Value of the New KW Share Component	(4)	559	967
Value of the KW Cash Component		300	_
Value of the KWE Special Distribution		250	
Total value of consideration in respect of one KWE Share		1,109	967
Less: market value of one KWE Share on 21 April 2017		979	979
Increase/(decrease) in capital value		130	(12)
Percentage increase/(decrease) in capital value		13.3%	(1.2)%
	Note	New Offer ⁽¹⁾	Original Offer ⁽²⁾
Illustrative effect on gross income ⁽³⁾			
Gross annual dividend income from New KW Shares Gross income from re-investment of the KW Cash	(5)	22.2	38.4
Component and the KWE Special Distribution	(6)	5.4	_
Total gross income in respect of consideration for one KWE			
Share		27.6	38.4
Less: gross annual dividend income from one KWE share	(7)	48	48
Increase/(decrease) in gross income		(20.4)	(9.6)
Percentage increase/(decrease) in gross income		(42.5)%	(20.0)%

Notes

⁽¹⁾ This column is based on the New Offer Consideration and Distribution, assumes no Mix and Match Election, and is based on the Closing Price of KW Shares and the US\$ / £ exchange rate on the Latest Practicable Date.

⁽²⁾ This column is based on the consideration available under the Original Offer, the Closing Price of KW Shares and the US\$ / £ exchange rate on the Latest Practicable Date.

- (3) In assessing the financial effects on the capital and income position of the Scheme Shareholders, no account has been taken of any potential liability to taxation of a Scheme Shareholder, or a beneficial owner of Scheme Shares, or of any timing differences in the payment of dividends. The attention of beneficial owners of Scheme Shares and Scheme Shareholders is drawn to Part Seven (*Taxation*) of this document. The tax implications of the financial effects of the Merger will depend on the individual circumstances of each beneficial owner of Scheme Shares and Scheme Shareholder. Beneficial owners of Scheme Shares and Scheme Shareholders should consult their own tax advisers.
- (4) The market value of KW Shares is based on the Closing Price per KW Share of US\$19.15 and a US\$ / £ exchange rate of US\$1.3209:£1 at the Latest Practicable Date.
- (5) The gross annual dividend income is based on an assumed quarterly dividend of US\$0.19 per KW Share and a US\$ / £ exchange rate of US\$1.3209:£1 at the Latest Practicable Date.
- (6) The gross income from re-investment of the KW Cash Component and the KWE Special Distribution has been calculated on the assumption that the KW Cash Component and the KWE Special Distribution are re-invested for a period of 12 months, to yield approximately 0.99% per annum, being the yield shown by UK Gilts of ten year maturities on the Latest Practicable Date.
- (7) The dividend income from one KWE Share is based on the aggregate of the quarterly dividend of 12 pence per quarter.

SCHEME SHAREHOLDERS SHOULD NOTE THAT:

- THE ILLUSTRATIVE INCREASE OR DECREASE IN CAPITAL VALUE AND GROSS INCOME ABOVE ARE CALCULATED BASED ON THE CLOSING PRICE OF A KW SHARE OF US\$19.15 AND THE EXCHANGE RATE OF US\$1.3209:£1, IN EACH CASE AS AT THE LATEST PRACTICABLE DATE.
- THE CLOSING PRICES OF KW SHARES, EXCHANGE RATES AND DIVIDEND PAYMENTS MAY FLUCTUATE. A SCHEME SHAREHOLDER MAY RECEIVE GREATER OR LOWER VALUE THAN THE VALUES REFERRED TO ABOVE.

22. Structure of the Merger

It is intended that the Merger will be implemented by way of a Court-sanctioned scheme of arrangement among KWE and the Scheme Shareholders under Article 125 of the Jersey Companies Law and KW will undertake to the Court to be bound by and comply with the Scheme.

The Scheme will be governed by Jersey law and will be subject to the jurisdiction of the Court and the Panel. The Scheme will be subject to the applicable requirements of the Code, the Listing Rules, the FCA, the London Stock Exchange and the Jersey Financial Services Commission.

This document includes full details of the Scheme (Part Four (*The Scheme of Arrangement*)), together with notices of the Scheme Court Meeting (Part Twelve (*Notice of Scheme Court Meeting*)) and the KWE General Meeting (Part Thirteen (*Notice of KWE General Meeting*)).

23. Description of the Scheme

The purpose of the Scheme is to cause KW to become the owner of the entire issued and to be issued ordinary share capital of KWE (other than the Excluded Shares).

If the Scheme is approved and sanctioned in the manner required and becomes effective, it will be binding on all holders of Scheme Shares, irrespective of whether or not they attended or voted at the KWE Meetings and, if they attended and voted, whether or not they voted in favour of the Scheme at the Scheme Court Meeting or the Special Resolution at the KWE General Meeting. The Scheme will also be binding in law.

The Scheme will become effective at the time at which the Court Order is filed with the Registrar of Companies. Assuming the Court sanctions the Scheme at the Scheme Court Sanction Hearing, the Effective Date is expected to be the second Business Day immediately after the date of the Scheme Court Sanction Hearing.

KW has reserved the right, subject to the consent of the Panel, to implement the Merger by means of a Takeover Offer.

24. The KWE Meetings

The Scheme Court Meeting

The Scheme Court Meeting is being convened at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN for 10.00 a.m. on 10 October 2017 to seek the approval of Scheme Shareholders for the Scheme.

Each Scheme Shareholder present, in person or by proxy, will be entitled to one vote for each Scheme Share held at the Voting Record Time. The Scheme will require the approval at the Scheme Court Meeting (and at any separate class meeting which may be required), or any adjournment thereof, by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the Register at the Voting Record Time, and present and voting (either in person or by proxy) at the Scheme Court Meeting, representing three-quarters or more of the votes attached to Scheme Shares which are cast at the Scheme Court Meeting (or at the meeting of each class of Scheme Shareholders, as the case may be).

A Scheme Shareholder may vote for or against the Scheme in respect of all of its Scheme Shares or may vote for the Scheme in respect of some Scheme Shares held and against the Scheme in respect of others (this may be of particular relevance to Scheme Shareholders who are nominees or trustees for more than one person).

You are strongly urged to complete and return your BLUE Form of Proxy for use at the Scheme Court Meeting as soon as possible and in any event so as to be received by no later than 10.00 a.m. on 8 October 2017 (or, in the case of any adjournment, not later than 48 hours before the time fixed for the adjourned meeting). Alternatively, the BLUE Form of Proxy will still be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or, if relevant, any adjournment thereof). Detailed instructions on the action to be taken are set out on pages 14 to 22 of this document.

Alternatively, you can log on to the website www.investorcentre.co.uk/je to submit your proxy or proxies electronically so as to be received by not later than 48 hours before the Scheme Court Meeting (or any adjournment thereof). You will need your unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on your Form of Proxy) in order to submit your proxies in this manner.

If you hold Scheme Shares through CREST and wish to appoint a proxy or proxies, you can do so by using the CREST electronic proxy appointment service. Proxies submitted via CREST must be received by the KWE Registrar (ID is 3RA50) not later than 10.00 a.m. on 8 October 2017 (or, in the case of any adjournment, not later than 48 hours before the time fixed for the holding of the adjourned meeting).

A proxy, however appointed, need not be a KWE Shareholder and may be the Chairman of the relevant meeting.

The KWE General Meeting

The KWE General Meeting is being convened at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN for 10.15 a.m. on 10 October 2017 (or, if later, as soon as the Scheme Court Meeting has concluded or been adjourned).

At the KWE General Meeting, each KWE Shareholder present, in person or by proxy (including holders of Excluded Shares), will be entitled to one vote for each KWE Share held at the Voting Record Time. In order for the Special Resolution to be passed, it must be approved by a majority of not less than two-thirds of votes cast by KWE Shareholders present and voting (and entitled to vote) in accordance with the KWE Articles, either in person or by proxy (including holders of Excluded Shares).

The Special Resolution contains provisions to amend the KWE Articles to ensure that any KWE Shares issued (other than to KW, its nominees or any member of the KW Group) (i) between the date of the KWE General Meeting and on or prior to the Scheme Record Time will be subject to the Scheme; and (ii) after the Scheme Record Time will automatically be acquired by KW on the same terms (other than terms as to timings and formalities and without any ability to make any Elections with respect to the Mix and Match Facility or the Original Offer) as under the Scheme. These provisions will avoid any person (other than a member of the KW Group) holding KWE Shares after dealings in such shares have ceased on the London Stock Exchange. The Special Resolution also contains provisions to authorise the KWE Directors to take all such actions as they may consider necessary or appropriate for carrying the Scheme into full effect. The terms of the Special Resolution are set out in their entirety in Part Thirteen (Notice of KWE General Meeting) of this document.

Entitlement to vote at the KWE Meetings

Each Scheme Shareholder who is on the Register at the Voting Record Time will be entitled to attend and vote, in person or by proxy, at the Scheme Court Meeting. Each KWE Shareholder who is on the Register at the Voting Record Time will be entitled to attend and vote, in person or by proxy, at the KWE General Meeting.

At both of the KWE Meetings, voting on the resolutions set out in the notices at Part Twelve (Notice of Scheme Court Meeting) and Part Thirteen (Notice of KWE General Meeting) of this document will be conducted on a poll rather than on a show of hands.

As at the Latest Practicable Date, KW and its subsidiaries held approximately 23.8% of the issued ordinary shares of KWE. Any KWE Shares held by or on behalf of any member of the KW Group (including, for the avoidance of doubt, any KWE Shares held by or on behalf of the Manager) will be Excluded Shares (i.e. not subject to the Scheme) and no member of the KW Group (including the Manager), or its nominee, is entitled to vote at the Scheme Court Meeting in respect of the Excluded Shares held by it.

The 23.8% of KWE Shares held by KW and/or its subsidiary undertakings as at the Latest Practicable Date included 532,230 KWE Shares held for the purpose of satisfying RSUs which will vest in connection with the Merger. It is intended that these RSUs will vest in consequence of the Merger after the KWE General Meeting but before the Scheme Record Time and that the 532,230 KWE Shares underlying these RSUs (less a certain number of KWE Shares withheld by KW to satisfy tax withholding obligations in accordance with the terms of the underlying award agreements) will be transferred to RSU holders (in accordance with the terms of the underlying award agreements) before the Scheme Record Time, i.e. in time for the relevant KWE Shares to become Scheme Shares and to be acquired by KW pursuant to the Scheme. However, irrespective of whether RSU holders or KW or any of its subsidiaries are holders of record of any KWE Shares to which these arrangements apply at any time when votes may be cast or proxies appointed for the Scheme Court Meeting, KW has undertaken to KWE to take all reasonable steps to procure that no proxies are appointed and no votes are cast at the Scheme Court Meeting in respect of any such underlying KWE Shares. However, votes attached to these underlying KWE Shares may be freely cast at the KWE General Meeting and Elections in respect of the underlying KWE Shares may be made prior to the Election Return Time.

As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.

If either KWE Meeting is adjourned, only those Scheme Shareholders (in the case of the Scheme Court Meeting) or KWE Shareholders (in the case of the KWE General Meeting) on the Register not later than 6.00 p.m. on the last date to fall two Jersey Working Days before the date of the relevant adjourned KWE Meeting will be entitled to attend and vote.

If you propose to attend the KWE Meetings, please detach and bring with you the attendance cards to assist with your admission.

The completion and return of a Form of Proxy, the electronic appointment of a proxy or the submission of a proxy via CREST will not prevent an eligible Scheme Shareholder from attending and voting in person at the KWE Meetings, or any adjournment thereof, or an eligible KWE Shareholder from attending and voting in person at the KWE General Meeting, or any adjournment thereof. Attending and voting in person will result in the invalidation of any previously lodged proxy.

Each Scheme Shareholder and KWE Shareholder is entitled to appoint a proxy in respect of some or all of the shares held. Each Scheme Shareholder and KWE Shareholder is also entitled to appoint more than one proxy, as long as each proxy is appointed to exercise rights attached to different Scheme Shares or KWE Shares. A space has been included on each Form of Proxy for specifying the number of shares in respect of which a proxy is appointed. A Scheme

Shareholder or KWE Shareholder who returns a Form of Proxy duly executed but leaves this space blank will be deemed to have appointed a proxy in respect of all of its Scheme Shares or KWE Shares, respectively.

A Scheme Shareholder or KWE Shareholder who wishes to appoint more than one proxy in respect of its shareholding should contact the KWE Registrar for further Forms of Proxy or photocopy the Forms of Proxy as required.

The person appointed as a proxy in relation to the Scheme Court Meeting will vote for or against the Scheme as per your instructions in the BLUE Form of Proxy. For any other business arising at the Scheme Court Meeting (including any proper procedural resolution not listed in the notice of the Scheme Court Meeting), the person appointed as a proxy will vote at his or her discretion.

The person appointed as a proxy in relation to the KWE General Meeting will vote for or against, or abstain from voting in relation to, the Special Resolution as per your instructions in the GREEN Form of Proxy. In the absence of instructions and for any business arising at the KWE General Meeting (including any proper procedural resolution not listed in the notice of the KWE General Meeting), the person appointed may vote, or abstain from voting, as he or she thinks fit on any business which may properly come before the KWE General Meeting.

The Chairman of the KWE Meetings shall be at liberty to accept a Form of Proxy (and to accept as having voted the number of Scheme Shares or, where relevant, KWE Shares, in respect of which the relevant holder seeks to vote) notwithstanding that the relevant Form of Proxy has not been completed in accordance with the instructions contained in the Form of Proxy, provided that the Chairman considers that the information contained in the Form of Proxy is sufficient to establish the entitlement of the holder of the relevant Scheme Shares or, where relevant, the KWE Shares, to vote and is satisfied as to the authority of the person signing it to do so.

A proxy, however appointed, need not be a KWE Shareholder and may be the Chairman of the relevant meeting.

Further information on the action to be taken is set out on pages 14 to 22 ("Action to be Taken") of this document.

Shareholder Helpline

If you require assistance, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

25. Sanction of the Scheme by the Court

Under Article 125 of the Jersey Companies Law, the Scheme requires the sanction of the Court. A Scheme Court Sanction Hearing is scheduled to be held at 10.00 a.m. on 18 October 2017 at the Royal Court of Jersey, Royal Court Building, St Helier, JE1 1BA, Jersey, Channel Islands. The time and date for the Scheme Court Sanction Hearing is indicative only and will depend on, among other matters, the date on which the Conditions are satisfied or (to the extent capable of waiver) waived. If the expected time and/or date of the Scheme Court Sanction Hearing is changed, KWE will give notice of this change by issuing an announcement via a Regulatory Information Service. Scheme Shareholders may, if they wish, attend and be heard at the Scheme Court Sanction Hearing, in person or through counsel, to support or oppose the Court's sanctioning of the Scheme.

KW has confirmed that it will be represented by counsel at the Scheme Court Sanction Hearing, so as to consent to the Scheme and to undertake to the Court to be bound by and comply with the Scheme. KW agrees to undertake to the Court to execute and do and procure to be executed and done all such documents, acts and things as may be necessary or desirable to be executed or done by each of them respectively for the purpose of giving effect to the Scheme.

If the Scheme is sanctioned by the Court at the Scheme Court Sanction Hearing, it will become effective only on delivery of the Court Order to the Registrar of Companies for registration which, under Jersey law, is required to take place not later than 14 days after the Court sanctions the Scheme. KWE expects to file the Court Order on the second Business Day after the date on which the Court sanctions the Scheme.

If the Scheme becomes effective, it will be binding on all holders of Scheme Shares, irrespective of whether or not they attended or voted in favour of the Scheme at the Scheme Court Meeting or in favour of the Special Resolution at the KWE General Meeting. It will also be binding in law. If the Scheme has not become effective on or before the Longstop Date, the Scheme will not be implemented and the Merger will not proceed (unless (a) KWE and KW otherwise agree to a later date and (b) the Panel and, if required, the Court permit such later date).

The last day of dealings in, and for registration of transfers of, KWE Shares is expected to be 19 October 2017, following which the KWE Shares will be suspended from the premium listing segment of the Official List and from trading on the London Stock Exchange's main market for listed securities. No transfers of KWE Shares will be registered after that date.

26. Modifications to the Scheme

Clause 13 of the Scheme provides for KWE and KW jointly to consent on behalf of all persons concerned to any modification of, or addition to, the Scheme or to any condition approved or imposed by the Court. The Court would be unlikely to approve any modification of, or additions to, or impose a condition to the Scheme which might be material and adverse to the interests of KWE Shareholders unless KWE Shareholders were informed of such modification, addition or condition. It would be a matter for the Court to decide, in its discretion, whether or not any further KWE shareholder meetings should be held in these circumstances.

27. Delisting of KWE Shares

It is intended that, if the Scheme is approved at the Scheme Court Meeting, the Special Resolution is passed at the KWE General Meeting and the Scheme is sanctioned by the Court at the Scheme Court Sanction Hearing, the Court Order will be delivered to the Registrar of Companies for registration on 20 October 2017. It is expected that dealings in KWE Shares will be suspended at 7.30 a.m. (London time) on the Effective Date. It is also intended that an application will be made to the London Stock Exchange for the cancellation of the trading of the KWE Shares on the Main Market and that the UKLA will be requested to cancel the listing of KWE Shares on the Official List, to take effect on or after the first Business Day to fall after the Effective Date.

28. Takeover Offer

KW reserves the right to elect, subject to the Panel's consent, to implement the Merger by way of a Takeover Offer as an alternative to the Scheme. In such an event, the Takeover Offer will be implemented on the same terms (subject to appropriate amendments), so far as applicable, as those which would apply to the Scheme.

If the Merger is effected by way of a Takeover Offer and such Takeover Offer becomes or is declared unconditional in all respects and sufficient acceptances are received, KW intends to: (i) make a request to the UKLA to cancel the listing of the KWE Shares from the Official List; (ii) make a request to the London Stock Exchange to cancel trading in KWE Shares on the Main Market; and (iii) exercise its rights to apply the relevant provisions of Part 18 of the Jersey Companies Law, if applicable, to acquire compulsorily the remaining KWE Shares in respect of which the Takeover Offer has not been accepted.

If KW elects, with the Panel's consent, to implement the Merger by way of a Takeover Offer, such offer will be made in compliance with applicable US tender offer and securities laws and regulations, including any applicable exemptions therefrom.

29. Return of documents of title

If the Scheme is withdrawn or lapses, or an Election Form is validly withdrawn, documents of title submitted and other documents lodged with either Form of Proxy or the Election Form will be returned to the relevant Scheme Shareholder as soon as practicable and in any event within 14 days of such lapse or withdrawal.

30. Offer-related arrangements

(a) Confidentiality Agreement

On 23 March 2017, KW, the Manager and KWE entered into the Confidentiality Agreement, pursuant to which the parties have undertaken, among other things, to (a) keep confidential all confidential information relating to the Merger and the other parties and not to disclose it to third parties (other than to certain permitted recipients), unless such disclosure is permitted pursuant to the Confidentiality Agreement (for example, where disclosure is required by law or regulation), and (b) use the confidential information for the sole purpose of considering, facilitating or advising on, or seeking advice in relation to, the Merger. These confidentiality obligations remain in force (i) for 24 months from the date of the Confidentiality Agreement, or (ii) until completion of the Merger, whichever is earlier.

Under the Confidentiality Agreement, KW and the Manager undertook to KWE to instruct the Ring Fenced Employees not to communicate to KW any confidential information regarding certain matters relating to KWE and its preparations for responding to KW's proposals in respect of the Merger and promptly to supply to the KWE Independent Directors all information belonging to KWE requested by any KWE Independent Director and to instruct the named executives to ensure that such information was, when supplied, complete, accurate and not misleading.

In addition, KW also agreed that, other than KWE Shares acquired pursuant to the IMA, neither it nor any person acting in concert with it may acquire any interest in any KWE Shares without the prior written consent of the Independent Directors of KWE. This restriction will fall away immediately if KW indicates to KWE that it is no longer actively considering the Merger.

(b) Separate Manager's Powers Agreement

On 10 April 2017, KWE and the Manager entered into the Separate Manager's Powers Agreement, pursuant to which the Manager agreed that, during the offer period, it will make all arrangements requested by the Independent Directors of KWE to allow the Ring Fenced Employees to make all payments out of the bank accounts of KWE or any member of the KWE Group which are specified by the Independent Directors of KWE. For the purposes of the Separate Manager's Powers Agreement, "offer period" means the period starting on the date of the Separate Manager's Powers Agreement and ending on the earlier of (i) completion of the Merger, (ii) the date on which KW publicly announces that its offer for KWE has been withdrawn or has lapsed, and (iii) the date on which KW has confirmed in writing to the Independent Directors of KWE that it is no longer actively considering the Merger.

In addition, the Manager agreed that, during the offer period and notwithstanding anything to the contrary in the IMA, it will not take certain actions without the prior written approval of the Independent Directors of KWE. These actions include, among other things, (i) causing KWE or any member of the KWE Group to issue or acquire any KWE Shares, (ii) causing any payments in excess of (a) £2,000,000 on any single day or (b) £5,000,000 in any period of three consecutive days to be made out of the bank accounts of KWE or any member of the KWE Group, (iii) entering into any agreement to sell or acquire, on behalf of KWE or any member of the KWE Group, any assets having a value in excess of £15,000,000, and (iv) taking any action or engaging in any conduct, or causing KWE or any member of the KWE Group to enter into any arrangement, which would be restricted under Rule 21 of the Code.

(c) Transaction Agreement

On 24 April 2017, KW and KWE entered into the Transaction Agreement, pursuant to which they have agreed to certain mutual commitments to regulate the basis on which they are willing to implement the Merger.

In particular, KW and KWE agreed that KW shall have primary responsibility for obtaining all regulatory consents and approvals (the "Regulatory Consents"), specified in Conditions 2(b) and 2(c) of Part Three (Conditions and Certain Further Terms of the Scheme) of this document (the "Regulatory Conditions"), and that each party will promptly make all filings and supply all information necessary to obtain the Regulatory Consents. Each party also undertook to provide copies of, and incorporate reasonable comments on, all material filings, notifications, submissions and communications in relation to satisfaction of the Regulatory Conditions. The parties further undertook to notify the other parties of, and allow persons nominated by the

other to attend, all meetings or material telephone calls in connection with obtaining the satisfaction of the Regulatory Conditions and the implementation of the Merger and to make reasonable oral submissions during such meetings or telephone calls.

The parties also undertook, to the extent not prohibited by law or regulation, to keep the other informed of the progress towards satisfaction of the Regulatory Conditions and of any development that might reasonably be considered to be likely to prevent any of the Regulatory Conditions from being satisfied.

Pursuant to the Transaction Agreement, KW undertook to provide KWE with all such cooperation, information and assistance as it may reasonably request in connection with the preparation and verification of this document and to procure that the KW Directors take responsibility for the information relating to KW, the KW Group, the Manager or any persons acting in concert with KW.

KW agreed, to the extent not prohibited by law or regulation or disclosed generally to the public, to notify the KWE Independent Directors in writing if, on or after the date of the Transaction Agreement, KW or any of its subsidiaries (other than KWE and its subsidiaries) take certain actions, including, but not limited to:

- issuing, or becoming obliged to issue, any KW Shares (to the extent material in the context of the Merger);
- declaring or paying any dividend or other distribution, other than its ordinary course quarterly dividends;
- entering into any material agreement outside the ordinary course of its business (to the extent material in the context of the Merger); and
- incurring any indebtedness which (if payable in full) would increase the borrowings of the KW Group (excluding KWE and its subsidiaries) by more than 20%.

(d) Indemnity Agreement and KWE Independent Directors' Additional Remuneration Agreements

On 23 June 2017, KW, KWE, the Manager and the KWE Independent Directors entered into an indemnity agreement pursuant to which KW and the Manager, subject to certain exceptions and limitations, agreed to hold harmless and indemnify KWE and the KWE Independent Directors for liability arising from any inaccuracies, deficiencies and omissions relating to certain KWE information communicated externally or filed with a regulatory body.

The Indemnity Agreement further provides that the Manager will, when requested, promptly furnish the KWE Independent Directors with KWE information relating to KWE's financial condition and use reasonable efforts to ensure such information is accurate and not misleading.

KW agreed that KWE information communicated by it to other persons or filed by it with any regulatory body will be accurate and not misleading and to accept responsibility for such information to the exclusion of the KWE Independent Directors.

In addition, KW has agreed to supply the KWE Independent Directors with copies of KWE information prior to the repetition, publication, dissemination, communication, filing or submission of such information and to refrain from releasing such information if the KWE Independent Directors provide written notice of their objections.

Pursuant to the terms of separate agreements entered into, with the consent of KW, between KWE and each of Charlotte Valeur, Mark McNicholas and Simon Radford dated 23 April 2017 (the "KWE Independent Directors' Additional Remuneration Agreements"), KWE agreed to pay each KWE Independent Director extra remuneration for any services rendered in his/her capacity as a KWE Director in relation to the Merger. For more information on the Independent Directors' Additional Remuneration Agreements, please see paragraph 12 of Part Eight (Additional Information) of this document.

31. Settlement

Subject to (i) the Scheme becoming effective; and (ii) the payment of the KWE Special Distribution if, when and to the extent required by the Scheme (with the effect that the remaining arrangements provided for under the Scheme are required to be implemented), settlement of the consideration to which any Scheme Shareholder is entitled under the Scheme

and of the KWE Special Distribution will be effected as soon as practicable and, in any event, not later than 14 days after the Effective Date in the manner explained in this paragraph 31. The Closing Dividend will be payable not later than 10 January 2018.

31.1 New KW Shares

(a) Scheme Shares in uncertificated form (i.e. in CREST)

Unlike KWE Shares, New KW Shares are not capable of being held, transferred or settled directly through the CREST settlement system. For this reason, a Scheme Shareholder who holds Scheme Shares in uncertificated form through CREST (directly or through a broker or other nominee with a CREST account) immediately prior to the Scheme Record Time will not be issued New KW Shares directly but will be issued KW CDIs through an existing CDI programme administered through CREST in respect of KW Shares. Under the arrangements, one KW CDI will represent one KW Share. The KW CDIs will reflect the economic rights attached to the New KW Shares. However, while the holders of KW CDIs will have an interest in the underlying New KW Shares, they will not be the registered holders of the New KW Shares.

The KW CDIs to which such Scheme Shareholders will be entitled under the Scheme will be delivered, held and settled in CREST and linked to the underlying New KW Shares by means of the CREST International Settlement Links Service and, in particular, CREST's established link with DTC. This link operates via the services of CREST International Nominees Limited, which is a participant in DTC.

Under the CREST International Settlement Links Services, CREST Depository Limited, a subsidiary of Euroclear, issues dematerialised depository interests representing entitlements to non-UK securities (such as the New KW Shares) called CREST Depository Interests or CDIs, which may be held, transferred and settled exclusively through CREST.

The terms on which CDIs are issued and held in CREST are set out in the CREST Manual (and, in particular, the deed poll set out in the CREST International Manual) and the CREST Terms and Conditions issued by Euroclear.

On settlement, the KW Transfer Agent will cause the credit of the New KW Shares through DTC to the securities deposit account of CREST International Nominees Limited, as nominee for CREST Depository Limited in DTC. CREST Depository Limited will then issue the KW CDIs through CREST to the CREST Receiving Agent, as receiving agent for delivery to the securities deposit account in CREST in which each such uncertificated Scheme Shareholder previously held Scheme Shares. A custody fee, as determined by CREST from time to time, is charged at the user level (i.e. to the holder of KW CDIs) for the CREST International Settlement Links Service.

KW and KWE will enter into arrangements with the CREST Receiving Agent pursuant to which Euroclear will be instructed to credit the appropriate stock account in CREST of the relevant Scheme Shareholder with such relevant Scheme Shareholder's entitlement to KW CDIs as soon as practicable after the Effective Date and in any event within 14 days thereof.

Notwithstanding the above, KW reserves the right to settle all or part of such consideration in accordance with paragraph 31.1(c) below of this Part Two if, for reasons outside KW's reasonable control, it is not able to effect settlement in accordance with this paragraph 31.1(a).

With effect from close of trading on the last day of dealings in KWE Shares prior to the Effective Date, each holding of Scheme Shares credited to any stock account in CREST will be disabled and all Scheme Shares will be removed from CREST in due course. Euroclear will be instructed to credit the appropriate stock account in CREST of the CREST shareholder concerned with such CREST shareholder's entitlement to KW CDIs representing New KW Shares. The stock account concerned will be an account under the same participant ID and member account ID under which the relevant CREST shareholder holds the relevant Scheme Shares.

Holders of KW CDIs will be able to cancel their KW CDIs by settling a cross-border delivery transaction in respect of the underlying New KW Shares through CREST to a DTC participant, in accordance with the rules and practices of CREST and DTC.

(b) Rights attaching to KW CDIs

The registered holder of the New KW Shares represented by KW CDIs will be Cede & Co., a nominee of DTC. The custodian of those New KW Shares will be CREST International Nominees Limited, who will hold them, either directly or indirectly through a sub-custodian, through book entry interests within the DTC system as nominee for CREST Depository Limited. CREST Depository Limited will hold those New KW Shares on trust (as bare trustee under English law) for the holders of Scheme Shares in uncertificated form to whom it will issue KW CDIs through CREST.

Under an agreement with Euroclear, Euroclear will make a copy of the register of names and addresses of holders of KW CDIs available to KW (and/or its voting agent) to enable KW (or its voting agent) to (a) send out notices of shareholder meetings and forms of proxy to KW CDI holders; and (b) produce a definitive list of KW CDI holders as at the record date of the meeting.

In order to allow holders of KW CDIs to exercise rights relating to the underlying New KW Shares, KW will enter into arrangements pursuant to which holders of KW CDIs will be able to (a) receive notices of general shareholder meetings of KW; (b) give directions as to voting at general shareholder meetings of KW; and (c) have made available to them and be sent, at their request, copies of the annual report and accounts of KW and any other documents issued by KW to holders of KW Shares generally.

In addition, Cede & Co. and Euroclear have omnibus proxy arrangements pursuant to which CREST International Nominees Limited (the custodian of the New KW Shares underlying the KW CDIs) will be able to grant each KW CDI holder the right to vote in respect of such holder's underlying New KW Shares.

Holders of KW CDIs will otherwise be treated in the same manner as if they were registered holders of the New KW Shares underlying the KW CDIs, in each case in accordance with applicable law and, so far as is possible, in accordance with the CREST arrangements.

Scheme Shareholders who wish to have the full ease of dealing and other benefits associated with holding KW Shares in DTC in the same way as existing KW Shareholders should consider transferring the New KW Shares they receive in CDI form under the terms of the Merger from CREST to DTC as soon as conveniently possible in accordance with the procedure described above.

(c) Scheme Shares in certificated form (i.e. not in CREST)

Each Scheme Shareholder who holds Scheme Shares in certificated form will be issued New KW Shares to which he or she is entitled under the Scheme in respect of Scheme Shares held in certificated form directly, so that the name of such Scheme Shareholder will be entered as the registered owner of the relevant number of New KW Shares on the KW Register and the KW Transfer Agent will deliver share certificates to such Scheme Shareholder in respect of the New KW Shares to which the Scheme Shareholder is entitled. Definitive certificates for the New KW Shares will be despatched by first class post (or such other method as may be approved by the Panel) within 14 days of the Effective Date. In the case of joint Scheme Shareholders, certificates will be despatched to the joint Scheme Shareholder whose name appears first in the Register.

The KWE Registrar will forward details of Scheme Shareholders who hold Scheme Shares in certificated form to the KW Transfer Agent. Subject to satisfactory completion of the KW Transfer Agent's new client procedures, the KW Transfer Agent will make arrangements for such holder to hold its New KW Shares directly in registered form.

The information requested in any form(s) provided to Scheme Shareholders holding their Scheme Shares in certificated form by the KW Transfer Agent, as part of the KW Transfer Agent's new client procedures, is needed in order to permit the New KW Shares to which the relevant Scheme Shareholder is entitled to be properly recorded in that Scheme Shareholder's name within the KW register. Failure to return such form(s) provided by the KW Transfer Agent as part of its new client procedures may affect the ability of the KW Transfer Agent to properly: (i) record New KW Shares in the name of the relevant Scheme Shareholder; (ii) send to that Scheme Shareholder any documentation relating to his or her holding; or (iii) pay that Scheme Shareholder any dividends or other revenues he or she may be entitled to, and may result in withholding tax being suffered on such payments to a greater extent than would otherwise be the case.

On the Effective Date, each certificate issued by KWE and representing a holding of Scheme Shares will cease to be valid. Following settlement under the Scheme, all Scheme Share certificates should be destroyed.

The New KW Shares will be listed on the NYSE, and holders of Scheme Shares in certificated form may find that holding and trading the New KW Shares directly involves US market practices and formalities that may be unfamiliar to such holders. Further information about the rights attaching to KW Shares is set out in Part Ten (*Description of KW Shares*) of this document.

31.2 KW Cash Component, KWE Special Distribution and Closing Dividend

(a) KW Cash Component and Fractional Entitlement Payments

In the case of Scheme Shares which, at the Scheme Record Time, are in uncertificated form, the KW Cash Component and any cash sum due in respect of fractional entitlements to New KW Shares or KW CDIs will be paid through CREST in pounds sterling, in accordance with the CREST payment arrangements. However, KW reserves the right to make payment of the KW Cash Component and any cash sum due in respect of fractional entitlements to New KW Shares or KW CDIs by cheque in each case in the manner described below with respect to Scheme Shares in certificated form if, for reasons outside its reasonable control, it is not able to effect settlement via CREST.

Where KW settles the KW Cash Component or any cash sum due in respect of fractional entitlements to New KW Shares or KW CDIs to a Scheme Shareholder via CREST, KW will procure that an instruction is given to Euroclear to create an assured payment obligation in favour of the appropriate CREST account. The creation of such assured payment obligation will be a complete discharge of the obligations of KW under the Scheme with reference to payments made through CREST. The instruction to Euroclear to create an assured payment obligation in favour of the appropriate CREST account will be given in such time as to enable such payments to be made within 14 days after the Effective Date.

In the case of Scheme Shares which at the Scheme Record Time are in certificated form, settlement of the KW Cash Component and any cash sum due in respect of fractional entitlements to New KW Shares or KW CDIs will be made in pounds sterling by cheque, or if practicable in the opinion of KW and KWE, via BACS.

Cheques shall be despatched within 14 days after the Effective Date. All deliveries of cheques required to be made pursuant to the Scheme will be effected by posting the same day in prepaid envelopes addressed to the persons entitled thereto at their respective addresses as appearing in the Register at the Scheme Record Time or, in the case of joint holders, at the address of that one of the joint holders whose name stands first in the Register in respect of such joint holding at the Scheme Record Time or in accordance with any special instructions regarding communications previously given to KWE, and none of KWE, KW nor any person or nominee appointed by KWE or KW in accordance with this Scheme nor their respective agents, will be responsible for any loss or delay in the transmission or delivery of any share certificates or cheques sent in this way and shall be sent at the risk of the person entitled thereto.

KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, all payments required from KW by the Scheme in respect of the KW Cash Component and any payment in respect of a fractional entitlement will be made free and clear of any and all deductions and/or withholdings in respect of any US federal income tax, provided that US federal backup withholding may apply under the circumstances described in Clause 8.11 of the Scheme. Instructions to Scheme Shareholders for completing and providing these forms to the Receiving Agent are described in the section entitled "Action to be Taken" and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

(b) Payment of the KWE Special Distribution

Subject to compliance with the provisions of the Jersey Companies Law, Scheme Shareholders on the Register at the Scheme Record Time who are entitled to receive the New Offer Consideration and Distribution will be entitled to receive the KWE Special Distribution under

the terms of the Scheme. Payment of the KWE Special Distribution will be made by KWE in accordance with the KWE Articles and the provisions of the Jersey Companies Law in pounds sterling by cheque, or, where practicable in the opinion of KWE, via CREST or BACS.

Payments made by cheque will be payable to the Scheme Shareholders who are entitled to the payment. Cheques shall be despatched within 14 days after the Effective Date. All deliveries of cheques required to be made pursuant to the Scheme will be effected by posting the same day in pre-paid envelopes addressed to the persons entitled thereto at their respective addresses as appearing in the Register at the Scheme Record Time or, in the case of joint holders, at the address of that one of the joint holders whose name stands first in the Register in respect of such joint holding at the Scheme Record Time or in accordance with any special instructions regarding communications previously given to KWE, and none of KWE, KW nor any person or nominee appointed by KWE or KW in accordance with this Scheme nor their respective agents will be responsible for any loss or delay in the transmission or delivery of any share certificates or cheques sent in this way and shall be sent at the risk of the person entitled thereto.

In circumstances where KWE decides to pay the KWE Special Distribution to a Scheme Shareholder via CREST, KWE will procure that an instruction is given to Euroclear to create an assured payment obligation in favour of the appropriate CREST account. The creation of such assured payment obligation will be a complete discharge of the obligations of KWE under the Scheme with reference to payments made through CREST. The instruction to Euroclear to create an assured payment obligation in favour of the appropriate CREST account will be given in such time as to enable such payments to be made within 14 days after the Effective Date.

In circumstances where KWE decides to pay the KWE Special Distribution via BACS, KWE will make appropriate arrangements for such payments having regard, where practicable, to the existing arrangements under which KWE remits dividend payments to KWE Shareholders and such payments will be made within 14 days after the Effective Date.

KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, all payments required from KWE by the Scheme in respect of the KWE Special Distribution will be made free and clear of any and all deductions and/or withholdings in respect of any US federal income tax, provided that US federal backup withholding may apply under the circumstances described in Clause 8.13 of the Scheme. Instructions to Scheme Shareholders for completing and providing these forms to the Receiving Agent are described in the section entitled "Action to be Taken" and Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of this document.

(c) Payment of the Closing Dividend

Subject to compliance with the provisions of the Jersey Companies Law, all KWE Shareholders on the Register at the Closing Dividend Record Time will be entitled to receive the Closing Dividend from KWE under the terms of the Scheme. The Closing Dividend will be paid on the First KW Dividend Payment Date (i.e. by 10 January 2018) and in accordance with the KWE Articles and the provisions of the Jersey Companies Law and in pounds sterling by cheque, or, where practicable in the opinion of KWE, via CREST or BACS.

Payments made by cheque will be payable to the KWE Shareholders who are entitled to the payment. All deliveries of cheques required to be made pursuant to the Scheme will be effected by posting the same day in pre-paid envelopes addressed to the persons entitled thereto at their respective addresses as appearing in the Register at the Closing Dividend Record Time or, in the case of joint holders, to the address of that one of the joint holders whose name stands first in the Register in respect of such joint holding at the Closing Dividend Record Time or in accordance with any special instructions regarding communications previously given to KWE, and none of KWE, KW nor any person or nominee appointed by KWE or KW in accordance with this Scheme nor their respective agents shall be responsible for any loss or delay in the transmission or delivery of any share certificates or cheques sent in this way and shall be sent at the risk of the person entitled thereto.

In circumstances where KWE agrees to pay the Closing Dividend to a KWE Shareholder via CREST, the relevant KWE Shareholder will receive the Closing Dividend from KWE through CREST. KWE will procure that an instruction is given to Euroclear to create an assured

payment obligation in favour of the appropriate CREST account. The creation of such assured payment obligation will be a complete discharge of the obligations of KWE under the Scheme with reference to payments made through CREST.

In circumstances where KWE decides to pay the Closing Dividend via BACS, KWE will make appropriate arrangements for such payments having regard, where practicable, to the existing arrangements under which KWE remits dividend payments to KWE Shareholders.

31.3 General

All documents and remittances sent by post will be sent at the risk of the person(s) entitled thereto.

Except with the consent of the Panel, settlement of the consideration to which any Scheme Shareholder is entitled under the Scheme will be implemented in full in accordance with the terms of the Scheme, free of any lien, right of set-off, counterclaim or other analogous right to which KW might otherwise be, or claim to be, entitled against such Scheme Shareholder.

32. Taxation

Your attention is drawn to Part Seven (*Taxation*) of this document, which contains a summary of certain aspects of the Jersey, UK and US tax treatment of the Merger. That summary relates only to the position of certain categories of beneficial owners of Scheme Shares and Scheme Shareholders (as explained further in Part Seven (*Taxation*) of this document), does not constitute tax advice and does not purport to be a complete analysis of all potential Jersey, UK or US tax consequences of the Merger.

If KWE were to elect to be treated as a partnership for US tax purposes at or before the time at which KW acquires the Scheme Shares pursuant to the Scheme, beneficial owners of Scheme Shares and Scheme Shareholders who are US Holders and who have made a QEF election with respect to their Scheme Shares may incur additional US tax liabilities in connection with the Scheme. Such beneficial owners of Scheme Shares and Scheme Shareholders may also incur other tax liabilities if this occurs after that time. KW undertakes for the benefit of all beneficial owners of Scheme Shares and Scheme Shareholders that it will not take any step, and will procure that none of its Affiliates will take any step, calculated to cause KWE to be treated as a partnership for US tax purposes effective on or prior to the date on which it acquires the Scheme Shares pursuant to the Scheme, except with the prior written consent of the KWE Independent Directors. The KWE Independent Directors will not give their consent to this unless, seven (7) days prior to the Scheme Court Meeting, they have announced their intention to do so through a Regulatory Information Service and unless KW undertakes not to take, and to prevent its Affiliates from taking, any such step until after the Court agrees to sanction the Scheme on the basis that this will occur.

Any US Holder (as defined in Part Seven (*Taxation*) of this document) who has made, or is considering making, a QEF election with respect to his or her Scheme Shares or which is a tax-exempt organisation, or a non-US Holder which is an entity controlled by a foreign government under the US federal income tax laws should note, in particular, the matters discussed in paragraphs 2.1(3), 2.1(4) and 2.2(3) of Part Seven (*Taxation*) of this document.

Beneficial owners of KWE Shares and KWE Shareholders who believe that they may prefer alternative tax treatments to those described in Part Seven (*Taxation*) should consider seeking independent professional advice on alternative courses of action that may be available to them (including disposing of their shares in the market rather than pursuant to the Scheme) in the event that KWE were to be treated as a partnership for US tax purposes.

The payment of the KW Cash Component, the KWE Special Distribution and/or any other amount (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer to Scheme Shareholders entitled to the New Offer may be subject to US federal backup withholding (currently, at a rate of 28%). An explanation of this backup withholding is given in paragraphs 2.1(6) and 2.2(5) of Part Seven (Taxation) of this document.

In accordance with Clause 8.10 of the Scheme, KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free and clear and without deduction for any US federal income tax

withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-8 or IRS Forms W-9 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-9 or, as applicable, IRS Forms W-8 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed IRS Forms W-9 or applicable IRS Forms W-8, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of IRS Form W-9 and applicable IRS Form W-8 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A SEPARATE APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 OR TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 OR IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 OR IRS FORM W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 OR IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

<u>US federal backup withholding is not an additional tax.</u> Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability) or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner.

FAILURE TO COMPLETE AND RETURN IRS FORMS W-9 OR, IF APPLICABLE, IRS FORMS W-8 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

The various versions of IRS Forms W-8 and W-9 for use by different categories of taxpayer are generally as follows:

- W-9 for use by beneficial owners of Scheme Shares which are "United States persons";
- W-8BEN for use by individual beneficial owners of Scheme Shares that are not "United States persons";
- W-8BEN-E for use by corporate and certain other non-individual beneficial owners of Scheme Shares that are not "United States persons"; and
- W-8IMY for use by non-US intermediaries and non-US partnerships (or entities treated as partnerships for US federal income tax purposes).

Beneficial owners of Scheme Shares and Scheme Shareholders should also review the discussion of US federal backup withholding in Part Seven (*Taxation*) of this document and consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e. Form W-9 or applicable Form W-8) and any supporting documentation to be provided to avoid potential US federal backup withholding.

In addition to the backup withholding regime, there are special rules regarding US taxation in relation to the KW Cash Component of the New Offer. For a more complete explanation of these, beneficial owners of KWE Shares and KWE Shareholders should carefully review Part Seven (*Taxation*) of this document and seek advice from their tax advisers.

In overview, these rules could, in certain circumstances, cause non-US Holders of Scheme Shares to be subject to US federal income tax in respect of the KW Cash Component despite the fact that KW pays the KW Cash Component without deduction or withholding of US federal income tax. In general, these rules could only apply if beneficial owners of Scheme Shares that own at least 50% of KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme come to own at least 50% of KW immediately after such time, in each case taking into account constructive ownership rules.

In that case, all or a portion of the cash received by a non-US Holder in respect of the KW Cash Component could be subject to US tax on deemed dividends by KW or treated as a gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme.

- Any such deemed dividend would be treated as paid by KW to the extent of its 2017 "earnings and profits" for US federal income tax purposes. This tax would not exceed 30% of the lesser of (a) the KW Cash Component received by the relevant non-US Holder and (b) the non-US Holder's attributable proportion of such earnings and profits (and may, in appropriate cases, be eligible for a reduced rate of US income taxation under an applicable tax treaty). Any earnings and profits of KW would be allocated ratably among actual and deemed dividends paid by KW in 2017. KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profits (for US federal income tax purposes) for the financial year ending 31 December 2017.
- In addition, even if, contrary to KW's stated expectation, KW were to have a material amount of earnings and profits for US federal income tax purposes for the financial year ending 31 December 2017, any non-US Holder whose percentage interest (taking into account the constructive ownership rules described below) in all of the outstanding KW Shares immediately after the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme (including, but not limited to, New KW Shares acquired pursuant to the Scheme, as well as any KW Shares otherwise acquired in connection therewith) is less than 80% of such non-US Holder's relative interest in KWE Shares immediately prior to such time (the "Less Than 80% Test") will not be treated as receiving a deemed dividend (and will thus not be subject to the 30% deemed dividend tax referred to above on any portion of the KW Cash Component received pursuant to the Scheme).

For example, a non-US Holder that held a 5% interest in KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme could not be subject to the 30% deemed dividend tax unless such non-US

Holder holds a 4% or greater interest in KW Shares immediately after such time. Accordingly, the 30% deemed dividend tax would not apply to any non-US Holder who does not actually or constructively hold any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme, but who only acquires KW Shares pursuant to the Scheme. Under constructive ownership rules, a non-US Holder generally will be treated as owning stock owned by certain related persons (such as an individual's spouse, children, grandchildren and parents), as well as, in certain circumstances, such holder's allocable share of stock held by any corporation (such as KWE shares owned by KW immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme) or any partnership, estate or trust, in each case, in which the non-US Holder or its relatives hold an interest. In addition, a non-US Holder which actually or constructively holds an option to purchase stock, is treated as owning the underlying stock.

A non-US Holder will not be subject to this form of 30% deemed dividend tax as a result of choosing to receive the consideration available under the New Offer or as a result of making any Additional Share Elections under the Mix and Match Facility if that non-US Holder does not own, actually or constructively, any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme. Accordingly, non-US Holders of Scheme Shares may wish to consider whether to avoid (or limit) any further acquisitions of KW Shares prior to such time. Those non-US Holders who already own KW Shares may also wish to avoid (or limit) any Additional Share Elections under the Mix and Match Facility, which could jeopardise their ability to satisfy the Less Than 80% Test, or to consider disposing of some or all of their KW Shares, or to reduce the KW Shares they will receive under the Scheme by reducing their holdings of KWE Shares in time, in either case insofar as that would remove or mitigate their potential liability to this form of deemed dividend taxation.

Assuming that a substantial number of beneficial owners of Scheme Shares satisfy the Less Than 80% Test and thus are not treated as if they had received deemed dividends, the remaining recipients of the KW Cash Component payable to non-US Holders (i.e. those which do not satisfy the Less Than 80% Test) would be allocated a significantly larger portion of KW's earnings and profits. However, a non-US holder will never be allocated earnings and profits in excess of the KW Cash Component it receives and would thus never be subject to the 30% deemed dividend tax in respect of an amount greater than the KW Cash Component it receives.

To the extent that the KW Cash Component is not treated as a deemed dividend on the basis described above, it may nonetheless be treated as gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme and subjected to US federal income tax if the non-US Holder owned (or was treated as owning) more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the deemed redemption and (ii) the non-US Holder's holding period in such shares (a "5% Holder"). This tax would be payable on the graduating scale of US federal income tax rates (including, if applicable, reduced tax rates on long-term capital gains derived by certain non-corporate holders, including individuals) on the amounts deemed to be received after allowing for certain reductions in respect of tax basis. The Less than 80% Test will not avoid or mitigate this tax.

Any beneficial owners of Scheme Shares who are also holders of KW Shares and who wish to avoid the potential application of these rules should consult their tax advisers, including as to the desirability and tax consequences of disposing of some or all their Scheme Shares or KW Shares in the market prior to the Merger.

Any person (whether or not a holder of shares in KW or KWE) who is considering acquiring shares in KW or KWE prior to the Effective Date should consider whether to refrain from doing so if, as a result, it may become liable to additional US tax as a result of the requirements described above.

This summary is, of necessity, a simplification of numerous complex tax considerations. Beneficial owners of Scheme Shares and Scheme Shareholders should not rely on it as a comprehensive statement of the potential tax implications of their receipt of the KW Cash

Component under the New Offer or the First KW Dividend and should in addition read and consider carefully the explanations in Part Seven (*Taxation*) of this document in full and seek independent tax advice on them.

33. Overseas Shareholders

Where legally permissible, Scheme Shareholders, and other KWE Shareholders, in all jurisdictions will receive this document and the accompanying Forms of Proxy to notify them of, and permit them to attend and/or vote for or against the Scheme at, the meetings at which they are entitled to vote. However, there may be restrictions on the extent to which certain shareholders of KWE in jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands and the Cayman Islands can receive either this document or an Election Form, make Elections and/or receive New KW Shares pursuant to the Scheme. The arrangements for such shareholders are summarised below and set out more fully in Clause 9 of the Scheme. Any persons who are resident in, or citizens or nationals of, or otherwise subject to the law of, any jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands should inform themselves about, and observe, any applicable legal or regulatory requirements. Any failure to comply with the applicable requirements may constitute a violation of the securities laws of any such jurisdiction. To the fullest extent permitted by applicable law, the companies and persons involved in the Merger disclaim any responsibility or liability for the violation of such requirements by any person.

The release, publication or distribution of this document and/or the Election Form in or into jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands may be restricted or prohibited by the laws of those jurisdictions and therefore persons into whose possession any Election Form comes should inform themselves about, and observe, any such restrictions or applicable requirements. Any failure to comply with such restrictions or applicable requirements may constitute a violation of the securities laws of any such jurisdiction. Neither this document nor any Election Form should be forwarded or transmitted in, into or from any jurisdiction in which to do so would give rise to a violation of the relevant laws in such jurisdiction.

The availability of the Scheme and the Merger to certain Overseas Shareholders (including the right to make an Election under the Mix and Match Facility or for the Original Offer) may be restricted or prohibited by the laws of the relevant jurisdiction. Overseas Shareholders should inform themselves about, and observe, any applicable legal or regulatory requirements. KWE Shareholders who are in any doubt regarding such matters should consult an appropriate independent professional adviser in the relevant jurisdiction without delay.

It is the responsibility of each Overseas Shareholder to satisfy itself as to the full observance of the laws of the relevant jurisdiction in connection therewith, including the obtaining of any governmental, exchange control or other consents which may be required, or the compliance with other necessary formalities which are required to be observed and the payment of any issue, transfer or other taxes due in such jurisdiction. In respect of any Scheme Shareholder who is resident, located or has a registered address in a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, or who KW or KWE reasonably believes to be a citizen, resident or national of, or subject to the law of, a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, if the law of a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands precludes (or precludes except after compliance by KW or KWE with any governmental or other consent or any registration, filing or other formality with which KW or KWE, as applicable, is unable to comply or compliance with which KW or KWE (in each case acting reasonably) regards as unduly onerous), or KW or KWE is advised that such law so precludes (i) the extension or availability of the Merger or the Election in, into or from such jurisdiction; (ii) an Election or the granting of a right to make an Election pursuant to the Merger; or (iii) the issue of New KW Shares or KW CDIs pursuant to the Merger, then KW or KWE may, in its discretion, determine that:

• one or more such Election shall not be valid or accepted in respect of any relevant Scheme Shareholder, in which case such Scheme Shareholder shall receive the New Offer Consideration and Distribution; or

- such New KW Shares (or KW CDIs) shall be sold, in which event the relevant New KW Shares (or KW CDIs) shall be issued to (or for the benefit of) such Scheme Shareholder and KW shall appoint a person who shall be authorised to act on behalf of such Scheme Shareholder to procure that any shares (or KW CDIs) in respect of which KW or KWE has made such determination shall, as soon as practicable following the Effective Date, be sold and the net proceeds of such sale shall be remitted to such Scheme Shareholder; or
- such New KW Shares (or KW CDIs) shall not be issued to such Scheme Shareholder but shall instead be issued to a nominee for such Scheme Shareholder appointed by KW on terms that the nominee shall, as soon as practicable following the Effective Date, sell the relevant New KW Shares (or KW CDIs) so issued and the net proceeds of such sale shall be remitted to such Scheme Shareholder.

Neither KW nor KWE will be liable to any Scheme Shareholder for making any determination or acting in accordance with this paragraph 33.

This document and the accompanying documents have been prepared for the purposes of complying with Jersey law, the Listing Rules and the Code and the information disclosed may not be the same as that which would have been disclosed if this document had been prepared in accordance with the laws of jurisdictions outside England and Wales and Jersey. Overseas Shareholders should consult their own legal and tax advisers with regard to the legal and tax consequences of the Scheme to their particular circumstances.

All KWE Shareholders (including, without limitation, nominees, trustees or custodians) who would, or otherwise intend to, forward this document and its accompanying documents to any jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands should seek appropriate independent professional advice before taking any action.

US Shareholders

The New KW Shares have not been and will not be registered under the US Securities Act or under the securities laws of any state or other jurisdiction of the United States and are expected to be issued in the United States in reliance upon the exemption from the registration requirements of the US Securities Act provided by Section 3(a)(10) thereof. Accordingly, the New KW Shares may not be offered, sold, resold, delivered, distributed or otherwise transferred, directly or indirectly, in or into the United States without an exemption from registration under the US Securities Act.

The New KW Shares generally should not be treated as "restricted securities" within the meaning of Rule 144(a)(3) under the US Securities Act and persons who receive securities under the Scheme (other than "affiliates" as described in the paragraph below) may resell them without restriction under the US Securities Act.

Under US securities laws, persons who are or will be deemed to be affiliates (as defined under the US Securities Act) of KW prior to or after the Effective Date may be subject to timing, manner of sale and volume restrictions on the resale in the United States of New KW Shares received in connection with the Scheme. Whether a person is an "affiliate" of a company for such purposes depends upon the circumstances, but affiliates of a company include individuals who, or entities that, control directly or indirectly, or are controlled by or are under common control with, that company, and may include certain officers and directors and significant shareholders of KW. Scheme Shareholders who believe they may be affiliates for the purposes of the US Securities Act should consult their own legal advisers prior to any resale of New KW Shares received under the Scheme.

For the purposes of qualifying for the exemption from the registration requirements of the US Securities Act afforded by Section 3(a)(10), the Court will be advised through counsel that KW will rely on the Section 3(a)(10) exemption based on the Court's sanctioning of the Scheme, which will be relied upon by KW as an approval of the Scheme following a hearing on the fairness of the terms and conditions of the Scheme to Scheme Shareholders, at which hearing all KWE Shareholders are entitled to attend in person or through counsel to support or oppose the sanctioning of the Scheme and with respect to which notification has been given to all KWE Shareholders.

The New KW Shares have not been approved or disapproved by the SEC, any state securities commission in the United States or any other US regulatory authority, nor have such authorities passed upon or determined the fairness or merits of the New KW Shares or upon the adequacy or accuracy of the information contained in this document. Any representation to the contrary is a criminal offence in the United States.

The Merger relates to the shares of a Jersey company and is being effected by means of a scheme of arrangement pursuant to the Jersey Companies Law. A transaction effected by means of a scheme of arrangement is not subject to the tender offer rules under the US Exchange Act and is exempt from the registration requirements under the US Securities Act. Accordingly, the Merger will be subject to disclosure requirements and practices applicable under the Code and Jersey law to schemes of arrangement, which are different from the disclosure and other requirements of the US tender offer and securities laws.

The shares of KWE are not registered under the US Exchange Act. The financial information in respect of KWE included in, or incorporated by reference into, the KW Prospectus (the publication and content of which are the responsibility solely of the KW Directors) and this document has been prepared in accordance with accounting standards applicable in the UK and therefore may not be comparable to the financial information of US companies or companies whose financial statements have been prepared in accordance with US generally accepted accounting principles ("US GAAP"). US GAAP differ in certain significant respects from International Financial Reporting Standards as adopted by the European Union. None of the financial information in, or incorporated by reference into, this document with respect to KWE has been audited in accordance with auditing standards generally accepted in the United States or the auditing standards of the Public Accounting Standards Oversights Board (United States).

It may be difficult for US holders of KWE Shares to enforce their rights and any claims they may have arising under US federal securities laws in connection with the Merger, since KWE is organised under the laws of a country other than the United States, and some of its officers and directors may be residents of countries other than the United States and most of the assets of KWE are located outside the United States. US holders of KWE Shares may not be able to sue a non-US company or its officers or directors in a non-US court for violations of US securities laws. Further, it may be difficult to compel a non-US company and its affiliates to subject themselves to a US court's jurisdiction or judgment.

The Merger may, in certain circumstances and subject to the consent of the Panel, be carried out by way of a Takeover Offer under the Jersey Companies Law rather than using a scheme of arrangement. If KW elects to implement the Merger by way of a Takeover Offer, such Takeover Offer will be subject to applicable US tender offer and securities laws and regulations, including any exemptions therefrom.

In accordance with normal practice under the Code and Jersey law, KW or its nominees, or its brokers (acting as agents), may from time to time make certain purchases of, or arrangements to purchase, KWE Shares outside the United States, other than pursuant to the Scheme of Arrangement, until the date on which the Merger becomes effective, lapses or is withdrawn. These purchases may occur either in the open market at prevailing prices or in private transactions at negotiated prices.

In addition, the Manager is entitled to an annual management fee (payable quarterly in arrear) equal to 1% of KWE's adjusted net asset value, half of which may be paid in certain circumstances in KWE Shares (for further information please see paragraph 15 of this Part Two). Any information about such purchases and/or fees will be disclosed, as required in the UK, will be reported to a Regulatory Information Service and will be available on the London Stock Exchange website at www.londonstockexchange.com.

34. Action to be taken

Your attention is drawn to the sections entitled "Important Notice about Voting on the Merger" commencing on page 12 of this document and "Action to be Taken" commencing on page 14 of this document.

35. Further information

Apart from completing, signing and returning the Forms of Proxy and Election Form, if you wish to make an Election, you do not need to take any further action at this stage.

36. Shareholder helpline

If you require assistance, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

The terms of the Scheme are set out in full in Part Four (*The Scheme of Arrangement*) of this document. Your attention is also drawn to the further information contained (or incorporated by reference) in this document which forms part of this Explanatory Statement.

Yours faithfully,

Alex Midgen
Duly authorised
For and on behalf of Rothschild
Dwayne Lysaght
Duly authorised
For and on behalf of J.P. Morgan Limited

PART THREE

CONDITIONS AND CERTAIN FURTHER TERMS OF THE SCHEME

Part A – Conditions to the Scheme

The Scheme will be conditional upon the Scheme becoming unconditional and effective, subject to the provisions of the Code, on or before the Longstop Date.

Scheme approval

- 1. The Scheme will be conditional upon:
 - (a) (i) approval of the Scheme at the Scheme Court Meeting (and at any separate class meeting which may be required or any adjournment thereof) by a majority in number of the Scheme Shareholders (or, if more than one class, of each class of Scheme Shareholders) on the register of members of KWE at the Scheme Voting Record Time, present and voting, either in person or by proxy, representing three-quarters or more of the votes attached to the Scheme Shares cast by those Scheme Shareholders (or class of Scheme Shareholders, as the case may be); and (ii) such Scheme Court Meeting being held on or before the later of (1) the 22nd day after the expected date of the Scheme Court Meeting set out in this document and (2) the fifth Business Day following the date on which the KW Shareholder Resolution has been duly passed (or, in each case, such later date (if any) as KW and KWE may agree);
 - (b) (i) all resolutions required to approve and implement the Scheme as set out in the notice of the KWE General Meeting (including, without limitation, the Special Resolution) being duly passed by the requisite majority of KWE Shareholders at the KWE General Meeting or at any adjournment thereof; and (ii) the KWE General Meeting being held on or before the later of (1) the 22nd day after the expected date of the KWE General Meeting set out in this document; and (2) the fifth Business Day following the date on which the KW Shareholder Resolution has been duly passed (or, in each case, such later date (if any) as KW and KWE may agree); and
 - (c) the sanction of the Scheme by the Court without modification, or with modification on terms acceptable to KW and KWE, at the Scheme Court Sanction Hearing and the delivery of the Court Order to the Registrar of Companies for registration.

Other conditions

2. In addition, KW and KWE have agreed that the Scheme will be conditional upon the following Conditions and, accordingly, the necessary actions to make the Scheme effective will not be taken unless such Conditions (as amended, if appropriate) have been satisfied or, where relevant, waived:

KW Shareholder approval

(a) the KW Shareholder Resolution being duly passed at the KW Special Meeting;

Listing of New KW Shares

(b) the NYSE having authorised the Listing upon official notice of issuance and not having withdrawn such authorisation;

Anti-trust

- (c) to the extent that a mandatory notification to the CCPC pursuant to section 18(1)(a) of the Irish Competition Act is required in respect of the Merger, the Merger having been notified to the CCPC and one of the following having occurred:
 - (i) the CCPC having informed KW of its determination pursuant to either section 21(2)(a) or section 22(3)(a) of the Irish Competition Act that the Merger may be put into effect;
 - (ii) the CCPC having informed KW of its determination pursuant to section 22(3)(c) of the Irish Competition Act that the Merger may be put into effect subject to condition(s) specified by the CCPC being complied with;

- (iii) the period specified in section 21(2) of the Irish Competition Act having lapsed without the CCPC having informed KW of its determination (if any) under section 21(2)(a) or (b) of the Irish Competition Act; or
- (iv) where applicable, if 120 working days after the appropriate date (as defined by section 19(6) of the Irish Competition Act) have elapsed without the CCPC having informed KW of its determination (if any) under section 22 of the Irish Competition Act in relation to the Merger;

Other Third Party clearances

- (d) (other than in respect of the anti-trust clearance referred to in Condition 2(c) above) all material notifications to and filings with Third Parties which are necessary having been made, and all necessary waiting and other time periods (including any extensions of such waiting and other time periods), in each case under any applicable legislation or regulation of any relevant jurisdiction having expired, lapsed or been terminated (as appropriate) and all statutory or regulatory obligations in any relevant jurisdiction having been complied with, in each case in connection with the Scheme or the Merger or the acquisition or proposed acquisition of any shares or other securities in, or control or management of, any member of the Wider KWE Group by any member of the Wider KW Group, where the direct consequence of a failure to make such a notification or filing or to wait for the expiry, lapse or termination of any such waiting or time period would be unlawful in any relevant jurisdiction and have a material adverse effect on the Wider KW Group or the Wider KWE Group, in each case, taken as a whole;
- (e) (other than in respect of the anti-trust clearance referred to in Condition 2(c) above) no Third Party having intervened (as defined below) and there not continuing to be outstanding any statute, regulation or order of any Third Party (in each case which is material in the context of the Wider KW Group or the Wider KWE Group, in each case, taken as a whole) which would:
 - (i) make the Scheme or the Merger or, in each case, its implementation or the acquisition or proposed acquisition by KW or any other member of the Wider KW Group of any shares or other securities in, or control or management of, any member of the Wider KWE Group void, illegal or unenforceable in any relevant jurisdiction, or otherwise materially restrain, prevent, prohibit, restrict or delay the same or impose material additional conditions or obligations with respect to the Scheme or the Merger or such acquisition, or otherwise materially impede, challenge or interfere with the Scheme or Merger or such acquisition, or require material amendment to the terms of the Scheme or Merger;
 - (ii) materially limit or delay, or impose any material limitations on, the ability of any member of the Wider KW Group or any member of the Wider KWE Group to acquire or to hold or to exercise effectively all or any rights of ownership in respect of shares or other securities in, or to exercise voting or management control over, any member of the Wider KWE Group;
 - (iii) require, prevent or materially delay the divestiture or materially alter the terms envisaged for any proposed divestiture by any member of the Wider KW Group of any shares or other equity securities in any member of the Wider KWE Group;
 - (iv) require, prevent or materially delay the divestiture or alter the terms envisaged for any proposed divestiture by any member of the Wider KW Group or by any member of the Wider KWE Group of all or any material portion of their respective businesses, assets or properties or materially limit the ability of any of them to conduct any material portion of their respective businesses or to own or control any material portion of their respective assets or properties or any part thereof;
 - (v) except pursuant to Articles 116 to 118 and Article 121 of the Jersey Companies Law, require any member of the Wider KW Group or of the Wider KWE Group to acquire, or to offer to acquire, any shares or other equity securities in any member of either group owned by any third party;

- (vi) materially limit the ability of any member of the Wider KW Group or of the Wider KWE Group to conduct or integrate or co-ordinate its business, or any material part of it, with the businesses or any material part of the businesses of any other member of the Wider KW Group or of the Wider KWE Group;
- (vii) result in any member of the Wider KWE Group or the Wider KW Group ceasing to be able to carry on business under any name under which it presently does so; or
- (viii) otherwise materially adversely affect any or all of the business, assets or profits of any member of the Wider KWE Group,
- and all applicable waiting and other time periods during which any Third Party could intervene under the laws of any relevant jurisdiction in respect of the Merger having expired, lapsed or been terminated;
- (f) (other than in respect of the anti-trust clearance referred to in Condition 2(c) above) all material Authorisations which are necessary in any relevant jurisdiction for or in respect of the Scheme or Merger or the acquisition or proposed acquisition of any shares or other securities in, or control or management of, any member of the Wider KWE Group by any member of the Wider KW Group or the carrying on by any member of the Wider KWE Group of its business having been obtained, in terms and in a form reasonably satisfactory to KW, from all appropriate Third Parties or from any persons or bodies with whom any member of the Wider KWE Group has entered into contractual arrangements, in each case where the absence of such material Authorisation would have a material adverse effect on the Wider KWE Group taken as a whole and all such material Authorisations remaining in full force and effect and there being no written notice of any intention to revoke, suspend, restrict, modify or not to renew any of the same;

Certain matters arising as a result of any arrangement, agreement etc.

- (g) except as Disclosed, there being no provision of any arrangement, agreement, licence, permit, franchise or other instrument to which any member of the Wider KWE Group is a party, or by or to which any such member or any of its assets is or are or may be bound, entitled or subject or any circumstance, which, in each case as a consequence of the Scheme or Merger or the acquisition or proposed acquisition of any shares or other securities in, or control or management of, any member of the Wider KWE Group by any member of the Wider KW Group would result in (in each case to an extent which is material in the context of the Wider KWE Group taken as a whole):
 - (i) any monies borrowed by or any other indebtedness or liabilities (actual or contingent) of, or any grant available to, any member of the Wider KWE Group being or becoming repayable or capable of being declared repayable immediately or prior to its stated maturity date or repayment date or the ability of any member of the Wider KWE Group to borrow monies or incur any indebtedness being withdrawn or inhibited or becoming capable of being withdrawn or inhibited;
 - (ii) the creation or enforcement of any mortgage, charge or other security interest over the whole or any material part of the business, property, assets or interests of any member of the Wider KWE Group or any such mortgage, charge or other security interest (wherever created, arising or having arisen) becoming enforceable;
 - (iii) any such arrangement, agreement, licence, permit, franchise or instrument, or the rights, liabilities, obligations or interests of any member of the Wider KWE Group thereunder, being, or becoming capable of being, terminated or adversely modified or affected or any adverse action being taken or any onerous obligation or liability arising thereunder;
 - (iv) any material asset or interest of any member of the Wider KWE Group being or falling to be disposed of or charged or ceasing to be available to any member of the Wider KWE Group or any right arising under which any such material asset or interest could be required to be disposed of or could cease to be available to any member of the Wider KWE Group, otherwise than, in each case, in the ordinary course of business;
 - (v) any member of the Wider KWE Group ceasing to be able to carry on business under any name under which it presently does so;

- (vi) the creation of material liabilities (actual or contingent) by any member of the Wider KWE Group, otherwise than in the ordinary course of business;
- (vii) the interests or business of any member of the Wider KWE Group in or with any other person, firm, company or body (or any arrangement or arrangements relating to any such interests or business) being adversely modified or affected; or
- (viii) the financial or trading position or the value of any member of the Wider KWE Group being prejudiced or adversely affected,

and no event having occurred which, under any provision of any such arrangement, agreement, licence, permit, franchise or other instrument, would, as a consequence of the Scheme or the Merger, result in any of the events or circumstances which are referred to in paragraphs (i) to (viii) of this Condition 2(g), in each case to an extent which is material in the context of the Wider KWE Group taken as a whole;

- (h) since 31 December 2016, except as Disclosed, no member of the Wider KWE Group having (in each case to an extent which is material in the context of the Wider KWE Group taken as a whole):
 - (i) issued or agreed to issue, or authorised the issue of, additional shares of any class, or securities convertible into or exchangeable for, or rights, warrants or options to subscribe for or acquire, any such shares or convertible securities or transferred or sold any shares out of treasury, other than as between KWE and wholly-owned subsidiaries of KWE or between such wholly-owned subsidiaries;
 - (ii) purchased or redeemed or repaid any of its own shares or other securities or reduced or made any other change to any part of its share capital;
 - (iii) recommended, declared, paid or made, or proposed to recommend, declare, pay or make, any bonus, dividend or other distribution (whether payable in cash or otherwise) or any bonus issue (other than (i) a Permitted KWE Distribution or (ii) to KWE or a wholly-owned subsidiary of KWE);
 - (iv) (other than any transaction in the ordinary course of business or a transaction as between KWE and its wholly-owned subsidiaries or between such wholly-owned subsidiaries) made or authorised any change in its loan capital;
 - (v) (other than any transaction or disposal in the ordinary course of business or a transaction as between KWE and a wholly-owned subsidiary of KWE or between such wholly-owned subsidiaries) merged with, demerged or acquired any body corporate, partnership or business or acquired or disposed of or, other than in the ordinary course of business, transferred, mortgaged, charged or created any security interest over any assets or any right, title or interest in any assets (including shares in any undertaking and trade investments) or authorised the same;
 - (vi) issued or authorised the issue of, or made any change in or to, any debentures or (except in the ordinary course of business or except as between KWE and its wholly-owned subsidiaries or between such wholly-owned subsidiaries) incurred or increased any indebtedness or liability (actual or contingent);
 - (vii) (other than any transaction in the ordinary course of business) entered into or varied, or authorised, proposed or announced its intention to enter into or vary, any material agreement, contract, transaction or commitment (whether in respect of capital expenditure or otherwise) which:
 - a. is of a long term, onerous or unusual nature or magnitude or which could reasonably be expected to involve an obligation of such nature or magnitude; or
 - b. would materially restrict the business of any member of the Wider KWE Group:
 - (viii) (other than in connection with the Merger and except as between KWE and its wholly-owned subsidiaries or between such wholly-owned subsidiaries) entered into, implemented, effected or authorised any merger, demerger, reconstruction, amalgamation, scheme, commitment or other transaction or arrangement in respect of itself or another member of the Wider KWE Group otherwise than in the ordinary course of business;

- (ix) entered into or varied to a material extent the terms of any contract, agreement or arrangement with any of the directors or senior executives of any member of the Wider KWE Group, save for salary increases, bonuses or variations of terms in the ordinary course of business or in connection with the Merger;
- (x) (other than in respect of a member of the Wider KWE Group which was solvent at the relevant time) taken any corporate action or had any legal proceedings instituted or threatened against it or petition presented or order made for its winding up (voluntarily or otherwise), dissolution or reorganisation or for the appointment of a receiver, administrator, administrative receiver, trustee or similar officer of all or a substantial part of its assets and revenues or any analogous proceedings in any jurisdiction or appointed any analogous person in any jurisdiction;
- (xi) been unable, or admitted in writing that it is unable, to pay its debts or having stopped or suspended (or threatened to stop or suspend) payment of its debts generally or ceased or threatened to cease carrying on all or a substantial part of its business;
- (xii) waived or compromised any claim, otherwise than in the ordinary course of business;
- (xiii) other than in connection with the Merger, made any material alteration to its memorandum or articles of association;
- (xiv) made or agreed or consented to:
 - a. any material change:
 - 1. to the terms of the trust deeds constituting the pension scheme(s) established for its directors, employees or their dependants;
 - 2. the contributions payable to any such scheme(s) or to the benefits which accrue or to the pensions which are payable thereunder;
 - 3. the basis on which qualification for, or accrual or entitlement to such benefits or pensions are calculated or determined;
 - 4. the basis upon which the liabilities (including pensions) or such pension schemes are funded, valued or made; or
 - b. any change to the trustees of such scheme(s), including the appointment of a trust corporation;
- (xv) proposed, agreed to provide or modified to a material extent the terms of any share option scheme, incentive scheme or other benefit relating to the employment or termination of employment of any person employed by the Wider KWE Group;
- (xvi) entered into any agreement, commitment or arrangement or passed any resolution or made any offer (which remains open for acceptance) or proposed or announced any intention with respect to any of the transactions, matters or events referred to in this Condition 2(h); or
- (xvii) entered into any contract, transaction or arrangement which would be materially restrictive on the business of any member of the Wider KWE Group or the Wider KW Group, other than to a nature and extent which is normal in the context of the business concerned;

No adverse change, litigation or regulatory enquiry

- (i) since 31 December 2016 and except as Disclosed:
 - (i) there having been no adverse change in the business, assets or profits of any member of the Wider KWE Group which, in any case, is material in the context of the Wider KWE Group taken as a whole;
 - (ii) no contingent or other liability of any member of the Wider KWE Group having arisen or increased (other than in the ordinary course of its business) which would materially adversely affect the business, assets or profits of any member of the Wider KWE Group to an extent which is material in the context of the Wider KWE Group taken as a whole:

- (iii) no litigation, arbitration proceedings, prosecution or other legal proceedings to which any member of the Wider KWE Group is or may become a party (whether as claimant, defendant or otherwise) having been threatened in writing, announced, implemented or instituted by or against or remaining outstanding against or in respect of any member of the Wider KWE Group, in each case to an extent which is material in the context of the Wider KWE Group taken as a whole; and
- (iv) (other than as a result of the Merger) no enquiry or investigation by, or complaint or reference to, any Third Party having been threatened in writing, announced, implemented, instituted by or against or remaining outstanding against or in respect of any member of the Wider KWE Group (in each case to an extent which is material in the context of the Wider KWE Group taken as a whole);

No discovery of certain matters

- (j) KW not having discovered, except as Disclosed, that:
 - (i) any financial or business or other information concerning the Wider KWE Group disclosed prior to the Original Announcement Date by or on behalf of any member of the Wider KWE Group publicly, or to any member of the Wider KW Group, is materially misleading or contains any material misrepresentation of fact or omits to state a fact necessary to make any information contained therein not materially misleading and which was not subsequently corrected by disclosure either publicly or otherwise to KW or its advisers, which in any case is material in the context of the Wider KWE Group taken as a whole;
 - (ii) any member of the Wider KWE Group is subject to any material liability (actual or contingent) which has not been Disclosed and which is material in the context of the Wider KWE Group taken as a whole;
 - (iii) any information which has not been Disclosed affects the import of any information which has been disclosed prior to the Original Announcement Date by or on behalf of any member of the Wider KWE Group and which is material in the context of the Wider KWE Group taken as a whole;
 - (iv) any past or present member of the Wider KWE Group has failed to comply with any and/or all applicable legislation or regulation of any jurisdiction with regard to the disposal, spillage, release, discharge, leak or emission of any waste or hazardous substance or any substance likely to impair the environment or harm human health or animal health or otherwise relating to environmental matters, or that there has otherwise been any such disposal, spillage, release, discharge, leak or emission (whether or not the same constituted a non-compliance by any person with any such legislation or regulations, and wherever the same may have taken place) any of which disposal, spillage, release, discharge, leak or emission would be likely to give rise to any material liability (actual or contingent) on the part of any member of the Wider KWE Group and which is material in the context of the Wider KWE Group taken as a whole; or
 - (v) there is, or is reasonably likely to be, any material liability (actual or contingent) of any past or present member of the Wider KWE Group to make good, repair, reinstate or clean up any property or any controlled waters now or previously owned, occupied, operated or made use of or controlled by any such past or present member of the Wider KWE Group, under any environmental legislation, regulation, notice, circular or order of any government, governmental, quasi-governmental, state or local government, supranational, statutory or other regulatory body, agency, court, association or any other person or body in any jurisdiction and which is material in the context of the Wider KWE Group taken as a whole; and

Anti-corruption, sanctions and criminal property

- (k) KW not having discovered (except as Disclosed) that:
 - (i) any past or present member, director, officer or employee of the Wider KWE Group is or has at any time during the course of such person's employment with any member of the Wider KWE Group engaged in any activity, practice or conduct

- which would constitute an offence under the Bribery Act 2010, the US Foreign Corrupt Practices Act of 1977 or any other anti-corruption legislation applicable to the Wider KWE Group:
- (ii) any asset of any member of the Wider KWE Group constitutes criminal property as defined by section 340(3) of the Proceeds of Crime Act 2002 (but disregarding paragraph (b) of that definition);
- (iii) any past or present member, director, officer or employee of the KWE Group is or has, at any time during the course of such person's employment with, or performance of services for, any member of the KWE Group, engaged in any business with, made any investments in, made any funds or assets available to or received any funds or assets from: (a) any government, entity or individual in respect of which US or European Union persons, or persons operating in those territories, are prohibited from engaging in activities or doing business, or from receiving or making available funds or economic resources, by US or European Union laws or regulations, including the economic sanctions administered by the United States Office of Foreign Assets Control or HM Treasury & Customs or (b) any government, entity or individual targeted by any of the economic sanctions of the United Nations, the United States, the European Union or any of its member states; or
- (iv) a member of the Wider KWE Group has engaged in any transaction which would cause any member of the Wider KW Group to be in breach of any law or regulation upon completion of the Merger, including the economic sanctions of the United States Office of Foreign Assets Control, or HM Treasury & Customs, or any government, entity or individual targeted by any of the economic sanctions of the United Nations, the United States, the European Union or any of its member states.

For the purpose of these Conditions:

- (a) "Third Party": means any central bank, government, government department or governmental, quasi-governmental, supranational, statutory, regulatory, environmental or investigative body, authority (including any national or supranational anti-trust or merger control authority), court, trade agency, association, institution or professional or environmental body or any other person or body whatsoever in any relevant jurisdiction, including, for the avoidance of doubt, the Panel;
- (b) a Third Party shall be regarded as having "intervened" if it has taken, instituted, implemented or threatened in writing any action, proceeding, suit, investigation, enquiry or reference or made, proposed or enacted any statute, regulation, decision or order or taken any measures or other steps (and, in each case, not having withdrawn the same) and "intervene" shall be construed accordingly; and
- (c) "Authorisations": means authorisations, orders, grants, recognitions, determinations, certificates, confirmations, consents, licences, clearances, provisions and approvals, in each case, of a Third Party.
- 3. In addition, the New Offer was, when originally announced, also conditional upon:
 - (a) the KW Prospectus being:
 - (i) approved by the UK Listing Authority; and
 - (ii) made available to the public in accordance with the prospectus rules made by the Financial Conduct Authority under section 73A of the Financial Services and Markets Act 2000 relating to the offer of securities to the public and admission of securities to trading on a regulated market; and
 - (b) to the extent required, the Jersey Financial Services Commission having consented to the circulation in Jersey of the offer of the New KW Shares pursuant to Article 8(2) of the Control of Borrowing (Jersey) Order 1958, as amended.

The Conditions in paragraphs 3(a) and (b) and the Condition in paragraph 2(c) of this Part Three have been satisfied in accordance with their terms as at the date of this document.

Part B: Certain further terms and conditions of the Merger

The Scheme will not become effective unless the Conditions have been satisfied or (to the extent capable of waiver) waived by KW or, where appropriate, have been determined by KW to be or remain satisfied by no later than the Longstop Date.

Subject to the requirements of the Panel, and to the extent permitted by law, KW reserves the right in its discretion to waive, in whole or in part, any of the Conditions in Part A of this Part Three above, except for Conditions 1(a)(i), 1(c), 2(a) and 2(b), which cannot be waived.

KW shall be under no obligation to waive (to the extent capable of waiver), to determine to be or remain satisfied or to treat as satisfied any of the Conditions in Part A of this Part Three above by a date earlier than the latest date specified above for the satisfaction of that Condition, notwithstanding that the other Conditions may at such earlier date have been waived or fulfilled and that there are, at such earlier date, no circumstances indicating that any Condition may not be capable of fulfilment.

Each of the Conditions shall be regarded as a separate condition and shall not be limited by reference to any other Condition.

KW reserves the right to elect to implement the Merger by way of a Takeover Offer, subject to the Panel's consent. In such event, the Takeover Offer will be implemented on the same terms and conditions, so far as applicable and subject to Part 18 of the Jersey Companies Law, as those which would apply to the Scheme subject to appropriate amendments to reflect the change in method of effecting the Merger, including (without limitation) an acceptance condition set at 90% (or such other percentage as KW may decide (subject to the Panel's consent)) (i) in nominal value of the shares to which such Offer relates; and (ii) of the voting rights attaching to those shares, provided that the acceptance condition will not be satisfied unless any member of the Wider KW Group shall have acquired or agreed to acquire (whether pursuant to the Offer or otherwise), directly or indirectly, KWE Shares carrying in aggregate more than 50% of the voting rights normally exercisable at a general meeting of KWE (including for this purpose, except to the extent otherwise agreed by the Panel, any such voting rights attaching to the KWE Shares that are unconditionally allotted or issued before the Offer becomes or is declared unconditional as to acceptances whether pursuant to exercise of any outstanding subscription rights or conversion rights or otherwise).

If the Panel requires KW to make an offer or offers for any KWE Shares under the provisions of Rule 9 of the Code, KW may make such alterations to the Conditions as are necessary to comply with the provisions of that Rule.

The Scheme will be governed by Jersey law and will be subject to the jurisdiction of the Court and the Panel. The Scheme will be subject to the applicable requirements of the Code, the Listing Rules, the FCA, the London Stock Exchange and the Jersey Financial Services Commission.

Under Rule 13.5(a) of the Code, KW may not invoke a Condition to the Merger so as to cause the Merger not to proceed, to lapse or to be withdrawn, unless the circumstances which give rise to the right to invoke the Condition are of material significance to KW in the context of the Merger. The Conditions contained in paragraphs 1(a), 1(c), 2(a) and 2(b) of Part A of this Part Three (and any acceptance condition if the Merger is implemented by means of a Takeover Offer) are not subject to this provision of the Code.

The KWE Shares will be acquired pursuant to the Merger fully paid and free from all liens, equitable interests, charges, encumbrances, rights of pre-emption and other third party rights of any nature whatsoever and together with all rights attaching to them as at the Original Announcement Date or subsequently attaching or accruing to them, save for any Permitted KWE Distributions and any dividend and/or distribution and/or other return of capital in respect of which a reduction has been made to the Consideration as specified below.

If, on or after the Original Announcement Date, any dividend, other distribution or other return of capital is declared, made or paid or becomes payable in respect of KWE Shares by reference to a record date falling prior to the Effective Date in excess of the Permitted KWE Distributions, KW shall have the right (without prejudice to any right of KW to invoke, with the consent of the Panel in accordance with Rule 13.5 of the Code, Condition 2(h)(iii) of this Part Three) to require the reduction of:

- (a) in the case of the New Offer:
 - (i) the KW Cash Component and, to the extent necessary, the KWE Special Distribution, by a cash amount equal to the gross amount of such dividend, other distribution or other return of capital; and/or

- (ii) the New KW Share Component by a number of New KW Shares having a value equal to not more than the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital); or
- (b) in the case of the Original Offer, the Original Offer Consideration by a number of New KW Shares having a value equal to not more than the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital),

in which case (i) any reference to the New Offer Consideration and Distribution and the Original Offer Consideration in this document shall be deemed to be a reference to such New Offer Consideration and Distribution or Original Offer Consideration as so reduced and (ii) KWE Shareholders will be entitled to receive and retain the amount by reference to which the consideration or distribution has been reduced.

To the extent that any such dividend and/or distribution and/or other return of capital is declared, made or paid or is payable in excess of the Permitted KWE Distributions and it is: (i) transferred pursuant to the Merger on a basis which entitles KW to receive the dividend or distribution and to retain it or (ii) cancelled prior to payment, the Consideration payable under the terms of the Merger will not be subject to change in accordance with this paragraph. Any exercise by KW of its rights referred to in this paragraph shall be the subject of an announcement and, for the avoidance of doubt, shall not be regarded as constituting any revision or variation of the terms of the Merger.

If, on or after the Original Announcement Date, any dividend, other distribution or other return of capital (other than a Permitted KW Dividend or an on-market share buyback of KW Shares at a price no greater than the then prevailing market price) is declared, made or paid or becomes payable in respect of KW Shares, KW shall be required to:

- (a) in the case of the New Offer, increase the New KW Share Component; or
- (b) in the case of the Original Offer, increase the Original Offer Consideration,

in each case by a number of New KW Shares to the extent necessary to ensure that the additional value from such increase is equal to the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital, less the gross amount of such dividend, other distribution or other return of capital).

Fractions of New KW Shares will not be issued to Scheme Shareholders. Instead, Scheme Shareholders who otherwise would have received a fraction of a New KW Share will instead receive an amount in cash rounded to the nearest cent, based on the amount obtained by multiplying such fraction by the average of the highest and lowest sales prices of KW Shares on the NYSE on each of the five consecutive trading days ending on the trading day that is two trading days prior to the Effective Date, except that individual entitlements of less than US\$5.00 will not be paid but will be retained for the benefit of the Combined Group.

The New KW Shares will be issued credited as fully paid and free from all Encumbrances and will rank pari passu in all respects with KW Shares in issue at the time the New KW Shares are issued pursuant to the Merger, including the right to receive and retain dividends and other distributions declared, made or paid on or after the Effective Date or by reference to a record date falling on or after the Effective Date by reference to a record date on or after the Effective Date (in each case whether or not wholly or partly in respect of a period which precedes the Effective Date). Application will be made to the NYSE for the New KW Shares to be listed on the NYSE.

The Merger is subject to, among other things, the satisfaction or waiver of the Conditions set out in this Part Three and such further terms as may be required to comply with provisions of the Code and any requirements of the Jersey Financial Services Commission.

The availability of the New KW Shares pursuant to the Merger to Scheme Shareholders who are persons resident in, or citizens or nationals of, or otherwise subject to the law of, jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands and the Cayman Islands or the ability of those persons to hold such shares may be restricted or prohibited by the laws of the relevant jurisdictions. Such persons should inform themselves about, and observe, any applicable legal

or regulatory requirements. KWE Shareholders who are in any doubt regarding such matters should consult an appropriate independent professional adviser in the relevant jurisdiction without delay.

PART FOUR

THE SCHEME OF ARRANGEMENT

ROYAL COURT OF JERSEY SAMEDI DIVISION

File No 2017/252

IN THE MATTER OF

KENNEDY WILSON EUROPE REAL ESTATE PLC (COMPANY NUMBER 114680)

AND IN THE MATTER OF

THE COMPANIES (JERSEY) LAW 1991

SCHEME OF ARRANGEMENT

(under Article 125 of the Companies (Jersey) Law 1991)

between

KENNEDY WILSON EUROPE REAL ESTATE PLC

and

THE SCHEME SHAREHOLDERS (as hereinafter defined)

PRELIMINARY

In this Scheme, unless inconsistent with the subject or context, the following expressions bear the following meanings:

"£", "pounds sterling" or "pence" the lawful currency of the UK;

"\$", "US\$", "US Dollars" or the lawful currency of the United States;

"cent"

"Additional Cash Election" an election under the Mix and Match Facility to receive a greater

proportion of the KW Cash Component for each Scheme Share than would otherwise have been payable under the terms of the

New Offer;

"Additional Share Election" an election under the Mix and Match Facility to receive a greater

proportion of the New KW Share Component for each Scheme Share than would otherwise have been required to be issued under

the terms of the New Offer;

"Applicable Exchange Ratio" the number of New KW Shares to be received pursuant to this

Scheme by a Scheme Shareholder in respect of one Scheme Share, calculated in accordance with such Scheme Shareholder's entitlements under the Original Offer or, as applicable, the New Offer and, if applicable, after satisfying elections under the Mix and Match Facility to the extent permissible in accordance with its

terms;

"Applicable Translation Rate" the number of US Dollars for each pound sterling under the US\$ / £

spot exchange rate derived from Bloomberg as at 5.00 p.m. on the Business Day immediately prior to the First KW Dividend

Payment Date;

"Business Day" a calendar day, other than a Saturday, Sunday or public or bank

holiday, when banks are open for business in London, Jersey and

New York;

"CDI" a CREST depository interest representing an entitlement to a share;

"certificated" or "in certificated in relation to a share or other security, means a share or other form" security which is not in uncertificated form (i.e. not in CREST); "Closing Dividend" the dividend referred to as the "Closing Dividend" payable by KWE on the basis specified in Clause 4.1; "Closing Dividend Record Time" the Scheme Record Time; "Closing Price" the closing middle market quotation of a share derived from the Daily Official List (in the case of KWE Shares) or the NYSE (in the case of KW Shares); "Code" the UK City Code on Takeovers and Mergers, as issued from time to time by the Panel; "Combined Group" KW and its subsidiary undertakings following the completion of the Merger, comprising the KW Group and the KWE Group; "Court" the Royal Court of Jersey; "Court Order" the Act of the Court sanctioning this Scheme under Article 125 of the Jersey Companies Law; "CREST" the computer system (as defined in the Uncertificated Securities Order) in respect of which Euroclear UK & Ireland Limited is the authorised operator (as defined in such order) in accordance with which securities may be held and transferred in uncertificated form; "CREST Receiving Agent" Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom, in its capacity as CREST receiving agent; "DTC" the Depository Trust Company; "Effective Date" the date upon which this Scheme becomes effective in accordance with its terms; "Election" a Mix and Match Election or an Original Offer Election; "Election Form" the form of Election which accompanies the Scheme Document; "Election Return Time" the latest time for lodging an Election Form or making an Electronic Election either under the Mix and Match Facility or for the Original Offer, being 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service); "Electronic Election" an Election validly made by a Scheme Shareholder who holds Scheme Shares in uncertificated form immediately prior to the Election Return Time; "Electronic Mix and Match a Mix and Match Election validly made by a Scheme Shareholder Election" in respect of Scheme Shares held in uncertificated form when the Mix and Match Election is submitted: "Electronic Original Offer an Original Offer Election validly made by a Scheme Shareholder in Election" respect of Scheme Shares held in uncertificated form when the Original Offer Election is submitted; any lien, equity, charge, hypothec, encumbrance, option, right of "Encumbrance" pre-emption and any other third party right and/or interest of any "Euroclear" Euroclear UK & Ireland Limited, a company incorporated under the laws of England and Wales with registered number 02878738; "Excluded Share" at any relevant time, a KWE Share beneficially owned at that time by, or on behalf of, KW, the Manager or any other member of the

KW Group (excluding, for the avoidance of doubt, any KWE Shares so held at the Latest Practicable Date after they have been transferred by any member of the KW Group to any director, officer or employee of any member of the KW Group prior to the

Scheme Record Time pursuant to the vesting of RSUs in connection with the Merger) and any KWE Shares held by KWE in treasury;

"Final KWE Dividend"

an amount of 12 pence per KWE Share;

"Final KWE Dividend Period"

the KWE Dividend Period in which the Effective Date falls;

"First KW Dividend"

the amount of the dividend per KW Share first paid by KW after the Effective Date in respect of the First KW Dividend Period or any part thereof;

"First KW Dividend Payment Date"

the date on which the First KW Dividend is paid, which shall be not later than 10 days after the expiration of the First KW Dividend Period;

"First KW Dividend Period"

the KW Dividend Period in which the Effective Date falls;

"Jersey Companies Law"

the Companies (Jersey) Law 1991, as amended from time to time;

"Jersey Working Day"

a calendar day (other than a Saturday, Sunday or public or bank holiday) when banks are open for business in Jersey;

"KW"

Kennedy-Wilson Holdings, Inc., a Delaware corporation;

"KW Cash Component"

300 pence per Scheme Share payable in cash by KW pursuant to the

New Offer;

"KW CDI"

a CREST depository interest representing an entitlement to a New KW Share;

"KW Dividend Period"

each period of three calendar months in the financial year of KW ending on (and including) 31 March, 30 June, 30 September and 31 December;

"KW Group"

KW, the Manager and any of their respective subsidiaries and subsidiary undertakings from time to time (excluding, for the avoidance of doubt, any member of the KWE Group);

"KW Payment"

the amount established as the KW Payment in accordance with Clause 4.1;

"KW Shares"

KW shares of common stock of US\$0.0001 each;

"KW Transfer Agent"

Continental Stock Transfer & Trust Company, One State Street Plaza, 30th Floor, New York, NY 10004-1561, United States of America:

"KWE"

Kennedy Wilson Europe Real Estate plc, incorporated in Jersey with registered number 114680;

"KWE Dividend Period"

each period of three calendar months in the financial year of KWE ending on (and including) 31 March, 30 June, 30 September and 31 December;

"KWE Group"

KWE, its subsidiaries and its subsidiary undertakings from time to

"KWE Meeting"

each of the Scheme Court Meeting and the KWE General Meeting;

"KWE Payment"

the amount established as the KWE Payment in accordance with Clause 4.1;

"KWE Shareholder"

a registered holder of KWE Shares from time to time;

"KWE Share"

an ordinary share of no par value in the capital of KWE;

"KWE Special Distribution"

the special distribution of 250 pence per Scheme Share payable in cash by KWE pursuant to the New Offer in respect of each Scheme Share but not, for the avoidance of doubt, any other KWE Shares;

"Latest Practicable Date"

the close of business on 8 September 2017, being the latest practicable Business Day before the date of this Scheme Document;

"Manager" KW Investment Management Ltd, a company incorporated in Jersey with registered number 114679, a wholly owned subsidiary of KW, which is the manager to KWE; the proposed acquisition of the entire issued and to be issued share "Merger" capital of KWE by KW (other than the Excluded Shares), to be implemented by way of this Scheme; "Mix and Match Election" an election validly made by a Scheme Shareholder who is entitled to receive the consideration to be delivered by KW under the New Offer to vary such consideration to the extent permitted by the Mix and Match Facility; "Mix and Match Facility" the arrangement under which a Scheme Shareholder entitled to receive the consideration to be delivered by KW under the New Offer may elect to vary the proportion in which it receives, pursuant to this Scheme, the New KW Share Component and the KW Cash Component (but, for the avoidance of doubt, not the KWE Special Distribution or the Closing Dividend) to the extent permitted by the Mix and Match Facility and except to the extent otherwise provided pursuant to Clause 9 of this Scheme; "New KW Share Component" 0.3854 New KW Shares in respect of each Scheme Share; "New KW Share" a KW Share which is to be issued pursuant to this Scheme; "New Offer" the offer pursuant to which a Scheme Shareholder is entitled to receive the New Offer Consideration and Distribution; "New Offer Consideration and the consideration and special distribution available pursuant to the Distribution" New Offer, consisting of the New KW Share Component, the KW Cash Component and the KWE Special Distribution in respect of each Scheme Share held at the Scheme Record Time, subject to any variation to the extent permitted by the terms of the Mix and Match Facility; "NYSE" the New York Stock Exchange; each dividend for any KWE Dividend Period, declared, paid or "Ordinary Quarterly Dividend" effected by KWE in an amount not exceeding 12 pence per KWE Share in aggregate in respect of a single KWE Dividend Period; "Original Offer" the offer pursuant to which a Scheme Shareholder is entitled to receive the Original Offer Consideration in respect of each Scheme Share held at the Scheme Record Time except to the extent otherwise provided pursuant to Clause 9 of this Scheme; "Original Offer Consideration" the consideration available under the Original Offer, consisting of 0.667 New KW Shares in respect of each Scheme Share held at the Scheme Record Time; "Original Offer Election" an election validly made by a Scheme Shareholder to receive the Original Offer Consideration except to the extent otherwise provided pursuant to Clause 9 of this Scheme; "Overseas Shareholders" KWE Shareholders who are resident in, ordinarily resident in, or citizens or nationals of, or otherwise subject to the law of, jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands; "Panel" the UK Panel on Takeovers and Mergers; "Penultimate KWE Dividend" a dividend of £0.12 per KWE Share to be paid by KWE in respect of the Penultimate KWE Dividend Period; "Penultimate KWE Dividend the KWE Dividend Period immediately preceding the Final KWE Period" Dividend Period:

quarterly dividends in amounts not exceeding, in aggregate, US\$0.17 per KW Share with respect to any KW Dividend Period preceding the First KW Dividend Period and US\$0.19 per KW

"Permitted KW Dividends"

Share in respect of the First KW Dividend Period which, in any such case, are payable by reference to a record date preceding the Effective Date, and any dividend paid by KW on or after the Effective Date to which Scheme Shareholders are entitled pursuant to this Scheme:

"Permitted KWE Distributions"

the KWE Special Distribution, the Closing Dividend and all Ordinary Quarterly Dividends;

"Pre-Effective Date Proportion"

the proportion which (a) the number of days from (and including) the first day of the Final KWE Dividend Period to (but excluding) the Effective Date bears to (b) the number of days from (and including) the first day of the Final KWE Dividend Period to (and including) the last day of the Final KWE Dividend Period;

"Receiving Agent"

Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom, in its capacity as receiving agent;

"Register"

the register of members of KWE;

"Registrar of Companies"

the registrar of companies for Jersey;

"Regulatory Information Service"

any information service authorised from time to time by the FCA for the purpose of disseminating regulatory announcements;

"Ring Fenced Employees"

certain employees of members of the KW Group who have been instructed by KW to assist KWE and the KWE Independent Directors in connection with the Merger;

"RSUs"

restricted share units issued by KW to directors, officers and employees of members of the KW Group entitling them to receive KWE Shares:

"Scheme" or "Scheme of Arrangement"

this scheme of arrangement under Article 125 of the Jersey Companies Law among KWE and the Scheme Shareholders, with or subject to any modification, addition or condition approved or imposed by the Court and agreed to by KWE and KW and sanctioned by the Court at the Scheme Court Sanction Hearing;

"Scheme Court Meeting"

the meeting or meetings of the Scheme Shareholders or any class or classes thereof to be convened by order of the Court pursuant to Article 125 of the Jersey Companies Law for the purpose of considering and, if thought fit, approving this Scheme and any adjournment, postponement or reconvention thereof;

"Scheme Court Sanction Hearing"

the hearing of the Court (and any adjournment thereof) to sanction this Scheme pursuant to Article 125 of the Jersey Companies Law, at which the Court Order is expected to be granted;

"Scheme Document"

the document in respect of this Scheme sent to (among others) KWE Shareholders, containing and setting out (among other matters) the full terms and conditions of this Scheme and containing the notices convening the KWE Meetings;

"Scheme Record Time"

6.00 p.m. on the most recent Business Day before the Effective Date:

"Scheme Share"

a KWE Share, which is not an Excluded Share and which is:

- a) in issue as at the date of the Scheme Document;
- transferred by any member of the KW Group to any director, officer or employee of any member of the KW Group after the KWE General Meeting and prior to the Scheme Record Time pursuant to the vesting of RSUs in connection with the Merger;
- c) (if any) issued after the date of the Scheme Document and prior to the Voting Record Time; or

d) (if any) issued on or after the Voting Record Time and before the Scheme Record Time, either on terms that the original or any subsequent holders thereof shall be bound by this Scheme or in respect of which the holders thereof shall have agreed in writing to be bound by this Scheme;

"Scheme Shareholders"

holders of Scheme Shares;

"subsidiary" and "subsidiary undertaking"

shall be construed in accordance with the UK Companies Act 2006;

"UK" or "United Kingdom"

the United Kingdom of Great Britain and Northern Ireland;

"uncertificated" or "in uncertificated form"

in relation to a share or other security, a share or other security which is recorded on the relevant register of the share or security concerned as being held in uncertificated form in CREST and title to which, by virtue of the Uncertificated Securities Order may be transferred by means of CREST;

"Uncertificated Securities Order"

the Companies (Uncertificated Securities) (Jersey) Order 1999, as amended;

"US" or "United States"

the United States of America, its territories and possessions, any state of the United States of America and the District of Columbia; and

"Voting Record Time"

6.00 p.m. on 6 October 2017 or, if the Scheme Court Meeting and/ or the KWE General Meeting are or is adjourned, 6.00 p.m. on the last date to fall two Jersey Working Days before the date of the adjourned meeting.

- (A) KWE was incorporated in Jersey with registered number 114680. As at the Latest Practicable Date, the issued share capital of KWE consisted of 126,133,407 ordinary shares each with no par value.
- (B) As at the Latest Practicable Date, the KW Group held 30,015,924 KWE Shares (or 29,483,694 KWE Shares, net of 532,230 KWE Shares underlying unvested RSUs that KW has granted to certain directors, officers and employees of members of the KW Group and which are unvested as at the Latest Practicable Date).
- (C) As at the Latest Practicable Date, KW and its subsidiaries held approximately 23.8% of the issued ordinary shares of KWE. Any KWE Shares held by or on behalf of any member of the KW Group (including, for the avoidance of doubt, any KWE Shares held by or on behalf of the Manager) will be Excluded Shares (i.e. not subject to the Scheme) and no member of the KW Group (including the Manager), or its nominee, is entitled to vote at the Scheme Court Meeting in respect of the Excluded Shares held by it.
- (D) The 23.8% of KWE Shares held by KW and/or its subsidiary undertakings as at the Latest Practicable Date included 532,230 KWE Shares held for the purpose of satisfying RSUs which will vest in connection with the Merger. It is intended that these RSUs will vest in consequence of the Merger after the KWE General Meeting but before the Scheme Record Time and that the 532,230 KWE Shares underlying these RSUs (less a certain number of KWE Shares withheld by KW to satisfy tax withholding obligations in accordance with the terms of the underlying award agreements) will be transferred to RSU holders (in accordance with the terms of the underlying award agreements) before the Scheme Record Time, i.e. in time for the relevant KWE Shares to become Scheme Shares and to be acquired by KW pursuant to the Scheme. However, irrespective of whether RSU holders or KW or any of its subsidiaries are holders of record of any KWE Shares to which these arrangements apply at any time when votes may be cast or proxies appointed for the Scheme Court Meeting, KW has undertaken to KWE to take all reasonable steps to procure that no proxies are appointed and no votes are cast at the Scheme Court Meeting in respect of any such underlying KWE Shares. However, votes attached to these underlying KWE Shares may be freely cast at the KWE General Meeting and Elections in respect of the underlying KWE Shares may be made prior to the Election Return Time.

- (E) As at the Latest Practicable Date, certain directors, officers and employees of members of the KW Group involved in KW's preparations for the Merger and the Ring Fenced Employees already held, in aggregate, 427,129 KWE Shares, representing approximately 0.34% of the issued KWE Shares as at the Latest Practicable Date. KWE Shares held by these KWE Shareholders will be Scheme Shares (i.e. subject to the Scheme), but KW has undertaken to KWE to take all reasonable steps to procure that these KWE Shareholders will not vote at the Scheme Court Meeting, with a view to preserving neutrality of voting at the Scheme Court Meeting. Each such KWE Shareholder can, however, choose to vote at the KWE General Meeting and to make any of the available Elections with respect to his or her Scheme Shares.
- (F) KW has agreed to appear by counsel at the Scheme Court Sanction Hearing and has agreed to consent to the Scheme, to be bound by, and to undertake to the Court to be bound by and comply with this Scheme and to execute and do and procure to be executed and done all such documents, acts and things as may be necessary or desirable to be executed or done by each of them respectively for the purpose of giving effect to this Scheme.
- (G) The provisions of this Scheme are subject to the Court sanctioning the Scheme at the Scheme Court Sanction Hearing and, accordingly, will not be effective until the Court Order has been delivered to the Registrar of Companies. The Court Order must be delivered to the Registrar of Companies within 14 days after the Scheme Court Sanction Hearing.
- (H) The Scheme Court Sanction Hearing has been convened for 10.00 a.m. on 18 October 2017 at the Royal Court of Jersey, Royal Court Building, St Helier, JE1 1BA, Jersey, Channel Islands. Scheme Shareholders may, if they wish, attend and be heard at the Scheme Court Sanction Hearing, in person or through counsel, to support or oppose the Court's sanctioning of the Scheme.

SCHEME OF ARRANGEMENT

1. Interpretation

- 1.1 In this Scheme, except where the context otherwise requires or otherwise as expressly provided:
 - (1) references to Recitals, Parts, Clauses and Sub-clauses are references to the Recitals, Parts, Clauses and Sub-clauses respectively of this Scheme;
 - (2) references to a "person" include references to an individual, firm, partnership, company, corporation, unincorporated body of persons or any state or state agency;
 - (3) references to a statute, statutory provision, enactment or subordinate legislation include the same as subsequently modified, amended or re-enacted from time to time;
 - (4) references to an agreement, deed or document shall be deemed also to refer to such agreement, deed or document as amended, supplemented, restated, verified, replaced and/or novated (in whole or in part) from time to time and to any agreement, deed or document executed pursuant thereto;
 - (5) any phrase introduced by the terms "including", "include", "in particular" or any similar expression is to be construed as illustrative only and does not limit the sense of the words preceding those terms;
 - (6) the singular includes the plural and vice-versa and words importing one gender shall include all genders;
 - (7) headings of Recitals, Parts, Clauses and Sub-clauses are for ease of reference only and shall not affect the interpretation of this Scheme; and
 - (8) all references to time are references to London time.

2. Transfer of the Scheme Shares

2.1 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme, KW shall acquire all the Scheme Shares fully paid up, free from all Encumbrances and together with all rights attaching to the Scheme Shares at the Effective Date or thereafter, including voting rights and the entitlement to all dividends and other distributions declared, paid or effected by KWE on or after the Effective Date, other than (i) the KWE Special Distribution, (ii) any Ordinary Quarterly Dividend in respect of a KWE Dividend Period up to and including the Final KWE Dividend Period payable on any Scheme Share and (iii) the Closing Dividend payable on any Scheme Share.

2.2 For the purposes of such acquisition, the Scheme Shares shall be transferred from the Scheme Shareholders to KW by means of a form of transfer or other instrument or instruction of transfer and, to give effect to such transfers, any person may be appointed by KW as attorney, agent or otherwise of each Scheme Shareholder and shall be authorised as such attorney, agent or otherwise on behalf of the relevant Scheme Shareholder to execute and deliver as transferor a form of transfer or other instrument or instruction of transfer of, or to procure the transfer by means of CREST or otherwise give any instructions to transfer (by deed or otherwise), the Scheme Shares and every form of transfer, other instrument or instruction of transfer executed or so given shall be effective as if it had been executed by the holder or holders of the Scheme Shares thereby transferred.

3. Consideration for the transfer of the Scheme Shares

- 3.1 Subject to the remaining provisions of this Clause 3 and subject to Clauses 5 to 7, where a Scheme Share of a Scheme Shareholder is transferred to KW as provided for in Clause 2.1:
 - (1) if no valid Original Offer Election has been made in respect of that Scheme Share, KW shall deliver the New KW Share Component and the KW Cash Component, and KWE shall (subject to compliance with the provisions of the Jersey Companies Law) pay the KWE Special Distribution to the Scheme Shareholder holding such Scheme Share; and
 - (2) if a valid Original Offer Election has been made in respect of that Scheme Share, KW shall deliver the Original Offer Consideration to the Scheme Shareholder holding such Scheme Share.
- 3.2 In addition, subject to the Scheme becoming effective and to compliance with the provisions of the Jersey Companies Law, KWE shall pay, and KW shall procure the payment by KWE of, the Closing Dividend to each KWE Shareholder at the Closing Dividend Record Time in accordance with Clause 4 and the other terms of this Scheme.
- 3.3 All payments and deliveries to a holder of a Scheme Share or a KWE Share (as applicable) will be to the holder whose name appears in the Register at the Scheme Record Time (or, in the case of the Closing Dividend, at the Closing Dividend Record Time), except that, in the case of joint holders, payments and deliveries shall be made to that one of the joint holders whose name stands first in the Register in respect of the relevant Scheme Shares at the Scheme Record Time (or, in the case of the Closing Dividend, at the Closing Dividend Record Time).
- 3.4 Notwithstanding anything to the contrary herein, if a Scheme Shareholder makes no valid Election in respect of any Scheme Shares held by such Scheme Shareholder in accordance with the terms of the Scheme, subject to the Scheme becoming effective and subject to the provisions regarding fractional entitlements set out in Clause 3.5, such Scheme Shareholder shall receive, in respect of such Scheme Shares, the New Offer Consideration and Distribution. Subject to the Scheme becoming effective, KWE Shareholders shall, in all cases, be entitled to receive the Closing Dividend.
- 3.5 No fraction of a New KW Share shall be issued to any Scheme Shareholder pursuant to this Scheme. A Scheme Shareholder who otherwise would have received a fraction of a New KW Share under Clause 3.1 shall instead receive an amount in cash rounded to the nearest cent, based on the amount obtained by multiplying such fraction by the average of the highest and lowest sales prices of KW Shares on the NYSE on each of the five consecutive trading days ending on the trading day that is two trading days prior to the Effective Date, provided that individual entitlements of less than US\$5.00 will not be paid to any Scheme Shareholder but will be retained by the Combined Group for its own benefit. For the purposes of determining fractional entitlements, each portion of a Scheme Shareholder's holding which is recorded in the Register by reference to a separate designation at the Scheme Record Time, whether in certificated or uncertificated form, shall be treated as a separate holding. Fractional entitlements will be determined after allowing for the outcome of Mix and Match Elections and Original Offer Elections.
- 3.6 The New KW Shares to be issued pursuant to Clause 3.1 shall be issued credited as fully paid and free from all Encumbrances, shall rank *pari passu* in all respects with all other fully paid KW Shares in issue on the Effective Date and shall carry the right to receive all dividends and other distributions declared, paid or made by KW on or after the Effective Date or by reference to any record date on or after the Effective Date (in each case whether or not wholly or partly in respect of a period which precedes the Effective Date).

4. Closing Dividend; Ordinary Quarterly Dividends and First KW Dividend

4.1 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme, each KWE Shareholder shall be entitled to receive, in respect of each KWE Share held by such KWE Shareholder at the Closing Dividend Record Time, a closing dividend (the "Closing Dividend") payable on the First KW Dividend Payment Date of an amount (rounded to the second decimal place) determined by the following formula:

KWE Payment less KW Payment

Where the amount of the **KWE Payment** is determined by the following formula:

$$A+(B \times D)$$

Where the amount of the KW Payment is determined by the following formula:

$$\frac{C \times D \times E}{F}$$

And where:

A = the amount of the Penultimate KWE Dividend per KWE Share;

B = the amount of the Final KWE Dividend per KWE Share;

C = the amount of the First KW Dividend per New KW Share (before any reduction for any US federal withholding tax);

D = the Pre-Effective Date Proportion;

E = the Applicable Exchange Ratio; and

F =the Applicable Translation Rate.

- 4.2 Subject to Clause 4.3, if, on or after 24 April 2017, any dividend, other distribution or other return of capital is declared, made or paid or becomes payable in respect of KWE Shares by reference to a record date falling prior to the Effective Date in excess of the Permitted KWE Distributions, KW shall have the right to require the reduction of:
 - (1) in the case of the New Offer:
 - (a) the KW Cash Component and, to the extent necessary, the KWE Special Distribution, by a cash amount equal to the gross amount of such dividend, other distribution or other return of capital; and/or
 - (b) the New KW Share Component by a number of New KW Shares having a value equal to not more than the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital); and
 - (2) in the case of the Original Offer, the Original Offer Consideration by a number of New KW Shares having a value equal to not more than the gross amount of such dividend, distribution or return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital),

in which case (i) any reference to the New Offer Consideration and Distribution and the Original Offer Consideration in this Scheme shall be deemed to be a reference to such New Offer Consideration and Distribution and Original Offer Consideration as so reduced, and (ii) KWE Shareholders will be entitled to receive and retain the amount by reference to which the New Offer Consideration and Distribution or the Original Offer Consideration (as applicable) has been reduced. KW shall make an announcement of any exercise of its rights under this Clause 4.2 via a Regulatory Information Service. For the avoidance of doubt, any exercise by KW of its rights under this Clause 4.2 shall not be regarded as constituting any revision or variation of the terms of the Merger.

4.3 To the extent that any dividend, distribution or other return of capital is declared, made or paid or is payable in respect of KWE Shares by reference to a record date falling prior to the Effective Date in excess of the Permitted KWE Distributions and it is: (i) transferred pursuant

to the Merger on a basis which entitles KW to receive the dividend or distribution and to retain it or (ii) cancelled prior to payment, KW shall not be entitled to reduce the New Offer Consideration and Distribution or the Original Offer Consideration, notwithstanding Clause 4.2.

- 4.4 Other than a Permitted KW Dividend or an on-market share buyback of KW Shares at a price no greater than the then prevailing market price, KW shall not, from the date of the Scheme Document until after the Effective Date, effect or announce any intention to effect any distribution, share redemption or capital reduction, whether in or in exchange for cash or other assets.
- 4.5 Without prejudice to Clause 4.4, if, on or after 24 April 2017, any dividend, other distribution or other return of capital (other than a Permitted KW Dividend or an on-market share buyback of KW Shares at a price no greater than the then prevailing market price) is declared, made or paid or becomes payable in respect of KW Shares, KW shall be required to:
 - (1) in the case of the New Offer, increase the New KW Share Component; or
 - (2) in the case of the Original Offer, increase the Original Offer Consideration,

in each case by a number of New KW Shares to the extent necessary to ensure that the additional value from such increase is equal to the gross amount of any such dividend, other distribution or other return of capital (for this purpose, the KW Shares shall be valued at their Closing Price on the Business Day immediately prior to the announcement of such dividend, other distribution or other return of capital, less the gross amount of such dividend, other distribution or other return of capital) and in which case any reference to the KW Share Component and the Original Offer Consideration in this Scheme shall be deemed to be a reference to such consideration as so increased. KW shall make an announcement of any event which requires any adjustment to the New Offer Consideration and Distribution and the Original Offer Consideration pursuant to this Clause 4.5, and of the adjustment so required, via a Regulatory Information Service.

4.6 If the KWE Special Distribution is not paid if, when and to the extent required by this Scheme, the arrangements under this Clause 4 will not have effect.

5. Original Offer Elections

- 5.1 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme, and subject to the provisions regarding fractional entitlements in Clause 3.5, a Scheme Shareholder that makes no Election in respect of a Scheme Share shall be entitled to receive the New Offer Consideration and Distribution for each such Scheme Share held at the Scheme Record Time.
- 5.2 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme, and except to the extent otherwise provided pursuant to Clause 9, by making a valid Original Offer Election, a Scheme Shareholder will become entitled to receive the Original Offer Consideration, and not the New Offer Consideration and Distribution, for each Scheme Share held at the Scheme Record Time in respect of which the valid Original Offer Election is made.
- 5.3 An Original Offer Election made by a Scheme Shareholder shall not affect the entitlements of any other Scheme Shareholder under the Scheme.
- 5.4 For the avoidance of doubt, any Scheme Shareholder who makes an Original Offer Election in respect of all or part of its holding of Scheme Shares shall not be entitled to make any Mix and Match Election in respect of the Scheme Shares to which the Original Offer Election relates. Any purported Mix and Match Election by a Scheme Shareholder who has elected to make an Original Offer Election shall be void in respect of the Scheme Shares to which such Original Offer Election relates.
- 5.5 Each Original Offer Election made by a Scheme Shareholder in respect of Scheme Shares held in certificated form shall be made by completing the section of the Election Form relating to the Original Offer Election, which Election Form shall be executed by the Scheme Shareholder or its duly authorised agent (or, in the case of a body corporate, executed by an authorised representative). To be effective, the relevant sections of the Election Form which relate to the Original Offer Election must be completed and such Election Form must be returned in accordance with the instructions printed thereon so as to arrive at the offices of the KWE Registrar by no later than the Election Return Time.

- 5.6 Each Original Offer Election made by a Scheme Shareholder in respect of Scheme Shares held in uncertificated form shall be made by submitting an Electronic Original Offer Election. To be effective, the Electronic Original Offer Election must be submitted by such a holder of Scheme Shares and received by the KWE Registrar by no later than the Election Return Time.
- 5.7 If an Election Form in relation to an Original Offer or an Electronic Original Offer Election is received by the KWE Registrar after the Election Return Time, or if such Election Form or Electronic Original Offer Election is received by the KWE Registrar before such time but is not, or is deemed not to be, valid and complete in all respects at such time, then such Original Offer Election shall, for all purposes, be void unless and to the extent that KWE and KW, in their absolute discretion, elect to treat as valid in whole or in part any such Original Offer Election.
- 5.8 Upon execution and delivery by a Scheme Shareholder of a valid Election Form in relation to the Original Offer or the making of a valid Electronic Original Offer Election, such Scheme Shareholder shall be bound by the terms and provisions contained in such Election Form or Electronic Original Offer Election (as the case may be) and by the terms and provisions contained in Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of the Scheme Document.
- 5.9 An Election Form in relation to the Original Offer duly completed and delivered or an Electronic Original Offer Election made in accordance with this Clause 5 may be withdrawn by notice to the KWE Registrar in writing to be received by the Election Return Time.
- 5.10 If a Scheme Shareholder delivers more than one Election Form in relation to the Original Offer Election or Electronic Original Offer Election in respect of its holding of Scheme Shares, in the case of an inconsistency between such Election Forms or Electronic Original Offer Elections, the last valid Election Form in relation to the Original Offer or Electronic Original Offer Election which is delivered by the Election Return Time shall prevail over any earlier Election Form in relation to the Original Offer or Electronic Original Offer Election. The delivery time for an Election Form in relation to the Original Offer or Electronic Original Offer Election shall be determined on the basis of which such Election Form or Electronic Original Offer Election is last sent or, if the KWE Registrar is unable to determine which is last sent, is last received. Election Forms in relation to the Original Offer which are sent in the same envelope shall be treated for these purposes as having been sent and received at the same time, and, in the case of an inconsistency between such Election Forms, none of them shall be treated as valid (unless KWE and KW otherwise determine in their absolute discretion).
- 5.11 Minor adjustments to the entitlements of a Scheme Shareholder pursuant to an Original Offer Election made under this Scheme may be made by the KWE Registrar with the prior consent of KWE and KW on a basis that KWE and KW consider to be fair and reasonable to the extent necessary to satisfy all entitlements pursuant to Original Offer Elections as nearly as may be practicable. Such adjustments shall be final and binding on such Scheme Shareholder.
- 5.12 No Original Offer Election may be made by a KWE Shareholder who is resident, located or has a registered address in, or is a citizen or national of, or otherwise subject to the law of, a jurisdiction where the extension or availability of the Merger, or any Original Offer Election or the granting of a right to make an Original Offer Election pursuant to Clause 3 and Clauses 5 to 7, would breach any applicable law. Any purported Offer Election by such a KWE Shareholder shall be void.
- 5.13 By submitting an Original Offer Election, the KWE Shareholder making such Original Offer Election undertakes not to sell or encumber any KWE Shares to which such Election relates unless and until that Election has been validly withdrawn.

6. Mix and Match Facility

6.1 A Mix and Match Election may be made only by a Scheme Shareholder who is entitled to receive the New Offer Consideration and Distribution except to the extent otherwise provided pursuant to Clause 9. Any Scheme Shareholder who makes an Original Offer Election in respect of all or part of its holding of Scheme Shares is not entitled to make any Mix and Match Election in respect of the Scheme Shares to which the Original Offer Election relates. Any purported Mix and Match Election by a Scheme Shareholder who has elected to make an Original Offer Election will be void in respect of the Scheme Shares to which such Original Offer Election relates.

- 6.2 The New KW Share Component and the KW Cash Component available under the New Offer, to the extent due to a Scheme Shareholder with respect to a Scheme Share, shall be subject to any valid Mix and Match Election made by such Scheme Shareholder. However, the Mix and Match Facility will not enable any Scheme Shareholder to vary the amount of the KWE Special Distribution receivable in connection with the New Offer. It will also not enable any Scheme Shareholder to vary the amount of cash receivable in respect of the Closing Dividend.
- 6.3 A Mix and Match Election made by a Scheme Shareholder hereunder shall not affect the entitlement to the New Offer Consideration and Distribution of a Scheme Shareholder who is entitled to receive the New Offer Consideration and Distribution but has not made any such Mix and Match Election or of a Scheme Shareholder who has not made any valid Election hereunder in respect of certain Scheme Shares (and is therefore entitled to the New Offer Consideration and Distribution in respect of those Scheme Shares), nor shall it affect the entitlement of any Scheme Shareholder to receive the Original Offer Consideration.
- 6.4 A Mix and Match Election shall only be accepted in respect of a whole number of Scheme Shares. Any Mix and Match Election which is made in respect of a number of Scheme Shares which is not a whole number shall be deemed to be made in respect of the nearest whole number of Scheme Shares when rounded down.
- 6.5 A Scheme Shareholder may make a Mix and Match Election in respect of all or part of the holding of Scheme Shares in respect of which the shareholder is entitled to receive the New Offer Consideration and Distribution.
- 6.6 The following provisions shall apply:
 - (1) the aggregate number of New KW Shares to be issued to Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution shall not be increased or decreased as a result of Mix and Match Elections, save to the extent required to accommodate rounding of individual entitlements to the nearest whole New KW Share;
 - (2) the aggregate amount of the KW Cash Component to be paid to Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution shall not be increased or decreased as a result of Mix and Match Elections made pursuant to this Clause 6, save to the extent required to accommodate rounding of individual entitlements to the nearest whole pence;
 - (3) Additional Share Elections validly made by Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution shall be satisfied:
 - (a) on the basis that for every 300 pence of cash which such Scheme Shareholders would be entitled to receive for one Scheme Share pursuant to the KW Cash Component absent such Mix and Match Election, they will be entitled to receive 0.2069 additional New KW Shares per Scheme Share (subject to Clause 6.6(3)(b)); but
 - (b) only to the extent that other Scheme Shareholders make valid countervailing Additional Cash Elections; and
 - (4) Additional Cash Elections validly made by Scheme Shareholders who are entitled to receive the New Offer Consideration and Distribution shall be satisfied:
 - (a) on the basis that for every 0.3854 New KW Shares which such Scheme Shareholders would be entitled to receive for one Scheme Share absent such Mix and Match Election, they will be entitled to receive 559 pence in cash per Scheme Share (subject to Clause 6.6(4)(b)); but
 - (b) only to the extent that other Scheme Shareholders make valid countervailing Additional Share Elections.
- 6.7 If there are Additional Share Elections which cannot be satisfied in full due to an insufficiency of countervailing Additional Cash Elections, the number of Scheme Shares to which each Additional Share Election relates will be reduced by the same proportion to the extent required to permit the unreduced Additional Share Elections to be satisfied in full (but not further); and Scheme Shares to which, as a result of such reduction, unsatisfied Additional Share Elections relate will be deemed to be Scheme Shares in respect of which no Mix and Match Election has been made.

- 6.8 If there are Additional Cash Elections which cannot be satisfied in full due to an insufficiency of countervailing Additional Share Elections, the number of Scheme Shares to which each Additional Cash Election relates will be reduced by the same proportion to the extent required to permit the unreduced Additional Cash Elections to be satisfied in full (but not further); and Scheme Shares to which, as a result of such reduction, unsatisfied Additional Cash Elections relate will be deemed to be Scheme Shares in respect of which no Mix and Match Election has been made.
- 6.9 Each Mix and Match Election made by a Scheme Shareholder in respect of Scheme Shares held in certificated form shall be made by completing the relevant sections of the Election Form relating to the Mix and Match Election, which Election Form shall be executed by the Scheme Shareholder or its duly authorised agent (or, in the case of a body corporate, executed by an authorised representative). To be effective, the relevant sections of the Election Form which relate to the Mix and Match Election must be completed and such Election From must be returned in accordance with the instructions printed thereon so as to arrive at the offices of the KWE Registrar by no later than the Election Return Time.
- 6.10 Each Mix and Match Election made by a Scheme Shareholder in respect of Scheme Shares held in uncertificated form shall be made by submitting an Electronic Mix and Match Election. To be effective, an Electronic Mix and Match Election must be submitted by such a holder of Scheme Shares and received by the KWE Registrar by no later than the Election Return Time.
- 6.11 If an Election Form in relation to the Mix and Match Election or an Electronic Mix and Match Election is received by the KWE Registrar after the Election Return Time, or if an Election Form in relation to the Mix and Match Election or an Electronic Mix and Match Election is received by the KWE Registrar before such time but is not, or is deemed not to be, valid and complete in all respects at such time, then such Mix and Match Election shall, for all purposes, be void unless and to the extent that KWE and KW, in their absolute discretion, elect to treat as valid in whole or in part any such Mix and Match Election.
- 6.12 Upon execution and delivery by a Scheme Shareholder of a valid Election Form in relation to the Mix and Match Election or the making of a valid Electronic Mix and Match Election, such Scheme Shareholder shall be bound by the terms and provisions contained in the Election Form or the Electronic Mix and Match Election (as the case may be) and by the terms and provisions contained in Part Five (Notes for Making Elections and Submitting IRS Forms W-8 or W-9) of the Scheme Document.
- 6.13 An Election Form in relation to the Mix and Match Election duly completed and delivered or an Electronic Mix and Match Election made in accordance with this Clause 6 may be withdrawn by notice to the KWE Registrar in writing to be received by the Election Return Time.
- 6.14 If a Scheme Shareholder delivers more than one Election Form in relation to the Mix and Match Election or Electronic Mix and Match Election in respect of any of its Scheme Shares, in the case of an inconsistency between such Election Forms or Electronic Mix and Match Elections, the last valid Election Form in relation to the Mix and Match Election or Electronic Mix and Match Election which is delivered by the Election Return Time shall prevail over any earlier Election Form in relation to the Mix and Match Election or Electronic Mix and Match Election. The delivery time for an Election Form in relation to the Mix and Match Election or Electronic Mix and Match Election shall be determined on the basis of which such Election Form or Electronic Mix and Match Election is last sent or, if the KWE Registrar is unable to determine which is last sent, is last received. Election Forms in relation to the Mix and Match Election which are sent in the same envelope shall be treated for these purposes as having been sent and received at the same time, and, in the case of an inconsistency between such Election Forms, none of them shall be treated as valid (unless KWE and KW otherwise determine in their absolute discretion).
- 6.15 Minor adjustments to the entitlements of Scheme Shareholders pursuant to Mix and Match Elections made under this Scheme may be made by the KWE Registrar with the prior consent of KWE and KW on a basis that KWE and KW consider to be fair and reasonable to the extent necessary to satisfy all entitlements pursuant to Mix and Match Elections under this Scheme as nearly as may be practicable. Such adjustments shall be final and binding on Scheme Shareholders.

- 6.16 No Mix and Match Election may be made by a KWE Shareholder who is resident, located or has a registered address in, or is a citizen or national of, or otherwise subject to the law of, a jurisdiction where the extension or availability of the Merger, or any Mix and Match Election or the granting of a right to make a Mix and Match Election pursuant to Clause 3 and Clauses 5 to 7, would breach any applicable law. Any purported Mix and Match Election by such a KWE Shareholder shall be void.
- 6.17 By submitting a Mix and Match Election, the KWE Shareholder making such Mix and Match Election undertakes not to sell or encumber any KWE Shares to which such Election relates unless and until that Election has been validly withdrawn.

7. Conflicting Elections and changes in shareholdings

- 7.1 Where a Scheme Shareholder has made (i) a valid Original Offer Election; or (ii) having made no valid Original Offer Election, has made a valid Mix and Match Election in respect of all Scheme Shares held by that Scheme Shareholder, the Scheme Shareholder must not transfer or encumber any Scheme Shares to which the Original Offer Election or Mix and Match Election relates other than Scheme Shares in respect of which it has first validly withdrawn the relevant Election.
- 7.2 The provisions of this Clause 7 are subject, and without prejudice, to any valid withdrawals of Original Offer Elections and/or Mix and Match Elections and to any replacement or supplemental Original Offer Elections or Mix and Match Elections validly made in respect of Scheme Shares held or acquired by a Scheme Shareholder prior to the Scheme Record Time.
- 7.3 Except in circumstances where Clause 7.4 and/or Clause 7.5 applies (when they will take precedence), where any Election or Elections is or are received from a Scheme Shareholder in respect of the Mix and Match Facility, it or they will be void to the extent (but not further) that it or they relate to a number of Scheme Shares which, in aggregate, exceed the number of Scheme Shares in respect of which that Scheme Shareholder is entitled to the New Offer Consideration and Distribution.
- 7.4 Where a Scheme Shareholder has made a valid Original Offer Election in respect of all that Scheme Shareholder's Scheme Shares and it has not been validly withdrawn, in whole or part, as at the Scheme Record Time:
 - (1) any increase, as at the Scheme Record Time, in the number of Scheme Shares held by that Scheme Shareholder will be deemed to be comprised of Scheme Shares in respect of which the Scheme Shareholder is entitled to receive the New Offer Consideration and Distribution with no adjustment under the Mix and Match Facility, except to the extent of any of the additional Scheme Shares in respect of which the Scheme Shareholder has by then made a valid supplemental Original Offer Election or Mix and Match Election; and
 - (2) any decrease, as at the Scheme Record Time, in the number of Scheme Shares held by that Scheme Shareholder, will not invalidate the Original Offer Election and the Original Offer Election will continue to apply to all Scheme Shares then held by that Scheme Shareholder.
- 7.5 Where a Scheme Shareholder makes a valid Original Offer Election in respect of a specified number of Scheme Shares which comprise less than all the Scheme Shares it holds or, due to a valid withdrawal of an Original Offer Election to which Clause 7.2 applies, its subsisting Original Offer Election relates to less than all the Scheme Shares it holds:
 - (1) if the number of Scheme Shares it holds at the Scheme Record Time exceeds the number of Scheme Shares to which its subsisting Original Offer Election relates, the excess will be deemed to be Scheme Shares in respect of which it is entitled to the New Offer Consideration and Distribution with no adjustment under the Mix and Match Facility, except to the extent of any of the additional Scheme Shares in respect of which the Scheme Shareholder has by then made a valid supplemental Mix and Match Election; and
 - (2) if the number of Scheme Shares it holds at the Scheme Record Time is less than the number of Scheme Shares to which its subsisting Original Offer Election relates, the shortfall will not invalidate the Original Offer Election and the Original Offer Election will apply to all Scheme Shares then held by that Scheme Shareholder.

- 7.6 Except in circumstances where Clause 7.7 and/or Clause 7.8 apply (when they will take precedence), if a Scheme Shareholder delivers an Additional Share Election and an Additional Cash Election in respect of numbers of Scheme Shares which, in aggregate, exceed the number of Scheme Shares in respect of which the Scheme Shareholder is entitled to the New Offer Consideration and Distribution, the Additional Share Election and Additional Cash Election will be scaled down *pro rata* to the number of Scheme Shares to which each of them applies until they relate, in aggregate, only to a number of Scheme Shares equal to the number of Scheme Shares in respect of which that Scheme Shareholder is entitled to the New Offer Consideration and Distribution.
- 7.7 Where a Scheme Shareholder has made one or more valid Mix and Match Elections in respect of all Scheme Shares it holds, and such Mix and Match Election(s) has/have not been validly withdrawn in whole or in part, as at the Scheme Record Time:
 - (1) any increase, as at the Scheme Record Time, in the number of Scheme Shares held by that Scheme Shareholder (other than any such Scheme Shares to which a valid and subsisting supplemental Original Offer Election applies), will be deemed to comprise Scheme Shares in respect of which the Scheme Shareholder is entitled to receive the New Offer Consideration and Distribution, with no adjustment under the Mix and Match Facility except to the extent of any of the additional Scheme Shares in respect of which the Scheme Shareholder has by then made a valid supplemental Mix and Match Election; and
 - (2) any decrease, as at the Scheme Record Time, in the number of Scheme Shares held by that Scheme Shareholder below the number of Scheme Shares in respect of which the Scheme Shareholder's subsisting valid Mix and Match Elections apply will not invalidate the Mix and Match Elections, and the Mix and Match Election(s) will continue to apply to all Scheme Shares then held by that Scheme Shareholder; and if the relevant Mix and Match Election(s) comprise one or more Additional Share Election(s) and one or more Additional Cash Election(s), the reduction shall apply *pro rata* to the aggregate numbers of Scheme Shares which are subject to, respectively, all subsisting valid Additional Share Election(s) and Additional Cash Election(s) of that Scheme Shareholder.
- 7.8 Where a Scheme Shareholder has made one or more valid Mix and Match Elections in respect of a specified number of Scheme Shares which comprise less than all the Scheme Shares it holds or, due to a valid withdrawal of a Mix and Match Election to which Clause 7.2 applies, its subsisting Mix and Match Election relates to less than all the Scheme Shares it holds:
 - (1) if the number of Scheme Shares it holds at the Scheme Record Time (other than any such Scheme Shares to which a valid and subsisting initial or supplemental Original Offer Election applies in accordance with Clause 7.4 or Clause 7.5) exceeds the aggregate number of Scheme Shares to which its subsisting Mix and Match Elections relate, the excess will be deemed to be Scheme Shares in respect of which it is entitled to the New Offer Consideration and Distribution, with no adjustment under the Mix and Match Facility except to the extent of any of the additional Scheme Shares in respect of which the Scheme Shareholder has by then made a valid supplemental Mix and Match Election; and
 - (2) if the number of Scheme Shares it holds at the Scheme Record Time (other than any such Scheme Shares to which a valid and subsisting initial or supplemental Original Offer Election applies in accordance with Clause 7.4 or Clause 7.5) is less than the number of Scheme Shares to which its subsisting Mix and Match Elections relate, the shortfall will not invalidate the Mix and Match Election(s) which will continue to apply to all Scheme Shares then held by that Scheme Shareholder; and if the relevant Mix and Match Election(s) comprise one or more Additional Share Elections and one or more Additional Cash Elections, the reduction shall apply *pro rata* to the aggregate numbers of Scheme Shares to which, respectively, all subsisting valid Additional Cash Elections and Additional Share Elections of that Scheme Shareholder apply.
- 7.9 KWE and KW reserve the right by mutual agreement to vary the operation of this Clause 7 and/or to give effect to initial Elections in priority to subsequent Elections and to treat them as unchanged by subsequent Elections or by acquisitions or disposals of Scheme Shares to which initial or subsequent Elections relate, in each case to the extent that, acting reasonably, they consider this necessary to permit the provisions of this Clause 7 to be applied equitably.

8. Settlement of consideration

- 8.1 As soon as practicable, and in any event not later than 14 days, after the Effective Date, KWE shall (subject to compliance with the provisions of the Jersey Companies Law) pay the KWE Special Distribution to each Scheme Shareholder pursuant to Clause 3.
- 8.2 Not later than 14 days after the Effective Date and subject to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme:
 - (1) KW shall issue the New KW Shares which it is required to issue to each Scheme Shareholder pursuant to Clause 3 and Clauses 5 to 7 and pay any cash sum due in respect of fractional entitlements pursuant to Clause 3.5; and
 - (2) KW shall pay the KW Cash Component which it is required to pay to each Scheme Shareholder pursuant to Clauses 3, 6 and 7.
- 8.3 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme, KWE shall pay, and KW shall procure the payment by KWE (subject in each case to compliance with the provisions of the Jersey Companies Law) of, the Closing Dividend to each KWE Shareholder on the First KW Dividend Payment Date pursuant to Clause 4.
- 8.4 KW shall not cause or permit any action to be taken by KWE or any of its subsidiaries which could impede or delay payment of all or part of the Closing Dividend.
- 8.5 Settlement of the KW Cash Component and any cash sum due in respect of fractional entitlements pursuant to Clause 3.5 to which a Scheme Shareholder is entitled shall be effected as follows:
 - (1) in the case of Scheme Shares which at the Scheme Record Time are in certificated form, by procuring the despatch to persons entitled thereto in accordance with the provisions of Clause 8.9 of cheques in pounds sterling for the sums payable to them in accordance with the terms of the Scheme, or if practicable in the opinion of KW and KWE, via BACS; and
 - (2) in the case of Scheme Shares which at the Scheme Record Time are in uncertificated form, by procuring that Euroclear is instructed to create an assured payment obligation in pounds sterling in favour of the payment bank of the persons entitled thereto in accordance with the CREST assured payment arrangements for the sums payable in accordance with the terms of the Scheme, *provided* that the payor reserves the right to make payment of such sums by cheque as set out in Clause 8.5(1) if, for reasons outside its reasonable control, it is not able to effect settlement in accordance with this Clause 8.5(2).
- 8.6 Settlement of the KWE Special Distribution to which a Scheme Shareholder is entitled, whether the Scheme Shares are in certificated or uncertificated form at the Closing Dividend Record Time, shall be effected by procuring the despatch to persons entitled thereto in accordance with the provisions of Clause 8.9 of cheques in pounds sterling for the sums payable to them in accordance with the terms of the Scheme, or if practicable in the opinion of KWE, via CREST by procuring that Euroclear is instructed to create an assured payment obligation in pounds sterling in favour of the payment bank of the persons entitled thereto in accordance with the CREST assured payment arrangements for the sums payable in accordance with the terms of the Scheme or BACS in accordance with any applicable mandate pursuant to Clause 11.
- 8.7 Settlement of the Closing Dividend to which a KWE Shareholder is entitled, whether the KWE Shares are in certificated or uncertificated form at the Closing Dividend Record Time, shall be effected by procuring the despatch to persons entitled thereto in accordance with the provisions of Clause 8.9 of cheques in pounds sterling for the sums payable to them in accordance with the terms of the Scheme, or if practicable in the opinion of KWE, via CREST by procuring that Euroclear is instructed to create an assured payment obligation in pounds sterling in favour of the payment bank of the persons entitled thereto in accordance with the CREST assured payment arrangements for the sums payable in accordance with the terms of the Scheme or BACS in accordance with any applicable mandate pursuant to Clause 11.

- 8.8 Settlement of the New KW Shares to which a Scheme Shareholder is entitled shall be effected as follows:
 - (1) in the case of Scheme Shares which at the Scheme Record Time are in uncertificated form, the New KW Shares to which the uncertificated Scheme Shareholder is entitled shall be issued to Cede & Co., which will be the registered holder of such New KW Shares. KW shall procure that Cede & Co. will hold the New KW Shares via an account with DTC. KW shall procure that the interest in such shares shall be credited by its transfer agent through DTC to the securities deposit account of CREST Depository Limited's nominee, CREST International Nominees Limited. Following the Effective Date, CREST Depository Limited shall issue KW CDIs, in CREST, to the CREST Receiving Agent and KW shall procure that the CREST Receiving Agent shall thereupon deliver, through CREST to the stock account in CREST in which each such uncertificated Scheme Shareholder held the relevant Scheme Shares, such uncertificated Scheme Shareholder's entitlement to KW CDIs; and
 - (2) in the case of Scheme Shares which at the Scheme Record Time are in certificated form, the KW Transfer Agent shall deliver to the relevant Scheme Shareholder (or as such Scheme Shareholder may direct) share certificates in respect of the New KW Shares to which such certificated Scheme Shareholder is entitled.

8.9 General

- (1) All deliveries of share certificates or cheques pursuant to this Scheme shall be effected by sending the same by first class post in prepaid envelopes addressed to the persons entitled thereto at their respective addresses as appearing in the Register at the Scheme Record Time or Closing Dividend Record Time (as applicable) or, in the case of joint holders, at the address of that one of the joint holders whose name stands first in the Register in respect of the relevant Scheme Shares at the Scheme Record Time or relevant KWE Shares at the Closing Dividend Record Time, and none of KWE, KW nor any person or nominee appointed by KWE or KW in accordance with this Scheme nor their respective agents shall be responsible for any loss or delay in the transmission or delivery of any share certificates or cheques sent in accordance with this Clause 8.9(1), which shall be sent at the risk of the persons entitled thereto.
- (2) All cheques shall be made payable to the persons respectively entitled to the moneys represented thereby (except that, in the case of joint holders, KW reserves the right to make such cheques payable to that one of the joint holders whose name stands first in the Register in respect of the relevant Scheme Shares at the Scheme Record Time or the relevant KWE Shares at the Closing Dividend Record Time).
- (3) The delivery of any cheque, the payment by BACS and/or the creation of an assured payment obligation in CREST each as is referred to in Clause 8.5, Clause 8.6 and Clause 8.7 shall be a complete discharge of the payment obligations of KW and KWE (as applicable) for the moneys represented thereby.
- (4) Notwithstanding the provisions of Clause 8.9(1) and Clause 8.9(2), any sums payable by KW in respect of fractional entitlements in accordance with Clause 3.5 (whether by cheque, BACS or CREST assured payment arrangements) shall be paid to the persons respectively entitled to the moneys represented thereby in accordance with any dividend mandates or other instructions to KWE in force at the Scheme Record Time relating to the Scheme Shares held by such persons.
- (5) The provisions of this Clause 8 shall be subject to any condition or prohibition imposed by law.
- 8.10 Subject to Clause 8.11, KW will instruct its paying agents to report all payments from KW required by this Scheme in respect of the KW Cash Component and any payment in respect of a fractional entitlement as paid as consideration for the Scheme Shares and KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, all payments required from KW by this Scheme in respect of the KW Cash Component and any payment in respect of a fractional entitlement will be made free and clear of any and all deductions and/or withholdings in respect of any US federal income tax, provided that US federal backup withholding may apply under the circumstances described in Clause 8.11.

- 8.11 Clause 8.10 will not prevent a deduction or withholding which is required by US federal income tax law to be made in respect of US federal backup withholding from a payment to a Scheme Shareholder in respect of the KW Cash Component or other payment from KW (including, without limitation, a payment in respect of a fractional entitlement) to the extent that Scheme Shareholder fails to either (i) provide a properly completed applicable IRS Form W-8 or, as applicable, IRS Form W-9 to the Receiving Agent at or before the Election Return Time showing that no US federal backup withholding is required, or (ii) otherwise establish that the payment can lawfully be made without such withholding, in each case (i) and (ii), in respect of itself (to the extent such Scheme Shareholder is a beneficial owner of Scheme Shares) or in respect of beneficial owners of Scheme Shares directly or indirectly holding through such Scheme Shareholder. This Clause 8.11 does not apply in relation to any deduction or withholding other than in respect of US federal backup withholding.
- 8.12 Subject to Clause 8.13, KWE will instruct the Receiving Agent to report all payments from KWE required by this Scheme in respect of the KWE Special Distribution as dividend paid by KWE and KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, all payments required from KWE by this Scheme in respect of the KWE Special Distribution will be made free and clear of any and all deductions and/or withholdings in respect of any US federal income tax, provided that US federal backup withholding may apply under the circumstances described in Clause 8.13.
- 8.13 Clause 8.12 will not prevent a deduction or withholding which is required by US federal income tax law to be made in respect of US federal backup withholding from a payment to a Scheme Shareholder in respect of the KWE Special Distribution or other payment from KWE to the extent that Scheme Shareholder fails to either (i) provide a properly completed applicable IRS Form W-8 or, as applicable, IRS Form W-9 to the Receiving Agent at or before the Election Return Time showing that no US federal backup withholding is required, or (ii) otherwise establish that the payment can lawfully be made without such withholding, in each case (i) and (ii), in respect of itself (to the extent such Scheme Shareholder is a beneficial owner of Scheme Shares) or in respect of beneficial owners of Scheme Shares directly or indirectly holding through such Scheme Shareholder. This Clause 8.13 does not apply in relation to any deduction or withholding other than in respect of US federal backup withholding.

9. Overseas Shareholders

- 9.1 The provisions of Clauses 2 to 8 (inclusive) shall be subject to any prohibition or condition imposed by law. Without prejudice to the generality of the foregoing, in respect of any Scheme Shareholder who is resident, located or has a registered address in a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, or who KW or KWE reasonably believes to be a citizen, resident or national of, or subject to the law of, a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, if the law of a jurisdiction other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands precludes (or precludes except after compliance by KW or KWE with any governmental or other consent or any registration, filing or other formality with which KW or KWE, as applicable, is unable to comply or compliance with which KW or KWE (in each case acting reasonably) regards as unduly onerous), or KW or KWE is advised that such law so precludes (i) the extension or availability of the Merger or the Election in, into or from such jurisdiction; (ii) any Election(s) or the granting of a right to make any Election(s) pursuant to Clause 3 and Clauses 5 to 7 (inclusive); or (iii) the issue of New KW Shares or KW CDIs pursuant to Clauses 2 and 3 and Clauses 5 to 8 (inclusive), then KW or KWE may, in its discretion, determine that:
 - (1) one or more such Election(s) shall not be valid or accepted in respect of any relevant Scheme Shareholder, in which case such Scheme Shareholder shall receive the New Offer Consideration and Distribution; or
 - (2) such New KW Shares (or KW CDIs) shall be sold, in which event the relevant New KW Shares (or KW CDIs) shall be issued to (or for the benefit of) such Scheme Shareholder and KW shall appoint a person who shall be authorised to act on behalf of such Scheme Shareholder to procure that any shares (or KW CDIs) in respect of which KW or KWE

has made such determination shall, as soon as practicable following the Effective Date, be sold and the net proceeds of such sale shall be remitted to such Scheme Shareholder in accordance with Clause 9.2; or

- (3) such New KW Shares (or KW CDIs) shall not be issued to such Scheme Shareholder but shall instead be issued to a nominee for such Scheme Shareholder appointed by KW on terms that the nominee shall, as soon as practicable following the Effective Date, sell the relevant New KW Shares so issued and the net proceeds of such sale shall be remitted to such Scheme Shareholder in accordance with Clause 9.2.
- 9.2 Any sale under Clause 9.1 shall be carried out at the best price which can reasonably be obtained at the time of sale and the net proceeds of such sale (after the deduction of all expenses and commissions incurred in connection with such sale) shall be paid to the relevant Scheme Shareholder, together with the KW Cash Component and the KWE Special Distribution by sending a cheque or creating an assured payment obligation in accordance with the provisions of Clause 8.
- 9.3 To give effect to any sale under Clause 9.1, the person appointed by KW in accordance with Clause 9.1(2) shall be authorised as agent on behalf of the holder concerned, and the nominee appointed by KW in accordance with Clause 9.1(3) shall be authorised, to execute and deliver as transferor a form of transfer or other instrument or instruction of transfer and to give such instructions and to do all other things which he or she may consider necessary or expedient in connection with such sale. In the absence of bad faith or wilful default, none of KWE, KW or the person or nominee so appointed shall have any liability for any loss or damage arising as a result of the timing or terms of such sale, or for any determination pursuant to Clause 9.1(1).
- 9.4 In the case of a Scheme Shareholder who is resident, located or has a registered address in, or is a citizen or national of, or otherwise subject to the law of, a jurisdiction where the extension or availability of the Merger, or any Election(s) or the granting of a right to make any Election(s) pursuant to Clause 3 and Clauses 5 to 8 (inclusive), would breach any applicable law, then no such Election(s) made by that Scheme Shareholder shall have effect and the omission to recognise any such Election(s) made by that Scheme Shareholder shall not constitute a breach by KW or KWE of any of their obligations under the Scheme.
- 9.5 Neither KW nor KWE will be liable to any Scheme Shareholder for making any determination or acting in accordance with this Clause 9.

10. Share certificates and transfer of entitlements

- 10.1 Subject to the Scheme becoming effective and to the payment of the KWE Special Distribution if, when and to the extent required by this Scheme:
 - (1) all certificates representing Scheme Shares shall cease to have effect and every Scheme Shareholder holding such certificated Scheme Shares shall be bound to destroy the share certificate(s) relating to the Scheme Shares it is required to transfer pursuant to the Scheme to KWE (or to any person appointed by KWE to receive the same);
 - (2) Euroclear shall be instructed by the individual authorised by KW pursuant to Clause 2.2 to transfer all entitlements to Scheme Shares held in uncertificated form to KW; and
 - (3) appropriate entries shall be made in the Register to reflect their transfer pursuant to Clause 2.

11. Mandates and other instructions

Save in respect of the payment of (i) the KWE Special Distribution; and (ii) the Closing Dividend (for which any relevant mandates and instructions shall, in each case, remain in place for such payments if, and to the extent that, KW and KWE are of the opinion that it is practicable for such payments to be made by CREST or BACS in accordance with Clause 8.5, Clause 8.6 and Clause 8.7 of this Scheme by means of such mandates and instructions), all mandates and other instructions relating to the payment of dividends or any other sums payable on any Scheme Shares and other instructions (including communication preferences) given to KWE by Scheme Shareholders in force at the Scheme Record Time relating to Scheme Shares shall, as from the Effective Date, cease to be valid.

12. The Effective Date

- 12.1 This Scheme shall become effective in accordance with its terms as soon as the Court Order shall have been delivered to the Registrar of Companies for registration, which must be a date no later than 14 days after the Scheme Court Sanction Hearing.
- 12.2 Unless this Scheme shall become effective on or before 30 November 2017 or such later date, if any, as KW and KWE may, with the consent of the Panel, agree and the Court may allow, this Scheme shall never become effective.
- 12.3 If this Scheme is not implemented in accordance with its terms, any Election made shall cease to be valid.

13. Modification

KW and KWE may jointly consent on behalf of all concerned to any modification of, or addition to, this Scheme or to any condition which the Court may approve or impose. Any such modification or addition may require the consent of the Panel.

14. Governing law

- 14.1 This Scheme is governed by Jersey law and is subject to the exclusive jurisdiction of the Jersey courts. The rules of the Code will apply to this Scheme on the basis provided in the Code.
- 14.2 The Election Form, Electronic Elections and all Elections thereunder, and all action taken or made, or deemed to be taken or made, pursuant to any of the terms of this Scheme, shall be governed by and interpreted in accordance with Jersey law and shall be subject to the jurisdiction of the Court.
- 14.3 Execution of an Election Form or the submission of an Electronic Election by or on behalf of a Scheme Shareholder will constitute such Scheme Shareholder's agreement that the courts of Jersey are (subject to Clause 14.4 below) to have exclusive jurisdiction to settle any dispute which may arise in connection with the creation, validity, effect, interpretation or performance of the Election Form or the submission of an Electronic Election, and for such purposes that he or she irrevocably submits to the jurisdiction of the Court.
- 14.4 Execution of an Election Form or the submission of an Electronic Election by or on behalf of a Scheme Shareholder will constitute such Scheme Shareholder's agreement that Clause 14.3 is included for the benefit of KWE, KW and their respective agents and, accordingly, notwithstanding Clause 14.3, each of KWE, KW and their respective agents shall retain the right to, and may in its absolute discretion, bring proceedings in the courts of any other country which may have jurisdiction, and that the electing Scheme Shareholder irrevocably submits to the jurisdiction of the courts of any such country.

Dated: 13 September 2017

PART FIVE

NOTES FOR MAKING ELECTIONS AND SUBMITTING IRS FORMS W-8 OR W-9

Whether or not you attend or vote in favour of the Scheme at the Scheme Court Meeting or in favour of the Special Resolution at the KWE General Meeting, you may submit the elections set out below. The attention of Overseas Shareholders is drawn to paragraph 3 of this Part Five below.

If you wish to receive the New Offer Consideration and Distribution, consisting of (i) the New KW Share Component of 0.3854 New KW Shares; (ii) the KW Cash Component of 300 pence in cash payable by KW; and (iii) the KWE Special Distribution of 250 pence in cash payable by KWE in the form of a special distribution in respect of each Scheme Share held, without variation pursuant to the Mix and Match Facility, DO NOT return the Election Form or send an Electronic Election via CREST.

If you wish to change the composition of the consideration available under the New Offer pursuant to the Mix and Match Facility or to elect for the Original Offer Consideration, please follow the instructions set out below in this Part Five.

Please note that if you hold Scheme Shares in certificated form and you need further copies of an Election Form, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

1. If you hold Scheme Shares in certificated form (i.e. not in CREST) and you wish to make one or more Elections:

- You (and all co-holders, if Scheme Shares are held jointly) must complete and sign, in the presence of an independent witness who must also sign, an Election Form in accordance with the instructions printed thereon and return it (as well as your original share certificate(s) and/or other document(s) of title) by post to: Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to: Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom. A pre-paid envelope is provided for use only in the UK.
- If your share certificate(s) and/or other document(s) of title are with your bank, stockbroker or other agent, you should complete and sign the Election Form and arrange for it to be lodged by such agent with the relevant document(s). If your certificate(s) and/or other document(s) of title are not readily available, please send the duly completed Election Form to the KWE Registrar at the address set out above with a note saying "Share certificate(s) to follow" and arrange for the share certificate(s) and/or other document(s) of title to be sent to the KWE Registrar as soon as practicable and in any event, prior to the Election Return Time.
- The Election Return Time (being the last time for lodging your Election Form) is 1.00 p.m. on 19 October 2017, as set out in the expected timetable of principal events on pages 10 to 11 of this document (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service).
- Any changes to the Election Return Time will be announced by KWE via a Regulatory Information Service in due course, with such announcement being made available on KWE's and KW's websites at www.kennedywilson.eu and ir.kennedywilson.com, respectively.
- If you wish to sell part of your holding of KWE Shares held in certificated form and also wish to make an Election but are unable to obtain a certificate reflecting your remaining holding by the Election Return Time, you should ensure that the stockbroker, bank or other agent through whom you make the sale obtains the appropriate endorsement or indication signed on behalf of the KWE Registrar in respect of the balance of your holding of Scheme Shares held in certificated form.

2. If you hold Scheme Shares in uncertificated form (i.e. in CREST) and you wish to make one or more Elections:

- You must submit your election electronically by taking (or procuring to be taken) the actions set out below to transfer the Scheme Shares in respect of which you wish to make an Election to an escrow balance, using an Electronic Election specifying the CREST Receiving Agent (in its capacity as a CREST participant under the ID 8RA25) as the escrow agent.
- The Election Return Time (being the last time for lodging an Election Form or making your Electronic Election) is 1.00 p.m. on 19 October 2017, as set out in the expected timetable of principal events on pages 10 to 11 of this document (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service).
- Any changes to the Election Return Time will be announced by KWE via a Regulatory Information Service in due course, with such announcement being made available on KWE's and KW's websites at www.kennedywilson.eu and ir.kennedywilson.com, respectively.
- If you wish to make an Election by completing an Election Form rather than making an Electronic Election, you must first materialise your Scheme Shares (i.e. withdraw them from CREST) by completing a CREST stock withdrawal form, and you may request an Election Form by contacting the Shareholder Helpline on the telephone number set out on page 22 of this document. Completing and returning an Election Form in relation to Scheme Shares held in uncertificated form will not constitute a valid Election and will be disregarded.
- If you hold Scheme Shares in both certificated and uncertificated form and/or if you hold Scheme Shares in two or more designated accounts and you wish to make an Election in respect of all such holdings, you must make a separate Election in respect of each holding.
- If you are a CREST personal member, you should refer to your CREST sponsor before taking any action. Your CREST sponsor will be able to confirm details of your participant ID and the member account ID under which your Scheme Shares are held. In addition, only your CREST sponsor will be able to send the Electronic Election to Euroclear in relation to your Scheme Shares.
- You should send (or, if you are a CREST personal member, procure that your CREST sponsor sends) an Electronic Election to Euroclear which must be properly authenticated in accordance with Euroclear's specifications and which must contain, in addition to the other information that is required for an Electronic Election to settle in CREST, the following details:
 - (a) the number of Scheme Shares in respect of which you are making an Election (such Scheme Shares to be transferred to an escrow balance);
 - (b) your member account ID;
 - (c) your participant ID;
 - (d) the participant ID of the escrow agent, Computershare Investor Services PLC, in its capacity as a CREST Receiving Agent; this is 8RA25;
 - (e) the relevant member account ID(s) of the escrow agent, Computershare Investor Services PLC, in its capacity as a CREST Receiving Agent:
 - (i) to make an Additional Share Election, this is "KENSHR02";
 - (ii) to make an Additional Cash Election, this is "KENCSH03"; and
 - (iii) to make an Original Offer Election, this is "KENORG01";
 - (f) the ISIN of the relevant Scheme Shares; this is JE00BJT32513;
 - (g) the intended settlement date (this should be as soon as possible and in any event by the Election Return Time);
 - (h) the corporate action number for the transaction: this is allocated by Euroclear and can be found by viewing the relevant corporate action details onscreen in CREST;
 - (i) CREST standard delivery instructions priority of 80; and

- (j) a contact name and telephone number (inserted in the shared note field of the Electronic Election).
- After making the Electronic Election, you will not be able to access the Scheme Shares concerned in CREST for any transaction or for charging purposes. You are recommended to refer to the CREST Manual published by Euroclear for further information on the CREST procedure outlined above. An Electronic Election is revocable. Please refer to the CREST Manual for information about how to withdraw an Electronic Election.

3. The availability of the Original Offer Election and the Mix and Match Facility to certain Overseas Shareholders may be restricted or prohibited by law:

- Overseas Shareholders who are resident, located or have a registered address in, or who are a citizen or national of, or otherwise subject to the law of, a jurisdiction where the extension or availability of the Merger, or an Election or the granting of a right to make an Election, would breach any applicable laws may not be entitled to make an Election.
- Overseas Shareholders should inform themselves of and observe any applicable legal or regulatory requirements in the jurisdiction in which they or the KWE Shareholder(s) on whose behalf they hold KWE Shares are located. If you are in any doubt about your position, you should consult an independent professional adviser in the relevant jurisdiction. The Mix and Match Facility and the ability to make an Original Offer Election may not be available to certain Overseas Shareholders.
- By signing and returning the Election Form or submitting your Electronic Original Offer Election or your Electronic Mix and Match Election, you are deemed to represent that you are not an Overseas Shareholder who is resident, located or has a registered address in, or is a citizen or national of, or otherwise subject to the law of, a jurisdiction where the extension or availability of the Merger, or an Election or the granting of a right to make an Election, would breach any applicable law.

4. Withdrawals

If you have returned an Election Form and subsequently wish to withdraw or amend that Election, please contact the KWE Registrar in writing by the Election Return Time, which is currently expected to be 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service). Please specify clearly whether you would like to withdraw or amend the Election that you have made, and ensure that your request contains an original signature. Any written requests of this nature should be sent by post to: Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or delivered by hand (during normal business hours) to: Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom. It is at the KWE Registrar's absolute discretion to require the submission of a new Election Form if an amendment is requested.

If your Election was made through an Electronic Election, you may withdraw your Election through CREST by sending (or, if you are a CREST sponsored member, procuring that your CREST sponsor sends) an ESA instruction to settle in CREST by no later than Election Return Time, which is currently expected to be 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service), in relation to each Election to be withdrawn. Each ESA instruction must, in order for it to be valid and to settle, include the following details:

- (a) the number of Scheme Shares to be withdrawn;
- (b) the ISIN number of the Scheme Shares; this is JE00BJT32513;
- (c) your member account ID;
- (d) your participant ID;
- (e) the participant ID of the escrow agent, Computershare Investor Services PLC, in its capacity as a CREST Receiving Agent; this is 8RA25;
- (f) the relevant member account ID(s) of the escrow agent, Computershare Investor Services PLC, in its capacity as a CREST Receiving Agent included in the relevant Mix and Match Election this is:

- (i) "KENSHR02" if an Additional Share Election was made,;
- (ii) "KENCSH03" if an Additional Cash Election was made; or
- (iii) "KENORG01" if an Original Offer Election was made);
- (g) the CREST transaction ID of the Election to be withdrawn;
- (h) the intended settlement date for the withdrawal;
- (i) the corporate action number for the transaction: this is allocated by Euroclear and can be found by viewing the relevant corporate action details on screen in CREST; and
- (j) CREST standard delivery instructions priority of 80.

Any such withdrawal will be conditional upon the CREST Receiving Agent verifying that the withdrawal request is validly made. Accordingly, the CREST Receiving Agent will, on behalf of KWE and KW, reject or accept the withdrawal or amendment by transmitting in CREST a receiving agent reject ("AEAD") or receiving agent accept ("AEAN") message.

5. Late, incomplete or inconsistent Elections

If any Election Form or Electronic Election is either received after the Election Return Time, which is currently expected to be 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service), or is received before such time and date but is not valid or complete in all respects at such time and date, such Election Form or Electronic Election shall, for all purposes, be void (unless and to the extent that KWE and KW, in their absolute discretion, elect to treat as valid, in whole or in part, any such Election Form or Electronic Election).

If a Scheme Shareholder delivers more than one Election Form or Electronic Election in respect of its Scheme Shares, in the case of an inconsistency between such Election Forms or Electronic Elections, the last Election Form or Electronic Election which is delivered by the Election Return Time shall prevail over any earlier Election Form or Electronic Election. The delivery time for an Election Form or Electronic Election shall be determined on the basis of which Election Form or Electronic Election is last sent or, if the KWE Registrar is unable to determine which is last sent, is last received. Election Forms which are sent in the same envelope shall be treated for these purposes as having been sent and received at the same time, and, in the case of an inconsistency between such Election Forms, none of them shall be treated as valid (unless KWE and KW otherwise determine in their absolute discretion).

6. Effect of Elections

By submitting an Election, the person doing so undertakes not to sell or encumber any KWE Shares to which such Election relates unless and until that Election has been validly withdrawn.

7. General

Without prejudice to any other provisions of this Part Five, KWE and KW reserve the right (subject to the terms of the Merger and the provisions of the Code) to treat as valid in whole or in part any Election which is not entirely in order.

No acknowledgements of receipt of any Election Form, Electronic Election or other documents will be given. All communications, notices, other documents and remittances to be delivered by, or sent to or from, holders of KWE Shares (or their designated agent(s)) or as otherwise directed will be delivered by or to, or sent to or from, such holders of KWE Shares (or their designated agent(s)) at their own risk.

KWE, KW and their respective agents reserve the right to notify any matter to all or any KWE Shareholders with registered addresses outside Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, or to the nominees, trustees or custodians for such KWE Shareholders, by announcement in Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, or paid advertisement in any daily newspaper published and circulated in Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands, or any part thereof, in which case such notice shall be deemed to have been sufficiently given, notwithstanding any failure by any such KWE Shareholders to receive or see such notice. All references in this document to notice in writing, or the provision of information in writing, by or on behalf of KWE, KW and their respective agents shall be construed accordingly. No such document shall be sent to an address outside Jersey, the United Kingdom,

the United States, the Netherlands or the Cayman Islands where it would or might infringe the laws of that jurisdiction or would or might require KWE or KW to comply with any governmental or other consent or any registration, filing or other formality with which KW or KWE, as applicable, is unable to comply or compliance with which KW or KWE (in each case acting reasonably) regards as unduly onerous.

The Election Form, Electronic Elections and all Elections thereunder, and all action taken or made, or deemed to be taken or made, pursuant to any of these terms, shall be governed by and interpreted in accordance with Jersey law and shall be subject to the jurisdiction of the Court.

Execution of an Election Form or the submission of an Electronic Election by or on behalf of a Scheme Shareholder will constitute his or her agreement that the courts of Jersey are (subject to the paragraph below) to have exclusive jurisdiction to settle any dispute which may arise in connection with the creation, validity, effect, interpretation or performance of the Election Form or the submission of an Electronic Election, and for such purposes that he or she irrevocably submits to the jurisdiction of the Jersey courts.

Execution of an Election Form or the submission of an Electronic Election by or on behalf of a Scheme Shareholder will constitute his or her agreement that the agreement in the paragraph above is included for the benefit of KWE, KW and their respective agents and, accordingly, notwithstanding the agreement in the paragraph above, each of KWE, KW and their respective agents shall retain the right to, and may in its absolute discretion, bring proceedings in the courts of any other country which may have jurisdiction, and that the electing Scheme Shareholder irrevocably submits to the jurisdiction of the courts of any such country.

If the Scheme is not implemented in accordance with its terms, any Election made shall cease to be valid.

None of KWE, KW, the KWE Registrar, the CREST Receiving Agent nor any of their respective advisers or any person acting on behalf of any one of them shall have any liability to any person for any loss or alleged loss arising from any decision as to the treatment of Elections on any of the bases set out in this Part Five or otherwise in connection therewith.

8. Unsettled trades

As at the close of trading on the last day of dealings in KWE Shares prior to the Effective Date, there may be unsettled, open trades for the sale and purchase of KWE Shares within CREST. The KWE Shares that are the subject of such unsettled trades will be treated under the Scheme in the same way as any other KWE Shares registered in the name of the relevant seller under that trade. Consequently, the KWE Shares will be transferred to KW under the Scheme and the seller will receive the appropriate consideration and other payments in accordance with the terms of the Scheme. The seller should consult the broker or other person through whom the trade was effected to inquire as to its obligations to account to the purchaser for the consideration and other payments it receives in connection with the Scheme. It is expected that CREST will automatically require the seller to settle the unsettled trade in KW CDIs and cash payments received in accordance with the terms of the Scheme. Consequently, the seller will need to ensure that it holds the appropriate number of KW CDIs and funds necessary to satisfy that trade at the relevant time. This position will be confirmed in due course by CREST to CREST participants.

9. US Tax Deductions

The payment of the KW Cash Component, the KWE Special Distribution and/or any other amount (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer to Scheme Shareholders entitled to the New Offer may be subject to US federal backup withholding (currently, at a rate of 28%). An explanation of this backup withholding is given in paragraphs 2.1(6) and 2.2(5) of Part Seven (*Taxation*) of this document.

In accordance with Clause 8.10 of the Scheme, KM will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the

Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017 the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 or applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct IRS Form W-9 or applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-8 or IRS Forms W-9 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-9 or, as applicable, IRS Forms W-8 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed IRS Forms W-9 or applicable IRS Forms W-8, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of IRS Form W-9 and applicable IRS Form W-8 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A SEPARATE APPROPRIATELY COMPLETED IRS FORM W-8 OR IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 OR TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 OR IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 OR IRS FORM W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 OR IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

<u>US federal backup withholding is not an additional tax.</u> Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability) or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner.

FAILURE TO COMPLETE AND RETURN IRS FORMS W-9 OR, IF APPLICABLE IRS FORMS W-8 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

The various versions of IRS Forms W-8 and W-9 for use by different categories of taxpayer are generally as follows:

- W-9 for use by beneficial owners of Scheme Shares which are "United States persons";
- W-8BEN for use by individual beneficial owners of Scheme Shares that are not "United States persons";
- W-8BEN-E for use by corporate and certain other non-individual beneficial owners of Scheme Shares that are not "United States persons"; and
- W-8IMY for use by non-US intermediaries and non-US partnerships (or entities treated as partnerships for US federal income tax purposes).

Beneficial owners of Scheme Shares and Scheme Shareholders should also review the discussion of US federal backup withholding in Part Seven (*Taxation*) of this document and consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e., Form W-9 or applicable Form W-8) and any supporting documentation to be provided to avoid potential US federal backup withholding.

In addition to the backup withholding regime, there are special rules regarding US taxation in relation to the KW Cash Component of the New Offer. For a more complete explanation of these, beneficial owners of KWE Shares and KWE Shareholders should carefully review Part Seven (*Taxation*) of this document and seek advice from their tax advisers.

In overview, these rules could, in certain circumstances, cause non-US Holders of Scheme Shares to be subject to US federal income tax in respect of the KW Cash Component despite the fact that KW pays the KW Cash Component without deduction or withholding of US federal income tax. In general, these rules could only apply if beneficial owners of Scheme Shares that own at least 50% of KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme come to own at least 50% of KW immediately after such time, in each case taking into account constructive ownership rules.

In that case, all or a portion of the cash received by a non-US Holder in respect of the KW Cash Component could be subject to US tax on deemed dividends by KW or treated as a gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme.

- Any such deemed dividend would be treated as paid by KW to the extent of its 2017 "earnings and profits" for US federal income tax purposes. This tax would not exceed 30% of the lesser of (a) the KW Cash Component received by the relevant non-US Holder and (b) the non-US Holder's attributable proportion of such earnings and profits (and may, in appropriate cases, be eligible for a reduced rate of US income taxation under an applicable tax treaty). Any earnings and profits of KW would be allocated ratably among actual and deemed dividends paid by KW in 2017. KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profits (for US federal income tax purposes) for the financial year ending 31 December 2017.
- In addition, even if, contrary to KW's stated expectation, KW were to have a material amount of earnings and profits for US federal income tax purposes for the financial year ending 31 December 2017, any non-US Holder whose percentage interest (taking into account the constructive ownership rules described below) in all of the outstanding KW Shares immediately after the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme (including, but not limited to, New KW Shares acquired pursuant to the Scheme, as well as any KW Shares otherwise acquired in connection therewith) is less than 80% of such non-US Holder's relative interest in KWE Shares immediately prior to such time (the "Less Than 80% Test") will not be treated as receiving a deemed dividend (and will thus not be subject to the 30% deemed dividend tax referred to above on any portion of the KW Cash Component received pursuant to the Scheme).

For example, a non-US Holder that held a 5% interest in KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme could not be subject to the 30% deemed dividend tax unless such non-US

Holder holds a 4% or greater interest in KW Shares immediately after such time. Accordingly, the 30% deemed dividend tax would not apply to any non-US Holder who does not actually or constructively hold any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme, but who only acquires KW Shares pursuant to the Scheme. Under constructive ownership rules, a non-US Holder generally will be treated as owning stock owned by certain related persons (such as an individual's spouse, children, grandchildren and parents), as well as, in certain circumstances, such holder's allocable share of stock held by any corporation (such as KWE shares owned by KW immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme) or any partnership, estate or trust, in each case, in which the non-US Holder or its relatives hold an interest. In addition, a non-US Holder which actually or constructively holds an option to purchase stock, is treated as owning the underlying stock.

A non-US Holder will not be subject to this form of 30% deemed dividend tax as a result of choosing to receive the consideration available under the New Offer or as a result of making any Additional Share Elections under the Mix and Match Facility if that non-US Holder does not own, actually or constructively, any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme. Accordingly, non-US Holders of Scheme Shares may wish to consider whether to avoid (or limit) any further acquisitions of KW Shares prior to such time. Those non-US Holders who already own KW Shares may also wish to avoid (or limit) any Additional Share Elections under the Mix and Match Facility, which could jeopardise their ability to satisfy the Less Than 80% Test, or to consider disposing of some or all of their KW Shares, or to reduce the KW Shares they will receive under the Scheme by reducing their holdings of KWE Shares in time, in either case insofar as that would remove or mitigate their potential liability to this form of deemed dividend taxation.

Assuming that a substantial number of beneficial owners of Scheme Shares satisfy the Less Than 80% Test and thus are not treated as if they had received deemed dividends, the remaining recipients of the KW Cash Component payable to non-US Holders (i.e. those which do not satisfy the Less Than 80% Test) would be allocated a significantly larger portion of KW's earnings and profits. However, a non-US holder will never be allocated earnings and profits in excess of the KW Cash Component it receives and would thus never be subject to the 30% deemed dividend tax in respect of an amount greater than the KW Cash Component it receives.

To the extent that the KW Cash Component is not treated as a deemed dividend on the basis described above, it may nonetheless be treated as gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme and subjected to US federal income tax if the non-US Holder owned (or was treated as owning) more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the deemed redemption and (ii) the non-US Holder's holding period in such shares (a "5% Holder"). This tax would be payable on the graduating scale of US federal income tax rates (including, if applicable, reduced tax rates on long-term capital gains derived by certain non-corporate holders, including individuals) on the amounts deemed to be received after allowing for certain reductions in respect of tax basis. The Less than 80% Test will not avoid or mitigate this tax.

Any beneficial owners of Scheme Shares who are also holders of KW Shares and who wish to avoid the potential application of these rules should consult their tax advisers, including as to the desirability and tax consequences of disposing of some or all their Scheme Shares or KW Shares in the market prior to the Merger.

Any person (whether or not a holder of shares in KW or KWE) who is considering acquiring shares in KW or KWE prior to the Effective Date should consider whether to refrain from doing so if, as a result, it may become liable to additional US tax as a result of the requirements described above.

This summary is, of necessity, a simplification of numerous complex tax considerations. Beneficial owners of Scheme Shares and Scheme Shareholders should not rely on it as a comprehensive statement of the potential tax implications of their receipt of the KW Cash

Component under the New Offer or the First KW Dividend and should in addition read and consider carefully the explanations in Part Seven (*Taxation*) of this document in full and seek independent tax advice on them.

10. Shareholder Helpline

If you require assistance, including in connection with IRS Forms W-8 or W-9, please telephone the Shareholder Helpline on 0370 707 4040 (from within the UK) or +44 370 707 4040 (from outside the UK) between 8.30 a.m. and 5.30 p.m. from Monday to Friday (except UK public holidays).

All calls may be recorded and monitored for security and training purposes. Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice.

PART SIX

FINANCIAL AND OTHER INFORMATION ON KWE AND KW

1. KWE

The following information is incorporated by reference into this document pursuant to Rule 24.15 of the Code and is available free of charge on KWE's website at www.kennedywilson.eu:

Information incorporated by reference	Document reference	Page number(s) in document
Unaudited condensed consolidated interim financial statements of KWE for the six months ended 30 June 2017	Independent Auditor's Report on Review of Condensed Consolidated Interim Financial Statements	18
	Condensed Consolidated Interim Income Statement	19
	Condensed Consolidated Interim Statement of Comprehensive Income	20
	Condensed Consolidated Interim Balance Sheet	21
	Condensed Consolidated Interim Statement of Changes in Equity	22-24
	Condensed Consolidated Interim Cash Flow Statement	25-26
	Notes to the Condensed Consolidated Interim Financial Statements	27-71
Audited consolidated financial statements of KWE for the year ended 31 December 2016	Independent Audit Report	82-84
	Consolidated Income Statement	85
	Consolidated Statement of Comprehensive Income	86
	Consolidated Balance Sheet	87
	Consolidated Statement of Changes in Equity	88-89
	Consolidated Cash Flow Statement	90-91
	Notes to the Consolidated Financial Statements	92-143
Audited consolidated financial	Independent Audit Report	82-84
statements of KWE for the year ended 31 December 2015	Consolidated Income Statement	85
	Consolidated Statement of Comprehensive Income	86
	Consolidated Balance Sheet	87
	Consolidated Statement of Changes in Equity	88-89
	Consolidated Cash Flow Statement	90-91
	Notes to the Consolidated Financial Statements	92-144

A KWE Shareholder may request a copy of such information in hard copy form (hard copies will not be provided unless requested). Hard copies may be requested by writing to the KWE company secretary at its registered office address: Corporate Services Team, Crestbridge, 47 Esplanade, St Helier, Jersey JE1 0BD or by contacting the Corporate Services Team during normal business hours on +44 (0) 1534 835600.

KWThe following information is incorporated by reference into this document pursuant to Rule 24.15 of the Code and is available free of charge on KW's website at <u>ir.kennedywilson.com</u>.

2.

Information incorporated by reference	Document reference	Page number(s) in document
Unaudited condensed consolidated financial statements of KW for the six months ended 30 June 2017	Consolidated Balance Sheets	1
	Consolidated Statements of Operations	2
	Consolidated Statements of Comprehensive Income	3
	Consolidated Statement of Equity	4
	Consolidated Statements of Cash Flows	5-6
	Notes to Consolidated Financial Statements	7-40
Audited consolidated financial statements of KW for the year ended 31 December 2016	Report of Independent Registered Public Accounting Firm	69-70
	Consolidated Balance Sheets	71
	Consolidated Statements of Operations	72
	Consolidated Statements of Comprehensive Income	73
	Consolidated Statement of Equity	74-75
	Consolidated Statements of Cash Flows	76-77
	Notes to Consolidated Financial Statements	78-119
Audited consolidated financial statements of KW for the year ended 31 December 2015	Report of Independent Registered Public Accounting Firm	71-72
	Consolidated Balance Sheets	73
	Consolidated Statements of Operations	74
	Consolidated Statements of Comprehensive Income	75
	Consolidated Statement of Equity	76-77
	Consolidated Statements of Cash Flows	78
	Notes to Consolidated Financial Statements	80-124

A KWE Shareholder may request a copy of such information in hard copy form (hard copies will not be provided unless requested). Hard copies may be requested by writing to Chris Emmerson, Goldman Sachs International, Peterborough Court, 133 Fleet Street, London EC4A 2BB, United Kingdom or by contacting +44 (0)20 7774 1000 during normal business hours.

PART SEVEN

TAXATION

THE CONTENTS OF THIS PART SEVEN ARE NOT TO BE CONSTRUED AS TAX ADVICE IN RESPECT OF THE IMPACT OF THE MERGER, THE OWNERSHIP AND DISPOSITION OF NEW KW SHARES OR OTHERWISE. SCHEME SHAREHOLDERS SHOULD CONSULT THEIR OWN TAX ADVISERS REGARDING THE TAX CONSEQUENCES OF THE MERGER AND OWNERSHIP AND DISPOSITION OF NEW KW SHARES IN LIGHT OF THEIR PARTICULAR CIRCUMSTANCES.

Under certain circumstances, non-US Holders that, actually or constructively, hold or during specified periods have held more than 5% of the outstanding KW Shares could be subject to tax with respect to the KW Cash Component, and are strongly urged to read paragraphs 2.1(1) and 2.2(1) of this Part Seven.

1. United Kingdom Taxation

The following statements are only a general guide to the principal UK tax consequences of the Merger, and the holding and disposing of New KW Shares, and do not constitute legal or tax advice.

They are based on UK tax legislation as applied in England and Wales and published practice of Her Majesty's Revenue & Customs ("HMRC") in force and effect at the date of this document, both of which are subject to change, possibly with retrospective effect, unless expressly provided otherwise.

These statements relate solely to Scheme Shareholders who:

- (a) are resident for tax purposes in (and only in) the UK at all relevant times, and, in the case of individuals, to whom "split year" treatment does not apply and who are domiciled for tax purposes only in the United Kingdom (except insofar as express reference is made to the treatment of non-United Kingdom residents);
- (b) hold their shares in KWE (and subsequently any shares in KW) as an investment (other than in an individual savings account or self investment pension plan) and not as trading stock;
- (c) who have not (and are not deemed to have) acquired their Scheme Shares (or New KW Shares) by reason of an office or employment; and
- (d) are the absolute beneficial owners thereof, who are beneficially entitled to the dividends thereon in circumstances where the dividends paid are regarded for UK tax purposes as that person's own income (and not the income of some other person) ("UK Holders").

The statements below may not apply to certain classes of Scheme Shareholders such as (but not limited to) dealers in securities, broker dealers, insurance companies, collective investment schemes, tax exempt organisations, financial institutions, persons who hold Scheme Shares or New KW Shares as part of hedging or conversion transactions, persons subject to the remittance basis, persons connected with KWE or KW, persons who hold investments in any HMRC-approved arrangements or schemes, or persons who control or hold (either alone or together with one or more associated or connected persons) directly or indirectly, 10% or more of the shares and/or voting power of KWE or KW.

The statements below in respect of UK stamp duty reserve tax ("SDRT") also assume that the SDRT requirements are met. The SDRT requirements are that KW is a body corporate not incorporated in the UK, that KW is not and will not be centrally managed and controlled in the UK, that there is no register, and that there will continue to be no register, in the UK in respect of the New KW Shares, that the New KW Shares are not and will not be paired with shares issued by a body corporate incorporated in the UK and that the New KW Shares are and will continue to be listed on a recognised stock exchange (within the meaning given by Section 1137(1) of the U.K. Corporation Tax Act 2010). KW expects that the SDRT requirements are and will continue to be met, though no assurances can be made.

IF SCHEME SHAREHOLDERS ARE IN ANY DOUBT AS TO THEIR TAX POSITION OR IF SCHEME SHAREHOLDERS ARE OR MAY BE SUBJECT TO TAX IN A JURISDICTION OTHER THAN THE UK, SUCH SCHEME SHAREHOLDERS SHOULD CONSULT THEIR OWN PROFESSIONAL ADVISERS.

1.1 Acquisition by KW of Scheme Shares under the Merger

(1) Taxation of Chargeable Gains

A UK Holder's liability to UK tax on capital gains will depend on the individual circumstances of that UK Holder and on the form of consideration received.

(a) UK Holders receiving the KW Cash Component under the Merger

To the extent a UK Holder receives the KW Cash Component from KW in respect of his or her (or its) Scheme Shares, that UK Holder will be treated as making a disposal (or where the UK Holder receives both the KW Cash Component and New KW Shares under the Merger, a part disposal) of Scheme Shares which may, depending on the UK Holder's individual circumstances (including the UK Holder's base cost in his or her (or its) holding of Scheme Shares, and the availability of exemptions, reliefs or allowable losses), give rise to a liability to UK capital gains tax (in the case of UK Holders who are individuals) or UK corporation tax on chargeable gains (in the case of UK Holders within the charge to UK corporation tax).

A UK Holder who is an individual and who disposes of Scheme Shares at a gain will be subject to United Kingdom capital gains tax to the extent that the gain (when aggregated with any other gains made by the individual in the current tax year) exceeds the annual exemption (£11,300, for the 2017-18 tax year) after taking account of any capital losses available to the individual and subject to any available reliefs.

In these circumstances, capital gains tax is charged at 10% to the extent that gains accruing to an individual in any tax year (calculated after deducting the annual exemption and capital losses) when aggregated with the individual's taxable income for that year (calculated after deduction of the personal allowance and other reliefs) are less than the upper limit of the income tax basic rate band (£33,500 for the 2017-18 tax year).

Capital gains tax is charged at 20% to the extent that gains accruing to an individual in any tax year (calculated after deducting the annual exemption and capital losses) when aggregated with the individual's taxable income for that year (calculated after deduction of the personal allowance and other reliefs) exceed the upper limit of the income tax basic rate band.

A UK Holder within the charge to UK corporation tax which disposes of Scheme Shares at a gain will be subject to UK corporation tax on chargeable gains after taking account of any losses available and subject to any available exemptions or reliefs. For such UK Holders, indexation allowance may be available to reduce any chargeable gain arising (but not create or increase any allowable loss) on the disposal of their Scheme Shares.

(b) UK Holders receiving New KW Shares under the Merger

To the extent that a UK Holder receives New KW Shares in exchange for his or her (or its) Scheme Shares, and provided that that UK Holder does not hold (either alone or together with persons connected with him or her (or it)) more than 5% of, or of any class of, shares in or debentures of KWE, that UK Holder will not be treated as having made a disposal of Scheme Shares. Instead, the New KW Shares should be treated as the same asset as those Scheme Shares, and as acquired at the same time and for the same consideration as those shares (except to the extent of any apportionment of base cost to the KW Cash Component as described in (c) below).

UK Holders who, alone or together with connected persons, hold more than 5% of, or of any class of shares in or debentures of KWE may be eligible for the treatment described in the preceding paragraph only if the transaction is effected for *bona fide* commercial reasons and does not form part of a scheme or arrangements of which the main purpose, or one of the main purposes, is the avoidance of liability to capital

gains tax or corporation tax, by virtue of section 137 of the Taxation of Chargeable Gains Act 1992. Such UK Holders are advised that no clearance has been sought from HMRC under section 138 of that Act that section 137 will not apply to prevent the treatment described in the preceding paragraph.

(c) Apportionment of base cost

A UK Holder who receives the KW Cash Component as well as New KW Shares will be treated as making a part disposal of Scheme Shares to the extent of the KW Cash Component received under the Merger. Any chargeable gain on the part disposal of a holding of Scheme Shares should be computed on the basis of an apportionment of the allowable cost of the total holding of Scheme Shares between the KW Cash Component and New KW Shares due by reference to the respective value of the KW Cash Component and the market value of the New KW Shares on the date on which the conditions for the disposal are satisfied (being that the Scheme becomes effective and the KWE Special Distribution is paid if, when and to the extent required by the Scheme) together with, in the case of shareholders within the charge to UK corporation tax, any indexation allowance available.

(2) Taxation of the KWE Special Distribution and Closing Dividend

UK Holders who accept the New Offer should be aware that the KWE Special Distribution is expected to be made under Article 115 of the Jersey Companies Law and is expected to be treated in consequence (although the analysis is not free from doubt) as a dividend for UK tax purposes. The Closing Dividend will be treated as a dividend for UK tax purposes.

(a) Individuals

UK Holders who are within the charge to UK income tax will pay no tax on their dividend income up to the annual dividend allowance of £5,000. The rates of income tax on dividends received above the dividend allowance are currently (i) 7.5% for basic rate taxpayers; (ii) 32.5% for higher rate taxpayers; and (iii) 38.1% for additional rate taxpayers. Dividend income that is within the dividend allowance counts towards an individual's basic or higher rate limits and will therefore affect the rate of tax that is due on any dividend income in excess of the annual dividend allowance. In calculating into which tax band any dividend income over the £5,000 allowance falls, savings and dividend income are treated as the highest part of an individual's income. Where an individual has both savings and dividend income, the dividend income is treated as the top slice. There is no longer any UK dividend tax credit

Special rules may apply to individuals who are resident but not domiciled in the UK for UK tax purposes.

(b) Other shareholders

UK Holders which are subject to UK corporation tax and are not categorised as small companies will prima facie be subject to UK corporation tax on any dividends on their Scheme Shares unless certain conditions for exemption are satisfied. The exemption for such UK Holders is of wide application and such UK Holders will therefore generally not be subject to UK corporation tax on the dividend. UK Holders which are subject to UK corporation tax and are categorised as small companies will be subject to UK corporation tax on any dividends on their Scheme Shares.

Such shareholders should consult their own professional advisers.

(3) UK Stamp Duty and Stamp Duty Reserve Tax ("SDRT")

No UK stamp duty or SDRT will be payable by the UK Holders in respect of the transfer of the Scheme Shares or on the issue of the New KW Shares, in each case, pursuant to the Merger.

No UK stamp duty or SDRT will be payable by UK Holders on the issue of KW CDIs by CREST Depository Limited.

1.2 Holdings of New KW Shares

(1) Taxation of Dividends on New KW Shares

KW is not required to withhold UK tax from dividends paid on the New KW Shares.

A UK Holder who receives New KW Shares and then receives a dividend on the New KW Shares may be subject to UK corporation or UK income tax (as the case may be) on that dividend.

As described in paragraph 2.2(4)(a) of this Part Seven below, US tax will generally be required to be withheld from dividends paid on New KW Shares. The normal rate of tax to be withheld is 30% of the gross amount of the dividend. This rate may, however, be reduced under an applicable double tax treaty. The rate of withholding on dividends for UK Holders who are entitled to claim (and who make a valid claim for) the benefit of the US-UK Double Tax Treaty is generally 15%.

If a UK Holder receives a dividend on his or her (or its) New KW Shares (including where represented by KW CDIs) and US tax is withheld from the payment of the dividend, credit for such US tax may be available for set-off against a liability to UK corporation tax or UK income tax on the dividend. The amount of the credit will normally be equal to the lesser of the amount withheld and the liability to UK tax on the dividend. The credit will not normally be available for set-off against a UK Holder's liability to UK tax other than on the dividend and, to the extent that the credit is not set off against UK tax on the dividend, the credit will be lost.

(a) Individuals

UK Holders who receive New KW Shares and who are within the charge to UK income tax will pay no tax on their dividend income up to the annual dividend allowance of £5,000. The UK government has proposed legislation to reduce the amount of the dividend allowance to £2,000 for dividends received in the 2018-19 tax year and subsequent years. The rates of income tax on dividends received above the dividend allowance are currently (i) 7.5% for basic rate taxpayers; (ii) 32.5% for higher rate taxpayers; and (iii) 38.1% for additional rate taxpayers. Dividend income that is within the dividend allowance counts towards an individual's basic or higher rate limits and will therefore affect the rate of tax that is due on any dividend income in excess of the annual dividend allowance. In calculating into which tax band any dividend income over the £5,000 (or £2,000, as the case may be) allowance falls, savings and dividend income are treated as the highest part of an individual's income. Where an individual has both savings and dividend income, the dividend income is treated as the top slice. There is no longer any UK dividend tax credit.

Special rules may apply to individuals who are resident but not domiciled in the UK for UK tax purposes.

(b) Other shareholders

UK Holders who receive New KW Shares and who are within the charge to UK corporation tax will *prima facie* be subject to UK corporation tax on any dividends on the New KW Shares unless certain conditions for exemption are satisfied. The exemption is of wide application and such UK Holders will therefore generally not be subject to UK corporation tax on the dividend.

(2) Taxation of Chargeable Gains

A disposal or deemed disposal of New KW Shares by a UK Holder for UK tax purposes may, depending on the UK Holder's particular circumstances and subject to any available exemption or relief, give rise to a chargeable gain or allowable loss for the purposes of capital gains tax or corporation tax on chargeable gains.

Individuals who are temporarily non-resident in the UK may, in certain circumstances, be subject to capital gains tax in respect of gains realised on a disposal of New KW Shares during their period of non-residence.

Special rules may apply to individuals who are resident but not domiciled in the UK for UK tax purposes.

(3) UK Stamp Duty and SDRT

No UK stamp duty will be payable in respect of a paperless transfer of New KW Shares in dematerialised form.

No UK stamp duty will be payable on a written transfer of New KW Shares if such transfer is executed and retained outside the UK and does not relate to any property situated in the UK or to any other matter or thing done or to be done in the UK (which may include, without limitation, the involvement of UK bank accounts in payment mechanics).

No UK SDRT will arise in respect of an agreement to transfer New KW Shares.

No UK stamp duty will arise on transfers of KW CDIs within the CREST system.

No UK SDRT will arise on transfers of KW CDIs within the CREST system.

2. US Federal Income Taxation

The following is a summary of the material US federal income tax consequences of (i) the Merger to US Holders and non-US Holders (each as defined below) who exchange Scheme Shares in the Merger and (ii) the ownership and disposition of New KW Shares received in the Merger to non-US Holders. This summary does not purport to consider all aspects of US federal income taxation that might be relevant to Scheme Shareholders. This discussion applies only to Scheme Shareholders that hold their Scheme Shares, and non-US Holders that will hold their New KW Shares received in the Merger, as capital assets for US federal income tax purposes (generally, property held for investment). This discussion does not purport to consider all aspects of US federal income taxation that may be relevant to Scheme Shareholders in light of their particular circumstances or that are subject to special rules under the US federal income tax laws, including but not limited to:

- Dealers in securities;
- Traders in securities that elect to use a mark-to-market method of accounting for securities holdings;
- Tax-exempt organizations;
- Life insurance companies;
- Financial institutions;
- Regulated investment companies;
- Real estate investment trusts;
- "Controlled foreign corporations";
- "Passive foreign investment companies";
- Partnerships (or entities or arrangements treated as partnerships for US federal income tax purposes);
- Scheme Shareholders that hold their Scheme Shares as part of a straddle, hedge or conversion transaction;
- Certain US expatriates;
- US Holders that own or have owned, actually or constructively, 10% or more of the voting shares of KWE;
- US Holders that have a functional currency other than the US dollar; or
- Scheme Shareholders that acquired their Scheme Shares upon the exercise of stock options or otherwise as compensation.

This discussion is based on the Internal Revenue Code of 1986, as amended (the "Internal Revenue Code"), its legislative history, final, temporary and proposed US Treasury Regulations, published rulings and administrative guidance from the Internal Revenue Service (the "IRS") and court decisions, all as of the date of this document. These laws and other authorities are subject to change, possibly on a retroactive basis.

This discussion addresses only US federal income taxation and does not address any aspect of foreign, state, local, alternative minimum, estate, gift or other tax law that may be applicable to a Scheme Shareholder.

This discussion is intended to provide only a general summary of the material US federal income tax consequences of the Merger and of the ownership and disposition of New KW Shares received pursuant to the Merger. KWE does not intend for this discussion to be a complete analysis or description of all potential US federal income tax consequences of the Merger or of the ownership and disposition of New KW Shares. The US federal income tax laws are complex and subject to varying interpretations. Accordingly, the IRS may not agree with the tax consequences described below.

If a partnership (including an entity or arrangement treated as a partnership for US federal income tax purposes) holds Scheme Shares or, following the Merger, New KW Shares received in the Merger, the tax treatment of a partner in such partnership generally will depend on the status of the partner and the activities of the partner and the partnership. A partner of a partnership holding Scheme Shares or, following the Merger, New KW Shares received in the Merger, should consult the partner's tax adviser regarding the US federal income tax consequences of the Merger and of the ownership and disposition of the New KW Shares to the partner.

Each Scheme Shareholder is urged to consult with its own tax adviser as to the tax consequences of the Merger and of the ownership and disposition of the New KW Shares in light of its own particular circumstances, including the applicability and effect of US federal (including the alternative minimum tax), state, local and foreign tax laws and of changes in those laws.

2.1 US Holders

For purposes of this discussion, "US Holder" means a beneficial owner of Scheme Shares that is, for US federal income tax purposes:

- An individual who is a citizen or resident of the United States;
- A corporation (or other entity taxable as a corporation for US federal income tax purposes) created or organised under the laws of the United States, any state thereof, or the District of Columbia;
- A trust that (i) is subject to the supervision of a court within the United States and the control of one or more US persons or (ii) has a valid election in effect under applicable US Treasury Regulations to be treated as a US person; or
- An estate, the income of which is subject to US federal income tax regardless of its source.

US Holders residing outside the United States may be subject to special tax rules and should consult with their own tax advisers as to the consequences of these rules.

(1) Exchange of Scheme Shares pursuant to the Merger

The exchange of Scheme Shares for New KW Shares (and, if applicable, the KW Cash Component) pursuant to the Merger will be a taxable transaction for US federal income tax purposes. Because KWE believes that it is and has been a PFIC, special tax rules that generally have a penalizing effect may apply to US Holders who exchange their Scheme Shares pursuant to the Merger. US Holders should review the discussion of these special tax rules under paragraph 2.1(3) (*PFIC Considerations*) of this Part Seven below.

In general, subject to the PFIC Considerations below, a US Holder who receives New KW Shares (and, if applicable, the KW Cash Component) in exchange for Scheme Shares pursuant to the Merger will recognise gain or loss for US federal income tax purposes equal to the difference, if any, between (i) the sum of the fair market value of the New KW Shares and the US dollar value of any cash (including cash in lieu of a fractional New KW Share) received in exchange for Scheme Shares pursuant to the Merger and (ii) the US Holder's adjusted tax basis in the Scheme Shares surrendered. A US Holder's adjusted tax basis in its Scheme Shares generally equals the price the US Holder paid for such shares. The US dollar value of any cash received in pounds sterling in exchange for Scheme Shares pursuant to the Merger will be equal to the US dollar value of such pounds sterling translated at the spot rate of exchange as of the date of receipt.

Gain or loss must be determined separately for each block of Scheme Shares (i.e., Scheme Shares acquired at the same cost in a single transaction). Such gain or loss will be long-term capital gain or loss if the US Holder's holding period in a particular block of Scheme

Shares exceeds one year as of the date on which the U.S. Holder exchanges its Scheme Shares, except to the extent treated as ordinary income pursuant to the PFIC rules discussed below. Because KWE believes it is a PFIC, as discussed below, a US Holder that has not made an effective QEF election in respect of its Scheme Shares that receives New KW Shares and/or the KW Cash Component in exchange for Scheme Shares pursuant to the Merger will generally be required (i) to pay tax on any gain described in this paragraph at ordinary income (rather than capital gains) tax rates and (ii) unless the US Holder has made an effective mark-to-market election, to pay a punitive interest charge. See paragraph 2.1(3) (*PFIC Considerations*) of this Part Seven below.

A US Holder's tax basis in the New KW Shares received pursuant to the Merger generally will equal the fair market value of such shares as of the date on which the U.S. Holder receives them, and the holding period of such shares will begin on the following day.

If a US Holder that receives pounds sterling in exchange for Scheme Shares pursuant to the Merger converts the pounds sterling so received into US dollars on the date of receipt, such US Holder generally will not recognise foreign currency gain or loss in respect of such pounds sterling. If the pounds sterling received are not converted into US dollars by the US Holder on the date of receipt, such US Holder may recognise foreign currency gain or loss on a subsequent conversion of such pounds sterling to US dollars. The amount of any such gain or loss recognised in connection with a subsequent conversion or disposal generally will be US-source ordinary income or loss. US Holders should consult their own tax advisers regarding the US federal income tax consequences of receiving, owning and disposing of foreign currency.

Under certain circumstances, the receipt of the KW Cash Component in exchange for Scheme Shares pursuant to the Merger by a US Holder that also actually or constructively owns KW common stock (including KW stock deemed received in the Merger) may be treated as having been distributed in redemption of KW shares deemed issued to such holder in connection with the transaction, which could give rise to deemed dividend income up to the amount of such KW Cash Component. If the receipt of the KW Cash Component in exchange for Scheme Shares pursuant to the Merger were treated as a distribution in this way, such distribution would be treated first as a dividend paid by KW to the extent of its current and accumulated earnings and profits and then as a dividend paid by KWE to the extent of its current and accumulated earnings and profits (in each case, as determined for US federal income tax purposes). KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profit (for US federal income tax purposes) for the financial year ending 31 December 2017. Any deemed dividend paid by KWE may be treated as an "excess distribution" from KWE for purposes of the PFIC rules described below. The possibility of distribution treatment depends upon each holder's particular circumstances, including the application of constructive ownership rules.

(2) Tax Consequences of the KWE Special Distribution

As with ordinary course dividends previously paid by KWE, KWE will report the KWE Special Distribution as a distribution with respect to Scheme Shares for US federal income tax purposes. KWE's reporting of the KWE Special Distribution in such manner will be based on the representation by KW to KWE and the KWE Independent Directors, in connection with the arrangements relating to the KWE Special Distribution, that KW currently has and at the Effective Date will have no intention, expectation or plan to contribute capital to, or otherwise fund, KWE in order to replenish, to any extent, the cash distributed pursuant to the KWE Special Distribution.

Consistent with KWE's reporting, the KWE Special Distribution should be treated as a distribution by KWE with respect to Scheme Shares and, subject to the discussion under "PFIC Considerations" below, the amount of the KWE Special Distribution paid to Scheme Shareholders electing to receive the New Offer Consideration and Distribution

generally will be treated first as a taxable dividend to the extent of KWE's current and accumulated earnings and profits (as determined for US federal income tax purposes), then as a non-taxable return of capital by KWE to the extent of the Scheme Shareholder's basis in its Scheme Shares, and finally as capital gain from the sale or exchange of Scheme Shares.

Certain US Holders could recognise dividend income with respect to the KWE Special Distribution even though they recognise a loss upon the exchange of their Scheme Shares for New KW Shares and the KW Cash Component. Any dividend income recognised by a US Holder in respect of the KWE Special Distribution will not be eligible for the lower capital gains rate applicable to "qualified dividend income" (as defined in the Internal Revenue Code). Any dividend income recognised in respect of the KWE Special Distribution generally will be treated as foreign source income for US foreign tax credit purposes.

For US federal tax purposes, the amount of the KWE Special Distribution paid in pounds sterling will be equal to the US dollar value of such pounds sterling, calculated by reference to the spot rate in effect on the date such distribution is received by the US Holder, regardless of whether and when such currency is in fact converted into US dollars. If the pounds sterling received in the KWE Special Distribution are converted into US dollars on the date received, the US Holder generally should not recognise foreign currency gain or loss on such conversion. If the pounds sterling are not converted into US dollars by the US Holder on the date of receipt, such US Holder may recognise foreign currency gain or loss on a subsequent conversion or other disposal of such pounds sterling. Such foreign currency gain or loss generally will be treated as US-source ordinary income or loss. US Holders should consult their own tax advisers regarding the US federal income tax consequences of receiving, owning and disposing of foreign currency.

(3) PFIC Considerations

If KWE were characterized as a PFIC, such characterization could result in adverse tax consequences to US Holders, and US federal income tax consequences different from those described above may apply. KWE believes that it is and has been a PFIC for each year during a US Holder's holding period in its Scheme Shares. A non-US corporation, such as KWE, will be a PFIC for any taxable year in which, after taking into account the income and assets of any subsidiaries which are at least 25% owned, either (i) at least 75% of its gross income consists of passive income (such as dividends, interest, rents, royalties and the excess of gains over losses from the disposition of assets which produce passive income) or (ii) at least 50% of the average quarterly value of its assets as determined on the basis of fair market value consists of assets that produce, or are held to produce, passive income. For each taxable year that a corporation is treated as a PFIC with respect to a US Holder, such US Holder is subject to special tax rules with respect to any "excess distribution" that it receives and any gain realised from a sale or other disposition of its PFIC shares, unless such US Holder makes a QEF election or mark-to-market election as discussed below.

Distributions received with respect to Scheme Shares, including the KWE Special Distribution and any portion of the KW Cash Component treated as a deemed dividend from KWE (as described above), in a taxable year that are greater than 125% of the average annual distributions received in respect of such Scheme Shares during the shorter of the three preceding taxable years or a US Holder's holding period for the Scheme Shares will be treated as an "excess distribution" and are subject to special tax rules. Gains from the disposition of Scheme Shares are also subject to these rules. Under these special tax rules:

- (i) the excess distribution and/or gain will be allocated ratably over a US Holder's holding period for the Scheme Shares;
- (ii) the amount allocated to the current taxable year, and any taxable years in a US Holder's holding period prior to the first taxable year in which KWE was a PFIC, will be treated as ordinary income; and

(iii) the amount allocated to each of the other taxable years will be subject to the highest tax rate in effect for individuals or corporations, as applicable, for each such year and an interest charge calculated based on the rates and methods applicable to underpayments of tax for each such year will be imposed on the resulting increase in tax attributable to each such year.

The tax liability for amounts allocated to taxable years prior to the year of disposition or excess distribution cannot be offset by any net operating losses for such years, and gains (but not losses) realised on the sale of the Scheme Shares cannot be treated as capital gains, even if the Scheme Shares are held as capital assets.

A US Holder that has made an effective QEF election with respect to its Scheme Shares applicable to each taxable year during the US Holder's holding period and during which KWE was a PFIC generally will recognise (subject to the discussion above regarding the possibility of deemed dividend treatment), upon the exchange of Scheme Shares for New KW Shares (and, if applicable, the KW Cash Component) pursuant to the Merger, capital gain or loss equal to the difference, if any, between the amount realised upon such exchange and its adjusted tax basis in such Scheme Shares. A US Holder's tax basis in its Scheme Shares is increased by amounts included in the US Holder's income pursuant to a QEF election and is decreased by the amount of distributions that the US Holder receives which are nontaxable as a result of the QEF election (including the KWE Special Distribution, subject to the discussion under paragraph 2.1(2) (Tax Consequences of the KWE Special Distribution) of this Part Seven above). The gain or loss recognised by such a US Holder generally will be long-term capital gain or loss if the US Holder held such Scheme Shares for more than one year as of the date on which the U.S. Holder exchanges such Scheme Shares. In certain circumstances, US Holders that are individuals may be entitled to preferential tax treatment for net long-term capital gains. The ability of US Holders to offset capital losses against ordinary income is limited.

A US Holder that has made an effective mark-to-market election with respect to its Scheme Shares generally will include (subject to the discussion above regarding the possibility of deemed dividend treatment) in income any gain realised on the exchange of the Scheme Shares for New KW Shares (and, if applicable, the KW Cash Component) pursuant to the Merger and will treat such gain as ordinary income. Ordinary loss treatment will also apply to any loss realised on the exchange, to the extent that the amount of such loss does not exceed the net mark-to-market gains previously included in income for such Scheme Shares. If a US Holder has made a mark-to-market election with respect to its Scheme Shares, the tax rules that apply to distributions by corporations which are not PFICs would apply to distributions by KWE (including the KWE Special Distribution, subject to paragraph 2.1(2) (*Tax Consequences of the KWE Special Distribution*) of this Part Seven, above), except that the lower capital gains rate applicable to "qualified dividend income" (as defined in the Internal Revenue Code) would not apply.

US Holders are urged to consult their tax advisers concerning the impact of the PFIC rules on the Merger, including, without limitation, the effect of making or having made a QEF election or mark-to-market election, if any.

(4) Potential Deemed Liquidating Distribution

KW and KWE are evaluating the possibility of causing KWE to be treated as a partnership for US federal income tax purposes effective on or immediately prior to the date on which KW acquires ownership of all Scheme Shares pursuant to the Merger. However, KW has undertaken for the benefit of all Scheme Shareholders that it will not take any step, and will procure that none of its Affiliates will take any step, calculated to cause KWE to be treated as a partnership for US tax purposes effective on or prior to the date on which it acquires the Scheme Shares pursuant to the Scheme, except with the prior written consent of the KWE Independent Directors. The KWE Independent Directors will not give their consent to this unless, seven (7) days prior to the Scheme Court Meeting, they have announced their intention to do so through a Regulatory Information Service and unless KW undertakes not to take, and to prevent its Affiliates from taking, any such step until after the Court agrees to sanction the Scheme on the basis that this will occur.

If KWE were treated as a partnership for US federal income tax purposes on the date on which KW acquires ownership of all Scheme Shares pursuant to the Merger, it is expected that a US Holder would be treated for US federal income tax purposes as (i) having received the KWE Special Distribution and its pro rata share of KWE's assets (subject to KWE's liabilities) in a liquidating distribution from KWE (the "Deemed Liquidating Distribution"), (ii) as having contributed such assets (subject to KWE's liabilities) to a newly formed foreign partnership (the "Deemed Contribution"), and (iii) as having sold its interest in such partnership to KW for New KW Shares (and, if applicable, the KW Cash Component) in a taxable transaction (the "Deemed Sale"). Subject to the discussion below regarding US Holders that have made a QEF election with respect to their KWE Shares, gain or loss on the Deemed Liquidating Distribution generally would be determined in the same manner as described in paragraph 2.1(1) (Exchange of Scheme Shares pursuant to the Merger) above (except that (i) the amount deemed realised would be equal to the sum of the fair market value of the KWE assets (subject to KWE's liabilities) deemed received, including the KWE Special Distribution, and (ii) if KWE were characterized as a PFIC, such liquidating distribution would be treated as an "excess distribution" for purposes of the PFIC rules). In this alternative, a US Holder would be allocated its pro rata share of KWE's income, gain, loss, credit and deduction for US federal income tax purposes for the period from the Deemed Contribution to the Deemed Sale. Such allocations generally would increase or decrease, as the case may be, a US Holder's adjusted tax basis in its partnership interests and, accordingly, no additional gain or loss is expected to be recognised by a US Holder with respect to the KWE Special Distribution, the Deemed Contribution or the Deemed Sale. Because KWE may have income that would be treated, for US federal income tax purposes, as "unrelated business taxable income" or "unrelated debt-financed income," US Holders that are tax-exempt organisations under US federal income tax law should consider selling their Scheme Shares prior to the effective date of the Deemed Liquidating Distribution (which would be expected to occur on or immediately prior to the date on which KW acquires ownership of all Scheme Shares pursuant to the Merger).

Because US Holders that have made a QEF election with respect to their Scheme Shares generally are required to include in gross income their *pro rata* share of the ordinary earnings and net capital gain of KWE for each taxable year of KWE, such holders could be adversely affected by a deemed liquidation of KWE (which would cause KWE to recognise gain or loss for US federal income tax purposes as if it had sold its assets for fair market value), even if such US Holders sell their Scheme Shares prior to the Deemed Liquidating Distribution (which would be expected to occur on or immediately prior to the date on which KW acquires ownership of all Scheme Shares pursuant to the Merger). If KWE were treated as a partnership for US federal income tax purposes, US Holders that have made a QEF election could, for example, recognise ordinary income in respect of their *pro rata* share of gain with respect to KWE's deemed sale of its assets, even though they recognise a capital loss (which generally cannot be used to offset ordinary income) in respect of the Deemed Liquidating Distribution or a sale of their Scheme Shares.

If KWE were to be treated as a partnership for US federal income tax purposes on any date during KWE's current taxable year subsequent to the date on which KW acquires ownership of all Scheme Shares pursuant to the Merger, in a manner that caused KWE to recognize gain under Section 336 of the Internal Revenue Code, then US Holders that have made a QEF election could still be subject to the treatment described in this paragraph. KW does not currently intend or expect to bring about the outcome described in the preceding sentence but expressly reserves the right to do so if it believes this to be appropriate in the circumstances at the relevant time.

US Holders who have made a QEF election with respect to their Scheme Shares or who are considering making such an election should consult their tax advisers concerning the impact of this possible change in the transaction treatment to them including, without limitation, the consequences of a deemed liquidation of KWE if KWE announces that consent has been given to allow KW to elect to treat KWE as a partnership for US federal income tax purposes.

(5) Medicare Tax

A US Holder that is an individual or estate, or a trust that does not fall into a special class of trusts that is exempt from such tax, is subject to a 3.8% tax on the lesser of (1) the US Holder's "net investment income" (or "undistributed net investment income" in the case of an estate or trust) for the relevant taxable year and (2) the excess of the US Holder's modified adjusted gross income for the taxable year over a certain threshold (which in the case of individuals is between US\$125,000 and US\$250,000, depending on the individual's circumstances). A US Holder's net investment income generally includes its dividend income and its net gains from the disposition of Scheme Shares (including pursuant to the Deemed Liquidating Distribution), unless such dividend income or net gains are derived in the ordinary course of the conduct of a trade or business (other than a trade or business that consists of certain passive or trading activities). A US Holder that is an individual, estate or trust is urged to consult its own tax advisers regarding the applicability of the Medicare tax to its income and gains in respect of the Merger.

(6) US Federal Backup Withholding and Information Reporting

A US Holder may be subject to information reporting and US federal backup withholding (currently, at a rate of 28%) with respect to the KW Cash Component, the KWE Special Distribution, any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer, the KW First Dividend, subsequent KW Dividends and/or the proceeds from the sale of New KW Shares, unless the US Holder properly establishes an exemption from backup withholding or provides its correct taxpayer identification number (generally, on an IRS Form W-9 provided to the solicitation agent or the US Holder's broker) and otherwise complies with the applicable requirements of the backup withholding rules.

In accordance with Clause 8.10 of the Scheme, KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct IRS Form W-9 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed IRS Form W-9 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct IRS Form W-9 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-9 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-9 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed IRS Forms W-9, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of IRS Form W-9 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A SEPARATE APPROPRIATELY COMPLETED IRS FORM W-9 IN THE MANNER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-9 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-9 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-9 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-9 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR

INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-9, OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

US federal backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability) or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner.

FAILURE TO COMPLETE AND RETURN IRS FORMS W-9 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

Scheme Shareholders should consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e. Form W-9) and any supporting documentation to be provided to avoid potential US federal backup withholding.

2.2 Non-US Holders

For purposes of this discussion, "non-US Holder" means a beneficial owner of Scheme Shares and, following the Merger, New KW Shares, that is neither a "US Holder" (as defined above) nor a partnership (including any entity or arrangement treated as a partnership for US federal income tax purposes).

(1) Exchange of Scheme Shares pursuant to the Merger

A non-US Holder generally will not be subject to US federal income tax on any gain recognised on the receipt of the Original Offer Consideration or the New KW Share Component (and, if applicable, the KW Cash Component) pursuant to the Merger unless:

- the gain is "effectively connected" with the non-US Holder's conduct of a trade or business in the United States (and, if required by an applicable income tax treaty, is attributable to a permanent establishment maintained by such non-US Holder in the United States); or
- the non-US Holder is an individual and is present in the United States for 183 or more days in the taxable year of the sale, and certain other conditions exist.

Gain that is effectively connected with the conduct of a trade or business in the United States generally will be subject to US federal income tax on a net income basis, at regular US federal income tax rates. If the Non-US Holder is a foreign corporation, such gain may also, under certain circumstances, be subject to an additional "branch profits tax" at a 30% rate or at a lower rate if such non-US Holder is eligible for the benefits of an income tax treaty that provides for a lower rate. An individual non-US Holder who is subject to US federal income tax because the non-US Holder was present in the United States for 183 days or more during the year of sale will be subject to a flat 30% tax on the gain derived from such sale, which may be offset by US-source capital losses.

As described in paragraph 2.1(1) (Exchange of Scheme Share pursuant to the Merger) above, under certain circumstances, the receipt of the KW Cash Component pursuant to the Merger may be treated as having been distributed to Scheme Shareholders in redemption of KW shares deemed issued to them in connection with the transaction. Under such treatment, a portion of the cash received by a non-US Holder could be treated as a dividend or as gain or loss from the sale or exchange of KW common stock as described in paragraph 2.2(4)(b). KW does not believe that such treatment is likely to apply and intends to treat the receipt of the KW Cash Component as sale proceeds as described above in this paragraph 2.2(1) that are not subject to US federal income tax withholding. KW has agreed, on the terms set out in Clause 8.10 and Clause 8.11 of the Scheme to procure that all payments of the KW Cash Component are made without withholding or deduction of US federal income tax, provided that US backup withholding may apply to a holder fails to establish that it is not subject to US federal backup withholding.

Notwithstanding that KW will not withhold US federal income tax from payments of the KW Cash Component, if such payments were in fact treated as having been distributed to Scheme Shareholders in redemption of KW Shares, a Non-US holder of Scheme Shares could potentially be subject to US federal income tax as a result of such treatment. The KW Cash Component would, however, only be treated in such manner if Scheme Shareholders that own at least 50% of KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme come to own at least 50% of KW immediately after such time, in each case taking into account constructive ownership rules.

In that case, all or a portion of the cash received by a non-US Holder in respect of the KW Cash Component could be subject to US tax on deemed dividends by KW or treated as a gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme.

- Any such deemed dividend would be treated as paid by KW to the extent of its 2017 "earnings and profits" for US federal income tax purposes. This tax would not exceed 30% of the lesser of (a) the KW Cash Component received by the relevant non-US Holder and (b) the non-US Holder's attributable proportion of such earnings and profits (and may, in appropriate cases, be eligible for a reduced rate of US income taxation under an applicable tax treaty). Any earnings and profits of KW would be allocated ratably among actual and deemed dividends paid by KW in 2017. KW has determined and has advised the KWE Independent Directors in writing that it does not have accumulated earnings and profits for 2017 to date and expects not to have current earnings and profits for the whole of 2017. On the basis that this is correct, no portion of the KW Cash Component would be treated as a deemed dividend from KW. KW has also undertaken to notify the KWE Independent Directors in writing immediately of any change prior to the Effective Date calculated to alter the expectation that it will have no earnings and profits (for US federal income tax purposes) for the financial year ending 31 December 2017.
- In addition, even if, contrary to KW's stated expectation, KW were to have a material amount of earnings and profits for US federal income tax purposes for the financial year ending 31 December 2017, any non-US Holder whose percentage interest (taking into account the constructive ownership rules described below) in all of the outstanding KW Shares immediately after the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme (including, but not limited to, New KW Shares acquired pursuant to the Scheme, as well as any KW Shares otherwise acquired in connection therewith) is less than 80% of such non-US Holder's relative interest in KWE Shares immediately prior to such time (the "Less Than 80% Test") will not be treated as receiving a deemed dividend (and will thus not be subject to the 30% deemed dividend tax referred to above on any portion of the KW Cash Component received pursuant to the Scheme).

For example, a non-US Holder that held a 5% interest in KWE immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme could not be subject to the 30% deemed dividend tax unless such non-US Holder holds a 4% or greater interest in KW Shares immediately after

such time. Accordingly, the 30% deemed dividend tax would not apply to any non-US Holder who does not actually or constructively hold any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme, but who only acquires KW Shares pursuant to the Scheme. Under constructive ownership rules, a non-US Holder generally will be treated as owning stock owned by certain related persons (such as an individual's spouse, children, grandchildren and parents), as well as, in certain circumstances, such holder's allocable share of stock held by any corporation (such as KWE shares owned by KW immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme) or any partnership, estate or trust, in each case, in which the non-US Holder or its relatives hold an interest. In addition, a non-US Holder which actually or constructively holds an option to purchase stock, is treated as owning the underlying stock.

A non-US Holder will not be subject to this form of 30% deemed dividend tax as a result of choosing to receive the consideration available under the New Offer or as a result of making any Additional Share Elections under the Mix and Match Facility if that non-US Holder does not own, actually or constructively, any KW Shares immediately prior to the time at which beneficial ownership of the Scheme Shares is acquired by KW pursuant to the Scheme. Accordingly, non-US Holders of Scheme Shares may wish to consider whether to avoid (or limit) any further acquisitions of KW Shares prior to such time. Those non-US Holders who already own KW Shares may also wish to avoid (or limit) any Additional Share Elections under the Mix and Match Facility, which could jeopardise their ability to satisfy the Less Than 80% Test, or to consider disposing of some or all of their KW Shares, or to reduce the KW Shares they will receive under the Scheme by reducing their holdings of KWE Shares in time, in either case insofar as that would remove or mitigate their potential liability to this form of deemed dividend taxation.

Assuming that a substantial number of beneficial owners of Scheme Shares satisfy the Less Than 80% Test and thus are not treated as if they had received deemed dividends, the remaining recipients of the KW Cash Component payable to non-US Holders (i.e. those which do not satisfy the Less Than 80% Test) would be allocated a significantly larger portion of KW's earnings and profits. However, a non-US holder will never be allocated earnings and profits in excess of the KW Cash Component it receives and would thus never be subject to the 30% deemed dividend tax in respect of an amount greater than the KW Cash Component it receives.

• To the extent that the KW Cash Component is not treated as a deemed dividend on the basis described above, it may nonetheless be treated as gain or loss from a notional redemption of KW Shares deemed to be issued in connection with the Scheme and subjected to US federal income tax if the non-US Holder owned (or was treated as owning) more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the deemed redemption and (ii) the non-US Holder's holding period in such shares (a "5% Holder"). This tax would be payable on the graduating scale of US federal income tax rates (including, if applicable, reduced tax rates on long-term capital gains derived by certain non-corporate holders, including individuals) on the amounts deemed to be received after allowing for certain reductions in respect of tax basis. The Less than 80% Test will not avoid or mitigate this tax.

Any beneficial owners of Scheme Shares who are also holders of KW Shares and who wish to avoid the potential application of these rules should consult their tax advisers, including as to the desirability and tax consequences of disposing of some or all their Scheme Shares or KW Shares in the market prior to the Merger.

Any person (whether or not a holder of shares in KW or KWE) who is considering acquiring shares in KW or KWE prior to the Effective Date should consider whether to refrain from doing so if, as a result, it may become liable to additional US tax as a result of the requirements described above.

(2) Tax Consequences of the KWE Special Distribution

As with ordinary course dividends previously paid by KWE, KWE will report the KWE Special Distribution as a distribution by KWE with respect to Scheme Shares for US federal income tax purposes. Accordingly, the KWE Special Distribution will be paid without deduction for any US federal income tax withholding (other than US federal backup withholding, if applicable). KWE's reporting of the KWE Special Distribution in such manner will be based on the representation by KW to KWE and the KWE Independent Directors, in connection with the arrangements relating to the KWE Special Distribution, that KW currently has and at the Effective Date will have no intention, expectation or plan to contribute capital to, or otherwise fund, KWE in order to replenish, to any extent, the cash distributed pursuant to the KWE Special Distribution.

Consistent with KWE's reporting, the KWE Special Distribution should be treated as a distribution with respect to Scheme Shares, and the amount of the KWE Special Distribution paid to Scheme Shareholders electing to receive the New Offer Consideration should not be treated first as a dividend to the extent of KWE's current and accumulated earnings and profits (as determined for US federal income tax purposes), then as a non-taxable return of capital by KWE to the extent of the Scheme Shareholder's basis in its Scheme Shares, and finally as capital gain from the sale or exchange of Scheme Shares.

In such a case, Non-US Holders generally will not be subject to US federal income tax on the KWE Special Distribution except with respect to any portion treated as a dividend that is effectively connected with the non-US Holder's conduct of a trade or business in the United States (and, if required by an applicable income tax treaty, is attributable to a permanent establishment maintained by such non-US Holder in the United States) or any portion treated as gain from the sale or exchange of Scheme Shares taxable in the manner described in paragraph 2.2(1) (Exchange of Scheme Shares pursuant to the Merger) of this Part Seven above.

(3) Potential Deemed Liquidating Distribution

The potential change to the transaction treatment described in paragraph 2.1(4) (Potential Deemed Liquidating Distribution) of this Part Seven above, if implemented, generally would not affect the US federal income tax treatment of non-US Holders, and the US federal income tax consequences to such holders generally would be the same as described under paragraph 2.2(1) (Exchange of Scheme Shares pursuant to the Merger) of this Part Seven above, except that any non-US Holders that would be subject to US federal income tax on any gain recognised on the receipt of the Original Offer Consideration or the New KW Share Component (and, if applicable, the KW Cash Component) by reason of a taxable presence in the United States would be subject to US federal income tax on any gain recognised on the Deemed Liquidating Distribution (determined in the manner described in paragraph 2.1(4) (Potential Deemed Liquidating Distribution) of this Part Seven above). In addition, because the receipt of a share of KWE's income could cause an entity controlled by a foreign government to be treated as a "controlled commercial entity" for US federal income tax purposes, any such entity should consider selling its Scheme Shares prior to the effective date of the Deemed Liquidating Distribution.

(4) Ownership of New KW Shares received by non-US Holders pursuant to the Merger

(a) Distributions

In general, the gross amount of any distribution KW makes to a non-US Holder with respect to the New KW Shares will be subject to US withholding tax at a rate of 30% to the extent the distribution constitutes a dividend for US federal income tax purposes, unless the non-US Holder is eligible for a reduced rate of withholding tax under an applicable income tax treaty and the non-US Holder provides proper certification of its eligibility for such reduced rate. A distribution will constitute a dividend for US federal income tax purposes to the extent of KW's current or accumulated earnings and profits as determined for US federal income tax purposes. To the extent any distribution does not constitute a dividend, it will be treated first as reducing the adjusted basis in the non-US Holder's New KW Shares and then, to

the extent it exceeds the adjusted basis, as gain from the sale or exchange of such stock. Any such gain will be subject to the treatment described in paragraph 2.2(4)(b) of this Part Seven below.

Dividends paid by KW to a non-US Holder that are effectively connected with the conduct of a trade or business by such non-US Holder within the United States (and, if required by an applicable income tax treaty, are attributable to a US permanent establishment of such non-US Holder) will not be subject to US withholding tax, as described above, if the non-US Holder complies with applicable certification and disclosure requirements. Instead, such dividends generally will be subject to US federal income tax on a net income basis, at regular US federal income tax rates. Dividends received by a foreign corporation that are effectively connected with its conduct of a trade or business within the United States may be subject to an additional branch profits tax at a rate of 30% (or such lower rate as may be specified by an applicable income tax treaty).

(b) Gain on Sale or other Disposition of New KW Shares

In general, a non-US Holder will not be subject to US federal income or, subject to paragraphs 2.2(5) (US Federal Backup Withholding and Information Reporting) and 2.3 (FATCA Withholding) of this Part Seven below, US federal withholding tax on any gain realised upon the sale or other disposition of New KW Shares unless:

- the gain is "effectively connected" with a trade or business carried on by the non-US Holder within the United States (and, if required by an applicable income tax treaty, is attributable to a US permanent establishment of such non-US Holder);
- the non-US Holder is an individual and is present in the United States for 183 days or more in the taxable year of disposition and certain other conditions are met; or
- KW is or has been a "US real property holding corporation" (a "USRPHC") and certain other requirements are met.

Gain that is effectively connected with the conduct of a trade or business in the United States generally will be subject to US federal income tax on a net income tax basis, at regular US federal income tax rates. If the non-US Holder is a foreign corporation, the branch profits tax described above may also apply. An individual non-US Holder who is subject to US federal income tax because the non-US Holder was present in the United States for 183 days or more during the year of sale or other disposition of the New KW Shares will be subject to a flat 30% tax on the gain derived from such sale or other disposition, which may be offset by US-source capital losses.

With respect to the third bullet point above, KW disclosed in the KW Prospectus that it believes that it currently is a USRPHC for US federal income tax purposes and has not made a determination as to whether, following the Merger, it will continue to so qualify. In the event that KW continues to be a USRPHC for US federal income tax purposes, as long as the KW Shares are regularly traded on an established securities market (within the meaning of applicable US Treasury Regulations), a non-US Holder will not be subject to US federal income tax on gain realised from a sale of New KW Shares unless the non-US Holder actually or constructively held more than 5% of the outstanding KW Shares at any time during the shorter of (i) the five-year period ending on the date of the non-US Holder's sale or disposition of such shares and (ii) the non-US Holder's holding period for such shares. KW disclosed in the KW Prospectus that it believes that the KW Shares are and will continue to be regularly traded on an established securities market, though no assurances can be made. The gain described in the third bullet point above generally will be taxed in the same manner as gain described in the first bullet point above, except that the branch profits tax will not apply. In addition, if KW Shares were not regularly traded on an established securities market, a purchaser of the KW Shares would generally be required to withhold and remit to the IRS 15% of the purchase price, which withholding would be creditable against any income tax liability described in the previous sentence and generally refundable to the extent in excess thereof.

(5) US Federal Backup Withholding and Information Reporting

The payment to Non-US Holders of the KW Cash Component, the KWE Special Distribution and/or any other amount (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer and/or the KW First Dividend (and subsequent KW dividends) may be subject to US information reporting and US federal backup withholding. US federal backup withholding (currently, at a rate of 28%) is imposed on certain payments to persons that fail to furnish the information required under the US information reporting rules. Payments to a non-US Holder generally will be exempt from backup withholding if the non-US Holder provides to the applicable withholding agent the required certification as to its non-US status (such as by providing a valid IRS Form W-8BEN or IRS Form W-8BEN-E), or otherwise establishes an exemption.

In accordance with Clause 8.10 of the Scheme, KW will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KW Cash Component and any other amount payable by KW under the New Offer (including, without limitation, a payment in respect of a fractional entitlement) will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment, and this Form is accompanied, where required by US law, by the correct applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US federal law.

In accordance with Clause 8.12 of the Scheme, KWE will procure that, except where the deduction or withholding is compulsory by virtue of a change in applicable law which comes into force after 13 September 2017, the KWE Special Distribution and any other amount payable by KWE in connection with the New Offer will be paid free and clear and without deduction for any US federal income tax withholding. This commitment will not however preclude US federal backup withholding by the Receiving Agent, or by a CREST nominee or other intermediary, unless the Receiving Agent and, where required by US federal tax law, the CREST nominee or other intermediary receives (or already holds) a properly executed applicable IRS Form W-8 in respect of the relevant Scheme Shareholder, intermediary or beneficial owner to whom it is making the payment and this Form is accompanied, where required by US law, by the correct applicable IRS Form W-8 in respect of the beneficial owner and, where relevant, its CREST nominee and/or other intermediaries. Accordingly, in the absence of submission of the correct IRS forms, Scheme Shareholders, intermediaries and beneficial owners of Scheme Shares should assume that these payments will be made after deduction for US federal backup withholding unless it is otherwise established that they can be made without deduction in accordance with US

Important information on how to submit your IRS forms. Each beneficial owner who holds Scheme Shares through a CREST nominee, bank, broker or other financial institution (an "intermediary") should contact its CREST nominee or other intermediary to determine the action the beneficial owner needs to take, including submitting duly competed IRS Forms W-8 to the CREST nominee or other intermediary, to avoid potential US federal backup withholding on the KW Cash Component, the KWE Special Distribution and/or any other payment (including, without limitation, a payment in respect of a fractional entitlement) under the New Offer. To avoid the potential for US federal backup withholding in

circumstances where it may apply, a beneficial owner should provide properly executed IRS Forms W-8 to its CREST nominee or other intermediary in order to enable the intermediary to submit relevant information to the Receiving Agent prior to the Election Return Time. Scheme Shareholders, including those who are CREST nominees or other intermediaries, should return all relevant information, including properly executed applicable IRS Forms W-8, by post to Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case by the Election Return Time. Copies of applicable IRS Form W-8 and accompanying instructions may be obtained from the Receiving Agent or from the IRS website at www.irs.gov.

IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KW CASH COMPONENT AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) BY KW UNDER THE NEW OFFER, YOU SHOULD SUBMIT AN APPROPRIATELY COMPLETED IRS FORM W-8 IN THE MANNER SET OUT ABOVE.

IN ADDITION, IN ORDER TO AVOID POTENTIAL US FEDERAL BACKUP WITHHOLDING ON THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT FROM KWE UNDER THE NEW OFFER, YOU SHOULD SUBMIT A SEPARATE APPROPRIATELY COMPLETED IRS FORM W-8 IN THE MANNER SET OUT ABOVE.

PLEASE NOTE THAT YOU MAY THEREFORE NEED TO RETURN TWO IRS FORMS W-8 IN ORDER TO AVOID A REDUCTION IN RESPECT OF US FEDERAL BACKUP WITHHOLDING OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER.

IF YOU BELIEVE THAT KWE IS ALREADY PERMITTED TO PAY DIVIDENDS AND OTHER DISTRIBUTIONS TO YOU WITHOUT DEDUCTION FOR US FEDERAL BACKUP WITHHOLDING (FOR EXAMPLE, IF YOU HAVE PREVIOUSLY SUBMITTED THE RELEVANT FORM AND RECEIVED THE MOST RECENT ORDINARY DIVIDEND FROM KWE WITHOUT ANY US FEDERAL BACKUP WITHHOLDING), PLEASE NONETHELESS SUBMIT THE APPROPRIATE IRS FORM W-8 AND, IF YOU HAVE A CREST NOMINEE OR OTHER INTERMEDIARY, ASK IT ABOUT HOW TO DO THIS.

EVEN IF YOU ELECT TO RECEIVE ONLY THE ORIGINAL OFFER, YOU SHOULD CONTACT YOUR INTERMEDIARY THROUGH WHICH YOU WILL HOLD NEW KW SHARES TO DETERMINE WHETHER YOU NEED TO SUBMIT THE RELEVANT IRS FORMS W-8 TO SUCH INTERMEDIARY WITH RESPECT TO PAYMENTS MADE BY KW TO AVOID US FEDERAL BACKUP WITHHOLDING AND, IF APPLICABLE, ESTABLISH YOUR ENTITLEMENT TO A REDUCED RATE OF US FEDERAL INCOME TAX WITHHOLDING ON DIVIDENDS PAYABLE ON KW SHARES AFTER THE SCHEME BECOMES EFFECTIVE INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND.

ALTHOUGH IRS FORMS W-8 AND/OR RELEVANT INFORMATION FROM INTERMEDIARIES MUST BE RECEIVED IN THE MANNER SET OUT ABOVE BY THE RECEIVING AGENT BY THE ELECTION RETURN TIME, IF IN DOUBT AS TO HOW TO FILL IN YOUR IRS FORMS, YOU SHOULD CONSULT YOUR INTERMEDIARY, OR THE RECEIVING AGENT, AND YOU ARE RECOMMENDED TO SUBMIT THEM AS EARLY AS POSSIBLE IN ORDER TO REDUCE THE RISKS OF THE CONSEQUENCES OF COMPLETING THEM INCORRECTLY.

IF YOU ARE IN ANY DOUBT ABOUT WHERE TO OBTAIN IRS FORMS W-8 OR HOW TO COMPLETE THEM, PLEASE TELEPHONE YOUR CREST NOMINEE OR OTHER INTERMEDIARY OR, IF YOU ARE A SCHEME SHAREHOLDER (INCLUDING A CREST NOMINEE OR OTHER INTERMEDIARY) THE

SHAREHOLDER HELPLINE ON 0370 707 4040 (FROM WITHIN THE UK) OR +44 370 707 4040 (FROM OUTSIDE THE UK) BETWEEN 8.30 A.M. AND 5.30 P.M. FROM MONDAY TO FRIDAY (EXCEPT UK PUBLIC HOLIDAYS).

Please note that, for legal reasons, the Shareholder Helpline cannot provide advice on the advantages or disadvantages of any aspect of the Merger or give any legal, tax or financial advice. All calls may be recorded and monitored for security and training purposes.

The relevant withholding agent (including, if applicable, KW) must report annually to the IRS and to each non-US Holder the amount of dividends paid to, and the tax withheld with respect to, each non-US Holder in respect of its New KW Shares. These reporting requirements apply regardless of whether withholding was reduced or eliminated by an applicable tax treaty. Copies of this information also may be made available under the provisions of a specific treaty or agreement with the tax authorities in the country in which the non-US Holder resides or is established.

Information reporting and backup withholding generally are not required with respect to payments of proceeds from the sale or other disposition of New KW Shares by a non-US holder effected outside the United States through a foreign office of a foreign broker that does not have certain specified connections to the United States. However, if a non-US Holder sells or otherwise disposes of New KW Shares through a US broker or the US offices of a foreign broker, the broker will generally be required to report to the IRS the amount of proceeds paid to the non-US Holder and also backup withhold on that amount unless such non-US holder provides appropriate certification to the broker of its status as a non-US person (generally, in the manner described in the immediately preceding paragraph) or otherwise establishes an exemption. Information reporting will also apply if a non-US Holder sells or otherwise disposes of New KW Shares through a foreign broker deriving more than a specified percentage of its income from US sources or having certain other connections to the United States, unless such broker has documentary evidence in its records that such non-US Holder is a non-US person and certain other conditions are met, or such non-US Holder otherwise establishes an exemption.

US federal backup withholding is not an additional tax. Any amounts withheld under the backup withholding rules can be refunded (for taxpayers which do not have any US federal tax liability and which do not use the amounts withheld to reduce their US federal tax liability or credited against a taxpayer's US federal income tax liability, if any, provided that the taxpayer furnishes the required information to the IRS in a timely manner

FAILURE TO COMPLETE AND RETURN IRS FORMS W-8 ON THE BASIS DESCRIBED ABOVE MAY RESULT IN A REDUCTION BEING MADE FOR US FEDERAL BACKUP WITHHOLDING ON THE PAYMENT OF THE KW CASH COMPONENT, THE KWE SPECIAL DISTRIBUTION AND/OR ANY OTHER PAYMENT (INCLUDING, WITHOUT LIMITATION, A PAYMENT IN RESPECT OF A FRACTIONAL ENTITLEMENT) UNDER THE NEW OFFER AND MAY ALSO IN THE FUTURE RESULT IN UNFAVOURABLE US TAX TREATMENT IN RESPECT OF KW DIVIDENDS (INCLUDING, WITHOUT LIMITATION, THE FIRST KW DIVIDEND) AFTER THE SCHEME BECOMES EFFECTIVE.

The various versions of IRS Forms W-8 for use by different categories of taxpayer are generally as follows:

- W-8BEN for use by individual beneficial owners of Scheme Shares that are not "United States persons";
- W-8BEN-E for use by corporate and certain other non-individual beneficial owners of Scheme Shares that are not "United States persons"; and
- W-8IMY for use by non-US intermediaries and non-US partnerships (or entities treated as partnerships for US federal income tax purposes).

Scheme Shareholders should consult their tax advisers, financial advisers and/or intermediaries regarding the appropriate IRS Forms (i.e. applicable Form W-8) and any supporting documentation to be provided to avoid potential US federal backup withholding.

2.3 FATCA Withholding

Pursuant to sections 1471 through 1474 of the Internal Revenue Code, commonly known as the Foreign Account Tax Compliance Act ("FATCA"), a 30% withholding tax ("FATCA Withholding") may be imposed on certain payments to non-US Holders or certain non-US financial institutions, investment funds and other non-US persons receiving payments on behalf of a non-US Holder if the non-US Holder or such institutions fail to comply with certain information reporting requirements. Such payments include US-source dividends and the gross proceeds from the sale or other disposition of stock that can produce US-source dividends. Dividend payments received by a non-US Holder could be subject to FATCA Withholding if the non-US Holder is subject to these information reporting requirements and fails to comply with them or if the non-US Holder holds New KW Shares through another person, such as a non-US bank or broker, that is subject to withholding because it fails to comply with these requirements. FATCA withholding will not apply to payments of gross proceeds from a sale or other disposition of New KW Shares before January 1, 2019. Non-US Holders should consult with their own tax advisers regarding the relevant US law and other official guidance on FATCA withholding.

The US federal income tax consequences described above are not intended to constitute a complete description of all tax consequences relating to the Merger or of the ownership and disposition of New KW Shares. Because individual circumstances may differ, each Scheme Shareholder should consult its own tax adviser regarding the applicability of the rules discussed above to it and the particular tax consequences of the Merger and of the ownership and disposition of the New KW Shares received in the Merger in light of the holder's particular circumstances and the application of state, local and foreign tax laws.

3. Jersey Taxation

The following summary of the anticipated tax treatment in Jersey of the holders of Scheme Shares is based on Jersey taxation law as it is understood to apply at the date of this document. It does not constitute legal or tax advice. Holders of Scheme Shares should consult their professional advisers on the implications of the Scheme under the laws of the jurisdictions(s) in which they may be liable to taxation. Holders of Scheme Shares should also be aware that tax laws, rules and practice and their interpretation may change.

No taxation or stamp duty will be payable in Jersey by holders of Scheme Shares (other than holders of Scheme Shares resident in Jersey who may be subject to tax depending on their circumstances) as a result of the transfer of Scheme Shares to KW or the implementation of the Scheme and no withholding is required under current law.

PART EIGHT

ADDITIONAL INFORMATION

1. Responsibility statements

- 1.1 Each KWE Independent Director accepts responsibility for the KWE Information (including any expressions of opinion) contained or incorporated by reference in this document, and none of the KWE Independent Directors accepts responsibility for any other information contained or incorporated by reference in this document, including, without limitation: paragraph 10.1 of Part One (Letter from the Independent Directors of Kennedy Wilson Europe Real Estate Plc), paragraphs 9, 10.1, 12, 13.2 and 20 of Part Two (Explanatory Statement), Part Seven (Taxation) (except any information, estimate, opinion or undertaking relating to KW or its tax position), paragraphs 7, 8.2, 10.2, 13.2 and 16.2 of this Part Eight, all the information contained in Part Fourteen (Profit Forecasts) and all other Covered Information. To the best of the knowledge and belief of the KWE Independent Directors (who have taken all reasonable care to ensure that such is the case), the information contained or incorporated by reference in this document for which they accept responsibility is in accordance with the facts and does not omit anything likely to affect the import of such information.
- 1.2 Each KW Director accepts responsibility for the information contained or incorporated by reference in this document (including any expressions of opinion therein), including, without limitation, paragraph 10.1 of Part One (Letter from the Independent Directors of Kennedy Wilson Europe Real Estate Plc), paragraphs 9, 10.1, 12, 13.2 and 20 of Part Two (Explanatory Statement), Part Seven (Taxation) (to the extent of any information, estimate, opinion or undertaking relating to KW or its tax position), paragraphs 7, 8.2, 10.2, 13.2 and 16.2 of this Part Eight, all the information contained in Part Fourteen (Profit Forecasts) and all other Covered Information but excluding all KWE Information, for which each KWE Independent Director accepts responsibility as set out in paragraph 1.1 of this Part Eight. To the best of the knowledge and belief of the KW Directors (who have taken all reasonable care to ensure that such is the case), the information contained or incorporated by reference in this document for which they accept responsibility is in accordance with the facts and does not omit anything likely to affect the import of such information.

2. Directors

2.1 The KWE Directors and their positions are as follows:

Directors	Position
Charlotte Valeur	Non-Executive Chair
Mark McNicholas	Non-Executive Director and Senior Independent Director
Simon Radford	Non-Executive Director
William J. McMorrow	Non-Executive Director
Mary L. Ricks	Non-Executive Director

As at the Latest Practicable Date, Charlotte Valeur, Mark McNicholas and Simon Radford are the KWE Independent Directors.

William J. McMorrow and Mary L. Ricks are representatives of the Manager, which is wholly owned and controlled by KW. For this reason, William J. McMorrow and Mary L. Ricks are not considered independent by the KWE Board.

The registered office of KWE, whose registered number is 114680, is 47 Esplanade, St Helier, Jersey JE1 0BD.

2.2 The KW Directors and their positions are as follows:

Directors	Position		
William J. McMorrow	Chairman and Chief Executive Officer		
Norman Creighton	Director		
Cathy Hendrickson	Director		
David A. Minella	Director		
Kent Y. Mouton	Director and Executive Vice President, General Counsel		
Jerry R. Solomon	Director		
Stanley R. Zax	Director		

The registered office of KW, whose Delaware file number is 4385645, is 151 South El Camino Drive, Beverly Hills, CA 90212, United States of America.

3. Persons acting in concert

3.1 In addition to KWE Group companies and their respective directors, and the close relatives and related trusts (as those terms are used in the Code) of those directors, for the purposes of the Code, the following persons and persons affiliated with them are deemed to be acting in concert with KWE in respect of the Merger:

Name	Type of company	Registered Office	Relationship with KWE
Rothschild	Financial services	N.M. Rothschild and Sons Limited, New Court, St Swithin's Lane, London, EC4N 8AL	Lead financial adviser
J.P. Morgan Cazenove	Financial services	J.P. Morgan Limited, 25 Bank Street, Canary Wharf, London, E14 5JP	Financial adviser

3.2 In addition to KW Group companies and their respective directors, and the close relatives and related trusts (as those terms are used in the Code) of those directors, for the purposes of the Code, the following persons and persons affiliated with them are deemed to be acting in concert with KW in respect of the Merger:

Name	Type of company	Registered Office	Relationship with KW
Goldman Sachs International	Financial services	Peterborough Court, 133 Fleet Street, London, EC4A 2BB	Lead financial adviser
Goldman Sachs & Co LLC	Financial services	200 West Street, New York, NY 10282, United States of America	Lead financial adviser
Merrill Lynch International	Financial services	2 King Edward Street, London, E14 5HP	Financial adviser
Merrill Lynch, Pierce, Fenner & Smith Incorporated	Financial services	One Bryant Park, New York, NY 10036, United States of America	Financial adviser

Name	Type of company	Registered Office	Relationship with KW
Deutsche Bank AG	Financial services	Winchester House, Great Winchester Street, London, EC2N 2DB	Financial adviser
Deutsche Bank Securities, Inc.	Financial services	60 Wall Street, New York, NY 10005, United States of America	Financial adviser

4. KWE Shareholder Approval

4.1 KWE Director irrevocable undertaking

The following irrevocable undertaking has been given to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept or procure acceptance of such Takeover Offer) in relation to the following Scheme Shares:

Name of KWE Independent Director giving undertaking	Number of Scheme Shares held	share capital of KWE in issue as at the Latest Practicable Date
Simon Radford	12,500	0.01%
Total	12,500	0.01%

KW has received an irrevocable undertaking from Simon Radford, who is the only KWE Independent Director who holds Scheme Shares to vote (or use reasonable endeavours to procure the voting) in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept or procure acceptance of the Takeover Offer) in respect of those Scheme Shares owned and/or controlled by him, as well as any further Scheme Shares he may acquire. As indicated above, such Scheme Shares amount to, in aggregate, 12,500 Scheme Shares, representing approximately 0.01% of the share capital of KWE in issue as at the Latest Practicable Date. As at the date of this document, Simon Radford intends to elect to receive the New Offer Consideration and Distribution in respect of the 12,500 Scheme Shares owned and/or controlled by him.

To the extent that any of the Scheme Shares held by Simon Radford are not registered in his name, he shall provide the registered holder(s) with a written irrevocable direction to vote, and shall otherwise use reasonable endeavours to ensure that such registered holder(s) vote, in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept or procure acceptance of such Takeover Offer).

The obligations of Simon Radford under his irrevocable undertaking will lapse and cease to have effect if:

- (a) KW announces, with the consent of the Panel, that it does not intend to proceed with the Merger;
- (b) in the event that the Merger is implemented by way of a Scheme, the Scheme terminates or lapses in accordance with its terms or otherwise becomes incapable of ever becoming effective, provided that KW has not announced that it intends to implement the Merger by way of a Takeover Offer;
- (c) in the event that the Merger is implemented by way of a Takeover Offer, the Takeover Offer lapses or is withdrawn; or

(d) any competing offer for KWE is made which is declared wholly unconditional or otherwise becomes effective.

As at the date of this document, all of the other KWE Directors who hold KWE Shares (being William J. McMorrow and Mary L. Ricks) intend to elect to receive the New Offer Consideration and Distribution in respect of all Scheme Shares owned and/or controlled by them, but for the avoidance of doubt, they are not bound to do so.

4.2 KWE Shareholder irrevocable undertakings

The following Scheme Shareholders have given irrevocable undertakings to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept the Takeover Offer) in relation to the following Scheme Shares currently held by them, as well as any further Scheme Shares they may acquire:

Percentage of

Number of Scheme Shares	share capital of KWE in issue as at the Latest Practicable Date (%)	
15,981,750	12.7	
11,534,900	9.1	
27,516,650	21.8	
	Scheme Shares 15,981,750 11,534,900	

The obligations of Quantum Strategic Partners Ltd. under its irrevocable undertaking will lapse and cease to have effect if:

- at any time before the date of the Scheme Court Meeting, a third party announces its intention to make an offer for KWE or substantially all of its business and assets which, in Quantum Strategic Partners Ltd.'s reasonable opinion, represents an improvement on the New Offer;
- KW's offer to implement the Merger by means of a Scheme lapses or is withdrawn (provided that KW has not announced, within 10 business days of the Scheme having so lapsed or been withdrawn, that it intends to implement the Merger by way of a Takeover Offer); or
- the Scheme has not become effective by the earlier of (i) midday Eastern Standard Time on 31 October 2017 and (ii) any date specified in this document as being the latest date by which the Scheme must have become effective.

Pursuant to the irrevocable undertaking, Quantum Strategic Partners Ltd. has also agreed to refrain from certain actions, including (a) disposing of any KWE Shares currently held by it or any further KWE Shares it may acquire (subject to certain exceptions), (b) acquiring any KWE Shares, (c) selling any KWE Shares short or entering into any short derivative position referenced to them, or (d) soliciting, directly or indirectly, or initiating discussions with a third party in connection with a competing offer for KWE.

The obligations of Franklin Templeton Institutional, LLC under its irrevocable undertaking will lapse and cease to have effect if:

- at any time before the date of the Scheme Court Meeting, a third party announces its intention to make an offer for KWE which is governed by the Code which, in Franklin Templeton Institutional, LLC's reasonable opinion, represents an improvement on the New Offer; or
- KW's offer to implement the Merger by means of a Scheme lapses or is withdrawn (provided that KW has not announced that it intends to implement the Merger by way of a Takeover Offer).

Pursuant to the irrevocable undertaking, Franklin Templeton Institutional, LLC has also agreed to refrain from certain actions, including (a) disposing of any KWE Shares currently held by it or any further KWE Shares it may acquire, (b) acquiring any KWE Shares, (c) selling any

KWE Shares short or entering into any short derivative position referenced to them or (d) soliciting, directly or indirectly, or initiating discussions with a third party in connection with a competing offer for KWE. However, Franklin Templeton Institutional, LLC will be permitted to sell KWE Shares if it is instructed to do so by clients on whose account it holds those KWE Shares.

4.3 KWE Shareholder letter of intent

The Värde Shareholders have confirmed their intention to vote in favour of the Scheme at the Scheme Court Meeting and in favour of the Special Resolution at the KWE General Meeting (or, in the event that the Merger is implemented by way of a Takeover Offer, to accept the Takeover Offer) in respect of their entire holdings of 5,452,722 KWE Shares, representing, in aggregate, approximately 4.3% of the share capital of KWE in issue as at the Latest Practicable Date. The letter of intent is non-binding and does not prevent the Värde Shareholders from modifying or withdrawing their support for the Merger in the event that a higher offer to acquire KWE is made or from transferring or disposing of their interests in the KWE Shares to which the letter of intent relates.

5. KW Shareholder Approval

Fairfax Financial Holdings Limited and Elkhorn Partners LP have indicated to KW that they intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 14.3% of KW's outstanding shares of common stock as at the Latest Practicable Date.

The KW Directors, as well as Mary L. Ricks, Matt Windisch and Justin Enbody (being all of the executive officers of KW who hold KW Shares) and In Ku Lee, also intend to vote in favour of the KW Shareholder Resolution at the KW Special Meeting in respect of their entire holdings, representing, in aggregate, approximately 17.1% of KW's outstanding shares of common stock as at the Latest Practicable Date.

6. Interests and dealings

6.1 Definitions

For the purposes of this paragraph 6 of this Part Eight:

"acting in concert" has the meaning given to it in the Code;

"arrangement" includes indemnity or option arrangements, and any agreement or understanding, formal or informal, of whatever nature, relating to relevant securities which may be an inducement to deal or refrain from dealing;

"dealings" has the meaning given to it in the Code (and "dealt" shall be construed accordingly);

"derivative" has the meaning given to it in the Code;

"disclosure date" means 7 September 2017;

"disclosure period" means the period commencing on 24 April 2016 (being the date 12 months prior to the commencement of the offer period) and ending on the disclosure date;

"interest" or "interests" has the meaning given to it in the Code;

"offer period" means the period commencing on 24 April 2017 and ending on the disclosure date;

"relevant securities" means:

- (A) KWE Shares and any other securities of KWE conferring voting rights;
- (B) equity share capital of KWE or, as the context requires, KW; and
- (C) securities of KWE or, as the context requires, KW carrying conversion or subscription rights into any of the foregoing; and

"short position" means any short position (whether conditional or absolute and whether in the money or otherwise) including any short position under a derivative any agreement to sell or any delivery obligation or right to require another person to purchase or take delivery.

6.2 Interests in relevant securities of KWE

(1) As at the close of business on the disclosure date, the KWE Directors held the following interests in, or rights to subscribe in respect of, relevant securities of KWE:

Name	Number of KWE Shares	Percentage of share capital of KWE in issue as at the Latest Practicable Date	
Simon Radford	12,500	0.01%	
Charlotte Valeur	Nil	0.00%	
Mark McNicholas	Nil	0.00%	
Mary L. Ricks ⁽¹⁾	259,411	0.20%	
William J. McMorrow ⁽²⁾	179,411	0.14%	

⁽¹⁾ This holding consists of 179,922 KWE Shares held by Mary L. Ricks and 79,489 RSUs which will vest after the KWE General Meeting (but before the Scheme Record Time) in connection with the Merger.

(2) As at the close of business on the disclosure date, the KW Directors held the following interests in, or rights to subscribe in respect of, relevant securities of KWE:

Name	Number of KWE Shares	Percentage of share capital of KWE in issue as at the Latest Practicable Date
William J. McMorrow ⁽¹⁾	179,411	0.14
Kent Y. Mouton ⁽²⁾	11,086	0.01

⁽¹⁾ This holding consists of 99,922 KWE Shares held by William J. McMorrow and 79,489 RSUs which will vest after the KWE General Meeting (but before the Scheme Record Time) in connection with the Merger.

⁽³⁾ As at the close of business on the disclosure date, members of the KW Group held the following interests in, or rights to subscribe in respect of, relevant securities of KWE:

Name	Number of KWE Shares	Percentage of share capital of KWE in issue as at the Latest Practicable Date
KW	576,954	0.46
KW Europe Investors Limited	25,350,895	20.10
Manager	1,439,660	1.14
Kennedy Wilson UK Limited	2,289,536	1.82
Kennedy Wilson Europe Limited	305,911	0.24
Kennedy Wilson España SLU	52,968	0.04

⁽²⁾ This holding consists of 99,922 KWE Shares held by William J. McMorrow and 79,489 RSUs which will vest after the KWE General Meeting (but before the Scheme Record Time) in connection with the Merger.

⁽²⁾ This holding consists of 2,139 KWE Shares held by Kent Y. Mouton and 8,947 RSUs which will vest after the KWE General Meeting (but before the Scheme Record Time) in connection with the Merger.

6.3 Dealings in relevant securities of KWE

(1) The following dealings in relevant securities of KWE by KWE have taken place during the offer period:

Name	Date	Transaction	Number of KWE Shares	Price (£) per KWE Share
KWE	5 May 2017	Acquisition pursuant to IMA ⁽¹⁾	186,194	10.35

⁽¹⁾ The KWE Shares were bought on the market by KWE as agent for the Manager and delivered to the Manager as part payment of the management fee due to the Manager from KWE for the provision of investment management services under the IMA.

(3) The following dealings in relevant securities of KWE by the KW Directors have taken place during the disclosure period:

Name	Date	Transaction	Number of KWE Shares	Price (£) per KWE Share
William J. McMorrow	1 March 2017	Vesting of RSUs	19,006	Nil
Kent Y. Mouton	1 March 2017	Vesting of RSUs	2,139	Nil

(4) The following dealings in relevant securities of KWE by members of the KW Group have taken place during the disclosure period:

Name	Date	Transaction	Number of KWE Shares	Price (£) per KWE Share
Kennedy Wilson UK Limited	8 May 2017	Acquisition pursuant to investment advisory agreement	130,336	Nil ⁽¹⁾
Kennedy Wilson Europe Limited	8 May 2017	Acquisition pursuant to investment advisory agreement	42,825	Nil ⁽¹⁾
Kennedy Wilson España SLU	8 May 2017	Acquisition pursuant to investment advisory agreement	5,586	Nil ⁽¹⁾
Manager	8 May 2017	Disposal pursuant to investment advisory agreement	178,747	Nil ⁽²⁾
Manager	5 May 2017	Acquisition pursuant to IMA	186,194	Nil ⁽³⁾
KW	1 March 2017	Transfer on vesting of RSUs	50,975	Nil
Kennedy Wilson UK Limited	1 March 2017	Transfer on vesting of RSUs	52,497	Nil
Kennedy Wilson Europe Limited	1 March 2017	Transfer on vesting of RSUs	26,040	Nil
Kennedy Wilson España SLU	1 March 2017	Transfer on vesting of RSUs	5,130	Nil
Kennedy Wilson UK Limited	28 February 2017	Acquisition pursuant to investment advisory agreement	136,612	Nil ⁽¹⁾
Kennedy Wilson Europe Limited	28 February 2017	Acquisition pursuant to investment advisory agreement	44,914	Nil ⁽¹⁾

⁽²⁾ No dealings in relevant securities of KWE by the KWE Directors have taken place during the offer period.

Name	Date	Transaction	Number of KWE Shares	Price (£) per KWE Share
Kennedy Wilson España SLU	28 February 2017	Acquisition pursuant to investment advisory agreement	5,614	Nil ⁽¹⁾
Manager	28 February 2017	Disposal pursuant to investment advisory agreement	187,140	Nil ⁽²⁾
Manager	24 February 2017	Acquisition pursuant to IMA	194,937	Nil ⁽³⁾
KW	4 November 2016	Acquisition pursuant to settlement of intragroup liabilities	627,929	Nil
Kennedy Wilson UK Limited	4 November 2016	Disposal pursuant to settlement of intragroup liabilities	627,929	Nil
Kennedy Wilson UK Limited	4 November 2016	Acquisition pursuant to investment advisory agreement	2,703,014	Nil ⁽¹⁾
Kennedy Wilson Europe Limited	4 November 2016	Acquisition pursuant to investment advisory agreement	244,212	Nil ⁽¹⁾
Kennedy Wilson España SLU	4 November 2016	Acquisition pursuant to investment advisory agreement	46,898	Nil ⁽¹⁾
Manager	4 November 2016	Disposal pursuant to investment advisory agreement	2,994,124	Nil ⁽²⁾
Manager	4 November 2016	Acquisition pursuant to IMA	206,460	Nil ⁽³⁾
Manager	5 August 2016	Acquisition pursuant to IMA	212,244	Nil ⁽³⁾
KW	28 June 2016	Open market purchase	500,000	9.20
KW	1 June 2016	Open market purchase	300,000	10.72
KW	19 May 2016	Open market purchase	250,000	10.82
KW	18 May 2016	Open market purchase	250,000	10.77
KW	5 May 2016	Open market purchase	314,373	10.92

Name	Date	Transaction	Number of KWE Shares	Price (£) per KWE Share	
Manager	5 May 2016	Acquisition pursuant to IMA	185,627	Nil ⁽³⁾	

⁽¹⁾ The KWE Shares were acquired from the Manager as part payment of the investment advisory fee due to this entity from the Manager for the provision of investment advisory services under an investment advisory agreement.

(5) The following dealings in relevant securities of KWE by persons acting in concert with KW (excluding the KW Directors and members of the KW Group) have taken place during the disclosure period:

Name	Date range	Transaction	Number of KWE Shares	Price (£) per KWE Share (Low)	Price (£) per KWE Share (High)
Deutsche Bank AG	25 January 2017 – 23 February 2017	Purchase	2,000	9.84	9.84
Deutsche Bank AG	25 January 2017 – 23 February 2017	Sale	2,000	9.84	9.84
Deutsche Bank AG	25 April 2016 – 24 July 2016	Purchase	13,992	10.61	10.80
Deutsche Bank AG	25 April 2016 – 24 July 2016	Sale	13,992	10.61	10.80
BofA Merrill Lynch	24 March 2017 – 23 April 2017	Purchase	286,017	9.39	9.85
BofA Merrill Lynch	24 March 2017 – 23 April 2017	Sale	282,303	9.39	9.84
BofA Merrill Lynch	24 February 2017 – 23 March 2017	Purchase	161,501	9.37	9.83
BofA Merrill Lynch	24 February 2017 – 23 March 2017	Sale	184,135	9.37	9.83
BofA Merrill Lynch	24 January 2017 – 23 February 2017	Purchase	1,619,241	9.40	9.86
BofA Merrill Lynch	24 January 2017 – 23 February 2017	Sale	1,621,106	9.37	9.79
BofA Merrill Lynch	24 December 2016 - 23 January 2017	Purchase	155,909	9.19	9.66
BofA Merrill Lynch	24 December 2016 - 23 January 2017	Sale	154,388	9.19	9.66
BofA Merrill Lynch	24 November 2016 - 23 December 2016	Purchase	445,261	9.51	10.14

⁽²⁾ The KWE Shares were transferred for nil consideration to members of the KW Group as payment of the investment advisory fees due to those members of the KW Group from the Manager for the provision of investment advisory services by those members of the KW Group to the Manager.

⁽³⁾ The KWE Shares were bought on the market by KWE as agent for the Manager and delivered to the Manager as part payment of the management fee due to the Manager from KWE for the provision of investment management services under the IMA.

Name	Date range	Transaction	Number of KWE Shares	Price (£) per KWE Share (Low)	Price (£) per KWE Share (High)
BofA Merrill Lynch	24 November 2016 - 23 December 2016	Sale	373,297	9.51	10.08
BofA Merrill Lynch	24 October 2016 – 23 November 2016	Purchase	480,874	9.99	10.52
BofA Merrill Lynch	24 October 2016 – 23 November 2016	Sale	459,754	9.99	10.43
BofA Merrill Lynch	24 September 2016 - 23 October 2016	Purchase	2,894,144	9.36	10.49
BofA Merrill Lynch	24 September 2016 - 23 October 2016	Sale	2,917,322	9.36	10.49
BofA Merrill Lynch	24 August 2016 – 23 September 2016	Purchase	3,990,408	9.19	9.67
BofA Merrill Lynch	24 August 2016 – 23 September 2016	Sale	3,948,755	9.20	9.67
BofA Merrill Lynch	24 July 2016 – 23 August 2016	Purchase	235,354	9.52	10.20
BofA Merrill Lynch	24 July 2016 – 23 August 2016	Sale	249,887	9.48	10.20
BofA Merrill Lynch	24 June 2016 – 23 July 2016	Purchase	275,457	8.86	10.26
BofA Merrill Lynch	24 June 2016 – 23 July 2016	Sale	275,727	8.81	10.06
BofA Merrill Lynch	24 May 2016 – 23 June 2016	Purchase	347,177	10.08	11.24
BofA Merrill Lynch	24 May 2016 – 23 June 2016	Sale	369,128	10.05	11.18
BofA Merrill Lynch	24 April 2016 – 23 May 2016	Purchase	208,116	10.53	11.10
BofA Merrill Lynch	24 April 2016 – 23 May 2016	Sale	175,467	10.54	11.07
Goldman, Sachs & Co. LLC	24 April 2017 to 30 August 2017	Purchase	7,206	11.49	11.49
Goldman, Sachs & Co. LLC	24 April 2017 to 30 August 2017	Sale	7,206	11.49	11.49

6.4 Interests in relevant securities of KW

(1) As at the close of business on the disclosure date, KWE held no interests in, or rights to subscribe in respect of, relevant securities of KW.

(2) As at the close of business on the disclosure date, the KWE Directors held the following interests in, or rights to subscribe in respect of, relevant securities of KW:

Name	Number of KW Shares	share capital of KW in issue as at the Latest Practicable Date
William J. McMorrow ⁽¹⁾	1,761,451	1.54
Leslie McMorrow (William J. McMorrow's wife)	90,851	0.08
Tyler McMorrow (William J. McMorrow's son)	387,821	0.34
William J. McMorrow and his close relatives (held by William J. McMorrow as trustee of the William J. McMorrow		
Revocable Trust)	11,270,894	9.87
William J. McMorrow and his close relatives (held by William J. McMorrow as trustee of the John & Sons Retirement		
Trust)	8,443	0.01
Mary L. Ricks ⁽²⁾	1,547,692	1.35

⁽¹⁾ Includes 583,334 RSGs. Whilst these RSGs have not yet vested, William J. McMorrow has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽²⁾ Includes 346,667 RSGs. Whilst these RSGs have not yet vested, Mary L. Ricks has the right to exercise the voting rights in respect of the underlying KW Shares.

(3) As at the close of business on the disclosure date, the KW Directors held the following interests in, or rights to subscribe in respect of, in relevant securities of KW:

Name	Number of KW Shares	Percentage of share capital of KW in issue as at the Latest Practicable Date
William J. McMorrow ⁽¹⁾	1,761,451	1.54
Leslie McMorrow (William J. McMorrow's wife)	90,851	0.08
Tyler McMorrow (William J. McMorrow's son)	387,821	0.34
William J. McMorrow and his close relatives (held by William J. McMorrow as trustee of the William J. McMorrow		
Revocable Trust)	11,270,894	9.87
William J. McMorrow and his close relatives (held by William		
J. McMorrow as trustee of the John & Sons Retirement		
Trust)	8,443	0.01
Norman Creighton ⁽²⁾	303,074	0.27
Cathy Hendrickson ⁽³⁾	57,718	0.05
Cathy Hendrickson and James Hendrickson (as trustees of		
Hendrickson Family Trust)	12,476	0.01
David A. Minella ⁽⁴⁾	2,362,032	2.07
Kent Y. Mouton ⁽⁵⁾	441,440	0.39
Jerry R. Solomon ⁽⁶⁾	39,500	0.03
Jerry R. Solomon and Anna Solomon (as trustees of the		
Solomon Family Trust)	75,500	0.07
Stanley R. Zax ⁽⁷⁾	242,300	0.21

⁽¹⁾ Includes 583,334 RSGs. Whilst these RSGs have not yet vested, William J. McMorrow has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽²⁾ Includes 24,000 RSGs. Whilst these RSGs have not yet vested, Norman Creighton has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽³⁾ Includes 24,000 RSGs. Whilst these RSGs have not yet vested, Cathy Hendrickson has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽⁴⁾ Includes 24,000 RSGs. Whilst these RSGs have not yet vested, David A. Minella has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽⁵⁾ Includes 166,668 RSGs. Whilst these RSGs have not yet vested, Kent Y. Mouton has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽⁶⁾ Includes 24,000 RSGs. Whilst these RSGs have not yet vested, Jerry R. Solomon has the right to exercise the voting rights in respect of the underlying KW Shares.

⁽⁷⁾ Includes 24,000 RSGs. Whilst these RSGs have not yet vested, Stanley R. Zax has the right to exercise the voting rights in respect of the underlying KW Shares.

(4) As at the close of business on the disclosure date, the following persons acting in concert with KW (excluding the KW Directors) held the following interests in, or rights to subscribe in respect of, relevant securities of KW:

Name	Nature of interest	Number of KW Shares	Percentage of share capital of KW in issue as at the Latest Practicable Date
Goldman, Sachs & Co. LLC	Holding	209,223	0.18
Goldman, Sachs & Co. LLC	Contract for Difference	48,892	0.04

(5) As at the close of business on the disclosure date, the following persons acting in concert with KW (excluding the KW Directors) held the following short positions (whether conditional or absolute and whether in the money or otherwise) relation to relevant securities of KW under derivatives, agreements to sell, or any delivery obligation or right to require another person to purchase or take delivery:

		N. I. CWW.	Percentage of share capital of KW in issue as at
Name	Nature of interest	Number of KW Shares	the Latest Practicable Date
Goldman, Sachs & Co. LLC	Physically settled short position	404,204	0.35

(6) Set out below are details of any relevant securities of KW which KW, or any person acting in concert with KW, has as at the close of business on the disclosure date, borrowed or lent (including for these purposes any financial collateral arrangements of the kind referred to in Note 4 on Rule 4.6 of the Code), save for any borrowed shares which have either been on-lent or sold:

Name	Number of KW Shares
William J. McMorrow ⁽¹⁾ David A. Minella ⁽¹⁾	1,500,000 1,750,000

⁽¹⁾ Pledged under a line of credit. Title and voting rights have not been transferred.

6.5 Dealings in relevant securities of KW

- (1) No dealings in relevant securities of KW by KWE have taken place during the offer period.
- (2) No dealings in relevant securities of KW by the KWE Directors have taken place during the offer period.

(3) The following dealings in relevant securities of KW by members of the KW Group have taken place during the disclosure period:

Name	Date	Transaction	Number of KW Shares	Price (US\$) per KW Share
KW	24 January 2017	On-market buyback	2,900	20.00
KW	23 January 2017	On-market buyback	5,224	20.00
KW	20 January 2017	On-market buyback	5,202	19.98
KW	19 January 2017	On-market buyback	63,829	20.00
KW	22 December 2016	On-market buyback	86,975	21.18
KW	21 December 2016	On-market buyback	82,861	21.45
KW	20 December 2016	On-market buyback	36,764	21.35
KW	19 December 2016	On-market buyback	60,274	21.21
KW	15 December 2016	On-market buyback	6,324	21.00
KW	14 December 2016	On-market buyback	15,838	21.00
KW	13 December 2016	On-market buyback	1,915	21.00
KW	7 December 2016	On-market buyback	50,000	20.70
KW	2 December 2016	On-market buyback	50,000	20.62
KW	1 December 2016	On-market buyback	50,000	20.65
KW	30 November 2016	On-market buyback	100,000	21.47
KW	14 November 2016	On-market buyback	35,000	22.49
KW	14 November 2016	On-market buyback	98,349	22.55
KW	11 November 2016	On-market buyback	91,500	22.52
KW	10 November 2016	On-market buyback	91,500	22.58
KW	9 November 2016	On-market buyback	37,919	21.96
KW	8 November 2016	On-market buyback	91,500	22.13
KW	6 July 2016	On-market buyback	2,431	17.99
KW	5 July 2016	On-market buyback	30,154	18.00
KW	29 June 2016	On-market buyback	24,970	17.97
KW	28 June 2016	On-market buyback	160,519	17.67
KW	27 June 2016	On-market buyback	159,690	17.60
KW	13 June 2016	On-market buyback	25,000	19.28
KW	10 June 2016	On-market buyback	50,000	20.16
KW	9 June 2016	On-market buyback	85,000	20.87
KW	23 May 2016	On-market buyback	16,922	20.57
KW	20 May 2016	On-market buyback	25,000	20.46
KW	19 May 2016	On-market buyback	50,000	20.04
KW	18 May 2016	On-market buyback	67,500	20.12
KW	17 May 2016	On-market buyback	75,000	20.04
KW	16 May 2016	On-market buyback	50,000	20.15
KW	13 May 2016	On-market buyback	90,000	20.07
KW	12 May 2016	On-market buyback	90,200	20.22
KW	11 May 2016	On-market buyback	80,000	20.60
KW	10 May 2016	On-market buyback	19,242	20.44
KW	9 May 2016	On-market buyback	68,035	20.39
KW	6 May 2016	On-market buyback	43,741	20.00
KW	5 May 2016	On-market buyback	433	20.00

(4) The following dealings in relevant securities of KW by KW Directors have taken place during the disclosure period:

Name	Date	Transaction	Number of KW Shares	Price (US\$) per KW Share
William J. McMorrow	15 April 2017	Vesting of RSGs	71,730	21.70
William J. McMorrow	15 March 2017	Vesting of RSGs	101,617	21.50
William J. McMorrow	26 January 2017	Vesting of RSGs	229,536	20.65
William J. McMorrow	31 August 2016	Grant of RSGs	212,500	$N/A^{(1)}$
William J. McMorrow	14 June 2016	Acquisition under a 10b5-1	10,000	18.78 – 18.87
William J. McMorrow (as trustee of the William J. McMorrow Revocable Trust)	12 December 2016	programme Gift	100,000	Nil
William J. McMorrow (as trustee of the William J. McMorrow Revocable Trust)	29 November 2016	Gift	450,000	Nil
William J. McMorrow (as trustee of the William J. McMorrow Revocable Trust)	14 June 2016	Acquisition under a 10b5-1 programme	10,000	18.78 – 18.87
William J. McMorrow (as trustee of the William J. McMorrow Revocable	18 May 2016	Acquisition under a 10b5-1 programme	25,000	19.87 – 20.27
Trust)	15 A:1 2017	Vesting of DCCs	5,000	21.70
Norman Creighton Norman Creighton	15 April 2017 15 March 2017	Vesting of RSGs Vesting of RSGs	5,000 5,500	21.70
Norman Creighton	26 January 2017	Vesting of RSGs Vesting of RSGs	5,000	20.65
Norman Creighton	2 November 2016	Grant of RSGs	7,500	N/A ⁽¹⁾
Cathy Hendrickson	15 April 2017	Vesting of RSGs	5,000	21.70
Cathy Hendrickson	15 April 2017 15 March 2017	Vesting of RSGs Vesting of RSGs	5,500	21.70
Cathy Hendrickson	26 January 2017	Vesting of RSGs Vesting of RSGs	5,000	20.65
Cathy Hendrickson	2 November 2016	Grant of RSGs	7,500	N/A ⁽¹⁾
David A. Minella	15 April 2017	Vesting of RSGs	5,000	21.70
David A. Minella	15 March 2017	Vesting of RSGs	5,500	21.50
David A. Minella	26 January 2017	Vesting of RSGs	5,000	20.65
David A. Minella	2 November 2016	Grant of RSGs	7,500	$N/A^{(1)}$
Kent Y. Mouton	15 April 2017	Vesting of RSGs	19,128	21.70
Kent Y. Mouton	15 March 2017	Vesting of RSGs	31,083	21.50
Kent Y. Mouton	26 January 2017	Vesting of RSGs	57,384	20.65
Kent Y. Mouton	31 August 2016	Grant of RSGs	65,000	$N/A^{(1)}$
Jerry R. Solomon	15 April 2017	Vesting of RSGs	5,000	21.70
Jerry R. Solomon	15 March 2017	Vesting of RSGs	5,500	21.50
Jerry R. Solomon	26 January 2017	Vesting of RSGs	5,000	20.65
Jerry R. Solomon	13 December 2016	Gift	2,500	Nil
Jerry R. Solomon	2 November 2016	Grant of RSGs	7,500	$N/A^{(2)}$
Stanley R. Zax	15 April 2017	Vesting of RSGs	5,000	21.70
Stanley R. Zax	15 March 2017	Vesting of RSGs	5,500	21.50
Stanley R. Zax	26 January 2017	Vesting of RSGs	5,000	20.65
Stanley R. Zax	2 November 2016	Grant of RSGs	7,500	N/A ⁽¹⁾
Stanley R. Zax	14 June 2016	Acquisition	50,000	18.62 - 18.85

⁽¹⁾ These RSGs vest over a three year period, but entitle their holder to exercise the voting rights attached to the underlying KW Shares prior to vesting. The price per KW Share is calculated by reference to the closing price of KW Shares on the relevant vesting date.

(5) The following dealings in relevant securities of KW by persons acting in concert with KW (excluding the KW Directors and members of the KW Group) have taken place during the disclosure period:

KW Shares

N	Determine	Townstian	Number of	Price (US\$) per KW	Price (US\$) per KW Share
Name	Date range	Transaction	KW Shares	Share (Low)	(High)
BofA Merrill Lynch	24 March 2017 – 23 April 2017	Purchase	161,802	20.96	22.65
BofA Merrill Lynch	24 March 2017 – 23 April 2017	Sale	134,175	20.90	22.63
BofA Merrill Lynch	24 February 2017 – 23 March 2017	Purchase	361,658	20.75	22.55
BofA Merrill Lynch	24 February 2017 – 23 March 2017	Sale	432,094	20.75	22.60
BofA Merrill Lynch	24 January 2017 – 23 February 2017	Purchase	811,350	20.07	21.45
BofA Merrill Lynch	24 January 2017 – 23 February 2017	Sale	805,167	20.03	21.45
BofA Merrill Lynch	24 December 2016 - 23 January 2017	Purchase	183,043	20.00	21.15
BofA Merrill Lynch	24 December 2016 - 23 January 2017	Sale	249,335	20.00	21.08
BofA Merrill Lynch	24 November 2016 - 23 December 2016	Purchase	424,387	20.42	22.85
BofA Merrill Lynch	24 November 2016 - 23 December 2016	Sale	466,925	20.40	22.85
BofA Merrill Lynch	24 October 2016 – 23 November 2016	Purchase	209,863	20.44	22.95
BofA Merrill Lynch	24 October 2016 – 23 November 2016	Sale	172,353	20.49	22.90
BofA Merrill Lynch	24 September 2016 - 23 October 2016	Purchase	286,178	20.37	23.05
BofA Merrill Lynch	24 September 2016 - 23 October 2016	Sale	409,548	20.31	23.05
BofA Merrill Lynch	24 August 2016 – 23 September 2016	Purchase	450,471	21.27	23.50
BofA Merrill Lynch	24 August 2016 – 23 September 2016	Sale	1,018,188	21.27	23.71
BofA Merrill Lynch	24 July 2016 – 23 August 2016	Purchase	148,572	20.51	21.76
BofA Merrill Lynch	24 July 2016 – 23 August 2016	Sale	124,295	20.47	21.76
BofA Merrill Lynch	24 June 2016 – 23 July 2016	Purchase	517,028	17.39	20.83
BofA Merrill Lynch	24 June 2016 – 23 July 2016	Sale	440,470	17.38	20.85
BofA Merrill Lynch	24 May 2016 – 23 June 2016	Purchase	306,165	18.50	21.39
BofA Merrill Lynch	24 May 2016 – 23 June 2016	Sale	294,538	18.50	21.39
BofA Merrill Lynch	24 April 2016 – 23 May 2016	Purchase	201,001	19.86	22.33
BofA Merrill Lynch	24 April 2016 – 23 May 2016	Sale	83,397	19.83	22.33

Name	Date range	Transaction	Number of KW Shares	Price (US\$) per KW Share (Low)	Price (US\$) per KW Share (High)
Goldman, Sachs & Co. LLC	24 April 2017 – 7 September 2017	Purchase	2,159,319	18.00	22.15
Goldman, Sachs & Co. LLC	24 April 2017 – 7 September 2017	Sale	2,259,883	18.00	21.73
Goldman, Sachs & Co. LLC	24 March 2017 – 23 April 2017	Purchase	651,869	20.90	22.625
Goldman, Sachs & Co. LLC	24 March 2017 – 23 April 2017	Sale	599,424	20.90	22.5373
Goldman, Sachs & Co. LLC	24 February 2017 – 23 March 2017	Purchase	361,506	20.75	22.50
Goldman, Sachs & Co. LLC	24 February 2017 – 23 March 2017	Sale	385,178	20.825	22.50
Goldman, Sachs & Co. LLC	24 January 2017 – 23 February 2017	Purchase	340,553	20.25	22.352
Goldman, Sachs & Co. LLC	24 January 2017 – 23 February 2017	Sale	344,556	20.25	22.15
Goldman, Sachs & Co. LLC	24 October 2016 – 23 January 2017	Purchase	2,125,781	19.95	22.90
Goldman, Sachs & Co. LLC	24 October 2016 – 23 January 2017	Sale	1,853,771	19.95	22.825
Goldman, Sachs & Co. LLC	24 July 2016 – 23 October 2016	Purchase	1,763,994	20.38	23.63
Goldman, Sachs & Co. LLC	24 July 2016 – 23 October 2016	Sale	1,612,767	20.38	23.73

Name	Date range	Transaction	Number of KW Shares	Price (US\$) per KW Share (Low)	Price (US\$) per KW Share (High)
Goldman, Sachs &	24 April 2017 –	Purchase	85,934	18.05	21.725
Co. LLC	7 September 2017				
Goldman, Sachs &	24 April 2017 –	Sale	41,285	18.05	20.9216
Co. LLC	7 September 2017				
Goldman, Sachs &	24 March 2017 –	Purchase	1,300	22.0524	22.0593
Co. LLC	23 April 2017				
Goldman, Sachs &	24 March 2017 –	Sale	4,700	21.9245	21.9245
Co. LLC	23 April 2017				
Goldman, Sachs &	24 July 2016 –	Purchase	20,000	20.0334	22.82
Co. LLC	23 October 2016				
Goldman, Sachs &	24 July 2016 –	Sale	22,500	20.5811	20.6339
Co. LLC	23 October 2016	_			
Goldman, Sachs &	24 April 2016 –	Purchase	43,800	17.6181	21.977
Co. LLC	23 July 2016	~ .		40.4400	21 1001
Goldman, Sachs &	24 April 2016 –	Sale	52,116	19.4120	21.4084
Co. LLC	23 July 2016				

Call Options

Name	Date range	Transaction	Number of KW Shares	price (US\$) per Call Option (Low)	price (US\$) per Call Option (High)	price (US\$) per Call Option (Low)	price (US\$) per Call Option (High)	Expiry date (Low)	Expiry date (High)
BofA Merrill	24 March 2017 -	Sale	1,500	22.50	22.50	0.30	0.30	19 May	19 May
Lynch	23 April 2017							2017	2017
BofA Merrill Lynch	24 February 2017 – 23 March 2017	Sale	800	22.50	22.50	0.65	0.65	16 June 2017	16 June 2017
BofA Merrill Lynch	24 January 2017 – 23 February 2017	Sale	100	22.50	22.50	0.50	0.50	16 June 2017	16 June 2017
BofA Merrill Lynch	24 October 2016 – 23 January 2017	Purchase	2,600	22.50	22.50	0.45	1.35	17 March 2017	17 March 2017
BofA Merrill Lynch	24 October 2016 – 23 January 2017	Sale	4,000	20.00	22.50	0.45	2.80	20 January 2017	17 March 2017
BofA Merrill Lynch	24 July 2016 – 23 October 2016	Sale	4,100	20.00	22.50	1.50	3.00	16 December 2016	17 March 2017
BofA Merrill Lynch	24 April 2016 – 23 July 2016	Purchase	2,400	22.50	22.50	0.20	0.77	16 September 2016	16 December 2016
BofA Merrill Lynch	24 April 2016 – 23 July 2016	Sale	2,500	20.00	22.50	0.74	0.85	16 December 2016	16 December 2016

Put Options

Name	Date range	Transaction	Number of KW Shares	price (US\$) per Call Option (Low)	price (US\$) per Call Option (High)	price (US\$) per Call Option (Low)	price (US\$) per Call Option (High)	Expiry date (Low)	Expiry date (High)
BofA Merrill	24 March 2017 -	Purchase	5,900	20.00	20.00	0.20	0.20	16 June	16 June
Lynch	23 April 2017							2017	2017
BofA Merrill	24 February 2017 -	Purchase	600	20.00	20.00	0.35	1.00	16 June	15 September
Lynch	23 March 2017							2017	2017
BofA Merrill	24 January 2017 -	Sale	1,000	20.00	20.00	0.40	0.40	17 February	17 February
Lynch	23 February 2017							2017	2017
BofA Merrill	24 October 2016 -	Sale	10,400	17.50	20.00	0.40	1.55	20 January	20 January
Lynch	23 January 2017							2017	2017
BofA Merrill	24 July 2016 -	Purchase	1,800	17.50	20.00	0.40	0.70	16 December	17 March
Lynch	23 October 2016							2016	2017
BofA Merrill	24 July 2016 -	Sale	500	17.50	17.50	0.40	0.40	17 March	17 March
Lynch	23 October 2016							2017	2017
BofA Merrill	24 April 2016 -	Sale	1,200	20.00	20.00	0.45	0.50	19 August	19 August
Lynch	23 July 2016							2016	2016

6.6 General

- (1) Save as disclosed in this document, as at the close of business on the disclosure date:
 - (a) no (i) member of the KWE Group, (ii) KWE Director, or (iii) person deemed to be acting in concert with KWE had any interest in, right to subscribe in respect of, short position in relation to, agreement to sell, delivery obligation or right to require

- another person to purchase or take delivery of, relevant securities of KW or relevant securities of KWE, nor has any such person dealt in any relevant securities of KWE or relevant securities of KW during the offer period;
- (b) no person who has an arrangement of the kind referred to in Note 11 on the definition of "acting in concert" set out in the Code with KWE, or any person acting in concert with KWE, had any interest in, right to subscribe in respect of, short position in relation to, agreement to sell, delivery obligation or right to require another person to purchase or take delivery of, relevant securities of KW or relevant securities of KWE, nor has any such person dealt in any relevant securities of KWE or relevant securities of KW during the offer period; and
- (c) neither KWE, nor any person acting in concert with KWE, has borrowed or lent any relevant securities of KWE or relevant securities of KW, save for any borrowed shares which have been either on-lent or sold.
- (2) Save as disclosed in this document, as at the disclosure date:
 - (a) no (i) member of the KW Group, (ii) KW Director, or (iii) person deemed to be acting in concert with KW had any interest in, right to subscribe in respect of, short position in relation to, agreement to sell, delivery obligation or right to require another person to purchase or take delivery of, relevant securities of KW or relevant securities of KWE, nor has any such person dealt in any relevant securities of KWE or relevant securities of KW during the disclosure period;
 - (b) no person who has an arrangement of the kind referred to in Note 11 on the definition of "acting in concert" set out in the Code with KW, or any person acting in concert with KW, had any interest in, right to subscribe in respect of, short position in relation to, agreement to sell, delivery obligation or right to require another person to purchase or take delivery of, relevant securities of KW or relevant securities of KWE, nor has any such person dealt in any relevant securities of KWE or relevant securities of KW during the disclosure period; and
 - (c) neither KW, nor any person acting in concert with KW, has borrowed or lent any relevant securities of KWE or relevant securities of KW, save for any borrowed shares which have been either on-lent or sold.
- (3) Save as disclosed in this document (including in paragraphs 4 and 5 of this Part Eight), no persons have given any irrevocable or other commitment to KW or KWE to vote in favour of the Scheme or the Special Resolution to be proposed at the KWE General Meeting.
- (4) Save as disclosed in this document, none of (i) KW or any person acting in concert with KW, or (ii) KWE or any person acting in concert with KWE has any arrangement of the kind referred to in Note 11 on the definition of "acting in concert" in the Code in relation to relevant securities of KWE or relevant securities of KW.

7. Major interests in KW

In so far as is known to KW, the following persons have (directly or indirectly) an interest of in excess of 5% in the share capital of KW on the basis set out in the table below as at the Latest Practicable Date. Accordingly, each of the persons named in the table below has (directly or indirectly) a potential interest in excess of 5% in the share capital of KWE (following the Effective Date):

Name	Number of KW Shares	Percentage of existing KW share capital	Estimated percentage of KW share capital following the Effective Date if all Scheme Shareholders were to receive the New Offer Consideration and Distribution ¹	Estimated percentage of KW share capital following the Effective Date if all Scheme Shareholders were to receive the Original Offer Consideration
Wellington Management Group				
LLP ²	12,717,888	11.13%	8.40%	7.12%
William J. McMorrow ³	13,519,460	11.84%	8.93%	7.57%
Fairfax Financial Holdings				
Limited and affiliates ⁴	12,321,921	10.79%	8.14%	6.90%
The Vanguard Group ⁵	7,750,075	6.79%	5.12%	4.34%

¹ Assumes no Mix and Match Election is made.

8. Significant change

8.1 <u>KWE</u>

The KWE Independent Directors are not aware of any significant change in the financial or trading position of KWE which has occurred since 30 June 2017, being the date to which the latest unaudited condensed consolidated financial statements of KWE were prepared.

Without prejudice to the preceding paragraph, in accordance with Rule 29.3 of the Code, the KWE Independent Directors are not aware of any significant change which has occurred in the deferred tax liability of KWE relating to investment properties, which amounted to £5.5 million as at 30 June 2017, as reported in KWE's unaudited condensed consolidated balance sheet as at 30 June 2017 incorporated by reference into this document.

8.2 KW

The KW Directors are not aware of any significant change in the financial or trading position of KW which has occurred since 30 June 2017, being the date to which the latest unaudited condensed consolidated financial statements of KW were prepared.

9. Effect of completion of the Merger

The KW Board believes that the Merger will create a leading real estate investment and asset management platform with increased scale and liquidity and provide flexibility to allocate capital globally across asset classes and geographic markets.

Wellington Management Group LLP serves as an investment adviser and strategic partner to investors worldwide. It is engaged in investment management and offers capabilities that span nearly all segments of the global capital markets.

³ Includes 90,851 KW Shares beneficially owned by Leslie McMorrow, Mr McMorrow's wife, and 387,821 KW Shares beneficially owned by Tyler McMorrow, Mr McMorrow's son. Mr McMorrow disclaims beneficial ownership of the shares owned by his wife and son. It also includes 1,500,000 pledged KW Shares.

⁴ Fairfax Financial Holdings Limited is a holding company which, through its subsidiaries, is engaged in property and casualty insurance and reinsurance and investment management. Fairfax's corporate objective is to achieve a high rate of return on invested capital and build long-term shareholder value.

⁵ The Vanguard Group is an American investment management company with over US\$4 trillion in assets under management. It is one of the largest provider of mutual funds and exchange-traded funds in the world.

The KW Board further believes that the Merger will be accretive to adjusted net income per share immediately following the completion of the Merger and provide the potential to enhance cash flows available to distribute to shareholders.³

As at 30 June 2017, KW had total assets of US\$8,284 million, net debt of US\$3,166 million and total KW Shareholders' equity of US\$1,028 million.

Under the Original Offer, assuming completion of the Merger occurred on 30 June 2017, the Combined Group would have total assets of US\$7,884 million (consolidated), net debt of US\$4,440 million (at share) and total KW Shareholders' equity of US\$2,110 million as at that date.

Under the New Offer, assuming completion of the Merger and payment of the KWE Special Distribution by KWE occurred on 30 June 2017, the Combined Group would have total assets of US\$7,551 million (consolidated), net debt of US\$5,123 million (at share) and total KW Shareholders' equity of US\$1,421 million as at that date.

10. Ratings

10.1 KWE

Prior to the commencement of the Offer Period, KWE's long-term debt was assigned a credit rating of "BBB (Stable)" by S&P. It is possible that the credit rating assigned by S&P may be downgraded as a result of the implementation of the Scheme due to the limited reduction of KWE's net assets from approximately £1,567.9 million (as at 30 June 2017) to approximately £1,326.3 million (as at 30 June 2017) resulting from the KWE Special Distribution and/or to the manner in which S&P assumes KW may cause KWE to deal with its assets after the Scheme is put into effect. On 27 April 2017, S&P placed KWE on negative outlook due to its view that a takeover by KW might result in a downgrading of KWE's credit risk profile from "BBB (Stable)" to "BB-" i.e. to S&P's significantly lower existing rating for KW. The negative outlook given on 27 April 2017 has not changed since then.

10.2 KW

Prior to the commencement of the Offer Period, KW's long-term credit ratings were as follows: S&P had a long-term credit rating of "BB- (Positive)" and Moody's had a credit rating of "B2 (Stable)". On 25 April 2017, S&P upgraded KW's outlook from Positive to CreditWatch Positive due to S&P's view that the addition of KWE's stable portfolio of commercial real estate should increase KW's diversification, boost recurring revenue, and reduce the volatility of earnings. The rating given by S&P on 25 April 2017 has not changed since. On 16 June 2017, Moody's upgraded KW's outlook from Stable to Positive due to Moody's view that the Merger will provide KW with a meaningfully larger asset base, reduced geographical concentration and enhanced product mix. The rating given by Moody's on 16 June 2017 has not changed since.

11. Middle market quotations

The following table shows the Closing Price of KWE Shares on:

- (A) the first Business Day of each of the six months immediately prior to the date of this document:
- (B) 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- (C) 12 June 2017 (the last Business Day before the New Offer Announcement); and

³ This statement is not intended as a profit forecast or estimate for any period and should not be interpreted to mean that adjusted earnings per share for KW, KWE or the Combined Group, as appropriate, for the current or future financial years would necessarily match or exceed the historical published adjusted net income per share for KW, KWE or the Combined Group, as appropriate.

(D) the Latest Practicable Date:

Date	KWE Share Closing Price (pence)
1 March 2017	962
3 April 2017	947
21 April 2017	979
2 May 2017	1,045
1 June 2017	1,040
12 June 2017	1,030
3 July 2017	1,105
1 August 2017	1,140
1 September 2017	1,114
8 September 2017	1,114

The following table shows the Closing Price of KW Shares on:

- (A) the first Business Day of each of the six months immediately prior to the date of this document;
- (B) 21 April 2017 (the last Business Day before the commencement of the Offer Period);
- (C) 12 June 2017 (the last Business Day before the New Offer Announcement); and
- (D) the Latest Practicable Date:

Date	KW Share Closing Price (US\$)
1 March 2017	22.40
3 April 2017	21.95
21 April 2017	22.50
1 May 2017	20.20
1 June 2017	19.40
12 June 2017	19.55
3 July 2017	18.80
1 August 2017	20.00
1 September 2017	19.20
8 September 2017	19.15

12. Letters of Appointment of the KWE Directors

The particulars of the letters of appointment that have been entered into between KWE and the KWE Directors are as follows. Save as disclosed, no such letter has been entered into or amended during the six months preceding publication of this document:

Director	Effective Date of Appointment	Expiration of Appointment	Employment Notice Period (from KWE)	Employment Notice Period (from individual)	Fee (£)
Charlotte Valeur	10 February 2014	10 February 2020	3 months	3 months	145,000
Mark McNicholas	10 February 2014	10 February 2020	3 months	3 months	100,000
Simon Radford	10 February 2014	10 February 2020	3 months	3 months	100,000
Mary L. Ricks ¹	10 February 2014	10 February 2020	3 months	3 months	_
William J. McMorrow ¹	10 February 2014	10 February 2020	3 months	3 months	_

¹ Mary L. Ricks and William J. McMorrow have waived their right to receive any fees in their capacity as directors on the KWE Board

Pursuant to the terms of separate agreements entered into with the consent of KW, between KWE and each of Charlotte Valeur, Mark McNicholas and Simon Radford dated 23 April 2017, KWE has agreed to pay each KWE Independent Director extra remuneration for any services rendered in his/her capacity as a KWE Director in relation to the Merger. Each KWE Independent Director will be paid a daily fee of £3,750.00 plus reasonable out of pocket expenses incurred for each day where he/she spends two hours or more performing work in his/her capacity as a KWE Director in relation to the Merger. Each KWE Independent Director will be entitled to the additional fee for the period from and including 1 March 2017 up to and including the first to occur of (1) the date of completion of the Merger or (2) the date on which the Merger lapses or is withdrawn. Each of William J. McMorrow and Mary L. Ricks has waived his/her right to receive any fees in his/her capacity as a KWE Director including, without limitation, in relation to the Merger.

As at the Latest Practicable Date, the following amounts have been paid or accrued under the terms of these letters:

Director	Fee (£)
Charlotte Valeur	406,325
Mark McNicholas	180,311
Simon Radford	150,000
Mary L. Ricks	_
William J. McMorrow	_

13. Material contracts

13.1 KWE

The following contracts have been entered into by members of the KWE Group otherwise than in the ordinary course of business in the two years prior to the commencement of the Offer Period and are or may be material:

(a) Confidentiality Agreement

The Confidentiality Agreement is described in paragraph 30(a) of Part Two (Explanatory Statement) of this document.

(b) Separate Manager's Powers Agreement

The Separate Manager's Powers Agreement is described in paragraph 30(b) of Part Two (Explanatory Statement) of this document.

(c) Transaction Agreement

The Transaction Agreement is described in paragraph 30(c) of Part Two (Explanatory Statement) of this document.

(d) Indemnity Agreement and KWE Independent Directors' Additional Remuneration Agreement

The Indemnity Agreement and the KWE Independent Directors' Additional Remuneration Agreements are described in paragraph 30(d) of Part Two (Explanatory Statement) of this document.

(e) Investment Management Agreement

The Manager, which is a wholly owned subsidiary of KW, serves as KWE's investment manager pursuant to the Investment Management Agreement, the Manager is responsible for acquiring and disposing of investments and managing all of the assets and investments of KWE and its intermediate holding or investing entities. The Manager is paid a management fee, equal to 1% of KWE's Adjusted NAV and payable quarterly in arrear. In the normal course, half of the management fee is settled in cash and the other half in KWE Shares (either by KWE issuing new KWE Shares to the Manager or by making market purchases of existing KWE Shares for delivery to the Manager). The management fee in respect of the first quarter of 2017 was £3.9 million and was paid on 8 May 2017, half of which was settled in KWE Shares purchased on the open market at market prices. The management fee in respect of the second quarter of 2017 was £3.9 million and was paid on 5 August 2017 entirely in cash. The management fee in respect of the third quarter of 2017 will be calculated and paid in accordance with the terms of the IMA, and if the management fee is paid before the Effective Date, the amount and form of payment of the management fee will be publicly announced.

The Investment Management Agreement also provides for the payment of a performance fee to the Manager, subject to certain thresholds being met. The performance fee is equal to 20% of the lesser of (i) the excess of the "shareholder return" for the relevant year (generally, the sum of the change in KWE's Adjusted NAV per ordinary share) over a 10% annual return hurdle and (ii) the excess of year-end Adjusted NAV per ordinary share over the relevant high water mark. The performance fee (if triggered) is payable annually in KWE Shares and such shares are subject to sale or transfer restrictions over a three-year period (with one-third of such KWE Shares being released each year from such sale restrictions).

(f) The 2022 Bonds

On 30 June 2015, KWE issued £300 million in principal amount of its 3.95% seven year unsecured bonds (the "2022 Bonds"). On 19 September 2016, KWE completed an additional issuance of the 2022 Bonds on the same terms and conditions for an aggregate principal amount of £200 million. This additional issuance was aggregated with the £300 million bonds issued by KWE on 30 June 2015, which together form the 2022 Bonds.

Following the issuance of the 2022 Bonds in June 2015, the KWE Group entered into a cross currency swap to convert a portion of the proceeds of that issuance into Euro. This transaction swapped £150 million of the 2022 Bonds into €210.8 million Euro-equivalent debt.

The 2022 Bonds will mature on 30 June 2022. The 2022 Bonds were rated "BBB" (outlook stable) by Standard & Poor's. Interest on the 2022 Bonds is payable annually on the anniversary of drawdown.

The 2022 Bonds are subject to the following covenants:

- Consolidated net indebtedness does not exceed 60% of the total asset value;
- Consolidated secured indebtedness (less cash and cash equivalents) does not exceed 50% of the total asset value;
- Interest coverage ratio is at least 1.5 to 1.0; and
- Unencumbered assets are not less than 125% of the unsecured indebtedness (less cash and cash equivalents).

(g) The 2025 Notes

On 5 November 2015, the KWE Group established a £2 billion multi-currency Euro Medium Term Note Programme (the "EMTN Programme"). Under the EMTN Programme, the KWE Group may issue, from time to time, up to £2 billion of various types of debt securities in certain markets and currencies. The EMTN Programme was rated "BBB" by Standard & Poor's.

In November 2015, KWE issued €400 million in principal amount of its 3.25% ten year unsecured notes under the EMTN Programme (the "2025 Notes"). On 19 April 2016, KWE completed an additional issuance of the 2025 Notes on the same terms and conditions for an aggregate principal amount of €150 million. This additional issuance was aggregated with the €400 million bonds issued on 12 November 2015, which together form the 2025 Notes. Therefore KWE currently has outstanding €550 million of 2025 Notes under the EMTN Programme that will mature on 12 November 2025. The 2025 Notes were rated "BBB" (outlook stable) by Standard & Poor's. Interest on the 2025 Notes is payable annually on the anniversary of drawdown.

The financial covenants applicable to the 2025 Notes are the same as the financial covenants applicable to the 2022 Bonds.

(h) The RCF

On 29 August 2014 KWE entered into a revolving credit facility agreement (as amended on 31 October 2014, 23 June 2015 and 6 July 2017) (the "RCF") with Bank of America Merrill Lynch International Limited as agent pursuant to which the lenders, which include Deutsche Bank AG, London Branch and JP Morgan Chase Bank, N.A. (London Branch), under the agreement have made available to KWE a revolving credit facility of up to £225 million (with an option for KWE to request that the lenders agree to increase the total commitments up to a maximum of £400 million) for general corporate purposes, including refinancing of existing indebtedness, working capital, payment of capital expenses and acquisitions of certain property and loan assets.

The original maturity date of the KWE Facility was 29 August 2017. However, on 6 July 2017, the RCF was extended to the earlier of 28 February 2018 and the Effective Date. In the event that the Scheme has not become effective by 28 February 2018, KWE will have the right, subject to certain customary conditions, to extend the maturity of the RCF for up to a further six months.

Drawdowns under the RCF are available in Sterling, Euro and, subject to certain conditions, other currencies. The RCF provides for an interest rate comprising LIBOR, or in relation to any loan in Euro, EURIBOR plus an applicable margin. Interest periods are, at the election of KWE, one, three or six months (or any other period as KWE and the agent, on instructions from the lenders, under the RCF may agree) and interest is payable on the last day of each interest period. If the interest period is longer than three months, the interest rate is payable at three month intervals.

Under the RCF, KWE has certain cancellation and prepayment rights. The RCF also contains certain mandatory prepayment and cancellation events, including a change of control provision. A change of control includes (i) the Manager or its affiliates, or any approved replacement investment manager, ceasing to act as sole investment manager of KWE, (ii) any persons other than the Manager (or its affiliates, or any approved replacement investment manager) gaining control of KWE, (iii) KW or its affiliates ceasing to control the Manager or (iv) the sale of the whole or substantially the whole of the business and assets of the KWE Group. The definition of the term "control" is broadly similar to the relevant Loan Market Association definition and it relates to control of either KWE or the Manager. The RCF contains various customary representations, warranties, undertakings and covenants given by KWE. The RCF is also subject to the following financial covenants:

- Consolidated net indebtedness does not exceed 60% of the total asset value;
- Consolidated earnings before interest, tax, depreciation and amortisation to consolidated fixed charges is not less than 1.9 to 1.0 for the last four quarters;
- Consolidated secured recourse debt does not exceed 2.5% of total asset value at any time;
 and

• Unencumbered assets are not less than 125% of unsecured indebtedness (less cash and cash equivalents).

At 31 December 2016 and the Latest Practicable Date the RCF was undrawn. The KWE Group incurs non-utilisation fees on the undrawn portion of the RCF. At 31 December 2016, the commitment fee on the undrawn facility was 55 basis points.

The RCF is governed by English law.

13.2 KW

The following contracts have been entered into by members of the KW Group otherwise than in the ordinary course of business in the two years prior to the commencement of the Offer Period and are or may be material:

(a) Confidentiality Agreement

The Confidentiality Agreement is described in paragraph 30(a) of Part Two (Explanatory Statement) of this document.

(b) Transaction Agreement

The Transaction Agreement is described in paragraph 30(c) of Part Two (Explanatory Statement) of this document.

(c) Indemnity Agreement

The Indemnity Agreement is described in paragraph 30(d) of Part Two (Explanatory Statement) of this document.

(d) The KW Revolving Facility

On 10 December 2015, Kennedy-Wilson, Inc., a wholly owned subsidiary of KW, entered into a US\$475 million revolving credit facility with a syndicate of lenders including JPMorgan Chase Bank, N.A., Deutsche Bank AG New York Branch, US Bank N.A., East West Bank, Fifth Third Bank, The Governor and Company of the Bank of Ireland, Compass Bank, City National Bank, Bank of America, N.A. and Merrill Lynch, Pierce, Fenner & Smith Incorporated (the "KW Revolving Facility").

Loans under the KW Revolving Facility bear interest at a rate equal to LIBOR plus 2.50% or 3.00%, depending on the consolidated leverage ratio as at the applicable measurement date, and have a maturity date of 10 December 2018. Subject to certain conditions precedent and at Kennedy-Wilson, Inc.'s option, the maturity date of the KW Revolving Facility may be extended by one year.

The KW Revolving Facility contains certain covenants that, among other things, limit KW's and certain of its subsidiaries' ability to incur additional indebtedness, repurchase capital stock or debt, sell assets or subsidiary stock, create or permit liens on assets, engage in transactions with affiliates, enter into sale/leaseback transactions, issue subsidiary equity and enter into consolidations or mergers. The credit agreement that governs the KW Revolving Facility requires KW to maintain (i) a maximum consolidated leverage ratio (as defined in the credit agreement) of not greater than 65%, measured as at the last day of each financial quarter, (ii) a minimum fixed charge coverage ratio (as defined in the credit agreement) of not less than 1.60 to 1.00, measured as at the last day of each financial quarter for the period of four full financial quarters then ended, (iii) a minimum consolidated tangible net worth equal to or greater than the sum of US\$920,660,504.65 plus an amount equal to 50% of net equity proceeds received by KW after the date of the 30 September 2015 financial statements, measured as at the last day of each financial quarter, (iv) a maximum recourse leverage ratio (as defined in the credit agreement) of not greater than an amount equal to consolidated tangible net worth as at the measurement date multiplied by 1.5, measured as at the last day of each financial quarter, (v) a maximum secured recourse leverage ratio (as defined in the credit agreement) of not greater than an amount equal to 3.5% of consolidated total asset value (as defined in the credit agreement) and US\$138.187.197. (vi) a maximum adjusted secured leverage ratio (as defined in the credit agreement) of not greater than 55%, measured as at the last day of each financial quarter and (vii) liquidity (as defined in the credit agreement) of at least US\$250 million.

The obligations of Kennedy-Wilson, Inc. pursuant to the KW Revolving Facility are guaranteed by KW and certain wholly owned subsidiaries of KW.

As at 30 June 2017, US\$125 million was undrawn under the KW Revolving Facility.

(e) The KW Senior Notes

In March 2014, Kennedy-Wilson, Inc. (the "Issuer") completed a public offering of US\$300.0 million aggregate principal amount of 5.875% senior notes due 2024 (the "2024 Notes"), for approximately US\$290.7 million, net of discount and estimated offering expenses. The 2024 Notes were issued pursuant to an indenture dated 25 March 2014 and entered into between the Issuer and Wilmington Trust National Association (as trustee), as supplemented by a supplemental indenture dated 25 March 2014 between the Issuer, KW (as parent guarantor), certain subsidiaries of the Issuer (as subsidiary guarantors) and Wilmington Trust National Association (as trustee) (the "2024 Indenture"). The Issuer's obligations under the 2024 Notes are fully and unconditionally guaranteed by KW and the subsidiary guarantors. At any time prior to 1 April 2019, the Issuer may redeem the 2024 Notes, in whole or in part, at a redemption price equal to 100% of their principal amount, plus an applicable "make-whole" premium and accrued and unpaid interest, if any, to the redemption date. At any time and from time to time on or after 1 April 2019, the Issuer may redeem the 2024 Notes, in whole or in part, at the redemption price specified in the 2024 Indenture, plus accrued and unpaid interest, if any, to the redemption date. Prior to 1 April 2017, the Issuer may also redeem up to 35% of the 2024 Notes from the proceeds of certain equity offerings. Interest on the 2024 Notes accrues at a rate of 5.875% per annum and is payable semi-annually in arrear on 1 April and 1 October of each year, commencing on 1 October 2014. The 2024 Notes will mature on 1 April 2024.

In August 2016, the Issuer completed an additional public offering of US\$250 million 2024 Notes (the "Additional Notes"). The Additional Notes have substantially identical terms as the 2024 Notes described above and are treated as a single series with the 2024 Notes under the 2024 Indenture. The Additional Notes were issued and sold at a public offering price of 100.0% of their principal amount, plus accrued interest. The amount of the 2024 Notes included in KW's consolidated balance sheets was US\$897.9 million as at 30 June 2017.

(f) The 2042 Notes

In November and December 2012, the Issuer completed a public offering of US\$55.0 million aggregate principal amount of senior notes due 2042 (the "2042 Notes"). The 2042 Notes were issued pursuant to an indenture dated 28 November 2012, by and among the Issuer, KW (as parent guarantor), certain subsidiaries of the Issuer (as subsidiary guarantors) and Wilmington Trust National Association (as trustee), as amended by various subsequent supplemental indentures. The Issuer's obligations under the 2042 Notes are fully and unconditionally guaranteed by KW and the subsidiary guarantors. At any time prior to 1 December 2017, the Issuer may redeem the 2042 Notes, in whole or in part, at a redemption price equal to 100% of their principal amount, plus an applicable "make-whole" premium and accrued and unpaid interest, if any, to the redemption date. At any time and from time to time on or after 1 December 2017, the Issuer may redeem the 2042 Notes, in whole or in part, at a redemption price equal to 100% of their principal amount, plus accrued and unpaid interest, if any, to the redemption date. Interest on the 2042 Notes accrues at a rate of 7.750% per annum and is payable quarterly in arrear on 1 March, 1 June, 1 September and 1 December of each year, commencing on 1 March 2013. The 2042 Notes will mature on 1 December 2042. The amount of the 2042 Notes included in KW's consolidated balance sheets was US\$55.0 million at 30 June 2017.

14. Other information

(a) Each of Rothschild, J.P. Morgan Cazenove, Goldman Sachs, BofA Merrill Lynch and Deutsche Bank has given and not withdrawn its written consent to the issue of this document with the references to its name in the form and context in which such references appear. In addition, each of KPMG and Goldman Sachs International has given and not withdrawn its written consent to the inclusion of its report on the KW 2017 Profit Forecast and the KWE 2017 Profit Forecast in the form and context in which it is included.

- (b) Each of CBRE and Colliers has given and not withdrawn its written consent to the issue of this document with the inclusion of: (i) the references to its name in the form and context in which such references appear; and (ii) its valuation report relating to the property portfolio in the form and context in which it is included.
- (c) There is no agreement, arrangement or understanding whereby the beneficial ownership of any of the Scheme Shares to be acquired by KW in pursuance of the Merger will be transferred to any other person, save that KW reserves the right to transfer any such shares to any member of the KW Group.
- (d) Save as disclosed in this document, there is no agreement, arrangement or understanding (including any indemnity or compensation arrangements) between KW or any person acting in concert with it and any of the directors, recent directors or shareholders or recent shareholders of KWE or any person interested or recently interested in KWE Shares having any connection with or dependence on, or which is conditional upon, the outcome of the Merger.
- (e) Save as disclosed in this document, the emoluments of the KW Directors will not be affected by the acquisition of KWE or by any other associated transaction.
- (f) No proposal exists in connection with the Merger that any payment or other benefit be made or given to any KWE Director as compensation for loss of office or as consideration for or in connection with his or her retirement from office.
- (g) Save as disclosed in this document, no management incentivisation arrangements, as envisaged by Rule 16.2 of the Code, are proposed in connection with the Merger.

15. No set-off of consideration

Except with the consent of the Panel, settlement of the consideration to which any Scheme Shareholder is entitled under the Scheme will be implemented in full in accordance with the terms of the Scheme, free of any lien, right of set-off, counterclaim or other analogous right to which KW might otherwise be, or claim to be, entitled against such Scheme Shareholder.

16. Fees and expenses

16.1 KWE

The aggregate fees and expenses expected to be incurred by KWE in connection with the Merger (excluding any applicable VAT) are expected to be approximately:

Category	Amount – £
Financing arrangements	Nil
Financial and corporate broking advice	11,000,000
Legal advice ¹	3,515,000
Accounting advice	210,000
Public relations advice	20,000
Other professional services	375,000
Other costs and expenses	760,000
Total	15,880,000

¹ Legal advice is charged based on hourly rates of legal advisers and this estimate is based on the time charged by legal advisers up to the Latest Practicable Date and an estimate of time required up to the Effective Date (such estimate is uncertain and depends on a number of factors).

16.2 KW

The aggregate fees and expenses expected to be incurred by KW in connection with the Merger (excluding any applicable VAT) are expected to be approximately:

Category	Amount – £
Financing arrangements	Nil
Financial and corporate broking advice	15,200,000
Legal advice ¹	5,850,000
Accounting advice	500,000
Public relations advice	80,000
Other professional services	675,000
Other costs and expenses	975,000
Total	23,280,000

¹ Legal advice is charged based on hourly rates of legal advisers and this estimate is based on the time charged by legal advisers up to the Latest Practicable Date and an estimate of time required up to the Effective Date (such estimate is uncertain and depends on a number of factors).

17. Documents on display

Copies of the documents in paragraph 17.1 below can be reviewed on KWE's website at www.kennedywilson.eu and copies of the documents in paragraph 17.2 below can be reviewed on KW's website at ir.kennedywilson.com up to and including the Effective Date or the date on which the Scheme lapses or is withdrawn, whichever is earlier.

17.1 KWE

- (A) the KWE Articles;
- (B) KWE's articles of association as proposed to be amended by the Special Resolution at the KWE General Meeting;
- (C) the material agreements referred to at paragraph 13.1 of this Part Eight above;
- (D) the irrevocable undertakings and letter of intent referred to at paragraph 4 of this Part Eight above;
- (E) each of Rothschild's and J.P. Morgan Cazenove's written consent referred to at paragraph 14 of this Part Eight above;
- (F) the Original Announcement;
- (G) the New Offer Announcement;
- (H) this document, the accompanying letter from the chair of the KWE board, the Forms of Proxy and the Election Form;
- (I) CBRE's Valuation Report set out in Annex I (Valuation Reports) to this document;
- (J) Colliers' Valuation Report set out in Annex I (Valuation Reports) to this document; and
- (K) each of CBRE's and Collier's written consents referred to at paragraph 14 of this Part Eight above.

17.2 KW

- (A) the KW Amended and Restated Articles;
- (B) the KW Bylaws;
- (C) the material agreements referred to at paragraph 13.2 of this Part Eight above;
- (D) the irrevocable undertakings and letter of intent referred to at paragraph 4 of this Part Eight above;
- (E) each of KPMG's, Goldman Sachs', BofA Merrill Lynch's and Deutsche Bank's written consents referred to at paragraph 14 of this Part Eight above;
- (F) each of KPMG's and Goldman Sachs International's reports on the KW 2017 profit forecast and the KWE 2017 Profit Forecast;

- (G) the Original Announcement;
- (H) the New Offer Announcement;
- (I) this document, the accompanying letter from the chair of the KWE board, the Forms of Proxy and the Election Form;
- (J) the KW Prospectus;
- (K) the KW Proxy Statement; and
- (L) a full list of the dealings by the persons acting in concert with KW referred to at paragraphs 6.3(5) and 6.5(5) of this Part Eight.

PART NINE

BASES AND SOURCES

- 1. In this document, unless otherwise stated, or the context otherwise requires, the following sources and bases of calculation have been used:
 - 1.1 Historical financial information relating to KW and KWE have been extracted or derived (without material adjustment) from their respective audited financial statements as at and for the financial years ended 31 December 2016 and 31 December 2015.
 - 1.2 Historical financial information relating to KW and KWE have been extracted or derived (without material adjustment) from their respective unaudited financial statements (and accompanying explanatory announcement via Regulatory Information Service) as at and for the six months ended 30 June 2017.
- 2. The implied value of the New Offer is calculated:
 - 2.1 by reference to the Closing Price of a KW Share of US\$19.15 at the Latest Practicable Date:
 - 2.2 on the basis of an exchange rate of US\$1.3209 / £1, taken from Bloomberg as at 5:00 p.m. on the Latest Practicable Date;
 - 2.3 on the basis of the exchange ratio of 0.3854 New KW Shares in exchange for each Scheme Share held; and
 - 2.4 on the basis of the existing number of KWE Shares in issue referred to in paragraph 4 below.
- 3. The implied value of the Original Offer is calculated:
 - 3.1 by reference to the Closing Price of 19.15 per KW Share on the Latest Practicable Date;
 - 3.2 on the basis of an exchange rate of US\$1.3209 / £1, taken from Bloomberg as at 5:00 p.m. on the Latest Practicable Date;
 - 3.3 on the basis of the exchange ratio of 0.667 New KW Shares in exchange for each Scheme Share held; and
 - 3.4 on the basis of the existing number of KWE Shares in issue referred to in paragraph 4
- 4. As at the Latest Practicable Date, KWE had 126,133,407 KWE Shares in issue and KW had 114,218,250 KW Shares in issue.
- 5. As at the Latest Practicable Date, KW and its subsidiaries held 30,015,924 KWE Shares.
- 6. As at the Latest Practicable Date, KWE had no dilutive options or awards outstanding over KWE Shares. KW has granted RSUs to certain KW Directors and certain of its employees, which convert into KWE Shares on a one-for-one basis on vesting. As at the Latest Practicable Date, there were 532,230 unvested RSUs.
- 7. Assuming that all Scheme Shareholders receive the consideration available under the New Offer, the share capital of the Combined Group (being 151,467,049 KW Shares) has been calculated as the sum of:
 - 7.1 a total number of 114,218,250 KW Shares in issue as referred to in paragraph 4 above; and
 - 7.2 37,248,799 KW Shares which would be issued pursuant to the terms of the New Offer (being the number of KWE Shares as referred to in paragraph 4 above, less the number of KWE Shares held by KW and its subsidiaries as referred to in paragraph 5 above plus the number of KWE Shares underlying the unvested RSUs that will vest after the KWE General Meeting but prior to the Scheme Record Time (assuming, for this purpose, that no such underlying KWE Shares are withheld to satisfy tax withholding obligations), multiplied by the exchange ratio of 0.3854 as referred to in paragraph 2.3 above).
- 8. Assuming that all Scheme Shareholders receive the consideration available under the Original Offer, the share capital of the Combined Group (being 178,683,609 KW Shares) has been calculated as the sum of:

- 8.1 a total number of 114,218,250 KW Shares in issue as referred to in paragraph 4 above; and
- 8.2 64,465,359 KW Shares which would be issued pursuant to the terms of the Original Offer (being the number of KWE Shares as referred to in paragraph 4 above, less the number of KWE Shares held by KW and its subsidiaries as referred to in paragraph 5 above plus the number of KWE Shares underlying the unvested RSUs that will vest after the KWE General Meeting but prior to the Scheme Record Time (assuming, for this purpose, that no such underlying KWE Shares are withheld to satisfy tax withholding obligations), multiplied by the exchange ratio of 0.667 as referred to in paragraph 3.3 above).
- 9. Assuming all KWE Shareholders receive the New Offer Consideration and Distribution, the percentage of the share capital of the Combined Group that will be owned by KWE Shareholders following the Merger (being approximately 25%) is calculated by dividing the number of KW Shares to be issued pursuant to the New Offer referred to in paragraph 7.2 above by the issued share capital of the Combined Group (as set out in paragraph 7 above) and multiplying the resulting sum by 100 to produce a percentage.
- 10. Assuming all KWE Shareholders elected to receive the Original Offer Consideration, the percentage of the share capital of the Combined Group that will be owned by KWE Shareholders following the Merger (being approximately 36%) is calculated by dividing the number of KW Shares to be issued pursuant to the Original Offer referred to in paragraph 8.2 above by the issued share capital of the Combined Group (as set out in paragraph 8 above) and multiplying the resulting sum by 100 to produce a percentage.
- 11. The market capitalisation of the Combined Group, assuming that all Scheme Shareholders elect to receive the consideration available under the New Offer, is calculated:
 - 11.1 by reference to the Closing Price of a KW Share of US\$19.15 at the Latest Practicable Date; and
 - 11.2 on the basis of the share capital of the Combined Group (being 151,467,049 KW Shares) following the Merger, referred to in paragraph 7 of this Part Nine above.
- 12. The market capitalisation of the Combined Group, assuming that all Scheme Shareholders elect to receive the consideration available under the Original Offer, is calculated:
 - 12.1 by reference to the Closing Price of a KW Share of US\$19.15 at the Latest Practicable Date: and
 - 12.2 on the basis of the share capital of the Combined Group (being 178,683,609 KW Shares) following the Merger, referred to in paragraph 8 of this Part Nine above.
- 13. The Combined Group's enterprise value, assuming that all Scheme Shareholders elect to receive the consideration available under the New Offer, is calculated:
 - 13.1 on the basis of the market capitalisation of the Combined Group as referred to in paragraph 11 of this Part Nine above;
 - 13.2 plus KW's net indebtedness (at share) of US\$3,166 million as at 30 June 2017;
 - 13.3 plus 76.2% of KWE's net indebtedness of US\$1,224 million as at 30 June 2017, converted from pounds sterling into US Dollars on the basis of an exchange rate of US\$1.3025 / £1 as at 30 June 2017, that KW did not own as at 30 June 2017; and
 - 13.4 plus expected incremental net indebtedness of US\$733 million related to the payment of the KW Cash Component and the KWE Special Distribution, as well as transaction costs expected to be incurred as a result of the Merger.
- 14. The Combined Group's enterprise value, assuming that all Scheme Shareholders elect to receive the consideration available under the Original Offer, is calculated:
 - 14.1 on the basis of the market capitalisation of the Combined Group as referred to in paragraph 12 of this Part Nine above;
 - 14.2 plus KW's net indebtedness (at share) of US3,166 million as at 30 June 2017;
 - 14.3 plus 76.2% of KWE's net indebtedness of US\$1,224 million as at 30 June 2017, converted from pounds sterling into US Dollars on the basis of an exchange rate of US\$1.3025 / £1 as at 30 June 2017, that KW did not own as at 30 June 2017; and

- 14.4 plus expected incremental net indebtedness of US\$50 million related to transaction costs expected to be incurred as a result of the Merger.
- 15. The premium calculations to the price of a KW Share have been calculated by reference to:
 - 15.1 the closing price of 979 pence of one KWE Share on 21 April 2017 (the last Business Day before the commencement of the Offer Period);
 - 15.2 the volume weighted average price of 960 pence of one KWE Share for the three month period that ended on 21 April 2017 (the last Business Day before the commencement of the Offer Period); and
 - 15.3 KWE's last reported Adjusted NAV of 1,241 pence of one KWE Share as at 30 June 2017.
- 16. The volume weighted average prices of a KWE Share are derived from data provided by Bloomberg.
- 17. Unless otherwise stated, all amounts converted from pounds sterling to US Dollars and vice versa are converted based on the exchange rate of US\$1.3209:£1 as at 5.00 p.m. on the Latest Practicable Date derived from Bloomberg.
- 18. Unless otherwise stated, all prices quoted for KWE Shares have been derived from the Daily Official List and represent closing middle market prices on the relevant date.
- 19. Unless otherwise stated, all prices quoted for UK Gilts have been derived from Bloomberg and represent the price at 5.00 p.m. on the relevant date.
- 20. Unless otherwise stated, all prices quoted for KW Shares have been derived from the NYSE and represent the Closing Price on the relevant date.
- 21. Unless otherwise stated, all figures and percentages in this document have been calculated on the assumption that all Scheme Shareholders made no Elections.
- 22. For the purposes of the financial comparisons contained in this document, no account has been taken of any liability to taxation or the treatment of fractions under the Merger.
- 23. Certain figures included in this document have been subject to rounding adjustments.

PART TEN

DESCRIPTION OF KW SHARES

1. Risk factors

The attention of KWE Shareholders is drawn to the risk factors set out in KW's filings with the SEC, including Item 1A of Part II of KW's quarterly report on Form 10-Q for its fiscal quarter ended 30 June 2017 and any subsequent reports on Forms 10-K, 10-Q and 8-K, and the risk factors set out in Part 2 "Risk Factors" of the KW Prospectus. KWE Shareholders should consider the balance of New KW Shares and cash consideration that they are entitled to receive under the Scheme in light of their own personal circumstances and are, therefore, strongly advised to seek their own independent financial, tax and legal advice in light of their own particular circumstances before deciding whether to make an Original Offer Election or a Mix and Match Election.

2. Rights attaching to KW Shares

The New KW Shares to be issued in connection with the Merger will be issued credited as fully paid and free from all Encumbrances and will rank pari passu in all respects with the existing issued KW Shares and will carry the right to receive all dividends and other distributions declared, paid or made by KW on or after the Effective Date or by reference to a record date on or after the Effective Date (in each case whether or not wholly or partly in respect of a period which precedes the Effective Date).

The following is a summary of the rights, preferences and restrictions attaching to KW Shares:

2.1 Voting rights

Holders of KW Shares have exclusive voting rights for the election of KW Directors and all other matters requiring shareholder action, except with respect to amendments to the KW Amended and Restated Articles that alter or change the powers, preferences, rights or other terms of any outstanding preferred stock if the holders of such affected series of preferred stock are entitled to vote on such an amendment. Holders of KW Shares are entitled to one vote per share on matters to be voted on by shareholders.

For further information on the voting rights of KW CDI holders, please see paragraph 31.1(b) of Part Two (*Explanatory Statement*).

2.2 Conversion, pre-emptive, subscription rights; redemption provisions

KW Shares have no conversion, preemptive or other subscription rights and there are no sinking fund or redemption provisions applicable to KW Shares.

2.3 Dividends

Dividends may be declared by the KW Board and may be paid in cash, property or in KW Shares. The payment of dividends, if ever, on KW Shares will be subject to (i) the prior payment of dividends on any outstanding shares of preferred stock and (ii) compliance with any applicable limitation in KW's debt agreements, including debt securities. Before the payment of any dividend, there may be set aside out of KW funds available for dividends such sum or sums as the KW Board from time to time, in their absolute discretion, think proper for equalising dividends. In order that KW may determine the shareholders entitled to receive payment of any dividend, the KW Board may fix a record date which shall not precede the date upon which the resolution fixing the record date is adopted or be more than 60 days prior to such action. If no record date is fixed, the record date shall be at the close of business on the day on which the KW Board adopts the resolutions relating thereto.

3. Comparison of shareholders' rights

As at the date of this document, your rights as a KWE Shareholder are governed by, among other things, applicable Jersey law, including the Jersey Companies Law, and by the KWE Articles and the memorandum of association of KWE. KWE has been established pursuant to the Collective Investment Funds (Jersey) Law 1988 (as amended) and the Jersey Listed Fund Guide published by the Jersey Financial Services Commission and is regulated by, and subject to the applicable requirements of, the Jersey Financial Services Commission. KWE is also required to comply with the requirements of the Listing Rules, the Code, the Disclosure and

Transparency Rules, the FCA and the London Stock Exchange. After the Merger, holders of New KW Shares will have the rights of a shareholder of KW, and such rights will be governed by, among other things, the requirements of Delaware law, US federal securities laws (including the US Exchange Act) and rules and regulations promulgated thereunder, NYSE rules and the KW Amended and Restated Articles and the KW Bylaws.

The table below provides a summary of material differences in your current rights as a KWE Shareholder and the rights you will have as a KW Shareholder as a result of the Merger, on the basis that following the Merger you will hold one or more New KW Shares.

This summary is indicative only and does not purport to cover all of the differences between the KWE Articles and the memorandum of association of KWE, and the KW Amended and Restated Articles and the KW Bylaws, or between laws and requirements applicable to the KWE Shares and the KW Shares. We encourage you to read the KW Amended and Restated Articles and the KW Bylaws which are available on KW's website at ir.kennedywilson.com for a more complete understanding of the rights conferred by the New KW Shares.

The following descriptions are qualified in their entirety by reference to, and are subject to, the detailed provisions of the KWE Articles and the memorandum of association of KWE, and the KW Amended and Restated Articles and the KW Bylaws, and the complete text of the relevant provisions of the Jersey Companies Law, the Listing Rules, the Code, the Disclosure and Transparency Rules, the regulations of the Jersey Financial Services Commission (including the Jersey Listed Fund Guide published by the Jersey Financial Services Commission), the Collective Investment Funds (Jersey) Law 1988 (as amended), the FCA and the London Stock Exchange, as well as the relevant provisions of Delaware law, US federal securities laws and the rules and regulations promulgated thereunder and the NYSE rules. We encourage you to read those laws and documents.

Copies of the KWE Articles and the memorandum of association of KWE, and the amendments to the KWE Articles as proposed by the Special Resolution, can be reviewed on KWE's website at www.kennedywilson.eu, and a copy of the KW Amended and Restated Articles and the KW Bylaws can be reviewed on KW's website at ir.kennedywilson.com, in each case up to and including the Effective Date or the date on which the Scheme lapses or is withdrawn, whichever is earlier.

Please note that holders of KW CDIs will have an interest in the underlying New KW Shares, but will not be the registered holder of the New KW Shares.

Provisions applicable to KW Shares

Provisions currently applicable to KWE Shareholders

Voting Rights

Voting, Generally

- Each KW Shareholder is entitled to one vote for each KW Share held, except as otherwise provided by applicable law or the KW Amended and Restated Articles.
- The KW Bylaws provide that, unless otherwise required by applicable law, the rules or regulations of any stock exchange applicable to KW, the KW Amended and Restated Articles or the KW Bylaws, a majority of votes cast on any question or matter (other than the election of directors) brought before a duly called meeting of KW Shareholders at which a quorum is present shall be sufficient to approve such question or matter.
- Except to the extent that the vote of a greater number or voting by classes is required by Delaware law or pursuant to certificate of incorporation, and except in the event of a "contested election" of directors (as explained below), each director shall be elected by a majority of the votes cast at any meeting of the KW Shareholders for the election of directors at which a quorum is present.

- Each KWE Shareholder is entitled to one vote for each KWE Share held by such shareholder.
- Any additional shares may be issued with such preferred, deferred or other special voting rights or restrictions as the KWE Shareholders may determine in accordance with the KWE Articles.
- Under the Jersey Companies Law and the KWE Articles, certain matters require "ordinary resolutions", which must be approved by at least a simple majority of the votes cast by shareholders, and certain other matters require "special resolutions", which require the affirmative vote of at least two thirds of the votes cast at the meeting.
- An ordinary resolution is needed to (among other matters): determine any preferred, deferred or other special rights or restrictions applicable to any new shares, determine the minimum and maximum

Provisions applicable to **KW Shares**

Provisions currently applicable to **KWE Shareholders**

- A "contested election" is defined as an election of directors in which the number of candidates for election exceeds the number of directors to be elected. In the case of a contested election, directors shall be elected by a plurality of the votes cast at any meeting at which a quorum is present, i.e. the director(s) with the most votes will be elected, even if they do not receive a majority of the votes cast.
- KW will enter into arrangements pursuant to which holders of KW CDIs will be able to give directions as to voting at meetings of KW Shareholders.

Record Date

- The KW Bylaws provide that the record date for determining the KW Shareholders entitled to attend and vote at any meeting of KW Shareholders shall be fixed by the KW Board. Unless otherwise required by law, notice of the meeting of KW Shareholders may not be given to KW Shareholders more than 60 days or less than 10 days prior to the meeting date.
- The record date for determining the holders of KW CDIs entitled to vote at any meeting of KW Shareholders will be the same as that of KW Shareholders.

- number of directors and appoint or remove directors from office.
- A special resolution is needed to (among other matters): amend the KWE Articles or KWE's memorandum of association, change the name of KWE, alter KWE's stated capital, change the company's status from a public to a private company, convert KWE from a no par value to a par value company, repurchase or redeem KWE Shares, reduce KWE's capital, ratify a transaction in which an interested KWE Director has failed to disclose his interest. commence or terminate a winding up of KWE and appoint or remove a liquidator.
- Certain other matters under the Jersey Companies Law require approval by a different majority of shareholders. For example, a scheme of arrangement requires the approval of a majority in number of shareholders representing three-fourths of the voting rights of the members or class of members present and voting at the relevant
- There is no comparable provision in the KWE Articles.
- The KWE Directors may exercise all powers of KWE not required by the Jersey Companies Law or the KWE Articles to be exercised by KWE Shareholders in general meeting.
- Under the Uncertificated Securities Order in relation to uncertificated KWE Shares, for the purposes of determining which KWE Shareholders are entitled to attend or vote at a meeting of KWE Shareholders, KWE may specify in the notice of the meeting a time (being not before the beginning of the period commencing 48 hours before the time fixed for the meeting) by which a KWE Shareholder must be entered on the Register in order to have the right to attend or vote at the meeting. For the purposes of serving notices of meetings, whether under the Jersey Companies Law, a provision in the KWE Articles, or any other instrument, KWE may determine that those KWE Shareholders entered on the Register at the close of business on a day determined by KWE shall be entitled to receive such notices (provided that such day is not more than 21 days before the day on which the notices of the meeting are sent).
- The quorum for a general meeting of KWE Shareholders (whether this be an "Annual General Meeting" or an "Extraordinary General Meeting" (the latter being the Jersey equivalent of a "special meeting" of KW Shareholders)) is not less than two

Quorum

Holders of at least a majority in voting power of all KW Shares, or where a separate vote is required by a class or classes, the holders of a majority of voting power of the KW Shares of such class or classes, issued and outstanding and entitled to vote, present in person at a meeting of KW Shareholders or

Provisions applicable to KW Shares

Provisions currently applicable to KWE Shareholders

- represented by proxy, shall constitute a quorum, except as otherwise provided by statute or by the KW Amended and Restated Articles.
- If a quorum is not present or represented at any meeting of KW Shareholders, the holders of a majority of the votes entitled to be cast by KW Shareholders present in person or represented by proxy shall have the power to adjourn the meeting.

KWE Shareholders present and entitled to vote but so that not less than two individuals will constitute the quorum.

The KWE Articles provide that two persons holding or representing at least one third in number of the issued shares of that class constitute a quorum at class meetings.

Action by Written Consent

- The KW Bylaws allow KW Shareholders to consent to corporate actions in writing without a meeting. If written consent is to be used, the KW Board may fix a record date which shall be not more than 10 days after the date upon which the resolution fixing the record date is adopted by the KW Board. Any KW Shareholder seeking to have KW Shareholders authorise or take corporate action by written consent shall, by written notice to the secretary of KW, request that the KW Board fix a record date.
- Each written consent must bear the date of signature of the KWE Shareholder who signs the consent.
- To be effective, written consents must be received from a sufficient number of KW Shareholders to take such action within 60 days of the earliest dated written consent received in the manner prescribed in the KW Bylaws and applicable law, and not revoked.

- KWE Shareholders can pass an ordinary or special resolution in writing without holding a meeting, except in the case of removing an auditor.
- All KWE Shareholders (or their duly appointed attorneys) who would be entitled to vote on the resolution if it was proposed at a general meeting must sign such a written resolution.

Shareholder Proposals and Shareholder Nominations of Directors

Shareholders' Ability to Call a Special Meeting

- The KW Bylaws do not permit KW Shareholders to call a special meeting of KW Shareholders.
- Under Delaware law, any stockholder may petition the Court of Chancery to order a meeting to elect directors if such meeting, or action to elect directors by written consent in lieu of a meeting, has not been held within thirteen months after the latest to occur of the last annual meeting or action by written consent to elect directors in lieu of an annual meeting.
- The KWE Directors must, if they receive a written requisition from KWE Shareholders representing at least 10% of the total voting rights of KWE Shareholders who have the right to vote at the meeting requisitioned, forthwith proceed to call a meeting of KWE Shareholders, to be held as soon as practicable and in any event not later than two months after the date of deposit of the requisition.
- Any such requisition must state the objects of the meeting, be signed by or on behalf of the requisitioners, and must be submitted to KWE's registered office in Jersey.
- If the KWE Directors do not, within 21 days of the date of deposit of a requisition from requisitioners, call a meeting of KWE Shareholders to be held within two months of the date of deposit of the requisition, the requisitioners (or those representing more than a majority of the total voting rights of the requisitioners) may call a meeting of KWE Shareholders to be held within three months of the date of deposit of the requisition. A meeting called by the requisitioners in this manner must be called in the same manner, as nearly as possible, as meetings are called by the KWE Directors. KWE must pay the requisitioners their

Shareholder Proposals: Director Nominations, Generally

- Under the KW Bylaws, KW Shareholders have an express right to nominate candidates for election to the KW Board and bring other business before an annual meeting, provided that the KW Shareholder was a KW Shareholder of record at the time notice was given of the meeting and is a KW Shareholder at the time of the meeting, is entitled to vote at the meeting and complies with the notice procedures set forth below as to such business or nomination. The KW Shareholder must give timely written notice and the notice must:
 - set forth, as to the KW Shareholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made (i) the name and address of such KW Shareholder and beneficial owner, (ii) the class or series and number of shares of capital stock of KW owned beneficially and of record by such KW Shareholder and such beneficial owner, (iii) a description of any agreement, arrangement or understanding between or among such KW Shareholder and/or beneficial owner, any of their affiliates or associates and any others acting in concert with any of the foregoing; (iv) a description of any arrangement, agreement or understanding that has been entered into by or on behalf of such KW Shareholder and beneficial owner, the effect or intent of which is to mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of such KW Shareholder or beneficial owner, with respect to the KW Shares; (v) a representation that the KW Shareholder is a holder of record KW Shares entitled to vote at such meeting and intends to appear in person or by proxy at the meeting; (vi) a representation whether the KW Shareholder or beneficial owner, intends or is part of a group which intends to deliver a proxy statement to the KW Shareholders or otherwise solicit votes from KW Shareholders; and (vii) any other information relating to such KW Shareholder and beneficial owner that would be required to be disclosed in a proxy statement or other filings required under Section 14(a) of the US Exchange Act and the rules and regulations promulgated thereunder;
 - (2) if the notice relates to any business other than a nomination of director(s), set forth (i) a brief

reasonable expenses incurred in calling the meeting if the KWE Directors have failed to properly call a requisitioned meeting. Only the business stated in the requisition as the objects of the meeting can be transacted at a requisitioned meeting.

- The KWE Articles provide that an "Extraordinary General Meeting" of KWE may be called by any KWE Shareholder or KWE Director if there are not sufficient KWE Directors to convene such a meeting.
- As noted above, KWE Shareholders representing at least 10% of the total voting rights of KWE Shareholders who have the right to vote at the meeting requisitioned have the right to requisition a meeting.
- Subject to the Jersey Companies Law and mandatory provisions of applicable law, a KWE Shareholder does not have the right to make nominations or other business proposals or to add proposals to the agenda before a general meeting of KWE Shareholders called at the instance of the KWE Directors.

description of the proposal or business desired to be brought before the meeting; (ii) the text of the proposal or business; (iii) the reasons for conducting such business at the meeting; and (iv) any material interest of such KW Shareholder and the beneficial owner, if any, on whose behalf the business is made; and

(3) if the notice relates to the nomination of a director, set forth, for each nominee (i) information relating to such person that would be required to be disclosed in a proxy statement, or otherwise required under Section 14(a) of the US Exchange Act and the rules and regulations promulgated thereunder and (ii) such nominee's written consent to being named in the proxy statement as a nominee and to serving as a director if elected.

References to a "beneficial owner" in the preceding paragraphs would, for the avoidance of doubt, include a holder of KW CDIs.

- To be timely, a KW Shareholder's notice must be delivered to the secretary of KW not later than the close of business on the 90th day and not earlier than the close of business on the 120th day prior to the first anniversary of the preceding year's annual meeting, subject to any other requirements of law; provided, however, that in the event that the date of the annual meeting is more than 30 days before or more than 70 days after such anniversary date, notice by the KW Shareholder to be timely must be so delivered not earlier than the close of business on the 120th day prior to the date of such annual meeting and not later than the close of business on the later of the 90th day prior to the date of such annual meeting or the 10th day following the day on which public announcement of the date of such meeting is first made by the corporation.
- Under the KW Bylaws the chairman of the meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was properly made and if not, to declare that such defective proposal or nomination shall be disregarded.

Sources and Payment of Dividends

Sources of Dividends

- Under Delaware law, subject to any restriction in the corporation's certificate of incorporation, the board of directors may declare and pay dividends out of:
 - surplus of the corporation, which is defined as net assets less statutory capital; or
 - (2) if no surplus exists, out of the net profits of the corporation for the year in which the dividend is declared and/or the preceding year;

provided, however, that if the capital of the corporation has been diminished to an amount less than the aggregate amount of capital represented by the outstanding stock of all classes having preference upon the distribution of assets, the board of directors

Under the Jersey Companies Law, KWE (being a no par value company) may pay a dividend out of any source other than its capital redemption reserve. However, KWE may only pay a dividend if the KWE Directors who are to authorise the dividend make a solvency statement.

may not declare and pay dividends out of the corporation's net profits until the deficiency in the capital has been repaired.

Declaration of Dividends

- The KW Bylaws provide that dividends (payable in cash, stock or property) may be declared by the Board pursuant to applicable law.
- Under Delaware law, if a dividend is to be paid in shares of unissued stock, the Board must designate a capital reserve of an amount no less than the aggregate par value of shares being declared as a dividend.

The KWE Articles provide that:

- subject to the provisions of the Jersey Companies Law, KWE may by ordinary resolution declare dividends in accordance with the respective rights of the KWE Shareholders, but no dividend shall exceed the amount recommended by the KWE Directors:
- the KWE Directors may pay interim dividends as they may determine from time to time;
- except as otherwise provided by the rights attached to the KWE Shares, all shares carry a pro rata entitlement to the receipt of dividends based on the amounts paid up on the shares;
- a general meeting declaring a dividend may, on the recommendation of the KWE Directors, direct that a dividend by satisfied wholly in in part by the distribution of specific assets. If any difficulty arises in relation to the distribution, the KWE Directors may resolve that difficulty in any way they consider expedient;
- no dividend bears interest; and
- if a dividend remains unclaimed for 10 years after it is declared, the KWE Directors may resolve that it be forfeited to, and no longer be owed by, KWE.

Record Date

- The KW Bylaws provide that, in order for KW to determine the KW Shareholders entitled to receive the payment of any dividend and other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action of the KW Board, the KW Board may fix a record date which shall not be more than 60 days prior to such action.
- The record date for determining the holders of KW CDIs entitled to receive the payment of any dividend and other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action of the KW Board shall be the same as that of other KW Shareholders.
- Under the NYSE listing rules, listed companies must notify the NYSE at least 10 calendar days in advance of all record dates set for any purpose.

The KWE Articles provide that any resolution declaring a dividend on the shares of any class, whether a resolution of KWE in general meeting or a resolution of the KWE Directors, or any resolution of the KWE Directors for the payment of a fixed divided on a date prescribed for the payment thereof may specify that the same shall be payable to KWE Shareholders registered as holders of KWE Shares of the class concerned on the Register at the close business on a particular notwithstanding that it may be a date prior to that on which the resolution is passed (or, as the case may be, that prescribed for payment of a fixed dividend).

Purchase and Redemption of Stock

Purchase and Redemption of Stock, Generally

- Under Delaware law, a corporation may purchase, redeem, receive, take or otherwise acquire, own and hold, sell, lend, exchange, transfer or otherwise dispose of, pledge, use and otherwise deal in and with its own shares; provided, however, that no corporation shall:
 - (1) purchase or redeem its own shares of capital stock for cash or other property when the capital of the corporation is impaired or when such purchase or redemption would cause any impairment of the capital of the corporation, except that a corporation may purchase or redeem out of capital any of its own shares which are entitled upon any distribution of its assets, whether by dividend or in liquidation, to a preference over another class or series of its stock, or, if no shares entitled to such a preference are outstanding, any of its own shares, if such shares will be retired upon their acquisition and the capital of the corporation reduced:
 - (2) purchase, for more than the price at which they may then be redeemed, any of its shares which are redeemable at the option of the corporation; or
 - (3) redeem any of its shares unless their redemption is authorised by subsection 151(b) of the DGCL and then only in accordance with such section and the certificate of incorporation.
- Under Delaware law, a corporation has a right to resell any of its shares theretofore purchased or redeemed out of surplus and which have not been retired, for such consideration as shall be fixed by the board of directors.

- Under the Jersey Companies Law, KWE may purchase its own fully paid shares, including any redeemable shares, from any source. However, the repurchase of shares cannot result in only redeemable or treasury shares being in issue. If KWE were to repurchase any of its shares on a stock exchange this would require:
 - (1) the approval of a special resolution by a majority of not less than two-thirds of the voting rights represented in person or by proxy at a meeting of KWE Shareholders at which the resolution is proposed. The resolution must specify the maximum number of shares to be purchased, the maximum and minimum prices which may be paid, and a date, not being later than five years after the passing of the resolution, on which the repurchase authority is to expire; and
 - (2) a cash-flow solvency statement from the directors who authorise the repurchase by reference to the repurchase monies and the repurchase payment date and forward-looking for 12 months.
- Similar but slightly different rules apply if KWE repurchases its shares off-market.
- KWE may, subject to the Jersey Companies Law, issue or convert existing nonredeemable shares (whether or not issued) into shares that are to be redeemed or are liable to be redeemed at the option of KWE or of the KWE Shareholder on such terms and in such manner as may be determined by special resolution of the KWE Shareholders.
- The Jersey Companies Law allows fully paid shares to be redeemed from any source provided that the directors who authorise the redemption shall make a solvency statement (referencing the date of redemption and the redemption monies).
- If KWE redeems or repurchases any of its shares, the Jersey Companies Law allows it to hold such shares as treasury shares, provided KWE is authorised to do so by ordinary resolution (passed by a simple majority of the votes cast).
- The Listing Rules impose certain restrictions on a company with a premium listing (such as KWE) in the event it purchases its own shares, purchases its own securities other than shares, sales or transfers of treasury shares and any purchases by KWE of its own securities from a related party (see

- "Related Party Transactions") below. In particular:
- (1) Unless a tender offer is made to all holders of the class, purchases by KWE of less than 15% of any class of its equity shares (excluding treasury shares) pursuant to a general authority granted by shareholders, may only be made if the price to be paid is not more than the higher of: (i) 5% above the average market value of KWE's equity shares for the 5 business days prior to the day the purchase is made; and (ii) that stipulated by article 5(6) of the Market Abuse Regulation.
- (2) Purchases by KWE of 15% or more of any class of its equity shares (excluding treasury shares) pursuant to a general authority by the shareholders must be by way of a tender offer to all shareholders of that class.
- (3) Purchases of 15% or more of any class of its own equity shares may be made, other than by way of a tender offer, provided that the full terms of the share buyback have been specifically approved by shareholders.
- (4) Where a series of purchases are made pursuant to a general authority granted by shareholders, which in aggregate amount to 15% or more of the number of equity shares of the relevant class in issue immediately following the shareholders meeting at which the general authority to purchase was granted, a tender offer need only be made in respect of any purchase that takes the aggregate to or above that level.
- (5) Any decision by the board to submit to shareholders a proposal for KWE to be authorised to purchase its own equity shares is required to be notified to a regulated information service, such as RNS, as soon as possible.
- (6) A meeting of the shareholders of the class of shares proposed to be purchased must (prior to entering into any agreement to purchase such shares): (i) be convened; and (ii) their approval must be obtained for the proposed purchase of equity shares by a special resolution. A circular convening a meeting must also be circulated to the relevant shareholders.
- Under the Jersey Companies Law, if and for so long as KWE holds any of its shares as treasury shares, KWE is not entitled to exercise any voting attaching to those shares.

Voting Treasury Stock

• Under Delaware law, shares of its own capital stock belonging to the corporation or to another corporation, if a majority of the shares entitled to vote in the election of directors of such other

Provisions applicable to KW Shares

corporation is held, directly or indirectly, by the corporation, are neither entitled to vote nor counted for quorum purposes. However, a corporation has a right to vote stock, including but not limited to its own stock, held by it in a fiduciary capacity.

Meetings of Shareholders

- The KW Bylaws provide that all meetings of KW Shareholders are to be held at any place designated by the Board. The KW Bylaws also state that the Board of Directors may, in its sole discretion, determine that a meeting shall not be held at any place, but may instead by held solely by means of remote communication in accordance with Delaware law.
- Written notice of the annual meeting stating the place, if any, date and hour of the meeting and the means of remote communication, if any, by which KW Shareholders and proxy holders may be deemed to be present in person and vote at such meeting, shall be given to each KW Shareholder entitled to vote at such meeting not less than 10 nor more than 60 days before the date of the meeting.
- KW will enter into arrangements pursuant to which holders of KW CDIs will be able to give directions as to voting at meetings of KW Shareholders.

The KWE Articles provide that:

- KWE will hold an "Annual General Meeting" every calendar year at such time and place determined by the KWE Directors, unless all KWE Shareholders agree in writing to dispense with the requirement to hold an "Annual General Meeting";
- written notice of every "Annual General Meeting" and of every "Extraordinary General Meeting", including every general meeting called for the passing of a special resolution, must be given at least 14 clear days' before the meeting;
- however, a meeting may be called at shorter written notice if it is so agreed: (i) in the case of an "Annual General Meeting", by all the KWE Shareholders entitled to attend and vote at that meeting; and (ii) in the case of any other meeting, by a majority in number of the KWE Shareholders having a right to attend and vote at the meeting, being a majority together holding not less than ninety-five per cent of the total voting rights of the KWE Shareholders who have that right;
- the notice of the meeting shall specify, among other things, the place, day and time of the meeting and the general nature of the business to be transacted; and
- "clear days" means calendar days and excludes (1) the date on which the notice is served or deemed served; and (2) the date of the meeting itself.

The Listing Rules impose certain obligations on companies with a premium listing (such as KWE) that require KWE Shareholders to receive circulars for notices of meetings. A circular for the purpose of Listing Rule 13 must contain, among others:

- a clear and adequate explanation of its subject matter giving due prominence to its essential characteristics, benefits and risks;
- the reason why the shareholder is being asked to vote or, if no vote is required, why the circular is being sent;
- if voting or other action is required, the circular will contain all information necessary to allow the shareholders to make a properly informed decision;
- if voting or other action is required, the circular will contain a heading drawing

- attention to the document's importance and advising shareholders who are in any doubt as to what action to take to consult appropriate independent advisers;
- if voting is required, the circular will contain a recommendation from the board as to the voting action shareholders should take for all resolutions proposed, indicating whether or not the proposal described in the circular is, in the board's opinion, in the best interests of shareholders as a whole:
- if new shares are being issued in substitution for existing ones, explain what will happen to existing documents of title;

if it relates to a transaction in connection with which shares are proposed to be listed, include a statement that application has been or will be made for the shares to be admitted and, if known, a statement of the following matters: the dates on which the shares are expected to be admitted and on which dealings are expected to commence; how the new shares rank for dividend or interest; whether the new shares rank equally with any existing listed shares; the nature of the document of title; the proposed date of issue; the treatment of any fractions; whether or not the security may be held in uncertificated form; and the names of the regulated information exchanges on which securities are to be traded.

Special Meetings of Shareholders

Calling a Special Meeting

- The KW Bylaws provide that special meetings of KW Shareholders may be called only on the order of a majority of the entire KW Board.
- The KWE Directors may call a general meeting of KWE whenever they think fit.
- As noted above, KWE Shareholders representing at least 10% of the total voting rights of KWE Shareholders who have the right to vote at the meeting requisitioned have the right to requisition a meeting.
- The KWE Articles provide that an "Extraordinary General Meeting" of KWE may be called by any KWE Shareholder or KWE Director if there are not sufficient KWE Directors to convene such a meeting.

Notice

- The KW Bylaws provide that KW Shareholders entitled to vote at such special meeting must receive notice of the meeting not less than 10 days and not more than 60 days prior to the meeting.
- This notice must state the place, if any, date and hour of the meeting, the means of remote communication, if any, by which KW Shareholders and proxy holders may be deemed to be present in person and vote at such meeting, and the purpose or purposes for which the meeting is called. Business transacted at any special meeting of KW Shareholders shall be limited to the purposes stated

The KWE Articles provide that:

- written notice of every special meeting shall be given at least 14 clear days' before such meeting;
- an "Annual General Meeting" may be called at shorter written notice if it is so agreed by all the KWE Shareholders entitled to attend and vote thereat;
- an "Extraordinary General Meeting" may be called at shorter written notice if it is so agreed by a majority in number of the

Provisions currently applicable to KWE Shareholders

in the notice.

 KW will enter into arrangements pursuant to which holders of KW CDIs will be able to receive notice of meeting of KW Shareholders. KWE Shareholders having a right to attend and vote at the meeting being a majority together holding not less than ninety-five per cent of the total voting rights of the KWE Shareholders who have that right; and

the notice shall specify, among other things, the place, day and time of the meeting and the general nature of the business to be transacted.

Appraisal Rights

To Whom are Appraisal Rights Available

• Under Delaware law, stockholders of a corporation involved in a merger who hold shares of stock on the date a demand is made with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, and who has neither voted in favor of the merger or consolidation nor consented thereto in writing generally have the right to demand and receive payment in cash of the fair value of their stock as determined by the Delaware Court of Chancery, in lieu of receiving the merger consideration.

There are no directly analogous appraisal rights under Jersey law in connection with mergers and acquisitions.

- Where an acquisition is taking place by means of a Jersey "scheme of arrangement", the Court is required to consider, among other things, whether the arrangement is such that an intelligent and honest person, a shareholder of the class concerned and acting in respect of his or her interest might reasonably approve.
- Similarly, in a takeover situation, a shareholder who objects to his or her shares being acquired compulsorily may apply to the Court for an order that the offeror shall not be entitled to acquire the shares or specifying terms of the acquisition different from those of the offer.

Exceptions to Appraisal Rights

- Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation if the holders are required by the terms of an agreement of merger or consolidation to accept for such stock anything except:
 - shares of stock of the corporation surviving or resulting from such merger or consolidation, or depositary receipts in respect thereof;
 - (2) shares of stock or depositary receipts of another corporation that, at the effective date of the merger, will be either (a) listed on a national securities exchange; or (b) held of record by more than 2,000 holders;
 - (3) cash in lieu of fractional shares of the stock or depositary receipts received; or
 - (4) any combination of the shares of stock, depositary receipts and cash in lieu of fractional shares or fractional depositary receipts described in the foregoing (1) (3).

In addition, appraisal rights are not available to the holders of shares of the surviving corporation in the merger, if the merger does not require the approval of the stockholders of that corporation.

Preemptive Rights

- Under Delaware law, a stockholder is not entitled to preemptive rights to subscribe for additional issuances of stock or any security convertible into stock unless they are specifically granted in the certificate of incorporation. No such rights are granted in the KW Amended and Restated Articles.
- The KWE Articles provide that KWE shall not allot and issue equity securities for cash to any person on any terms unless (i) it has first made an offer to allot and issue a prorata proportion of those equity securities on the same terms or more favourable terms to KWE Shareholders who hold ordinary shares in KWE; and (ii) the period during which any such offer may be accepted has expired or KWE has received written notice of the acceptance or refusal of every offer so made.
- However, these pre-emptive rights do not apply in certain limited circumstances set out in KWE's Articles, including where shares are allotted and issued (i) in connection with a performance or management fee under KWE's investment management agreement, (ii) as bonus shares, (iii) as a dividend in specie, (iv) in connection with a rights issue.
- KWE Shareholders have also suspended these pre-emptive provisions until the end of the 2018 Annual General Meeting of KWE or 30 June 2018 (whichever is earlier) in respect of 12,613,340 new KWE Shares (being approximately 10% of the ordinary KWE Shares in issue when the 2017 Annual General Meeting was called).
- In the absence of any protection under Jersey Companies Law, the Listing Rules protect KWE Shareholders in the sense that if KWE proposed to issue equity shares for cash or to sell treasury shares that are equity shares for cash, KWE must first offer those equity shares in proportion to their existing holdings to: existing shareholders of that class; and holders of other equity shares of KWE who are entitled to be offered them. However, this is subject to certain exceptions, including in the event KWE Shareholders approve the disapplication of their own pre-emption rights under Jersey Companies Law.

Amendment of Governing Instruments

- Under Delaware law, unless the certificate of incorporation requires a greater vote, an amendment to the certificate of incorporation requires:
 - (1) recommendation of the board of directors;
 - (2) the affirmative vote of a majority of the outstanding stock entitled to vote; and
 - (3) the affirmative vote of a majority of the outstanding stock of each class entitled to vote.
- Under Delaware law, any merger agreement approved by the board must state the amendments or changes to be made to the certificate of incorporation
- The KWE board of directors does not have the power to amend the KWE Articles without shareholder approval; a special resolution is required to do so.

pursuant to the merger.

- The KW Bylaws provide that the KW Bylaws may be adopted, amended or repealed by the KW Shareholders entitled to vote thereon at any regular or special meeting or, if the KW Amended and Restated Articles so provide, by the KW Board. However, the fact that such power has been conferred on the KW Board shall not divest the KW Shareholders of the power nor limit their power to adopt, amend or repeal the KW Bylaws.
- The KW Amended and Restated Articles states that the KW Board shall have the power to make, alter, amend, change, add to or repeal the KW Bylaws, subject to the power of KW Shareholders to alter or repeal any bylaw whether adopted by them or otherwise.

Preferred Stock/Additional Shares

- The KW Amended and Restated Articles authorise the KW Board:
 - to provide for the issuance of one or more series of preferred stock;
 - (2) to fix for each such series such voting powers and designations, preferences, and relative participating, optional, or other special rights, and any such qualifications limitations, and restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by KW provided for the issue of such series
 - (3) the number of authorised shares of preferred stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the KW Shares entitled to vote in the election of directors, voting together as a single class, except in certain limited circumstances.
- KW currently does not have any shares of preferred stock outstanding.

- Preferred shares can be issued which give the holders rights of priority over ordinary shareholders. However, this would constitute a variation of the class rights of the ordinary shareholders requiring their approval as specified under "Stock Class Rights" below.
- The KWE Articles provide that a share or class of shares in the capital of KWE may be issued with such preferred, deferred or other special rights or such restrictions whether in regard to dividends, return of capital, voting or otherwise as KWE may determine by ordinary resolution. This ordinary resolution is in addition to the approval of any class of shareholders (e.g. ordinary shareholders) whose rights are being varied by the creation of the preferred shares.
- Subject to the KWE Articles, the unissued shares in KWE are at the disposal of the KWE Directors, who may allot, grant options over or otherwise dispose of them to such persons at such times and generally on such terms and conditions as they see fit.

Stock Class Rights

- Under Delaware law, any change to the rights of holders of KW's common or preferred stock would require an amendment to the KW Amended and Restated Articles or the applicable certificate of designation. Holders of shares of a class are entitled to vote as a class upon a proposed amendment to the KW Amended and Restated Articles if the amendment will:
 - (1) increase or decrease the authorised shares of the class:
 - (2) increase or decrease the par value of the shares of the class; or
 - (3) alter or change the powers, preferences or
- Amendments affecting the rights of the holders of any class of shares requires: (i) the consent in writing of the holders of two-thirds in number of the issued shares of that class; or (ii) the sanction of a special resolution passed at a separate meeting of the holders of shares of that class (with the quorum at a class meeting being two persons holding or representing at least one third in number of the issued shares of that class (save on adjournment, where the quorum is those persons who are present in person or by proxy or corporate representative, as applicable, at such

Provisions currently applicable to KWE Shareholders

special rights of the shares of the class so as to affect them adversely.

meeting)).

Shareholders' Votes on Certain Transactions

Approval of Mergers and Acquisitions Generally

- Generally, under Delaware law, unless the certificate
 of incorporation provides for the vote of a larger
 portion of the stock, completion of a merger or
 consolidation of substantially all of a corporation's
 assets or dissolution requires:
 - (1) the approval of the board of directors; and
 - (2) approvals by the vote of the holders of a majority of the outstanding stock or, if the certificate of incorporation provides for more or less than one vote per share, a majority of the votes of the outstanding stock of a corporation entitled to vote on the matter.
- The KW Amended and Restated Articles do not provide for a higher vote for certain transactions than a majority vote of the KW Shareholders.

- Generally, the power to approve an acquisition or the sale or disposal of all or substantially all of KWE's assets is vested in the KWE Directors.
- However, under the Jersey Companies Law:
 - there is a statutory merger process by which a Jersey company can merge with another Jersey company or with an overseas company which is permitted by its jurisdiction of incorporation to do so. Subject to certain exceptions, the shareholders of each merging company must pass a special resolution approving a written merger agreement setting out the terms of the merger. The approval of the directors, the Registrar of Companies and, in some cases, the Jersey Financial Services Commission and the Court, is also required;
 - (2) in a "scheme of arrangement", the company would make an initial application to the Court to convene a meeting or meetings of its shareholders at which a majority in number of shareholders representing three-quarters of the voting rights of the shareholders present and voting either in person or by proxy at the meeting must agree to the arrangement by which they will sell their shares in exchange for the consideration being offered by the bidder; and
- KWE is also subject to the rules of the Code, which are a set of general principles and rules regulating the conduct of UK public takeovers, the primary purpose of which is to ensure protection of offeree company shareholders.
- The Listing Rules impose certain restrictions on companies with a premium listing (such as KWE) that wish to enter into transactions with or for the benefit of a "related party", other than transactions in the ordinary course of business.
 - For the purposes of the Listing Rules, "related party" includes (i) a person who is (or was within the twelve months before the date of the transaction) a shareholder who is entitled to exercise, or control the exercise of, 10% or more of the votes able to be cast on all or substantially all matters at general meetings of the company, (ii) a person who is (or was within the twelve months before

Related Party Transactions

- Under the rules of the NYSE, shareholder approval is required prior to the issuance of common stock, or of securities convertible into or exercisable for common stock, in any transaction or series of related transactions, to:
 - (1) a director, officer or substantial security holder of the company, referred to as a "related party;"
 - (2) a subsidiary, affiliate or other closely-related person of a related party; or
 - (3) any company or entity in which a related party has a substantial direct or indirect interest; if the number of shares of common stock to be issued, or if the number of shares of common

stock into which the securities may be convertible or exercisable, exceeds either one percent of the number of shares of common stock or one percent of the voting power outstanding before the issuance.

However, if the related party involved in the transaction is classified as such solely because such person is a substantial security holder, and if the issuance relates to a sale of stock for cash at a price at least as great as each of the book and market value of the issuer's common stock, then shareholder approval will not be required unless the number of shares of common stock to be issued, or unless the number of shares of common stock into which the securities may be convertible or exercisable, exceeds either five percent of the number of shares of common stock or five percent of the voting power outstanding before the issuance.

 Personal loans extended from or arranged by a listed company to directors and executive officers are prohibited under US securities laws

the date of the transaction) a director or shadow director of the company and (iii) a person or entity exercising significant influence over the company. As regards KWE, KW and its associates (including the Manager) would be "related parties".

- Transactions with or for the benefit of "related parties" are tested by reference to certain financial metrics known as the "class tests", being assets, profits, consideration and gross capital. Where the percentage ratio arrived at on the calculation of any one of the class tests is 5% or more, the company will usually need to (i) make a public announcement containing prescribed details of the transaction and the related party, (ii) send an explanatory circular to its shareholders and (iii) obtain shareholder approval for the transaction. For the purposes of the class tests, where a listed company enters into transactions with the same related party in any 12-month period, and the transactions have not received shareholder approval, those transactions must be aggregated.
- where each of the percentage ratios arrived at on the calculation of the class tests is less than 5% but one or more exceeds 0.25%, the usual requirements for a transaction with a related party are modified. In such cases, before entering into the transaction, the listed company must (i) obtain the written confirmation of a financial adviser, confirming that the terms of the proposed transaction are fair and reasonable as far as the shareholders of the company are concerned and (ii) make a public announcement containing certain details of the transaction and the related party.

Greater than 20 Percent Change in Common Shares or Voting Power

- Under the rules of the NYSE, stockholder approval is required prior to the issuance of common stock, or of securities convertible into or exercisable for common stock, in any transaction or series of related transactions if:
 - the common stock has, or will have upon issuance, voting power equal to or in excess of 20 percent of the voting power outstanding before the issuance of such stock or of securities convertible into or exercisable for common stock; or
 - (2) the number of shares of common stock to be issued is, or will be upon issuance, equal to or in excess of 20 percent of the number of shares of common stock outstanding before the issuance of the common stock or of securities convertible into or exercisable for common stock.

The same requirement does not apply to KWE.

Rights of Inspection

Rights of Inspection Generally

- Delaware law allows any stockholder in person or by attorney or other agent, upon written demand under oath stating the purpose thereof, during the usual hours for business to inspect for any proper purpose, and to make copies and extracts from:
 - the corporation's stock ledger, a list of its stockholders, and its other books and records; and
 - (2) a subsidiary's books and records, to the extent that:
 - (A) the corporation has actual possession and control of such records of such subsidiary; or
 - (B) the corporation could obtain such records through the exercise of control over such subsidiary, provided that as of the date of the making of the demand:
 - (i) the stockholder inspection of such books and records of the subsidiary would not constitute a breach of an agreement between the corporation or the subsidiary and a person or persons not affiliated with the corporation; and
 - (ii) the subsidiary would not have the right under the law applicable to it to deny the corporation access to such books and records upon demand by the corporation.

Stock List

- The KW Bylaws state that a complete list of the KW Shareholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each KW Shareholder and the number of shares registered in the name of each KW Shareholder shall be open to the examination of any KW Shareholder, for any purpose germane to the shareholder meeting, during ordinary business hours, for a period of at least 10 days prior to the meeting, (a) on a reasonably accessible electronic network; and (b) during ordinary business hours in the principal place of business of KW. If the meeting is to be held at a place the list shall also be produced and kept at the time and place of the meeting and may be inspected by any KW Shareholder who is present. If the meeting is to be held solely by means of remote communication, the list shall be available during the whole time of the meeting on a reasonably accessible electronic network.
- For the avoidance of doubt, holders of KW CDIs and the relevant details in relation to such holders shall not appear on the list described above. Rather, the depositary that is the registered holder of the KW Shares represented by the KW CDIs and its relevant details will be included.

- Under the Jersey Companies Law, the Register is open to inspection during business hours. Inspection is free for KWE Shareholders but KWE may charge a nominal fee to any other person who wishes to inspect the Register.
- In addition, it is open to anyone to request a copy of the Register on payment of a fee and on delivery to KWE of a declaration under oath confirming for which of the limited purposes stated in the Jersey Companies Law the copy Register will be used.
- The Jersey Companies Law also provides that the register of directors of KWE must be open for inspection for at least two hours on each business day. Inspection is free of charge to a KWE Shareholder or a KWE Director. The register of directors of KWE and any Jersey company which is its subsidiary is also open to inspection by anyone else on payment of a nominal fee to KWE.
- The accounting records of KWE must be open at all times to inspection by its directors, secretary and liquidator (if any). KWE Shareholders (as such) do not have a right to inspect the company's accounting records unless authorised by the KWE Directors or by an ordinary resolution passed by the KWE Shareholders.
- The Register contains details of the name and address of each KWE Shareholder and number of shares registered in their name.
- As noted above, under the Jersey Companies
 Law, the Register is open to inspection
 during business hours. Inspection is free for
 KWE Shareholders but KWE may charge a
 nominal fee to any other person who wishes
 to inspect the Register.
- In addition, it is open to anyone to request a copy of the Register on payment of a fee and on delivery to the company of a declaration under oath confirming for which of the limited purposes stated in the Jersey Companies Law the copy Register will be used.

Standard of Conduct for Directors

- The Delaware General Corporation Law does not contain any specific provisions setting forth the standard of conduct of a director. The scope of the fiduciary duties of the Board is thus determined by Delaware common law. In general, directors have a duty to act in good faith, on an informed basis and in a manner they reasonably believe to be in the best interests of the stockholders.
- Delaware law provides that a contract or a transaction in which one or more directors has an interest shall not be void or voidable solely because such director was present at or participated in the meeting approving such transaction if:
 - (1) The material facts of such director's interest were disclosed or known to the board of directors and the board of directors authorised the transaction in good faith by the affirmative vote of a majority of disinterested directors;
 - (2) The material facts of the director's interest are disclosed or known to the stockholders and the transaction is approved in good faith by the stockholders; or
 - (3) The transaction is fair as to the corporation at the time it is authorised by the board of directors or stockholders.
- The KW Bylaws state that the number of directors of the corporation shall be fixed from time to time by the KW Board but it shall not be less than 1 or more than 9. The Board currently consists of 7 directors, one of whom is also a KWE Director.

- The Jersey Companies Law requires a director to (i) act honestly and in good faith with a view to the best interests of the company (a subjective standard); and (ii) exercise the care, diligence and skill that a reasonably prudent person would exercise in comparable circumstances (an objective standard). This principal statutory duty is supplemented by certain customary law duties, in particular to act in good faith, to exercise the powers of a director for a proper purpose, to avoid a conflict of duty and interest and to account for profits from opportunities arising from his or her directorship.
- The KWE Directors also remain under an obligation to conduct the business of KWE in accordance with its memorandum and the KWE Articles.
- The KWE Articles state that KWE may by ordinary resolution determine the maximum and minimum number of directors and that, unless and until otherwise so determined, and subject to the provisions of the Jersey Companies Law, the minimum number of KWE Directors shall be two. Pursuant to the KWE Articles, a majority of the KWE Directors must be tax resident in Jersey and must not be resident in, or citizens of, the United States of America.
- The UK Corporate Governance Code (UKCGC), published by the Financial Reporting Council (FRC), is the primary governance code in the UK, and applies to companies with a premium listing of equity shares on the London Stock Exchange, regardless of whether they are incorporated in the UK or elsewhere. The UKCGC sets out standards of good practice for listed companies on board composition and development, remuneration, shareholder relations, accountability and audit.

Classification of the Board of Directors

- Delaware law permits the certificate of incorporation or a stockholder- adopted bylaw to provide that directors be divided into one, two or three classes, with the term of office of one class of directors to expire each year.
- The KW Amended and Restated Articles provide that the Board will be divided into three classes of directors, each class is elected to serve for a term of three years, with the term of only one class expiring every year.
- The KWE board of directors is not classified.

Removal of Directors

- Delaware law provides that a director or the entire board of directors may be removed with or without cause by the holders of a majority of the shares entitled to vote at an election of directors, except that:
 - shareholders of a classified board of directors may be removed only for cause, unless the certificate of incorporation provides otherwise; and
 - (2) directors may not be removed in certain situations in the case of a corporation having cumulative voting.
- Under the KW Bylaws, except as otherwise provided under Delaware law, the entire KW Board or any individual KW Director may be removed from office by the vote of the holders of a majority in voting power of the KW Shares issued and outstanding and then entitled to vote at an election of directors.

Under the KWE Articles, the office of a KWE Director shall be vacated if he or she:

- resigns his or her office by written notice to KWE;
- ceases to be a KWE Director by virtue of any provision of the Jersey Companies Law or he or she becomes prohibited or disqualified by law from being a director;
- becomes bankrupt or makes any arrangement or composition with his or her creditors generally;
- becomes of unsound mind;
- subsequent to his or her appointment becomes resident outside Jersey and would cause a majority of KWE Directors not be tax resident in Jersey;
- subsequent to his appointment becomes resident in, or a citizen of, the United States so as to cause a majority of KWE Directors to be resident in, or citizens of, the United States; or
- is removed from office by ordinary resolution.

Vacancies on the Board of Directors

Vacancies, Generally

- Under Delaware law, unless otherwise provided in the certificate of incorporation or the bylaws:
 - (1) vacancies on a board of directors; and
 - (2) newly created directorships resulting from an increase in the number of directors,

may, in each case, be filled by a majority of the directors in office, although less than a quorum, or by a sole remaining director. In the case of a classified board, directors elected to fill vacancies or newly created directorships will hold office until the next election of the class for which the directors have been chosen. If, at the time of filling any vacancy or any newly created directorship, the directors then in office shall constitute less than a majority of the whole board, the Court of Chancery may, upon application of any stockholder or stockholders holding at least 10 percent of the voting stock at the time outstanding having the right to vote for such directors, summarily order an election to be held to fill any such vacancies or newly created directorships, or to replace the directors chosen by the directors then in office.

• Under Delaware law, unless otherwise provided in the certificate of incorporation or the bylaws, when one or more directors shall resign from the board, effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation

- Under the KWE Articles, the KWE Directors have the power at any time to appoint any natural person (other than one disqualified or ineligible by law or under the KWE Articles to act as a director) to be a KWE Director either (i) to fill a casual vacancy or (ii) as an addition to the existing KWE Directors, provided that the appointment does not cause the number of KWE Directors to exceed any number fixed by or in accordance with the KWE Articles as the maximum number of directors.
- Pursuant to the KWE Articles, the KWE Shareholders may, by ordinary resolution, appoint any natural person (other than one disqualified or ineligible by law or under the KWE Articles to act as a director) to be a KWE Director.

or resignations shall become effective, and each director so chosen shall hold office as provided in this section in the filling of other vacancies.

- The KW Bylaws provide that:
 - (1) any vacancies on the Board; or
 - (2) newly created directorships,

shall be filled by the majority of the remaining directors, although less than a quorum, or by a sole remaining director.

Term of Service After Appointment to Fill a Vacancy

- The KW Bylaws further provide that any director appointed to fill a vacancy shall serve for the remainder of the then present term of office of the class to which he or she was appointed.
- Generally, once appointed, a KWE Director remains in office until he vacates his position in accordance with the KWE Articles (see "Removal of Directors" above).
- However, the KWE Articles provide that if any resolution(s) for the election or reelection of directors are put to a KWE Annual General Meeting but at the end of the meeting the number of directors is fewer than the minimum number required, all retiring KWE Directors who stood for reelection shall be deemed to have been reelected and shall remain in office but they may only (i) act for the purpose of filling vacancies and convening KWE general meetings, and (ii) perform such duties as are appropriate to maintain KWE as a going concern and comply with the company's legal and regulatory obligations.

Impact on Classification of the Board

- The KW Bylaws also provide that in the event of any change in the authorised number of directors, the newly created or eliminated directorships shall be subject to the same provisions as apply to vacancies generally as summarised above.
- The KWE board of directors is not classified.

Liability of Directors and Officers

- Delaware law permits a corporation's certificate of incorporation to include a provision eliminating or limiting the personal liability of a director to the corporation and its stockholders for damages arising from a breach of fiduciary duty as a director. However, no provision can limit the liability of a director for:
 - any breach of his or her duty of loyalty to the corporation or its stockholders;
 - acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law;
 - intentional or negligent payment of unlawful dividends or stock purchases or redemptions;
 - (4) any transaction from which he or she derives an improper personal benefit.
- The KW Amended and Restated Articles provide that a director of the company will not be personally liable to KW or the KW Shareholders for monetary

- The exculpation of directors is not a standard provision in a Jersey company's articles of association and no such provision is contained in the KWE Articles. The circumstances in which directors can be indemnified in Jersey are more restricted than in Delaware in that a Jersey company may not exempt or indemnify a director from general liabilities. However, this does not prevent a company from purchasing directors' and officers' insurance for such liabilities.
- In addition, the Jersey Companies Law enables a Jersey court to excuse a director from liability for negligence, default, breach of duty or breach of trust where the court is satisfied that the director acted honestly and that, having regard to all the circumstances of the case (including those connected with his or her appointment), he or she ought fairly to be excused. The director may apply for relief either during actual proceedings or

damages for breach of fiduciary duty as a director. However, a director will be liable (i) for any breach of the director's duty of loyalty, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) pursuant to Section 174 of the Delaware General Corporation Law, or (iv) for any transaction from which such director derived an improper personal benefit.

Provisions currently applicable to KWE Shareholders

where he or she has reason to believe that a claim will or might be made against him or her

- Separately, the Jersey Companies Law provides statutory relief for a breach of a director's statutory duty to act honestly and in good faith with a view to the best interests of the company and to exercise the care, diligence and skill that a reasonably prudent person would exercise in comparable circumstances. No act or omission of a director shall be treated as a breach of this duty if:
 - (1) the relevant shareholders of the company authorise or ratify the act or omission in accordance with the Jersey Companies Law; and
 - (2) after the act or omission, the company is able to discharge its liabilities as they fall due.
- However, this provision will only relieve the director of liability for a breach of the statutory duty in so far as the shareholders of the company are concerned; it will not otherwise relieve the director from liability to third parties or for any other breach of the law and the provision does not prevent the transaction potentially being challenged by a liquidator on any subsequent insolvency of the company.
- Shareholder ratification is not permitted where it would amount to an expropriation to the majority or would allow the majority to oppress the minority.

Indemnification of Directors and Officers

- Delaware law provides that a corporation may indemnify a person who is made a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding on account of being a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding if the person:
 - (1) acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the corporation; and
 - (2) in a criminal proceeding, had no reasonable cause to believe his or her conduct was unlawful.
- The KW Bylaws provide that the company will indemnify, to the fullest extent permitted by Delaware law, members of the Board and officers of the company from and against all expenses, liabilities or losses reasonably incurred or suffered if the person

- Under the Jersey Companies Law, directors and liquidators may not be indemnified by a company against any liability which by law would otherwise attach to them as directors and liquidators under Jersey law, except that they may be indemnified against:
 - any liabilities incurred in defending certain civil or criminal proceedings;
 - (2) any liability incurred otherwise than to the company if they acted in good faith with a view to the best interests of the company;
 - (3) any liability incurred in connection with a successful application by them to the Royal Court of Jersey for relief under the Jersey Companies Law; and
 - (4) any liability against which the company normally maintains insurance for persons other than directors.
- The KWE Articles provide that every present or former director, liquidator and company secretary of KWE shall, in so far

acted in good faith and in a manner they reasonably believed to be in, or not opposed to the best interests of, KW. Such determination shall be made:

- (a) By a majority vote of the KW Directors who were not parties to the relevant action, suit or proceeding, even though less than a quorum;
- (b) By a committee of such KW Directors designated by a majority of such KW Directors, even though less than a quorum;
- (c) If there are no such KW Directors, or if such KW Directors so direct, by independent legal counsel in a written opinion; or
- (d) By the KW Shareholders.
- Further, KW may purchase and maintain insurance to effect indemnification.

Provisions currently applicable to KWE Shareholders

as the Jersey Companies Law allows, be indemnified out of KWE's assets against any loss or liability incurred by him or her by reason of holding such office.

- KWE may, subject to the Jersey Companies Law, also advance funds to enable any such officer to defend a claim brought against him or her in that capacity.
- As noted above, KWE is permitted to purchase directors' and officers' insurance.

Shareholders' Suits

- Under Delaware law, a stockholder may initiate a derivative action to enforce a right of a corporation if the corporation fails to enforce the right itself. The complaint must:
 - (1) state that the plaintiff was a stockholder at the time of the transaction of which the plaintiff complains or that the plaintiff's shares thereafter devolved on the plaintiff by operation of law; and
 - (2) (A) allege with particularity the efforts made by the plaintiff to obtain the action the plaintiff desires from the directors; or (B) state the reasons for the plaintiff's failure to obtain the action or for not making the effort.
- Additionally, the plaintiff must remain a stockholder through the duration of the derivative suit. The action will not be dismissed or compromised without the approval of the Delaware Court of Chancery.
- An individual may also commence a class action suit on behalf of himself or herself and other similarly situated stockholders where the requirements for maintaining a class action under Delaware law have been met.

- As a general rule, in an action for a wrong done to a company the proper claimant is the company itself. One of the exceptions to this general rule is a derivative action, which allows a shareholder to bring a claim on the company's behalf where the persons against whom relief is sought hold and control the majority of the shares and will not permit an action to be brought in the company's name and:
 - (1) the act complained of is ultra vires by the company or illegal;
 - (2) the act complained of constitutes a fraud against the minority of the shareholders and the wrongdoers control the company;
 - (3) there is an irregularity in respect of which a qualified majority of votes is required; or
 - (4) the act complained of infringes the personal rights of an individual shareholder.
- As noted above, the Jersey Companies Law gives an aggrieved shareholder the ability to apply to the Court for relief on the ground that the company's affairs are being conducted in a manner which is unfairly prejudicial to the interests of some or all of its shareholders. In these circumstances, the court may, among other things, authorise civil proceedings to be brought in the name and on behalf of the company.

Share Acquisitions

- Section 203 of the Delaware General Corporation Law prohibits business combinations with interested stockholders. A corporation shall not engage in any business combination with any interested stockholder for a period of three years following the time that such stockholder became an interested stockholder,
- KWE is not subject to any comparable provision. However, please also see the section on "Related Party Transactions" above.

unless:

- (1) Prior to such time the board of directors of the corporation approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder;
- (2) Upon the consummation of the transaction which resulted in the stockholder becoming an stockholder, the interested stockholder held at least 85 percent of the voting stock of the corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned (i) by persons who are directors and also officers and (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- (3) At or subsequent to such time the business combination was approved by the board of directors and authorised at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least 66 2/3 percent of the outstanding voting stock which is not owned by the interested stockholder.
- KW has not expressly elected not to be governed by Section 203 of the Delaware General Corporation Law, and therefore the provisions of such section apply to KW.

Mandatory Offers

No comparable provision.

- A bid is required to be made for KWE under Rule 9 of the Code, broadly where: the bidder and any persons acting in concert acquire shares carrying 30% or more of the voting rights of KWE; or if a bidder, together with any concert parties, holding not less than 30% but not more than 50% of the voting rights in KWE increases its holding.
- Rule 9 requires a mandatory offer to be made in cash (or be accompanied by a cash alternative) and at the highest price paid by the bidder or any concert party for any interest in shares of the relevant class during the 12 months prior to the announcement of the offer.

Anti-Takeover Matters

- A Delaware court will generally uphold board of director decisions to adopt anti-takeover measures, such as shareholder rights plans, in the face of a potential takeover where the directors are able to show that:
- Jersey law does not expressly prohibit antitakeover measures.
- However, in approving or implementing any such measures, the KWE Directors would need to comply with their statutory and

Provisions currently applicable to KWE Shareholders

- they had reasonable grounds for believing that there was a danger to corporate policy and effectiveness from an acquisition proposal; and
- (2) the board action taken was reasonable in relation to the threat posed.

fiduciary duties, including the need to act in the best interests of KWE Shareholders.

• Additionally, KWE is subject to the rules of the Code, which are a set of general principles and rules regulating the conduct of UK public takeovers, since its securities are admitted to trading on a UK regulated market.

Disclosure of Interests

- Certain acquisitions of KW Shares may require disclosure under the US Exchange Act on Schedule 13D. Some acquisitions, however, may qualify for a short-form disclosure on Schedule 13G. Generally, an acquisition of more than a five percent interest in a US publicly-held issuer by:
 - certain types of persons, including a brokerdealer, a bank, an insurance company, an investment company and an investment advisor, or
 - (2) a "passive investor" who is not seeking to acquire or influence control of the issuer, so long as the investor owns less than 20 percent of the class of stock it is acquiring.

May be disclosed on a Schedule 13G.

- A buyer who files a Schedule 13G must amend it periodically:
 - (1) to report any change in the information previously reported; or
 - (2) if it acquires more than 10 percent of the class of stock and, thereafter, if it undergoes any change in ownership of five percent or more of the class of stock.
- KW is required by the rules of the SEC to disclose in the proxy statement relating to its annual meeting of stockholders the number of shares of KW voting securities beneficially owned by:
 - (1) each of its directors;
 - (2) its principal executive officer;
 - (3) its principal financial officer;
 - (4) each of its four most highly compensated executive officers other than its principal executive officer and its principal financial officer:
 - (5) all of its directors and executives as a group; and
 - (6) any beneficial owner of 5 per cent. or more of the KWH voting securities of which KWH is aware.

- The Jersey Companies Law does not provide a mechanism for a company to investigate the identity of those with interests in its shares
- However, the KWE Articles give the KWE Directors the power to require any KWE Shareholder to disclose to KWE the identity of any person who, in the previous three years, has/had any direct or indirect interest in the shares held by that shareholder, and the nature of such interest. The KWE Directors are obliged to exercise this power if required to do so by KWE Shareholders holding at least one-tenth of the voting rights in KWE.
- The KWE Articles also contain sanctions which, if the KWE board of directors so determines, apply to KWE Shareholders who fail to disclose interests in shares. These sanctions include the suspension of voting and other rights relating to meetings in the relevant shares and, where the relevant shares represent at least 0.25% of their class, the suspension of dividends, the right to receive the proceeds of a repurchase, redemption or repayment on shares, and share transfer rights.
- The KWE Articles also require any KWE Shareholder who acquires 5% or more of any class of KWE shares to notify KWE of his interest and any changes thereto.
- KWE is subject to the rules of the Code. The Code imposes obligations to disclose any dealings in offeree or offeror shares during an "offer period" (as defined in the Code).

Proxy Statements and Reports

Approval of Director Compensation

- KW Shareholders generally do not have the right to approve directors' compensation; however, KW is subject to SEC reporting requirements for director and executive officer compensation and shareholder
- The KWE Directors are entitled to such remuneration as they may determine, but subject to any limitation as the KWE Shareholders may, by way of ordinary

Provisions currently applicable to KWE Shareholders

non-binding advisory votes to approve named executive officer compensation and to determine the frequency of such non-binding advisory votes.

resolution, specify.

Notices

- The KW Bylaws provide that notice may be given by any means permitted by law including, without limitation, electronic transmission consented to by the KW Shareholder to whom the notice is given.
- The KWE Articles provide that whenever notice is required to be given to any shareholder, such notice may be given in writing:
 - (1) personally;
 - (2) by post, addressed to such shareholder, at his or her address as it appears on the records of KWE, with postage thereon prepaid; or
 - (3) in any other manner agreed in advance between the company and that shareholder receiving notice other than personally or by post as set out above.

Reporting Requirements

- As a US public company, KW must file with the SEC, among other reports and notices:
 - an Annual Report on Form 10-K within 60 days after the end of a fiscal year;
 - (2) a Quarterly Report on Form 10-Q within 40 days after the end of a fiscal quarter ending; and
 - (3) Current Reports on Form 8-K upon the occurrence of certain important corporate events. Unless otherwise specified, a report is to be filed or furnished within four business days after occurrence of the event.
- Since KWE Shares are listed on the London Stock Exchange, KWE Shares are subject to the regulatory requirements and ongoing reporting obligations imposed on companies listed on the Main Market of the London Stock Exchange. These obligations include a range of requirements, such as ensuring that information which might affect the share price is made available to all investors at the same time, and that financial results are released on a timely basis in an acceptable format.
- As a Jersey public company, KWE is also required to publish annual statements of its shareholders and directors, and its annual accounts, via the Jersey Companies Registry.

PART ELEVEN

DEFINITIONS

The following definitions apply throughout this document unless the context requires otherwise:

"£", "pounds sterling" or "pence" the lawful currency of the UK;

"\$", "US\$", "US Dollars" or the lawful currency of the United States;

"cent"

"2022 Bonds" has the meaning given in paragraph 13.1(f) of Part Eight

(Additional Information) of this document;

"2024 Notes" has the meaning given in paragraph 13.2(e) of Part Eight

(Additional Information) of this document;

"2025 Notes" has the meaning given in paragraph 13.1(g) of Part Eight

(Additional Information) of this document;

"2042 Notes" has the meaning given in paragraph 13.2(f) of Part Eight

(Additional Information) of this document;

"5% Holder" has the meaning given in the paragraph entitled "US Tax

Deductions" in the Section entitled "Action to be Taken" of this

document;

"Additional Cash Election" has the meaning given in the Scheme;

"Additional Notes" has the meaning given in paragraph 13.2(e) of Part Eight

(Additional Information) of this document;

"Additional Share Election" has the meaning given in the Scheme;

"Adjusted Final KWE Dividend" the amount resulting from the application of the formula in Clause

4.1 of the Scheme after deducting the value of "A" in such formula (being the amount of the Penultimate KWE Dividend per KWE

Share);

"Adjusted NAV" the IFRS net asset value adjusted (i) to include properties and other

investment interests at fair value, (ii) to exclude certain items not expected to be crystallised in normal circumstances (such as fair value of financial derivatives) and (iii) by deducting any management fee and performance fee accounted for in the share-

based payment reserve;

"Affiliate" means any entity under the control of another or under common

control, but KWE and its subsidiary undertakings are not, for this

purpose, Affiliates of KW;

"Applicable Exchange Ratio" the number of New KW Shares to be received pursuant to the

Scheme by a Scheme Shareholder in respect of one Scheme Share, calculated in accordance with such Scheme Shareholder's entitlements under the Original Offer or, as applicable, the New Offer and, if applicable, after satisfying elections under the Mix and Match Facility to the extent permissible in accordance with its

terms;

"Applicable Translation Rate" the number of US Dollars for each pound sterling under the US\$ / £

spot exchange rate derived from Bloomberg as at 5.00 p.m. on the Business Day immediately prior to the First KW Dividend

Payment Date;

"Authorisations" has the meaning set out in Part A of Part Three (Conditions and

Certain Further Terms of the Scheme) of this document;

"BLUE Form of Proxy" the Form of Proxy to be used in connection with the Scheme Court

Meeting;

"BofA Merrill Lynch" Merrill Lynch International and Merrill Lynch, Pierce, Fenner &

Smith Incorporated;

"Brexit" the outcome of the referendum on 23 June 2016 as to the withdrawal of the UK from membership of the European Union; "Business Day" a calendar day, other than a Saturday, Sunday or public or bank holiday, when banks are open for business in London, Jersey and New York: "CBRE" CBRE Limited, an MRICS qualified and registered valuer with the RICS, Henrietta House, Henrietta Place, London, W1G 0NB; "CCPC" the Irish Competition and Consumer Protection Commission; "CDI" a CREST depository interest representing an entitlement to a share; "certificated" or "in certificated in relation to a share or other security, means a share or other form" security which is not in uncertificated form (i.e. not in CREST); "Closing Dividend" has the meaning given in the Scheme; "Closing Dividend Record Time" the Scheme Record Time: "Closing Price" the closing middle market quotation of a share derived from the Daily Official List (in the case of KWE Shares) or the NYSE (in the case of KW Shares); "Code" the UK City Code on Takeovers and Mergers, as issued from time to time by the Panel; "Colliers" Colliers Real Estate Services Italia SRL, an MRICS qualified and registered valuer with the RICS, Via Durini 2, 20122 Milan, Italy; "Combined Group" KW and its subsidiary undertakings following the completion of the Merger, comprising the KW Group and the KWE Group; "Conditions" the conditions to the implementation of the Merger (including the Scheme) as set out in Part A of Part Three (Conditions and Certain Further Terms of the Scheme) of this document; "Confidentiality Agreement" the confidentiality agreement between KW and KWE, dated 23 March 2017; the consideration payable to Scheme Shareholders pursuant to the "Consideration" terms of the Scheme; "Court" the Royal Court of Jersey; "Court Order" the Act of the Court sanctioning the Scheme under Article 125 of the Jersey Companies Law; "Covered Information" has the meaning given in "Important Notice" commencing on page 4 of this document; "CREST" the computer system (as defined in the Uncertificated Securities Order) in respect of which Euroclear UK & Ireland Limited is the authorised operator (as defined in such order) in accordance with which securities may be held and transferred in uncertificated form; "CREST Manual" the CREST manual referred to in agreements entered into by Euroclear; "CREST Proxy Instruction" the appropriate CREST message properly authenticated in accordance with Euroclear's specifications and which contains the information required for such instructions, as described in the CREST Manual; "CREST Receiving Agent" Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom, in its capacity as CREST receiving agent; "Daily Official List" the daily official list of the London Stock Exchange;

an announcement pursuant to Rule 8 of the Code containing details

of dealings in relevant securities of a party to an offer;

"Dealing Disclosure"

"Deemed Contribution"

has the meaning given in paragraph 2.1(4) of Part Seven (*Taxation*) of this document;

"Deemed Liquidating Distribution"

has the meaning given in paragraph 2.1(4) of Part Seven (*Taxation*) of this document:

"Deemed Sale"

has the meaning given in paragraph 2.1(4) of Part Seven (*Taxation*) of this document;

"Deutsche Bank"

Deutsche Bank AG, acting through its London Branch, and Deutsche Bank Securities Inc.:

"Disclosed"

- (i) the information fairly disclosed by, or on behalf of, KWE:
 - a) in the Annual Report and Accounts of the KWE Group for the financial year ended 31 December 2016; or
 - b) in the Original Announcement; or
 - c) in this document; or
 - d) in any other public announcement made by KWE in accordance with the Market Abuse Regulation, the Listing Rules and the Disclosure and Transparency Rules after 31 December 2016 but prior to the Original Announcement Date; or
 - e) prior to the Original Announcement Date by or on behalf of KWE to KW (or its respective officers, advisers, employees or agents in their capacity as such); and
- (ii) all the information held by the Manager or within the actual knowledge of the Manager and its directors and employees in relation to the Wider KWE Group as at the Original Announcement Date;

"Disclosure and Transparency Rules"

the Disclosure Guidance and Transparency Rules made by the UKLA under the Financial Services and Markets Act 2000 and contained in the FCA's publication of the same name;

"DTC"

the Depository Trust Company;

"Effective Date"

the date upon which the Scheme becomes effective in accordance with its terms;

"Election"

a Mix and Match Election or an Original Offer Election;

"Election Form"

the form of Election which accompanies this document;

"Election Return Time"

the latest time for lodging an Election Form or making an Electronic Election either under the Mix and Match Facility or for the Original Offer, being 1.00 p.m. on 19 October 2017 (or such later time and/or date as KWE and KW may agree and KWE may announce through a Regulatory Information Service);

"Electronic Election"

an Election validly made by a Scheme Shareholder who holds Scheme Shares in uncertificated form immediately prior to the Election Return Time;

"Electronic Mix and Match Election"

a Mix and Match Election validly made by a Scheme Shareholder in respect of Scheme Shares held in uncertificated form when the Mix and Match Election is submitted;

"Electronic Original Offer Election"

an Original Offer Election validly made by a Scheme Shareholder in respect of Scheme Shares held in uncertificated form when the Original Offer Election is submitted;

"EMTN Programme"

has the meaning given in paragraph 13.1(g) of Part Eight (Additional Information) of this document;

"Encumbrance" any lien, equity, charge, hypothec, encumbrance, option, right of pre-emption and any other third party right and/or interest of any nature: "Euroclear" Euroclear UK & Ireland Limited, a company incorporated under the laws of England and Wales with registered number 02878738; "Excluded Share" at any relevant time, a KWE Share beneficially owned at that time by, or on behalf of, KW, the Manager or any other member of the KW Group (excluding, for the avoidance of doubt, any KWE Shares so held at the Latest Practicable Date after they have been transferred by any member of the KW Group to any director, officer or employee of any member of the KW Group prior to the Scheme Record Time pursuant to the vesting of RSUs in connection with the Merger) and any KWE Shares held by KWE in treasury; "Extended Period Projections" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "FATCA" has the meaning given in paragraph 2.3 of Part Seven (*Taxation*) of this document: "FATCA Withholding" has the meaning given in paragraph 2.3 of Part Seven (Taxation) of this document; "FCA" the UK Financial Conduct Authority; "Final KWE Dividend" an amount of 12 pence per KWE Share; "Final KWE Dividend Period" the KWE Dividend Period in which the Effective Date falls; "First KW Dividend" the amount of the dividend per KW Share first paid by KW after the Effective Date in respect of the First KW Dividend Period or any part thereof; "First KW Dividend Payment the date on which the First KW Dividend is paid, which shall be Date" not later than 10 days after the expiration of the First KW Dividend Period: "First KW Dividend Period" the KW Dividend Period in which the Effective Date falls; "Form of Proxy" each form of proxy in connection with each of the Scheme Court Meeting and the KWE General Meeting, which accompany this document; "Goldman Sachs" Goldman, Sachs & Co. and Goldman Sachs International (in their capacity as financial advisers to KW); "GREEN Form of Proxy" Meeting; "HMRC"

the form of proxy for use in connection with the KWE General

Her Majesty's Revenue and Customs;

"IFRS" International Financial Reporting Standards;

"Indemnity Agreement" the indemnity agreement among KW, KWE, the Manager and the

KWE Independent Directors, dated 23 June 2017;

"Internal Revenue Code" the Internal Revenue Code of 1986, as amended;

"Investment Management the investment management agreement, dated 25 February 2014, Agreement" or "IMA" between KWE and the Manager relating to the management by the

Manager of KWE;

"IPO" initial public offering;

"Irish Competition Act" the Irish Competition Act 2002 (as amended);

"IRS" the Internal Revenue Service:

"Issuer" has the meaning given in paragraph 13.2(e) of Part Eight

(Additional Information) of this document;

"J.P. Morgan Cazenove" J.P. Morgan Limited (which conducts its UK investment banking business as J.P. Morgan Cazenove); "Jersey Companies Law" the Companies (Jersey) Law 1991, as amended from time to time; "Jersey Working Day" a calendar day (other than a Saturday, Sunday or public or bank holiday) when banks are open for business in Jersey; "KPMG" KPMG LLP, in its capacity as KW's reporting accountants; "KW" Kennedy-Wilson Holdings, Inc., a Delaware corporation; "KW 2017 Profit Forecast" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "KW Amended and Restated KW's amended and restated articles, adopted on 19 June 2014, as Articles" amended from time to time: "KW Board" the board of directors of KW as at the date of this document or, where the context so requires, the board of directors of KW from time to time: KW's amended and restated bylaws, adopted on 22 February 2017, "KW Bylaws" as amended from time to time; "KW Cash Component" 300 pence per Scheme Share payable in cash by KW pursuant to the New Offer; "KW CDI" a CREST depository interest representing an entitlement to a New KW Share; "KW Directors" the directors of KW as at the date of this document or, where the context so requires, the directors of KW from time to time; "KW Dividend Period" each period of three calendar months in the financial year of KW ending on (and including) 31 March, 30 June, 30 September and 31 December; "KW EPP" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "KW Group" KW, the Manager and any of their respective subsidiaries and subsidiary undertakings from time to time (excluding, for the avoidance of doubt, any member of the KWE Group); "KW Prospectus" the prospectus to be published by KW to Scheme Shareholders in connection with the New Offer: "KW Proxy Statement" the proxy statement to be sent by KW to KW Shareholders summarising the background to and reasons for the Merger, which will include a notice convening the KW Special Meeting; "KW Revolving Facility" has the meaning given in paragraph 13.2(d) of Part Eight (Additional Information) of this document; "KW Senior Notes" has the meaning given in paragraph 13.2(e) of Part Eight (Additional Information) of this document; "KW Shareholder" a registered holder of KW Shares from time to time; "KW Shareholder Resolution" a resolution for the approval of the issuance of New KW Shares in connection with the Merger by a majority of the votes cast on such resolution at the KW Special Meeting, as required by the rules of the NYSE; "KW Shares" KW shares of common stock of US\$0.0001 each; the special meeting of KW Shareholders to be convened in "KW Special Meeting" connection with the Merger, notice of which will be sent to KW Shareholders, including any adjournment thereof; "KW Transfer Agent" Continental Stock Transfer & Trust Company, One State Street

America;

Plaza, 30th Floor, New York, NY 10004-1561, United States of

"KWE" Kennedy Wilson Europe Real Estate plc, incorporated in Jersey with registered number 114680; "KWE 2017 Profit Forecast" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "KWE Articles" the articles of association of KWE; "KWE Directors" the KWE directors from time to time; "KWE Dividend Period" each period of three calendar months in the financial year of KWE ending on (and including) 31 March, 30 June, 30 September and 31 December; "KWE EPP" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "KWE General Meeting" the extraordinary general meeting of KWE to be convened in connection with the Scheme and the Merger, notice of which is set out in Part Thirteen (Notice of KWE General Meeting) of this document, including any adjournment thereof; "KWE Group" KWE, its subsidiaries and its subsidiary undertakings from time to time: "KWE Independent Directors" the independent directors of KWE from time to time; "KWE Independent Directors' has the meaning given in paragraph 30(d) of Part Two (Explanatory **Additional Remuneration** Statement) of this document; Agreement" "KWE Information" all information in this document relating to KWE, the KWE Group, the recommendation of the KWE Independent Directors, the KWE Directors (but only insofar as the KWE Independent Directors are concerned) and their immediate families, related trusts and connected persons and persons acting in concert (as defined in the Code) with KWE, and in each case including all information relating to the foregoing which has been incorporated by reference into this document; "KWE Meeting" each of the Scheme Court Meeting and the KWE General Meeting; "KWE Registrar" Computershare Investor Services (Jersey) Limited, Queensway House, Hilgrove Street, St Helier, Jersey, JE1 1ES, in its capacity as the KWE registrar; "KWE Shareholder" a registered holder of KWE Shares from time to time; "KWE Share" an ordinary share of no par value in the capital of KWE; the special distribution of 250 pence per Scheme Share payable in "KWE Special Distribution" cash by KWE pursuant to the New Offer in respect of each Scheme Share but not, for the avoidance of doubt, any other KWE Shares; the close of business on 8 September 2017, being the latest "Latest Practicable Date" practicable Business Day before the date of this document; "Less than 80% Test" has the meaning given in the paragraph entitled "US Tax Deductions" in the Section entitled "Action to be Taken" of this document: "Listing" the listing of the New KW Shares on the NYSE, expected to be effective on the Effective Date;

"Listing Rules" the listing rules made by the FCA pursuant to Part 6 of the Financial Services and Markets Act 2000, referred to in section 73A(2) of the same, and contained in the FCA's publication of the

same name;

"London Stock Exchange" the London Stock Exchange plc;

"Longstop Date" 30 November 2017 or such later date as may be agreed in writing by KWE and KW (with the Panel's consent and as the Court may approve (if such approval(s) are required)); "Main Market" the Main Market of the London Stock Exchange; "Manager" KW Investment Management Ltd, a company incorporated in Jersey with registered number 114679, a wholly owned subsidiary of KW, which is the manager to KWE; the Market Abuse Regulation (2014/596/EU); "Market Abuse Regulation" "Merger" the proposed acquisition of the entire issued and to be issued share capital of KWE by KW (other than the Excluded Shares), to be implemented by way of the Scheme or (should KW so elect, subject to the consent of the Panel) by way of the Takeover Offer; "Mix and Match Election" an election validly made by a Scheme Shareholder who is entitled to receive the consideration to be delivered by KW under the New Offer to vary such consideration to the extent permitted by the Mix and Match Facility; "Mix and Match Facility" the arrangement under which a Scheme Shareholder entitled to receive the consideration to be delivered by KW under the New Offer may elect to vary the proportion in which it receives, pursuant to the Scheme, the New KW Share Component and the KW Cash Component (but, for the avoidance of doubt, not the KWE Special Distribution or the Closing Dividend) to the extent permitted by the Mix and Match Facility and except to the extent otherwise provided pursuant to Clause 9 of the Scheme; "New KW Share Component" 0.3854 New KW Shares in respect of each Scheme Share; "New KW Share" a KW Share which is to be issued pursuant to the Scheme; "New Offer" the offer pursuant to which a Scheme Shareholder is entitled to receive the New Offer Consideration and Distribution; "New Offer Announcement" the announcement by KW of the New Offer on the New Offer Announcement Date; "New Offer Announcement Date" 13 June 2017, being the date on which the New Offer Announcement was released; "New Offer Consideration and the consideration and special distribution available pursuant to the Distribution" New Offer, consisting of the New KW Share Component, the KW Cash Component and the KWE Special Distribution in respect of each Scheme Share held at the Scheme Record Time, subject to any variation to the extent permitted by the terms of the Mix and Match Facility; "non-US Holder" has the meaning given in paragraph 2.2 of Part Seven (Taxation) of this document; "NYSE" the New York Stock Exchange; "Offer Period" the period commencing on the Original Announcement Date; "Official List" the official list maintained by the UKLA; "Opening Position Disclosure" an announcement containing details of interests or short positions in, or rights to subscribe for, any relevant securities of a party to the offer if the person concerned has such a position; "Ordinary Quarterly Dividend" each dividend for any KWE Dividend Period, declared, paid or effected by KWE in an amount not exceeding 12 pence per KWE Share in aggregate in respect of a single KWE Dividend Period; "Original Announcement" the original announcement of the Merger made in accordance with

was released;

"Original Announcement Date"

Rule 2.7 of the Code on the Original Announcement Date;

24 April 2017, being the date on which the Original Announcement

"Original Offer" the offer pursuant to which a Scheme Shareholder is entitled to receive the Original Offer Consideration in respect of each Scheme Share held at the Scheme Record Time except to the extent otherwise provided pursuant to Clause 9 of the Scheme; "Original Offer Consideration" the consideration available under the Original Offer, consisting of 0.667 New KW Shares in respect of each Scheme Share held at the Scheme Record Time; "Original Offer Election" an election validly made by a Scheme Shareholder to receive the Original Offer Consideration except to the extent otherwise provided pursuant to Clause 9 of the Scheme; "Overseas Shareholders" KWE Shareholders who are resident in, ordinarily resident in, or citizens or nationals of, or otherwise subject to the law of, jurisdictions other than Jersey, the United Kingdom, the United States, the Netherlands or the Cayman Islands; "Panel" the UK Panel on Takeovers and Mergers; "Penultimate KWE Dividend" a dividend of £0.12 per KWE Share to be paid by KWE in respect of the Penultimate KWE Dividend Period; "Penultimate KWE Dividend the KWE Dividend Period immediately preceding the Final KWE Period" Dividend Period: "Permitted KW Dividends" quarterly dividends in amounts not exceeding, in aggregate, US\$0.17 per KW Share with respect to any KW Dividend Period preceding the First KW Dividend Period and US\$0.19 per KW Share in respect of the First KW Dividend Period which, in any such case, are payable by reference to a record date preceding the Effective Date, and any dividend paid by KW on or after the Effective Date to which Scheme Shareholders are entitled pursuant to the Scheme; "Permitted KWE Distributions" the KWE Special Distribution, the Closing Dividend and all Ordinary Quarterly Dividends; "PFIC" passive foreign investment company; "Pre-Effective Date Proportion" the proportion which (a) the number of days from (and including) the first day of the Final KWE Dividend Period to (but excluding) the Effective Date bears to (b) the number of days from (and including) the first day of the Final KWE Dividend Period to (and including) the last day of the Final KWE Dividend Period; "Profit Forecast RIS the RIS announcement released by KW on 27 June 2017, Announcement" publishing, among other things, the Profit Forecasts; "Profit Forecasts" has the meaning given in section A of Part Fourteen (Profit Forecasts) of this document; "QEF" a qualified electing fund within the meaning of paragraph 19 of Part One (Letter from the Independent Directors of Kennedy Wilson Europe Real Estate Plc);

"OEF election"

a qualified electing fund election for US tax purposes;

"RCF" has the meaning given in paragraph 13.1(h) of Part Eight

(Additional Information) of this document;

"Receiving Agent" Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom, in its capacity as receiving agent;

"Register" the register of members of KWE; "Registrar of Companies" the registrar of companies for Jersey;

"Regulatory Conditions" has the meaning given in paragraph 30 of Part Two (Explanatory Statement) of this document;

"Regulatory Consents" "Regulatory Information Service"

"Ring Fenced Employees"

"Rothschild"

"RSG"

"RSUs"

"S&P"

"Scheme" or "Scheme of Arrangement"

"Scheme Court Meeting"

"Scheme Court Sanction Hearing"

"Scheme Document"

"Scheme Record Time"

"Scheme Share"

has the meaning given in paragraph 30 of Part Two (Explanatory Statement) of this document;

any information service authorised from time to time by the FCA for the purpose of disseminating regulatory announcements;

certain employees of members of the KW Group who have been instructed by KW to assist KWE and the KWE Independent Directors in connection with the Merger;

N M Rothschild & Sons Limited;

restricted stock grant;

restricted share units issued by KW to directors, officers and employees of members of the KW Group entitling them to receive KWE Shares;

Standard & Poor's Global Ratings

the scheme of arrangement proposed to be effected under Article 125 of the Jersey Companies Law among KWE and the Scheme Shareholders, with or subject to any modification, addition or condition approved or imposed by the Court and agreed to by KWE and KW and sanctioned by the Court at the Scheme Court Sanction Hearing;

the meeting or meetings of the Scheme Shareholders or any class or classes thereof to be convened by order of the Court pursuant to Article 125 of the Jersey Companies Law for the purpose of considering and, if thought fit, approving the Scheme and any adjournment, postponement or reconvention thereof;

the hearing of the Court (and any adjournment thereof) to sanction the Scheme pursuant to Article 125 of the Jersey Companies Law, at which the Court Order is expected to be granted;

the document in respect of the Scheme sent to (among others) KWE Shareholders, containing and setting out (among other matters) the full terms and conditions of the Scheme and containing the notices convening the KWE Meetings;

6.00 p.m. on the most recent Business Day before the Effective Date:

- a KWE Share, which is not an Excluded Share and which is:
- in issue as at the date of the Scheme Document; a)
- transferred by any member of the KW Group to any director, b) officer or employee of any member of the KW Group after the KWE General Meeting and prior to the Scheme Record Time pursuant to the vesting of RSUs in connection with the Merger;
- (if any) issued after the date of the Scheme Document and c) prior to the Voting Record Time; or
- d) (if any) issued on or after the Voting Record Time and before the Scheme Record Time, either on terms that the original or any subsequent holders thereof shall be bound by the Scheme or in respect of which the holders thereof shall have agreed in writing to be bound by the Scheme;

registered holders of Scheme Shares;

the Voting Record Time in relation to the Scheme Court Meeting; UK stamp duty reserve tax;

the US Securities and Exchange Commission;

"Scheme Shareholders"

"Scheme Voting Record Time"

"SDRT"

"SEC"

"Separate Manager's Powers Agreement"	the agreement between KWE and the Manager dated 10 April 2017 and containing certain undertakings by the Manager as regards the exercise of its powers over KWE and the KWE Group;
"Shareholder Helpline"	the helpline operated by the KWE Registrar and the Receiving Agent;
"Special Resolution"	the special resolution of which notice is set out in Part Thirteen (<i>Notice of KWE General Meeting</i>) of this document, to be proposed by KWE at the KWE General Meeting in connection with, among other things, the approval of the Scheme, the amendment of KWE's articles of association and the authorisation of the KWE Directors to take all such action as may be necessary to implement the Scheme;
"subsidiary" and "subsidiary undertaking"	shall be construed in accordance with the UK Companies Act 2006;
"Substantial Interest"	a direct or indirect interest in 20% or more of the voting equity capital of an undertaking;
"Takeover Offer"	if (subject to the consent of the Panel) KW elects to effect the Merger by way of a takeover offer (as defined in Article 116 of the Jersey Companies Law), the takeover offer to be made by or on behalf of KW to acquire the issued and to be issued ordinary share capital of KWE (other than the Excluded Shares) on the terms and subject to the conditions to be set out in the related offer document and, where the context admits, any subsequent revision, variation, extension or renewal of such takeover offer;
"Third Party"	has the meaning given in Part A of Part Three (Conditions and Certain Further Terms of the Scheme) of this document;
"Transaction Agreement"	the transaction agreement between KW and KWE dated 24 April 2017;
"UK" or "United Kingdom"	the United Kingdom of Great Britain and Northern Ireland;
"UK Holders"	has the meaning given in paragraph 1 of Part Seven (Taxation) of this document;
"UKLA"	the FCA acting in its capacity as the competent authority for listing under the Financial Services and Markets Act 2000;
"uncertificated" or "in uncertificated form"	in relation to a share or other security, a share or other security which is recorded on the relevant register of the share or security concerned as being held in uncertificated form in CREST and title to which, by virtue of the Uncertificated Securities Order may be transferred by means of CREST;
"Uncertificated Securities Order"	the Companies (Uncertificated Securities) (Jersey) Order 1999, as amended;
"US" or "United States"	the United States of America, its territories and possessions, any state of the United States of America and the District of Columbia;
"US Exchange Act"	the United States Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder;
"US GAAP"	means the US generally accepted accounting principles;
"US Holders"	has the meaning given in paragraph 2.1 of Part Seven (<i>Taxation</i>) of this document;
"USRPHC"	has the meaning given in 2.2(4)(b) of Part Seven (<i>Taxation</i>) of this document:

the United States Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder;

"US Securities Act"

"Värde Shareholders"

the Värde Fund X (Master), L.P., the Värde Fund XI, (Master) L.P., the Värde Fund VI-A, L.P., Värde Investment Partners, L.P., Värde Investment Partners (Offshore) Master, L.P., Värde Credit Partners Master, L.P. and the Värde Skyway Master Fund, L.P. all of which are ultimately controlled by Värde Partners, Inc.;

"Voting Record Time"

6.00 p.m. on 6 October 2017 or, if the Scheme Court Meeting and/ or the KWE General Meeting are or is adjourned, 6.00 p.m. on the last date to fall two Jersey Working Days before the date of the adjourned meeting;

"Wider KW Group"

KW and the subsidiaries and subsidiary undertakings of KW and associated undertakings (including any joint venture, partnership, firm or company in which any member of the KW Group is interested or any undertaking in which KW and such undertakings (aggregating their interests) have a Substantial Interest), provided that, notwithstanding the foregoing, no member of the Wider KWE Group shall be regarded as a member of the Wider KW Group; and

"Wider KWE Group"

KWE and the subsidiaries and subsidiary undertakings of KWE and associated undertakings (including any joint venture, partnership, firm or company in which any member of the KWE Group is interested or any undertaking in which KWE and such undertakings (aggregating their interests) have a Substantial Interest), provided that, notwithstanding the foregoing, no member of the Wider KWE Group shall be regarded as a member of the Wider KWE Group.

For the purposes of this document, "subsidiary", "subsidiary undertaking", "parent undertaking", "undertaking" and "associated undertaking" have the respective meanings given thereto by the UK Companies Act 2006.

All times referred to in this document are London times unless otherwise stated.

References to the singular include the plural and vice versa.

Any phrase introduced by the terms "including", "include", "in particular" or any similar expression is to be construed as illustrative only and does not limit the sense of the words preceding those terms.

PART TWELVE

NOTICE OF SCHEME COURT MEETING

IN THE ROYAL COURT OF JERSEY SAMEDI DIVISION

File No. 2017/252

13 September 2017

IN THE MATTER OF KENNEDY WILSON EUROPE REAL ESTATE PLC

IN THE MATTER OF ARTICLE 125 OF THE COMPANIES (JERSEY) LAW 1991

NOTICE IS HEREBY GIVEN that, by an Act of Court dated 12 September 2017 made in the above matters, the Court has directed a meeting (the "Scheme Court Meeting") to be convened of the holders of Scheme Shares (as defined in the Scheme of Arrangement referred to below, of which this Notice forms a part) for the purpose of considering and, if thought fit, approving (with or without modification) a scheme of arrangement (the "Scheme of Arrangement") pursuant to Article 125 of the Companies (Jersey) Law 1991 (as amended) proposed to be made between Kennedy Wilson Europe Real Estate Plc (the "Company") and the Scheme Shareholders (as defined in the Scheme of Arrangement) and that such Scheme Court Meeting will be held at 10.00 a.m. on 10 October 2017 at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN, at which place and time all Scheme Shareholders are requested to attend either in person or by proxy.

A copy of the Scheme of Arrangement and a copy of the explanatory statement required to be furnished pursuant to Article 126 of the Companies (Jersey) Law 1991 (as amended) are incorporated in the document of which this Notice forms a part.

The BLUE Form of Proxy (together with any power of attorney or authority under which it is signed, or a notarially certified copy of such power or authority) should be lodged with the KWE Registrar by post to: Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom or by hand (during normal business hours) to: Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, in each case so as to be received by no later than 10.00 a.m. on 8 October 2017, or in the case of any adjournment, no later than 48 hours before the time fixed for the holding of the adjourned meeting. Alternatively, the BLUE Form of Proxy relating to the Scheme Court Meeting will be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or any adjournment thereof).

Scheme Shareholders entitled to attend and vote at the Scheme Court Meeting may vote in person at the Scheme Court Meeting or they may appoint another person or persons, whether a member of the Company or not, as their proxy or proxies to attend and vote in their stead. A BLUE Form of Proxy for use at the Scheme Court Meeting is enclosed with this Notice. Completion and return of the Form of Proxy will not prevent an eligible holder of the Scheme Shares from attending and voting at the Scheme Court Meeting (or any adjournment thereof) in person, if he wishes to do so.

Each Scheme Shareholder is entitled to appoint a proxy in respect of some or all of the Scheme Shares held. Each Scheme Shareholder is also entitled to appoint more than one proxy, as long as each proxy is appointed to exercise rights attached to different Scheme Shares. A space has been included on the BLUE Form of Proxy to allow such Scheme Shareholders to specify the number of Scheme Shares in respect of which that proxy is appointed. A Scheme Shareholder who returns a BLUE Form of Proxy duly executed but leaves this space blank will be deemed to have appointed a proxy in respect of all of their Scheme Shares.

A Scheme Shareholder who wishes to appoint more than one proxy in respect of his, her or its shareholding should contact the KWE Registrar for further BLUE Forms of Proxy or photocopy BLUE Forms of Proxy as required. Such shareholders should also read the notes in respect of the appointment of multiple proxies set out in the Notice of KWE General Meeting included at Part Thirteen (Notice of KWE General Meeting) of the document of which this Notice forms a part.

Proxies can be appointed electronically by logging on to the website <u>www.investorcentre.co.uk/je</u>. To appoint a proxy electronically, a Scheme Shareholder will need its unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on the BLUE Form of Proxy). For an electronic proxy appointment to be valid, the appointment must be received by no later than 10.00 a.m.

on 8 October 2017, or in the case of any adjournment, no later than 48 hours before the time fixed for the holding of the adjourned meeting.

If Scheme Shares are held in uncertificated form, a proxy may be lodged via the CREST Proxy Voting Service (the Company's agent ID number is 3RA50) by 10.00 a.m. on 8 October 2017 (or, in the case of any adjournment, not less than 48 hours before the time fixed for the holding of the adjourned meeting). In respect of proxies submitted using the CREST Proxy Voting Service, the time of receipt will be taken to be the time from which the KWE Registrar is able to retrieve the message by enquiry to CREST.

In the case of joint holders of any Scheme Share (as defined in the Scheme of Arrangement), such persons shall not have the right of voting individually in respect of such Scheme Share but shall elect one of their number to represent them and to vote in person or by proxy. In default of such election, the person whose name appears first in order on the register of members of the Company in respect of such Scheme Share shall be the only person entitled to vote in respect of such Scheme Share.

Entitlement to attend and vote at the Scheme Court Meeting or any adjournment thereof and the number of votes which may be cast thereat will be determined by reference to the register of members of the Company as at the Voting Record Time (as defined in the Scheme of Arrangement) or, in the event that the Scheme Court Meeting is adjourned, the register of members of the Company at 6.00 p.m. on the last date to fall two Jersey Working Days, before the date of the adjourned meeting (as the case may be). In each case, changes to the register of members of the Company after such time will be disregarded.

By the said order, the Court has appointed Charlotte Valeur or, failing her, Mark McNicholas or, failing him, Simon Radford to act as Chairman of the Scheme Court Meeting and has directed the Chairman to report the results thereof to the Court.

The Scheme of Arrangement will be subject to the subsequent sanction of the Court.

Dated: 13 September 2017

Appleby
13-14 Esplanade,
St Helier,
JE1 1BD,
Jersey

Advocates and Solicitors for the Company

PART THIRTEEN

NOTICE OF KWE GENERAL MEETING

KENNEDY WILSON EUROPE REAL ESTATE PLC

(the "Company")

(Registered in Jersey No. 114680)

NOTICE IS HEREBY GIVEN that an extraordinary GENERAL MEETING of the Company shall be held at the offices of Sullivan & Cromwell LLP at 1 New Fetter Lane, London, England, EC4A 1AN at 10.15 a.m. on 10 October 2017 (or as soon thereafter as the Scheme Court Meeting (as defined in the document of which this Notice forms a part) shall have been concluded or adjourned for the purpose of considering and, if thought fit, passing the following resolution, which will be proposed as a special resolution of the shareholders of the Company:

SPECIAL RESOLUTION

THAT:

for the purpose of giving effect to the scheme of arrangement (the "Scheme") dated 13 September 2017 between the Company and the holders of the Scheme Shares (as defined in the Scheme), a print of which has been produced to this meeting and for the purposes of identification signed by the Chairman hereof, in its original form in the circular sent to shareholders of the Company dated 13 September 2017 or subject to such modification, addition or condition as may be agreed between the Company and Kennedy-Wilson Holdings, Inc. and approved or imposed by the Royal Court of Jersey (the "Court"):

- the directors of the Company be authorised to take all such action as they may consider necessary or appropriate for carrying the Scheme into full effect; and
- with effect from the passing of this resolution, the articles of association of the Company be and are hereby amended by the adoption and inclusion of the following new article 21A after article 21:

"Scheme of Arrangement

- (a) In this Article 21A, references to the scheme circular (the "Scheme Document") relating to the scheme of arrangement (the "Scheme") under Article 125 of the Law between the Company and the holders of the Scheme Shares (as defined in the Scheme Document) as it may be modified or amended in accordance with its terms, and expressions defined in the Scheme Document shall have the same meanings in this Article 21A.
- (b) Notwithstanding either any other provision of these Articles or the terms of any resolution, whether ordinary or special, passed by the Company in any general meeting, if the Company issues any Ordinary Shares or any other securities (other than to Kennedy-Wilson Holdings, Inc. or its nominee(s) and/or designated subsidiary) at any time on or after the Scheme Voting Record Time and on or prior to the Scheme Record Time (each as defined in the Scheme), such securities shall be issued subject to the terms of the Scheme (and shall be Scheme Shares for the purposes thereof) and the original or any subsequent holder or holders of such securities shall be bound by the Scheme accordingly.
- (c) Notwithstanding any other provision of these Articles, if any Ordinary Shares or any other securities are issued to any person (other than Kennedy-Wilson Holdings, Inc. or its nominee(s) and/or designated subsidiary) (the "New Member") at any time after the Scheme Record Time, such New Member (or any subsequent holder or any nominee and/or designated subsidiary of such New Member or any such subsequent holder) (the New Member or any such subsequent holder or any such nominee and/or designated subsidiary being the "New Transferor") will, provided the Scheme shall have become effective, be obliged to transfer forthwith all the securities held by the New Transferor (the "Disposal Securities", and, individually, a "Disposal Security"), free from all encumbrances, to Kennedy-Wilson Holdings, Inc. (or as Kennedy-Wilson Holdings, Inc. may otherwise direct) who shall be obliged to acquire all of the Disposal Securities in consideration of and conditional on the payment by or on behalf of Kennedy-Wilson Holdings, Inc. to the New Transferor of an amount in cash for each Disposal Security equal to the cash and share consideration that the New Transferor would have been entitled to had each Disposal Security been a Scheme Share (the "Relevant Consideration"). The Relevant

- Consideration means a cash payment equal to the value of (a) 550 pence plus (b) 0.3854 multiplied by the closing price of a New KW Share on the Effective Date (using the exchange rate prevailing on the relevant date and expressed in pounds sterling) for each Disposal Security.
- (d) On any reorganisation of, or material alteration to, the share capital of the Company (including, without limitation, any subdivision and/or consolidation), the value of the Relevant Consideration per Disposal Security to be paid under Article 21A(c) shall be adjusted by the Directors in such manner as the auditors of the Company or an independent investment bank selected by the Directors may determine to be fair and reasonable to the New Transferor to reflect such reorganisation or alteration. References in this Article 21A to Ordinary Shares shall, following such adjustment, be construed accordingly.
- To give effect to any transfer required by this Article 21A, the Company may appoint (and separately, to the extent necessary, each New Transferor shall therefore also appoint) any person as the Company may determine as attorney (under the Powers of Attorney (Jersey) Law 1995 any such appointment shall be irrevocable for a period of one year from the date upon which such New Transferor is issued the relevant Disposal Securities for that New Transferor), agent or otherwise for the New Transferor to execute and deliver as transferor a form of transfer or instructions of transfer on behalf of the New Transferor in favour of Kennedy-Wilson Holdings, Inc. and/or its nominee and/or designated subsidiary and do all such other things and execute and deliver all such documents as may in the opinion of the attorney, agent or otherwise be necessary or desirable to vest the Disposal Securities in Kennedy-Wilson Holdings, Inc. and/or its nominee and/or designated subsidiary and, pending such vesting, to exercise all such rights in and attaching to the Disposal Securities as Kennedy-Wilson Holdings, Inc. may direct. If an attorney is so appointed, the New Transferor shall not thereafter (except to the extent that the attorney fails to act in accordance with the directions of Kennedy-Wilson Holdings, Inc.) be entitled to exercise any rights attaching to the Disposal Securities unless so agreed by Kennedy-Wilson Holdings, Inc. The Company may give good receipt for the purchase price of the Disposal Securities and may register Kennedy-Wilson Holdings, Inc. and/or its nominee and/or designated subsidiary as holder thereof and issue to it certificates for the same. The Company shall not be obliged to issue a certificate to the New Transferor for any Disposal Securities. Kennedy-Wilson Holdings, Inc. shall send a cheque to the New Transferor drawn on a UK clearing bank (or shall procure that such a cheque is sent) in favour of the New Transferor for the purchase price of such Disposal Securities within 14 days of the date on which the Disposal Securities are transferred to Kennedy-Wilson Holdings, Inc. and/or its nominee and/or designated subsidiary.
- (f) This Article 21A shall cease to be effective if the Scheme shall not have become effective on or before 30 November 2017 (or such later date, if any, as Kennedy-Wilson Holdings, Inc. and the Company may agree and the Court and the Panel may allow).
- (g) Notwithstanding any other provision of these Articles, both the Company and the Directors may refuse to register the transfer of any Ordinary Shares between the Scheme Record Time and the Effective Date.

(h) Notwithstanding any other provision of these Articles, both the Company and the Directors may refuse to register the transfer of any Ordinary Shares other than pursuant to the Scheme or as provided by this Article 21A, but neither the Company nor the Directors may refuse to register the transfer of any Ordinary Shares pursuant to the Scheme or as provided by this Article 21A."

Dated: 13 September 2017

BY ORDER OF THE BOARD Crestbridge Fund Administrators Limited Company Secretary

> Registered Office: 47 Esplanade, St Helier, Jersey JE1 0BD

> > Registrar:

Computershare Investor Services (Jersey) Limited,
Queensway House,
Hilgrove Street,
St Helier,
Jersey,
JE1 1ES

Notes

In relation to the Scheme Court Meeting, the Form of Proxy is the BLUE Form of Proxy and, in relation to the KWE General Meeting, the Form of Proxy is the GREEN Form of Proxy.

The following notes to the Notice of the KWE General Meeting and the GREEN Form of Proxy to be used in connection with the KWE General Meeting apply mutatis mutandis to the Notice of the Scheme Court Meeting that can be found at Part Twelve (Notice of Scheme Court Meeting) of the document of which this Notice forms a part (the "Scheme Document") and the BLUE Form of Proxy to be used in connection with the Scheme Court Meeting; provided that

- in relation to the Scheme Court Meeting, notes 4 and 5 below shall not apply. Instead, the enclosed BLUE Form of Proxy is pre-paid and addressed. It should be sent, in accordance with its instructions, and together with any power of attorney or other authority under which it is signed (or a duly certified copy of such authority) so as to be received by the Company's Registrar, Computershare Investor Services (Jersey) Limited, at Computershare Investor Services PLC, Corporate Actions Projects, Bristol BS99 6AH, United Kingdom if sent by post and at Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom if delivered by hand (during normal business hours), by no later than 10.00 a.m. on 8 October 2017 (or, if the Scheme Court Meeting is adjourned, not less than 48 hours before the time fixed for the adjourned meeting). If the BLUE Form of Proxy is not so lodged, it will still be valid if handed to a representative of the KWE Registrar at the venue of the Scheme Court Meeting, or to the Chairman of the Scheme Court Meeting, in each case before the start of the Scheme Court Meeting (or, if relevant, any adjournment thereof). Alternatively, Scheme Shareholders (as defined in the Scheme Document) may appoint proxies electronically by logging on to the website www.investorcentre.co.uklje. To appoint a proxy electronically, a Scheme Shareholder will need its unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on the BLUE Form of Proxy). For an electronic proxy appointment to be valid, the appointment must be received by no later than 10.00 a.m. on 8 October 2017, or in the case of any adjournment, no later than 48 hours before the time fixed for the holding of the adjourned meeting. If Scheme Shares are held in uncertificated form, the appointment of a proxy or proxies may be lodged via the CREST Proxy Voting Service (the Company's agent ID number is 3RA50) by 10.00 a.m. on 8 October 2017 (or, in the case of any adjournment, not less than 48 hours before the time fixed for the holding of the adjourned meeting);
- (ii) in relation to the Scheme Court Meeting, note 12 below shall not apply. Instead, only those Scheme Shareholders registered in the register of members of the Company as at the Voting Record Time (as defined in the Scheme Document) shall be entitled to attend and vote at the Scheme Court Meeting in respect of the number of Scheme Shares registered in their names at the relevant time. Changes to entries on the register of members of the Company after the relevant time will be disregarded in determining the rights of any person to attend or vote at the Scheme Court Meeting;
- (iii) in relation to the Scheme Court Meeting, note 15 below shall not apply. Instead, the revocation notice must be received by the KWE Registrar, Computershare Investor Services (Jersey) Limited at clo Computershare Investor Services PLC, Corporate Actions Projects, Bristol BS99 6AH, United Kingdom, if sent by post, and at Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, if delivered by hand (during normal business hours), by no later than 10.00 a.m. on 8 October 2017;
- (iv) in relation to the Scheme Court Meeting, note 19 below shall not apply. Instead only Scheme Shares (as such term is defined in the Scheme), comprising 96,117,483 ordinary shares of no par value each in the Company, carry the right to vote at the Scheme Court Meeting, resulting in the total number of voting rights of 96,117,483 for the purposes of the Scheme Court Meeting as at 8 September 2017 (being the latest practicable date prior to the publication of this notice); and
- (v) in relation to the Scheme Court Meeting, note 21 below shall not apply.

Capitalised terms used in this notice that are not otherwise defined have the meaning assigned to them in the Scheme Document.

- 1. A member entitled to attend and vote at the meeting convened by this Notice is entitled to appoint one or more proxies to attend and vote in his, her or its place (and at any adjournment thereof). A proxy, however appointed, need not be a member of the Company and may be the Chairman of the relevant meeting. A member can appoint a proxy only using the procedures set out in these notes and the notes to the Form of Proxy enclosed with this document. Details of how to appoint the Chairman of the meeting or another person as your proxy using the Form of Proxy are set out in the notes to the Form of Proxy. A member may appoint more than one proxy, provided that each proxy is appointed to exercise the rights attaching to a different share or shares held by such member.
- 2. Voting on the resolution contained in this notice will be by way of a poll. This will result in a more accurate reflection of the views of members by ensuring that every vote is recognised, including the votes of members who are unable to attend the meeting, but who appoint a proxy.
- 3. In the case of joint holders of any share, such persons shall not have the right of voting individually in respect of such share but shall elect one of their number to represent them and to vote in person or by proxy. In default of such election, the person whose name appears first in order on the Company's register of members in respect of such share shall be the only person entitled to vote in respect of such share.
- 4. The enclosed GREEN Form of Proxy is pre-paid and addressed. It should be sent, in accordance with its instructions, and together with any power of attorney or other authority under which it is signed (or a duly certified copy of such authority) so as to be received by the Company's Registrar, Computershare Investor Services (Jersey) Limited, at Computershare Investor Services PLC, Corporate Actions Projects, Bristol BS99 6AH, United Kingdom if sent by post and at Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom if delivered by hand (during normal business hours), by no later than 10.15 a.m. on 8 October 2017 (or, if the KWE General Meeting is adjourned, not less than 48 hours before the time fixed for the holding of the adjourned meeting).
- 5. Alternatively members can appoint proxies electronically by logging on to the website www.investorcentre.co.uk/je and following the instructions given there. You will need your unique voting reference numbers (the Control Number, PIN and Shareholder Reference Number shown on your Form of Proxy). For an electronic proxy appointment to be valid, the appointment must be received by no later than 10.15 a.m. on 8 October 2017 (or, if the KWE General Meeting is adjourned, not less than 48 hours before the time fixed for the holding of the adjourned meeting).
- 6. CREST members who wish to appoint a proxy or proxies through the CREST electronic proxy appointment service may do so for the meeting convened by this notice and any adjournment(s) of such meeting by using the procedures described in the CREST Manual. CREST Personal Members or other CREST sponsored members, and those CREST members who have appointed a voting service provider(s), should refer to their CREST sponsor or voting service provider(s), who will be able to take the appropriate action on their behalf.
- 7. In order for a proxy appointment or instruction made using the CREST service to be valid, the appropriate CREST Proxy Instruction must be properly authenticated in accordance with Euroclear specifications and must contain the information required for such instructions, as described in the CREST Manual. The CREST Manual can be viewed at www.euroclear.com.
 - The message, regardless of whether it constitutes the appointment of a proxy or an amendment to the instruction given to a previously appointed proxy must, in order to be valid, be transmitted so as to be received by the KWE Registrar (3RA50) by the latest time(s) for receipt of proxy appointments specified in this document. For this purpose, the time of receipt will be taken to be the time (as determined by the timestamp applied to the message by the CREST Applications Host) from which the KW Registrar is able to retrieve the message by enquiry to CREST in the manner prescribed by CREST. After this time any change of instructions to proxies appointed through CREST should be communicated to the appointee through other means.
- 8. CREST members and, where applicable, their CREST sponsors or voting service providers should note that Euroclear does not make available special procedures in CREST for any particular messages. Normal system timings and limitations will therefore apply in relation to the input of CREST Proxy Instructions. It is the responsibility of the CREST member concerned to take (or, if the CREST member is a CREST personal member or sponsored member or has appointed a voting service provider(s), to procure that his, her or its CREST sponsor or voting service provider(s) take(s)) such action as shall be necessary to ensure that a message is transmitted by means of the CREST system by any particular time. In this connection, CREST members and, where applicable, their CREST sponsors or voting service providers are referred, in particular, to those sections of the CREST Manual concerning practical limitations of the CREST system and timings.
- 9. The Company may treat a CREST Proxy Instruction as invalid in the circumstances set out in Article 34 of the Companies (Uncertificated Securities) (Jersey) Order 1999.
- 10. A Form of Proxy must be executed by or on behalf of the member making the appointment. A corporation may execute a Form of Proxy either under its common seal or under the hand of a duly authorised officer, attorney or other representative.
- 11. Members who return a Form of Proxy will still be able to attend the meeting (or any adjournment thereof) and vote in person if they so wish.
- 12. Only those members registered in the register of members of the Company as at 6.00 pm on 6 October 2017 or, in the event that the KWE General Meeting is adjourned, in such register at 6.00 pm on the last date to fall two Jersey Working Days, before the date of the adjourned meeting, shall be entitled to attend and vote at the KWE General Meeting in respect of the number of KWE Shares registered in their names at the relevant time. Changes to entries on the register of members of the Company after the relevant time will be disregarded in determining the rights of any person to attend or vote at the KWE General Meeting.

- 13. To change your proxy, simply submit a new proxy appointment using the methods set out in notes 4 to 9 above. Where you have appointed a proxy using the hard copy Form of Proxy and would like to make a change using another hard copy Form of Proxy, please contact the KWE Registrar, Computershare Investor Services (Jersey) Limited at c/o Computershare Investor Services PLC, Corporate Actions Projects, Bristol BS99 6AH, United Kingdom. If you submit more than one valid proxy appointment, the appointment last received before the latest time for the receipt of proxies will take precedence.
- 14. In order to revoke a proxy appointment you will need to inform the Company by sending a signed hard copy notice clearly stating your intention to revoke your proxy. In the case of a member which is a company or corporation, the revocation notice must be executed under its common seal or signed on its behalf by an officer of the company or corporation or an attorney or other representative of the company or corporation. Any power of attorney or any other authority under which the revocation notice is signed (or a duly certified copy of such power or authority) must be included with the revocation notice. No other methods of communication will be accepted. In particular, you may not use any electronic address provided either in this notice or in any related documents (including the Form of Proxy).
- 15. The revocation notice must be received by the KWE Registrar, Computershare Investor Services (Jersey) Limited at c/o Computershare Investor Services PLC, Corporate Actions Projects, Bristol BS99 6AH, United Kingdom, if sent by post, and at Computershare Investor Services PLC, The Pavilions, Bridgwater Road, Bristol, BS13 8AE, United Kingdom, if delivered by hand (during normal business hours), by no later than 10.15 a.m. on 8 October 2017.
- 16. If you attempt to revoke your proxy appointment but the revocation is received after the time specified at note 15 above, then your proxy appointment will remain valid.
- 17. Any corporation which is a member can appoint a corporate representative who may exercise on its behalf all of its powers as a member.
- 18. A quorum consisting of not less than two members (being not less than two individuals) entitled to vote and attending in person or by attorney or proxy (or in the case of a corporation, by a duly appointed representative) is required for the meeting convened by this notice.
- 19. As at 8 September 2017, the Company's issued share capital comprised 126,133,407 Ordinary Shares of no par value each. Each Ordinary Share carries the right to one vote at the meeting of the Company convened by this notice and, therefore, the total number of voting rights in the Company as at 8 September 2017 were 126,133,407. As at 8 September 2017, no shares were held in treasury.
- 20. Except as provided above, members who wish to communicate with the Company in relation to the Scheme Court Meeting or the KWE General Meeting should do so using the following means: (1) by writing to the Company Secretary at the Company's registered office at 47 Esplanade, St Helier, Jersey JE1 0BD, Channel Islands; or (2) by writing to the Company's Registrar, Computershare Investor Services (Jersey) Limited at c/o Computershare Investor Services PLC, Corporate Actions Projects, Bristol, BS99 6AH, United Kingdom. No other methods of communication will be accepted. In particular, you may not use any electronic address provided either in this document or in any related documents (including the Form of Proxy) for any purposes other than those expressly stated.
- 21. Copies of the Company's existing articles of association and copies of the articles of association as proposed to be amended by the special resolution set out in this notice are available for inspection at Computershare Investor Services (Jersey) Limited at Queensway House, Hilgrove Street, St Helier, Jersey, JE1 1ES and at the Company's registered office at 47 Esplanade, St Helier, Jersey JE1 0BD, Channel Islands, until the opening of business on the day on which the KWE General Meeting is held and will also be available for inspection at the place of the KWE General Meeting for at least 15 minutes prior to the KWE General Meeting.

ISIN/SEDOL number

The London Stock Exchange Daily Official List (SEDOL) code is BJT3251. The International Security Identification Number (ISIN) is JE00BJT32513.

Website

Details of the Company's share price, together with other information about the Company, can be found at www.kennedywilson.eu.

Shareholder information

Persons who are not members will not be admitted to the meeting convened by this notice unless prior arrangements have been made with the Company. Investors holding Ordinary Shares through nominees are welcome to attend provided that they bring proof of their holding with them to the meeting.

The KWE Directors ask all those present at the meeting to facilitate the orderly conduct of the meeting and reserve the right, if orderly conduct is threatened by a person's behaviour, to require that person to leave.

Members should note that the doors to the meeting will open at 9.45 a.m. on 10 October 2017 with respect to the Scheme Court Meeting and 10.00 a.m. on 10 October 2017 with respect to the KWE General Meeting (or as soon thereafter as the Scheme Court Meeting is concluded or adjourned).

Shareholder enquiries

The Company's Ordinary Share register is maintained by the Company's Registrar:

Computershare Investor Services (Jersey) Limited Queensway House, Hilgrove Street, St Helier, Jersey, JE1 1ES. Telephone: 0370 707 4040.

Enquiries about the administration of holdings of shares, such as change of address, change of ownership or dividend payments, should be directed to the Company's Registrar at the address and telephone number above.

PART FOURTEEN

PROFIT FORECASTS

SECTION A

GENERAL

KW has published certain projections and certain financial measures of performance in respect of KW for the financial year ending 31 December 2017, covering net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share (the "KW 2017 Profit Forecast").

KW has also published certain projections and certain financial measures of performance in respect of KWE for the financial year ending 31 December 2017, covering Recurring Adjusted Net Income attributable to KWE Shareholders and Recurring Adjusted Earnings per share (the "KWE 2017 Profit Forecast" and, together with the KW 2017 Profit Forecast, the "Profit Forecasts").

The Profit Forecasts constitute "profit forecasts" for the purposes of Rule 28 of the Code. As a result, and as required by Rule 28.1(a) of the Code, KPMG, KW's reporting accountants, and Goldman Sachs International, KW's lead financial adviser, have each prepared a report in respect of the KW 2017 Profit Forecast and the KWE 2017 Profit Forecast. The Profit Forecasts, together with the bases and principal assumptions on which they were prepared and the reports thereon produced by KPMG and Goldman Sachs International, are set out in Section B and Section C of this Part Fourteen.

KW has also published certain long-range unaudited financial projections relating to (i) KW for the financial years ending 31 December 2018, 31 December 2019, 31 December 2020 and 31 December 2021 (namely net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share) (the "KW EPP") and (ii) KWE for the financial years ending 31 December 2018, 31 December 2019, 31 December 2020 and 31 December 2021 (namely Recurring Adjusted Net Oncome attributable to KWE Shareholders and Recurring Adjusted Earnings per share) (the "KWE EPP" and, together with the KW EPP, the "Extended Period Projections"). For the reasons set out in Section D of this Part Fourteen, KW considers that the Extended Period Projections are no longer valid.

For the purposes of this Part Fourteen "Recurring Adjusted Net Income attributable to KWE Shareholders" means net income attributable to KWE Shareholders before depreciation and amortization and gain on sale of real estate and "Recurring Adjusted Earnings per share" means Recurring Adjusted Net Income per KWE Share.

For the purposes of Rule 28 of the Code, the Profit Forecasts are the responsibility of KW and the KW Directors (and not KWE nor the KWE Independent Directors). The Profit Forecasts have been prepared solely by KW's management in connection with KW's consideration and evaluation of the Merger. Neither KWE nor the KWE Independent Directors have been involved in preparing, nor have they approved, the Profit Forecasts or the Extended Period Projections.

SECTION B

THE KW 2017 PROFIT FORECAST

1. KW 2017 Profit Forecast, including bases and assumptions

1.1 Basis of preparation and principal assumptions

In the KW Proxy Statement and the Profit Forecast RIS Announcement, KW published the KW 2017 Profit Forecast, which was as follows:

(Dollars in millions except per share data)	2017E \$
Net income attributable to KW Shareholders	40.4
Adjusted net income attributable to KW Shareholders	204.1
Adjusted earnings per share	1.77

The KW 2017 Profit Forecast has been properly compiled on the basis of the assumptions stated below, and on a basis consistent with the accounting policies of KW. The KW 2017 Profit Forecast constitutes a profit forecast under Rule 28 of the Takeover Code and is still considered valid as at the date of publication of this document. Paragraphs 2 and 3 of Section B of this Part Fourteen contain KPMG's and Goldman Sachs International's reports on the KW 2017 Profit Forecast.

1.2 Basis of preparation

The KW 2017 Profit Forecast has been prepared on a consistent basis with the accounting policies of KW adopted in its consolidated financial statements for the financial year ended 31 December 2016 and in its second quarter results for the three months ended 30 June 2017.

The KW 2017 Profit Forecast does not take into account any effects of the Merger, including related transaction fees and expenses, funding of the KW Cash Component and the payment of the KWE Special Distribution.

1.3 Principal assumptions

The principal assumptions upon which the KW 2017 Profit Forecast is based are set out below:

Assumptions the KW Directors can influence

- trading performance in respect of rental revenue, hotel revenue, rental operating expenses, hotel operating expenses, commission and marketing expenses, general & administrative expenses, compensation & benefits expenses, acquisition expenses, net interest expenses and depreciation and amortisation for the six-month period ended 30 June 2017 is indicative of, and will generally continue at the same run-rate over, the forecast trading performance for the six months ending 31 December 2017;
- KW expects to achieve approximately \$220 million of gains (including KW shareholder's proportionate share of KWE gains) on asset disposals in 2017. Approximately \$70 million of the gains have been achieved during the six-month period ending 30 June 2017 and the balance are projected to be achieved in the six-month period ending 31 December 2017 and may be concentrated in a limited number of assets of which, some are currently under negotiation;
- property leasing activity will proceed materially as the KW Directors would reasonably expect based on KW's past experience;
- no material current tenant contract issues or changes will arise in the relevant period;
- the full impact in the financial year ending 31 December 2017 of acquisition, disposition, capital expenditure and development activity undertaken in the financial year ended 31 December 2016 will be as expected;
- KW will continue to undertake substantial levels of investment (asset acquisitions and capital expenditure) and asset dispositions, realising returns in line with historical performance;

- there will be no material change in momentum of KW's fund raising activities related to its investment management business or in its property services and research business;
- there will be no significant change in performance of smaller businesses;
- there will be no material change in operating margins, compensation structures or general and administrative expense; and
- the effective tax rate will be approximately 19-20%, with the US portion of this being a non-cash expense.

Assumptions the KW Directors cannot influence

- there will be no changes, beyond what is already contemplated, in property market conditions, demand for real estate and real estate services, competitive environment or economic conditions in the markets and asset classes in which KW operates;
- there will be no material changes in property valuations or market demand which could hinder KW's ability to purchase, dispose of and let buildings;
- there will be no changes in exchange rates, interest rates, bases of taxes or legislative or regulatory requirements from those currently forecast that would have a material impact on KW's operations or its accounting policies;
- there will be no material adverse weather events or natural catastrophes that affect KW's assets or operations;
- there will be no material change from any political or economic events in the countries in which KW or its tenants operate;
- there will be no material changes in costs of labour, goods and services necessary to operate KW's business; and
- there will be no business interruptions that materially adversely affect KW, its key tenants or its key suppliers.

2. KPMG Report on the KW 2017 Profit Forecast

The Directors Kennedy-Wilson Holdings, Inc. 151 S. El Camino Drive Beverly Hills, CA 90212 United States of America

Goldman Sachs International Peterborough Court, 133 Fleet St, London EC4A 2BB United Kingdom

Ladies and Gentlemen

13 September 2017

Kennedy-Wilson Holdings, Inc.

We report on the profit forecast comprising net income attributable to stockholders, adjusted net income and adjusted earnings per share of Kennedy-Wilson Holdings, Inc. ('the Company') and its subsidiaries ('the Group') for the year ending 31 December 2017 (the 'Profit Forecast'). The Profit Forecast, and the material assumptions upon which it is based, are set out on pages 222 to 223 of the scheme document published by Kennedy Wilson Europe Real Estate plc dated 13 September 2017 ('the Document'). This report is required by Rule 28.1 of The City Code on Takeovers and Mergers ('the City Code') and is given for the purpose of complying with that rule and for no other purpose.

Responsibilities

It is the responsibility of the directors of the Company ('the Directors') to prepare the Profit Forecast in accordance with the requirements of the City Code.

It is our responsibility to form an opinion as required by the City Code as to the proper compilation of the profit forecast and to report that opinion to you.

Save for any responsibility which we may have to those persons to whom this report is expressly addressed, to the fullest extent permitted by law we do not assume any responsibility and will not accept any liability to any other person for any loss suffered by any such other person as a result of, arising out of, or in connection with this report or our statement, required by and given solely for the purposes of complying with Rule 23.2 of the City Code, consenting to its inclusion in the Document.

Basis of Preparation of the Profit Forecast

The Profit Forecast has been prepared on the basis stated on pages 222 to 223 of the Document and is based on a forecast to 31 December 2017. The Profit Forecast is required to be presented on a basis consistent with the accounting policies of the Group.

Basis of opinion

We conducted our work in accordance with Standards for Investment Reporting issued by the Auditing Practices Board in the United Kingdom. Our work included evaluating the basis on which the historical financial information included in the Profit Forecast has been prepared and considering whether the Profit Forecast has been accurately computed based upon the disclosed assumptions and the accounting policies of the Group. Whilst the assumptions upon which the Profit Forecast are based are solely the responsibility of the Directors, we considered whether anything came to our attention to indicate that any of the assumptions adopted by the Directors which, in our opinion, are necessary for a proper understanding of the Profit Forecast have not been disclosed or if any material assumption made by the Directors appears to us to be unrealistic.

We planned and performed our work so as to obtain the information and explanations we considered necessary in order to provide us with reasonable assurance that the Profit Forecast has been properly compiled on the basis stated.

Since the Profit Forecast and the assumptions on which it is based relate to the future and may therefore be affected by unforeseen events, we can express no opinion as to whether the actual results reported will correspond to those shown in the Profit Forecast and differences may be material.

Our work has not been carried out in accordance with auditing or other standards and practices generally accepted in the United States of America and accordingly should not be relied upon as if it had been carried out in accordance with those standards and practices.

Opinion

In our opinion the Profit Forecast has been properly compiled on the basis stated and the basis of accounting used is consistent with the accounting policies of the Group.

Yours faithfully

KPMG LLP

3. GS Report on the KW 2017 Profit Forecast

The Board of Directors of Kennedy-Wilson Holdings, Inc. Kennedy-Wilson Holdings, Inc. 151 S. El Camino Drive Beverly Hills, CA 90212 United States of America

Ladies and Gentlemen:

13 September 2017

Recommended Merger of Kennedy-Wilson Holdings, Inc. ("KW") and Kennedy Wilson Europe Real Estate plc

We refer to the profit forecast made by KW in respect of net income attributable to KW shareholders, adjusted net income attributable to KW shareholders and adjusted earnings per KW share for the financial year ending 31 December 2017 (the "KW 2017 Profit Forecast"). The KW 2017 Profit Forecast and the material assumptions on which it is based are set out in the scheme circular published by Kennedy Wilson Europe Real Estate plc on 13 September 2017 (the "Scheme Circular"), in which our report is included. The board of directors of KW (the "Directors") are solely responsible for the KW 2017 Profit Forecast under Rule 28 of the City Code on Takeovers and Mergers (the "Code").

We have discussed the KW 2017 Profit Forecast (including the bases, assumptions and source of information referred to therein) with the Directors and those officers and employees of KW who developed the underlying plans. The KW 2017 Profit Forecast is subject to the uncertainty as described in the Scheme Circular and our work did not involve independent examination of any of the financial or other information underlying the KW 2017 Profit Forecast.

For purposes of rendering this report, we have relied upon and assumed the accuracy and completeness of all the financial, legal, regulatory, tax, accounting and other information provided to, discussed with or reviewed by, us, without assuming any responsibility for independent verification thereof. You have confirmed to us that all information material to the KW 2017 Profit Forecast has been disclosed to us.

We have also reviewed the work carried out by KPMG LLP, the reporting accountants to KW, and have discussed with them the opinion set out in the Scheme Circular addressed to you and us and the accounting policies and bases of calculation for the 2017 KW Profit Forecast.

This report is provided to you solely in connection with Rule 28(1)(a)(ii) of the Code and for no other purpose. We accept no responsibility to KW or its shareholders or any person other than the Directors in respect of the contents of this report. We are acting exclusively as lead financial adviser to KW and no one else in connection with the recommended merger, and it was solely for the purposes of complying with Rule 28(1)(a)(ii) of the Code that KW requested Goldman Sachs International to prepare a report on the KW 2017 Profit Forecast. No person other than the Directors can rely on the contents of this report and, to the fullest extent permitted by law, we exclude all liability (whether in contract, tort or otherwise) to any other person in respect of this report, its contents, or the work undertaken by us in connection with this report or any of the results that can be derived from this report or any written or oral information provided in connection with this report.

This report is not an opinion regarding, and we express no view as to, the reasonableness or achievability of the KW 2017 Profit Forecast. Instead, this report will address solely the matters required to be addressed by Rule 28(1)(a)(ii) of the Code and does not address any other matter; it does not address, among other things, the relative merits of the recommended merger as compared to any other transaction or business strategy, the merits of the decision by KW to engage in the recommended merger, the fairness to KW shareholders of the exchange ratios in the recommended merger, or the price at which KW's shares or the shares of the enlarged KW would trade at any time.

Furthermore, this report has been prepared independent of publication of the KW 2017 Profit Forecast and may not be relied on by you in recommending that KW's shareholders vote to approve the recommended merger. This report is not intended to and does not constitute a recommendation

to any shareholder as to how such shareholder should vote or act with respect to the proposed merger or any matter related thereto.

On the basis of the foregoing, we consider that the KW 2017 Profit Forecast, for which you as Directors are solely responsible, has been made with due care and consideration.

Yours faithfully,

Goldman Sachs International

SECTION C

THE KWE 2017 PROFIT FORECAST

1. KWE 2017 Profit Forecast, including bases and assumptions

1.1 Basis of preparation and principal assumptions

In the KW Proxy Statement and the Profit Forecast RIS Announcement, KW published the KWE 2017 Profit Forecast, which was as follows:

(Dollars in millions except per share data)	2017E \$
Recurring Adjusted Net Income attributable to KWE Shareholders	102.3
Recurring Adjusted Earnings per share	0.81

The KWE 2017 Profit Forecast has been properly compiled on the basis of the assumptions stated below, and on a basis consistent with the accounting policies of KW. The KWE 2017 Profit Forecast constitutes a profit forecast under Rule 28 of the Takeover Code and is still considered valid as at the date of publication of this document. Paragraphs 2 and 3 of Section C of this Part Fourteen contain KPMG's and Goldman Sachs International's reports on the KWE 2017 Profit Forecast.

1.2 Basis of preparation

KWE Shareholders should note that the KWE 2017 Profit Forecast set out in this document has been prepared in accordance with KW's US GAAP accounting policies and not prepared and presented in accordance with KWE's IFRS accounting policies. These accounting policies differ materially in certain respects.

The KWE 2017 Profit Forecast for the financial year ending 31 December 2017 is presented on a US GAAP basis and has been prepared on a consistent basis with the accounting policies of KW adopted in its consolidated financial statements for the financial year ended 31 December 2016 and in its second quarter results for the three months ended 30 June 2017. The KWE 2017 Profit Forecast does not take into account any effects of the Merger, including related transaction fees and expenses, and the payment of the KWE Special Distribution.

The KWE 2017 Profit Forecast has been prepared by KW on the basis of publicly available information and the KW Directors' knowledge of KWE's assets. The KWE 2017 Profit Forecast has been prepared by KW as a potential acquirer of KWE and not as the manager of KWE.

1.3 Principal assumptions

The principal assumptions upon which the KWE 2017 Profit Forecast is based are set out below:

Assumptions the KW Directors can influence, in light of KW's role as controller of the Investment Manager

- the trading performance in respect of rental revenue, hotel revenue, loan purchases, loan originations and other, net operating income, general & administrative expenses, compensation & benefits expenses, acquisition expenses, net interest expenses and depreciation and amortisation for the six-month period ended 30 June 2017 is indicative of, and will generally continue at the same run-rate over, the forecast trading performance for the six months ending 31 December 2017;
- property leasing activity will proceed materially as the KW Directors would reasonably expect based on KWE's past performance;
- no material current tenant contract issues or changes will arise in the relevant period;
- the full impact in the financial year ending 31 December 2017 of acquisition, disposition, capital expenditure and development activity undertaken in the financial year ended 31 December 2016 will be as expected;
- KWE will continue to undertake substantial levels of investment (asset acquisitions and capital expenditure) and asset dispositions, realising returns in line with historical performance;

- there will be no material change in operating margins, investment management fees, or general and administrative expense; and
- the effective tax rate will be approximately 10%;

Assumptions the KW Directors cannot influence

- there will be no changes, beyond what is already contemplated, in property market conditions, demand for real estate and real estate services, competitive environment or economic conditions in the markets and asset classes in which KWE operates;
- there will be no material changes in property valuations or market demand which could hinder KWE's ability to purchase, dispose of and let buildings;
- there will be no changes in exchange rates, interest rates, bases of taxes or legislative or regulatory requirements from those currently forecast that would have a material impact on KWE's operations or its accounting policies;
- there will be no material adverse weather events or natural catastrophes that affect KWE's assets or operations;
- there will be no material change from any political or economic events in the countries in which KWE or its tenants operate;
- there will be no material changes in costs of labour, goods and services necessary to operate KWE's business; and
- there will be no business interruptions that materially adversely affect KWE, its key tenants or its key suppliers.

2. KPMG Report on the KWE 2017 Profit Forecast

The Directors Kennedy-Wilson Holdings, Inc. 151 S. El Camino Drive Beverly Hills, CA 90212 United States of America

Goldman Sachs International Peterborough Court, 133 Fleet St, London EC4A 2BB United Kingdom

13 September 2017

Ladies and Gentlemen

Kennedy Wilson Europe Real Estate plc

We report on the profit forecast comprising recurring adjusted net income and recurring adjusted earnings per share of Kennedy Wilson Europe Real Estate plc for the year ending 31 December 2017 (the 'Profit Forecast'). The Profit Forecast, and the material assumptions upon which it is based, are set out on pages 228 to 229 of the scheme document published by Kennedy Wilson Europe Real Estate plc dated 13 September 2017 ('the Document'). This report is required by Rule 28.1 of The City Code on Takeovers and Mergers ('the City Code') and is given for the purpose of complying with that rule and for no other purpose.

Responsibilities

It is the responsibility of the directors of Kennedy-Wilson Holdings, Inc. ('the Directors') to prepare the Profit Forecast in accordance with the requirements of the City Code.

It is our responsibility to form an opinion as required by the City Code as to the proper compilation of the profit forecast and to report that opinion to you.

Save for any responsibility which we may have to those persons to whom this report is expressly addressed, to the fullest extent permitted by law we do not assume any responsibility and will not accept any liability to any other person for any loss suffered by any such other person as a result of, arising out of, or in connection with this report or our statement, required by and given solely for the purposes of complying with Rule 23.2 of the City Code, consenting to its inclusion in the Document.

Basis of Preparation of the Profit Forecast

The Profit Forecast has been prepared on the basis stated on pages 228 to 229 of the Document and is based on a forecast to 31 December 2017. The Profit Forecast is required to be presented on a basis consistent with the accounting policies of Kennedy-Wilson Holdings, Inc.

Basis of opinion

We conducted our work in accordance with Standards for Investment Reporting issued by the Auditing Practices Board in the United Kingdom. Our work included evaluating the basis on which the historical financial information included in the Profit Forecast has been prepared and considering whether the Profit Forecast has been accurately computed based upon the disclosed assumptions and the accounting policies of Kennedy-Wilson Holdings, Inc. Whilst the assumptions upon which the Profit Forecast are based are solely the responsibility of the Directors, we considered whether anything came to our attention to indicate that any of the assumptions adopted by the Directors which, in our opinion, are necessary for a proper understanding of the Profit Forecast have not been disclosed or if any material assumption made by the Directors appears to us to be unrealistic.

We planned and performed our work so as to obtain the information and explanations we considered necessary in order to provide us with reasonable assurance that the Profit Forecast has been properly compiled on the basis stated.

Since the Profit Forecast and the assumptions on which it is based relate to the future and may therefore be affected by unforeseen events, we can express no opinion as to whether the actual results reported will correspond to those shown in the Profit Forecast and differences may be material.

Our work has not been carried out in accordance with auditing or other standards and practices generally accepted in the United States of America and accordingly should not be relied upon as if it had been carried out in accordance with those standards and practices.

Opinion

In our opinion the Profit Forecast has been properly compiled on the basis stated and the basis of accounting used is consistent with the accounting policies of Kennedy-Wilson Holdings, Inc.

Yours faithfully

KPMG LLP

3. Goldman Sachs International Report on the KWE 2017 Profit Forecast

The Board of Directors of Kennedy-Wilson Holdings, Inc. Kennedy-Wilson Holdings, Inc. 151 S. El Camino Drive Beverly Hills, CA 90212 United States of America 13 September 2017

Ladies and Gentlemen:

Recommended Merger of Kennedy-Wilson Holdings, Inc. ("KW") and Kennedy Wilson Europe Real Estate plc ("KWE")

We refer to the profit forecast made by KW in respect of recurring adjusted net income attributable to KWE shareholders and recurring adjusted earnings per KWE share (the "KWE 2017 Profit Forecast"). The KWE 2017 Profit Forecast and the material assumptions on which it is based are set out in the scheme circular published by KWE on 13 September 2017 (the "Scheme Circular"), in which our report is included. The board of directors of KW (the "Directors") are solely responsible for the KWE 2017 Profit Forecast under Rule 28 of the City Code on Takeovers and Mergers (the "Code").

We have discussed the KWE 2017 Profit Forecast (including the bases, assumptions and sources of information referred to therein) with the Directors and those officers and employees of KW who developed the underlying plans. The KWE 2017 Profit Forecast is subject to uncertainty as described in the Scheme Circular and our work did not involve an independent examination of any of the financial or other information underlying the KWE 2017 Profit Forecast.

For purposes of rendering this report, we have relied upon and assumed the accuracy and completeness of all the financial, legal, regulatory, tax, accounting and other information provided to, discussed with or reviewed by, us, without assuming any responsibility for independent verification thereof. You have confirmed to us that all information material to the KWE 2017 Profit Forecast has been disclosed to us.

We have also reviewed the work carried out by KPMG LLP, the reporting accountants to KW, and have discussed with them the opinion set out in the Scheme Circular addressed to you and us and the accounting policies and bases of calculation for the 2017 KWE Profit Forecast.

This report is provided to you solely in connection with Rule 28(1)(a)(ii) of the Code and for no other purpose. We accept no responsibility to KW or its shareholders or any person other than the Directors in respect of the contents of this report. We are acting exclusively as lead financial adviser to KW and no one else in connection with the recommended merger, and it was solely for the purposes of complying with Rule 28(1)(a)(ii) of the Code that KW requested Goldman Sachs International to prepare a report on the KWE 2017 Profit Forecast. No person other than the Directors can rely on the contents of this report and, to the fullest extent permitted by law, we exclude all liability (whether in contract, tort or otherwise) to any other person in respect of this report, its contents or the work undertaken by us in connection with this report or any of the results that can be derived from this report or any written or oral information provided in connection with this report.

This report is not an opinion regarding, and we express no view as to, the reasonableness or achievability of the KWE 2017 Profit Forecast. Instead, this report will address solely the matters required to be addressed by Rule 28(1)(a)(ii) of the Code and does not address any other matter; it does not address, among other things, the relative merits of the recommended merger as compared to any other transaction or business strategy, the merits of the decision by KW to engage in the recommended merger, the fairness to KW shareholders of the exchange ratios in the recommended merger, or the price at which KW's shares or the shares of the enlarged KW would trade at any time.

Furthermore, this report has been prepared independent of publication of the KWE 2017 Profit Forecast and may not be relied on by you in recommending that KW's shareholders vote to approve the recommended merger. This report is not intended to and does not constitute a recommendation to any shareholder as to how such shareholder should vote or act with respect to the proposed combination or any matter related thereto.

On the basis of the foregoing, we consider that the KWE 2017 Profit Forecast, for which you as Directors are solely responsible, has been made with due care and consideration.

Yours faithfully,

Goldman Sachs International

SECTION D

THE EXTENDED PERIOD PROJECTIONS

1. The Extended Period Projections

KW disclosed the Extended Period Projections in the KW Proxy Statement and in the Profit Forecast RIS Announcement dated 27 June 2017.

KW does not, as a matter of course, make public projections as to future revenues, earnings or other results due to, among other reasons, the uncertainty of the underlying assumptions and estimates. However, in connection with KW's consideration and evaluation of the Merger, and in order to provide a basis for discussions on a range of possible future outcomes, KW's management prepared projections of certain financial measures of performance which KW's management provided to the KW Board and to Goldman Sachs. KW prepared the Extended Period Projections based on the assumptions and estimates considered reasonable by KW as at the date that the projections were prepared.

The Extended Period Projections were prepared for internal use and to assist KW and its financial adviser with their consideration and evaluation of the Merger. The Extended Period Projections were not prepared with a view toward public disclosure or toward complying with IFRS, US GAAP, the published guidelines of the SEC regarding projections or the guidelines established by the American Institute of Certified Public Accountants for preparation and presentation of prospective financial information. The Extended Period Projections have not been reported on by KW's financial adviser or reporting accountants under Rule 28 of the Code. In the Profit Forecast RIS Announcement dated 27 June 2017, the KW Directors provided the confirmation required under Rule 28.2(a) of the Code with respect to the Extended Period Projections. The KW 2017 Profit Forecast and the KWE 2017 Profit Forecast have been reported on by Goldman Sachs International and KPMG, as required by the Panel and in compliance with Rule 28 of the Code. The KW 2017 Profit Forecast and the KWE 2017 Profit Forecast have also been reported on by KPMG in accordance with the Prospectus Rules.

The quantification of the Extended Period Projections has been included in the KW Proxy Statement solely to comply with KW's obligations under US federal and state laws to give KW Shareholders access to certain nonpublic information that was made available to the KW Board for the purpose of considering and evaluating the Merger and to Goldman Sachs for the purposes of performing its financial analyses in connection with rendering its fairness opinion.

Except as described above with respect to the KW 2017 Profit Forecast and the KWE 2017 Profit Forecast, neither KW's independent registered public accounting firm nor Goldman Sachs nor any other firm has examined, compiled nor performed any procedures with respect to the Extended Period Projections and, accordingly, do not express an opinion or any other form of assurance with respect thereto.

The inclusion of this information should not be regarded as an indication that the KW Board, KW, Goldman Sachs or any other recipient of this information considered, or now considers, the Extended Period Projections to be material information of KW.

The summaries of the Extended Period Projections are not included in the KW Proxy Statement or the RIS Announcement in order to induce any KW Shareholder to vote in favour of the proposal to approve the issuance of New KW Shares in connection with the Merger or to influence any Scheme Shareholder to make any investment decision with respect of the Merger.

A summary of the Extended Period Projections was included in the KW Proxy Statement and, consequently, the Profit Forecast RIS Announcement solely pursuant to specific requirements of the applicable rules and regulations promulgated by the SEC. The Profit Forecast RIS Announcement also included the KW Directors' confirmation regarding the Extended Period Projections in accordance with Rule 28.2(a) of the Code, as well as the reports on the KW 2017 Profit Forecast and the KWE 2017 Profit Forecast prepared by Goldman Sachs International and KPMG, as required by the Panel and in compliance with Rule 28 of the Code.

2. Long-term projections

The KW EPP and the KWE EPP are forward-looking in nature. The KW EPP for net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share are extended extrapolations based on the KW 2017 Profit Forecast

and a limited number of high level assumptions. The KWE EPP for net income attributable to KWE Shareholders, Recurring Adjusted Net Income attributable to KWE Shareholders and Recurring Adjusted Earnings per share are extended extrapolations based on the KWE 2017 Profit Forecast and a limited number of high level assumptions.

Although the KW EPP and the KWE EPP are presented with numerical specificity, the extrapolations reflect numerous estimates and assumptions as to future events that KW's management believed were reasonable at the time the extrapolations contained in the KW EPP and the KWE EPP were prepared and used, taking into account the relevant information available to KW's management at the time. The KW EPP and the KWE EPP do not take into account any circumstances or events occurring after the date on which they were prepared. As a result, the KW EPP and the KWE EPP may not be comparable to future results of the Combined Group, which will reflect the impact of the items not taken into account in the preparation of the KW EPP and the KWE EPP. Except to the extent required by applicable law, KW does not intend, and expressly disclaims any responsibility, to update or otherwise revise the Extended Period Projections to reflect the occurrence of future events or changes in general economic or industry conditions

3. In the view of KW, the KW EPP are no longer valid

KW is of the opinion that the KW EPP are no longer valid in the context of the Merger. Please refer to Part 16 of the KW Prospectus for a full explanation as to why KW is of this opinion.

KW is also of the opinion that the KW EPP are no longer valid on a stand-alone basis for the following reasons. The principal changes expected to affect the financial results of KW are summarised below. Each of these changes will affect, in each of 2018, 2019, 2020 and 2021, net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share, compared to the KW EPP.

Q2 performance

As set out in the KW Proxy Statement and the Profit Forecast RIS Announcement, the KW EPP were prepared as longer term extrapolations of the KW 2017 Profit Forecast and are based on a number of key assumptions including constant segment revenue growth, constant segment costs as a percentage of revenue, growth in gain on sale and constant tax rate.

The operating and investment activity (pricing and volume) in the second quarter of 2017 differed from the expectations of KW's management at the time of preparing the KW 2017 Profit Forecast. Whilst the overall KW 2017 Profit Forecast (net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholder and adjusted earnings per share) remains unchanged, the segmental results of the second quarter of 2017 were not as previously forecast. If the KW EPP were updated to reflect the latest trading information, the segmental changes would result in a different outcome for the individual line items that make up the KW EPP, and therefore would result in a different outcome for net income attributable to KW Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share, compared to the KW EPP.

Investment activity

KW undertakes significant ordinary course investment activity (both acquisitions and disposals) in any given year which may impact gains on sale of real estate and net operating income in future periods. Following the announcement of the New Offer, given the significant cash component, in order to strengthen its balance sheet following the Merger, KW (along with KWE) announced that it anticipated \$500 million of additional sales by July 2018. In order to achieve these targets, KW has already made changes to its 2017 investment activity. These changes in investment activity, both increased disposals and reduction in investments, mean that the assumptions on which the KW EPP were based are no longer valid. However, they do not affect the validity of the KW 2017 Profit Forecast.

4. In the view of KW, the KWE EPP are no longer valid

KW is of the opinion that the KWE EPP are no longer valid in the context of the Merger. Please refer to Part 16 of the KW Prospectus for a full explanation as to why KW is of this opinion.

KW is also of the opinion that the KWE EPP are no longer valid on a stand-alone basis for the following reasons. The principal changes expected to affect the financial results of KW are summarised below. Each of these changes will affect, in each of 2018, 2019, 2020 and 2021, net income attributable to KWE Shareholders, adjusted net income attributable to KW Shareholders and adjusted earnings per share, compared to the KWE EPP.

Q2 performance

As set out in the KW Proxy Statement and the Profit Forecast RIS Announcement, the KWE EPP are longer term extrapolations of the KWE 2017 Profit Forecast and are based on the key assumptions of constant segment revenue growth, constant segment costs as a percentage of revenue, growth in gain on sale and constant tax rate.

The operating and investment activity (pricing and volume) in the second quarter of 2017 differed from to the expectations of KW's management at the time of preparing the KW 2017 Profit Forecast. Whilst the overall KWE 2017 Profit Forecast (Recurring Adjusted Net Income attributable to KWE Shareholders and Recurring Adjusted Earnings per share) remains unchanged, the segmental results of the second quarter of 2017 were not as previously forecast. If the KWE EPP were updated to reflect the latest trading information, the segmental changes would result in a different outcome for the individual line items that make up the KWE EPP and therefore would result in a different outcome for adjusted net income attributable to KW Shareholders and adjusted earnings per share, compared to the KWE EPP.

Acquisitions

The KWE EPP reflect assumptions as to the level of acquisitions and do not take into account the KWE Special Distribution payable in the context of the Merger. The KWE EPP assumed that a significant amount of KWE's available cash would be used to fund new acquisitions by the end of 2017 and not be used towards the KWE Special Distribution. Since the New Offer was announced in June 2017, and in anticipation of the payment by KWE of the KWE Special Distribution, KWE reduced its level of planned acquisitions, to take into account the fact that a large part of KWE's available cash will be required to fund the KWE Special Distribution and will, therefore, not be available to fund planned acquisitions. As a result, even if the Merger does not complete, the original assumptions as to the level of acquisitions in the third and fourth quarters of 2017 on which the KWE EPP were based will no longer be met. These new acquisitions would have generated net operating income assumed in the KWE EPP in each of 2018, 2019, 2020 and 2021. As a result, the level of acquisitions will be materially lower than originally assumed and therefore the KWE EPP are no longer valid.

Disposals

The reasons discussed in detail in paragraph 3 above in relation to the KW EPP also apply to the KWE EPP.

ANNEX I VALUATION REPORTS



PROJECT JEWEL

OTOTOTOTOOTOOTOOTOO

O70707070101011011010007111000007070707

KENNEDY WILSON EUROPE REAL ESTATE PLC

Valuation Date: 30 June 2017 Report Date: 13 September 2017

CBRE



VALUATION REPORT

Table of Contents

Content	Page Number

Valuation Report	1
Scope of Work & Sources of Information	9
Valuation Assumptions	11

Appendix A - Schedule of 10 largest assets by value in UK and Europe Appendix B - Portfolio Details Appendix C – List of Properties



CBRE Limited
Henrietta House
Henrietta Place
London W1G 0NB
Switchboard +44 (0)20 7182 2000

Valuation Report

Report Date 13 September 2017

Addressee Kennedy Wilson Europe Real Estate Plc

47 Esplande St Helier

Jersey ("the Company")

and

N M Rothschild & Sons Limited

New Court, St Swithin's Lane

London EC4N 8AL

(in their capacity as Advisor)

J.P. Morgan Limited 25 Bank Street London E14 5JP

(in their capacity as Advisor)

(collectively referred to as the "Addressees")

The PropertiesThe properties held by the Company as described in the

Appendices.

Ownership Purpose Investment

InstructionTo value on the basis of Fair Value the relevant interests in the

Properties as at the Valuation Date in accordance with our instruction letter dated 16 August 2017 and the extension of scope

letter dated 6 September 2017.

Valuation Date 30 June 2017

Capacity of Valuer External.

Purpose of Valuation

The Valuation has been prepared for a Regulated Purpose as defined in the RICS Valuation – Professional Standards (January 2014) ("Red Book"). We understand that our valuation report and the Appendices to it (together the "Valuation Report") is required for inclusion in a scheme circular (the "circular") which is to be published by Kennedy Wilson Europe Real Estate plc pursuant to an agreed merger of the Company and Kennedy-Wilson Holdings, Inc. (the "Merger").

The effective date of valuation is 30 June 2017.

In our opinion, having regard to the information provided to us by the Company, the aggregate value of the Properties which are still owned by the Company as at today's date, would not, in accordance with Rule 29.4 of the UK City Code on Takeovers and Mergers, be materially different from the valuation of those same Properties at the Valuation Date (30 June 2017).

In accordance with the RICS Valuation – Professional Standards (January 2014) ("Red Book") we have made certain disclosures in connection with this valuation instruction and our relationship with the Company.

Basis of Valuation and Fair Value

In respect of the UK Properties:

£1,517,810,000 (One Billion Five Hundred Seventeen Million Eight Hundred and Ten Thousand pounds) exclusive of VAT.

In respect of the European Properties:

€1,299,489,000 (One Billion Two Hundred and Ninety Nine Million, Four Hundred and Eight Nine Thousand Euros) exclusive of VAT.

Adopting the same exchange rate as adopted in the Company's accounts of \in 1.14 to the pound, this equates for the European Properties to £1,139,781,802.

The valuations have been prepared in accordance with the Red Book.

For the avoidance of doubt, we have valued the Properties as real estate assets and the values reported above represent 100% of the Fair Values of the assets.

There are no negative values to report. We confirm that the "Fair Value" reported above, for the purpose of financial reporting under International Financial Reporting Standards is effectively the same as "Market Value".

Our opinion of Fair Value is based upon the Scope of Work and Valuation Assumptions attached, and has been primarily derived using comparable recent market transactions on arm's length terms.

We have valued the Properties individually and no account has been taken of any discount or premium that may be negotiated in the market if all or part of the portfolio was to be marketed simultaneously, either in lots or as a whole.

Where a property is owned by way of a joint tenancy in a trust for sale, or through an indirect investment structure, our valuation represents the relevant apportioned percentage of ownership of the value of the whole property, assuming full management control. Our valuation does not necessarily represent the 'Fair Value' (as defined by IFRS 13 or FRS 102) of the interests in the indirect investment structure through which the property is held.

We are required to show the split of values between freeholdequivalent, leasehold and short leasehold property, and to report the following categories of property separately. The following values are:

	Freehold	Leasehold	Short Leasehold*	Total
UK	£1,102,580,000	£413,305,000	£1,925,000	£1,517,810,000
Europe	**€1,277,959,000	€21,530,000		€1,299,489,000
TOTAL in £***	£2,223,477,839	£432,188,963	£1,925,000	£2,657,591,802

^{*}Short leasehold is less than 50 years unexpired

Split by property number:

	Freehold	Leasehold	Short Leasehold*	Total
UK	133	15	1	149
Europe	44**	1		45
TOTAL	177	16	1	194

^{*}Short leasehold is less than 50 years unexpired

^{**} Includes three assets which are held on both freehold and long leasehold tenure, where the leasehold elements are considered to be virtual freehold interests

^{***} adopting an exchange rate of £1:€1.14 as at 30 June 2017

^{**} Includes three assets which are held on both freehold and long leasehold tenure, where the leasehold elements are considered to be virtual freehold interests

Report Format

Appendix A of this Valuation Report contains details of the 10 largest assets by value in the UK and Europe. Appendix B provides the Portfolio Details and Fair Value of the Portfolio. Appendix C contains a full property list of all properties owned by the Company at the valuation date.

The Company has expressly instructed us not to disclose certain information which is considered commercially sensitive, namely the individual values of the properties.

Material Change

We hereby confirm that as at the date of this report:

- (i) We have not become aware (after having made enquiry of the Company and KW Investment Management Limited) of any material change since 30 June 2017 in any matter relating to any property covered by our Valuation Report which in our opinion would have a material effect on the aggregate value as at today's date of the properties still owned by the Company, and
- (ii) In relation to market conditions and movements in the property markets in which the properties covered by our Valuation Report are located, based on observed transactions involving comparable properties which have occurred and independent data published, in each case, since 30 June 2017, we do not consider that the movement in respect of the aggregate value of the retained subject properties constitutes material change.

Compliance with Valuation Standards

The valuations have been prepared in accordance with the RICS Valuation – Professional Standards Global – January 2014 and the RICS Valuation Professional Standards UK January 2014 (revised April 2015), ("the Red Book").

The valuations are compliant with the International Valuation Standards, and are in accordance with paragraphs 128 to 130 of the ESMA update of the Committee of European Securities Regulators' (CESR) recommendations for the consistent implementation of the European Commission Regulation (EC) no. 809/2004 implementing the Prospectus Directive and the London Stock Exchange requirements.

We confirm that we have sufficient current local and national knowledge of the particular property market involved, and have the skills and understanding to undertake the valuations competently. Where the knowledge and skill requirements of the Red Book have been met in aggregate by more than one valuer within CBRE Ltd, we confirm that a list of those valuers has been retained within the working papers, together with confirmation that each named valuer complies with the requirements of the Red Book.

Assumptions

The property details on which each valuation is based are as set out in this report. We have made various assumptions as to tenure, letting, taxation, town planning, and the condition and repair of buildings and sites — including ground and groundwater contamination — as set out below.

If any of the information or assumptions on which the valuation is based are subsequently found to be incorrect, the valuation figures may also be incorrect and should be reconsidered.

Variation from standard Assumptions

None

Market Conditions

The values stated in this report represent our objective opinion of Fair Value in accordance with the definition set out above as of the valuation date. Amongst other things, this assumes that the properties had been properly marketed and that exchange of contracts took place on this date.

On 14 June, a fire at the Grenfell Tower apartment block in London became out of control and destroyed the building, with significant loss of life. A number of Inquiries have been commissioned to investigate the event, and it is likely that recommendations will be made as to the construction and management of similar properties. In the light of this, we are aware that market participants with tower buildings and / or buildings with cladding construction may also be affected by the same or similar issues are reviewing details of construction, health and safety – and particularly fire prevention, mitigation and means of escape. At this point it is too early to assess the longer-term consequences. In the short-term, however, it is likely that potential investors and occupiers will be more cautious - and the liquidity and pricing of Pioneer Point, Ilford may be impacted. We would therefore draw your attention to the fact that in the case of Pioneer Point within the subject portfolio as at the date of our valuation, there is greater uncertainty concerning the valuation figure than would normally be the case. We would recommend that specialist advice is taken on these issues.

Valuer

The Properties have been valued by a valuer who is qualified for the purpose of the valuation in accordance with the Red Book.

Independence

The total fees, including the fee for this assignment, earned by CBRE Ltd (or other companies forming part of the same group of companies within the UK) from the Addressee (or other companies forming part of the same group of companies) is less than 5.0% of the total UK revenues.

We confirm that we do not have any material interest in the Company or the Properties.

Disclosure

In accordance with the Red Book we make the following disclosures:

- The Portfolio has been assembled through the acquisition of a number of property portfolios and individual assets.
 CBRE Ltd has been involved in some of these transactions or otherwise provided advice.
- In respect of the Tiger and Artemis Portfolios two separate departments within CBRE Ltd were retained by the two vendors to publicly market the Tiger and Artemis Portfolios and a number of bids were received for each portfolio.
- We subsequently prepared valuations for the acquisition of the Tiger and Artemis Portfolios following acting for KW Investment Management Limited (as the Investment Manager) in agreeing terms for their purchase in December 2013. We further valued the Tiger and Artemis Portfolios for your Initial Public Offering in February 2014. We have subsequently carried out the valuations required for the Company's Report and Accounts as at 30 June 2014.
- For Jupiter we advised the Company on the original loan purchase and have valued the portfolio for the Group's Interim Financial Statements as at 30 June 2014.
- In August 2014 we also provided valuation advice, for loan security purposes, to Royal Bank of Scotland plc in respect of the Jupiter Portfolio.
- In respect of the Fairmont St Andrews Hotel we advised the vendor on the sale to the Company. We further advised the Company on the purchase of the asset. Both instructions were undertaken by different valuers.
- We valued the entire, wider Company Portfolio as at 31 August 2014 for your secondary offering in October 2014.

- For Gatsby our Capital Markets team advised the Company on the original purchase from Aviva. We understand that the Company have obtained an independent valuation from another qualified valuer as a result we are able to provide you with this report in respect of these assets.
- CBRE Capital Markets team acted on behalf of the vendor in respect of the SEO portfolio that was acquired by the Company in July 2015.
- CBRE provided advice in relation to the portfolio of office properties in Italy that was purchased in December 2015.
- CBRE Capital Markets team (Manchester) acted on behalf
 of the vendor in respect of Towers Business Park that was
 acquired by the Company in May 2016. The Company was
 advised by CBRE Capital Markets team (London). The asset
 was valued by a qualified third party prior to the valuation
 reported herein being undertaken.
- CBRE Capital Markets team provides advice to the Company on an ad hoc basis.

We confirm that we have had no previous material involvement with any of the properties other than as set out above, and that copies of our conflict of interest checks have been retained within the working papers. We have disclosed the relevant facts to you and have received confirmation that it is in order for us to carry out your valuation instruction.

CBRE Ltd provide some agency and/or professional services to some of the occupiers of the properties and where this occurs, any conflict arising is managed through an information barrier.

We do not consider that any of the above provides a conflict of interest preventing us from preparing this Valuation Report and the Company has confirmed to us that it also considers this to be the case.

Responsibility

For the purposes of the UK City Code on Takeovers and Mergers as issued from time to time by the UK Panel on Takeovers and Mergers (the "Code") we are responsible for this Valuation Report and we will accept responsibility for the information contained in this Valuation Report and confirm that to the best of our knowledge (having taken all reasonable care to ensure that such is the case), the information contained in this Valuation Report is in accordance with the facts and contains no omissions likely to affect its import. This Valuation Report complies with, and is prepared in accordance with, and on the basis of the Code.

Reliance

This report is for the use only of the party to whom it is addressed for the specific purpose set out herein and no responsibility is accepted to any third party for the whole or any part of its contents save as set out in "Responsibility" above.

No reliance may be placed upon the contents of this Valuation Report by any party for any purpose other than in connection with the Purpose of Valuation.

Publication

We consent to the inclusion of the Valuation Report in the circular, and in any further announcement to be published or made available by the Company or by Kennedy-Wilson Holdings, Inc. in accordance with the Code, and to the publication and reproduction of this report as required by Rules 26 and 29 of the Code.

Save as set out above, neither the whole nor any part of our report nor any references thereto may be included in any published document, circular or statement nor published in any way without our prior written approval of the form and context in which it will appear. Before this Valuation Report, or any part thereof, is disclosed orally or otherwise to a third party, CBRE's written approval of the form and context of such publication or disclosure must first be obtained. Such publication or disclosure will not be permitted unless where relevant it incorporates the Assumptions referred to herein. For the avoidance of doubt, such approval is required whether or not CBRE is referred to by name and whether or not the contents of our Valuation Report are combined with others.

Such publication of, or reference to this report will not be permitted unless it contains a sufficient contemporaneous reference to any departure from the Royal Institution of Chartered Surveyors Valuation Standards or the incorporation of the special assumptions referred to herein.

Yours faithfully

Nick Knight MRICS Executive Director RICS Registered Valuer

NJhungt

For and on behalf of CBRE Ltd

T: 020 7182 2897 E: nick.knight@cbre.com Peter Stoughton-Harris MRICS Executive Director RICS Registered Valuer

For and on behalf of CBRE Ltd

T: 020 7182 2675

E: peter.stoughton-harris@cbre.com

Scope of Work & Sources of Information

Sources of Information We have carried out our work based upon information supplied to us

by the Company and its advisers as set out within this report, which we

have assumed to be correct and comprehensive.

The Properties Our report contains a brief summary of the property details on which

our valuation has been based.

Inspections As instructed, we have inspected at least 80% of the properties by value

for the purpose of this revaluation. As part of our wider valuation instruction, we have inspected all the Properties on an ad hoc basis

over the last twelve months.

Areas We have not measured the Properties but have relied upon the floor

areas provided to us by the Company and its advisers which we have assumed to be correct and comprehensive and which you have advised us have been calculated using the appropriate market practice. All

areas quoted in this Valuation Report are approximate.

Environmental MattersWe have not undertaken, nor are we aware of the content of, any environmental audit or other environmental investigation or soil survey

which may have been carried out on the Properties and which may draw attention to any contamination or the possibility of any such

contamination.

We have not carried out any investigations into the past or present uses of the Properties, nor of any neighbouring land, in order to establish whether there is any potential for contamination and have therefore assumed that none exists. Where we have been informed of the

presence or potential for contamination we have reflected this

information in our valuations.

Repair and ConditionWe have not carried out building surveys, tested services, made independent site investigations, inspected woodwork, exposed parts of

the structure which were covered, unexposed or inaccessible, nor arranged for any investigations to be carried out to determine whether or not any deleterious or hazardous materials or techniques have been used, or are present, in any part of the Properties. We are unable, therefore, to give any assurance that the Properties are free from

defect.

Town Planning We have not undertaken planning enquiries.

Titles, Tenures and Lettings

Details of title/tenure under which the Properties are held and of lettings to which they are subject are as supplied to us. We have not generally examined nor had access to all the deeds, leases or other documents relating thereto. Where information from deeds, leases or other documents is recorded in this report, it represents our understanding of the relevant documents. We should emphasise, however, that the interpretation of the documents of title (including

relevant deeds, leases and planning consents) is the responsibility of your legal adviser.

We have not conducted credit enquiries on the financial status of any tenants. We have, however, reflected our general understanding of purchasers' likely perceptions of the financial status of tenants.

Valuation Assumptions

Introduction

An Assumption is defined in the Red Book Glossary and Appendix 3 to be a "supposition taken to be true" (an "Assumption").

Assumptions are facts, conditions or situations affecting the subject of, or approach to, a valuation that it has been agreed need not be verified by the valuer as part of the valuation process. Assumptions are made when it is reasonable for the valuer to accept that something is true without the need for specific investigation.

The Company and its advisers has confirmed and we confirm that our Assumptions are correct as far as the Company and we, respectively, are aware. In the event that any of these Assumptions prove to be incorrect then our valuations should be reviewed. The principal Assumptions which we have made are stated within this Valuation Report.

For the avoidance of doubt, the Assumptions made do not affect compliance with the approach to Fair / Market Value under the Red Book.

Capital Values

Each valuation has been prepared on the basis of "Fair Value", which is defined as:

"The price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date."

"Fair Value", for the purpose of financial reporting under International Financial Reporting Standards is effectively the same as "Market Value", which is defined as:

"The estimated amount for which an asset or liability should exchange on the date of valuation between a willing buyer and a willing seller in an arm's-length transaction after proper marketing and where the parties had each acted knowledgeably, prudently and without compulsion."

The valuation represents the figure that would appear in a hypothetical contract of sale of the property interest at the valuation date. No adjustment has been made to this figure for any expenses of acquisition or realisation - nor for taxation which might arise or crystallise in the event of a disposal.

In the event that the Properties were to be sold at the valuations contained in this valuation report, any gains realised on such disposals over the book value for tax purposes would be subject to taxation in the applicable jurisdiction.

No account has been taken of any inter-company leases or arrangements, nor of any mortgages, debentures or other charge.

No account has been taken of the availability or otherwise of capital based Government or European Community grants.

Taxation, Costs and Realisation Costs

As stated above, no allowances have been made for any expenses of realisation nor for taxation which might arise or crystallise in the event of a disposal.

Our valuations reflect purchasers' statutory and other normal acquisition costs.

VAT

We have not been advised whether the properties are elected for VAT.

All rents and capital values stated in this report are exclusive of VAT.

Gross Contracted Rent

Passing rents quoted in this report are the rents which are currently payable under the terms of the leases.

Estimated Rental Value

The rental value reflects the terms of the leases where the Properties, or parts thereof, are let at the date of valuation. Where the Properties, or parts thereof, are vacant at the date of valuation, the rental value reflects the rent we consider would be obtainable on an open market letting as at the date of valuation.

Rental values indicated in our report are those which have been adopted by us as appropriate in assessing the capital value and are not necessarily appropriate for other purposes, nor do they necessarily accord with the definition of Market Rent.

Lease Expiries

Fixed-term leases frequently incorporate either tenants' options to extend or tenants' break clauses; other leases are rolling to indeterminate, subject to stated notice periods. For the purposes of our valuations, we have made assumptions as to appropriate presumed expiry dates.

Any weighted average unexpired terms indicated in our Valuation report are calculated to lease expiry allowing for break options.

The Property

Where appropriate we have regarded the shop fronts of retail and showroom accommodation as forming an integral part of the building.

Landlord's fixtures such as lifts, escalators, central heating and other normal service installations have been treated as an integral part of the building and are included within our valuations.

Process plant and machinery, tenants' fixtures and specialist trade fittings have been excluded from our valuations.

All measurements, areas and ages quoted in our report are approximate.

Environmental Matters

In the absence of any information to the contrary, we have assumed that:

- (a) the Properties are not contaminated and are not adversely affected by any existing or proposed environmental law;
- (b) any processes which are carried out on the Properties which are regulated by environmental legislation are properly licensed by the appropriate authorities;
- the properties comply with the relevant local and European (c) legislation in respect of Energy Performance. In respect of the UK properties, we assume they possess current Energy Performance Certificates (EPCs) as required under the Government's Energy Performance of Buildings Directive, and that they have an energy efficient standard of 'E', or better. We would draw your attention to the fact that the Energy Act 2011 is due to come into force in England and Wales no later than 1 April 2018 (although it may be earlier), and in Scotland, no earlier than April 2015 (although it may be later). From such date, it will be unlawful for landlords to rent out a residential or business premise unless they have reached a minimum energy efficient standard – most likely, 'E' – or carried out the maximum package of measures funded under the 'Green Deal' or the Energy Company Obligation (ECO); and
- (d) the properties are either not subject to flooding risk or, if they are, that sufficient flood defences are in place and that appropriate building insurance could be obtained at a cost that would not materially affect the capital value.

High voltage electrical supply equipment may exist within, or in close proximity of, the Properties. The National Radiological Protection Board (NRPB) has advised that there may be a risk, in specified circumstances, to the health of certain categories of people. Public perception may, therefore, affect marketability and future value of the property. Our valuation reflects our current understanding of the market and we have not made a discount to reflect the presence of this equipment.

Repair and Condition

In the absence of any information to the contrary, we have assumed that:

- there are no abnormal ground conditions, nor archaeological remains, present which might adversely affect the current or future occupation, development or value of the Properties;
- (b) the Properties are free from rot, infestation, structural or latent defect;

- (c) no currently known deleterious or hazardous materials or suspect techniques have been used in the construction of, or subsequent alterations or additions to, the Properties; and
- (d) the services, and any associated controls or software, are in working order and free from defect.

We have otherwise had regard to the age and apparent general condition of the Properties. Comments made in the property details do not purport to express an opinion about, or advise upon, the condition of uninspected parts and should not be taken as making an implied representation or statement about such parts.

Title, Tenure, Lettings, Planning, Taxation and Statutory & Local Authority requirements Unless stated otherwise within this report, and in the absence of any information to the contrary, we have assumed that:

- (a) the Properties possess a good and marketable title free from any onerous or hampering restrictions or conditions;
- (b) all buildings have been erected either prior to planning control, or in accordance with planning permissions, and have the benefit of permanent planning consents or existing use rights for their current use;
- (c) the Properties are not adversely affected by town planning or road proposals;
- (d) all buildings comply with all statutory and local authority requirements including building, fire and health and safety regulations;
- (e) only minor or inconsequential costs will be incurred if any modifications or alterations are necessary in order for occupiers of each Property to comply with the provisions of the relevant local and European disability discrimination legislation
- (f) all rent reviews are upward only and are to be assessed by reference to full current market rents;
- (g) there are no tenant's improvements that will materially affect our opinion of the rent that would be obtained on review or renewal;
- tenants will meet their obligations under their leases, and are responsible for insurance, payment of business rates, and all repairs, whether directly or by means of a service charge;
- (i) there are no user restrictions or other restrictive covenants in leases which would adversely affect value;

- (i) in respect of the UK where more than 50% of the floorspace of a property is in residential use, the Landlord and Tenant Act 1987 (the "Act") gives certain rights to defined residential tenants to acquire the freehold/head leasehold interest in the property. Where this is applicable, we have assumed that necessary notices have been given to the residential tenants under the provisions of the Act, and that such tenants have elected not to acquire the freehold/head leasehold interest. Disposal on the open market is therefore unrestricted;
- (k) where appropriate, permission to assign the interest being valued herein would not be withheld by the landlord where required;
- (I) vacant possession can be given of all accommodation which is unlet or is let on a service occupancy; and
- (m) Stamp Duty Land Tax (SDLT) in the UK or, the appropriate transfer tax in Europe, will apply at the rate currently applicable.

Appendix A: Top 10 Holdings in UK and Europe as valued by CBRE

PROPERTY	LOCATION AND DESCRIPTION, AGE	FLOOR AREA	TENURE	GROSS CONTRACTED RENT PER ANNUM
111 Buckingham Palace Road, London SW1W OSR	Landmark headquarters building of Grade A office space in a key strategic area of London's West End. The building, constructed in 1984, sits on a podium structure above London Victoria Station and provides office accommodation arranged over ground and five upper floors. The principal floor plates range in size from approximately 3,901.8 sq m (42,000 sq ft) – 6,317.2 sq m (68,000 sq ft), providing some of the largest in the West End of London. The building was comprehensively refurbished in 2003 with the reception areas extensively refurbished more recently in 2016.	20,820.5 sq m (224,117 sq ft)	LLH — 149 years from 3 rd December 1984 expiring 1 st July 2134 at a peppercom rent	811,216,838
	Tenants include Telegraph Media, Scripps Networks, Regus and Metalogics. Part of the third floor (1,847.6 sq m (19,888 sq ft)) is currently vacant representing a vacancy rate of 9.82% by ERV.			
	Weighted average unexpired lease term is 4.52 years to expiry. There are no break options within any of the leases. All leases are on standard institutional terms.			
Baggot Plaza, 27 / 33 Baggot Street Upper, Dublin 4	Located on the south western side of Baggot Street Upper in close proximity to the Grand Canal and just south of Mespil Road. Baggot Plaza consists of a second generation office building built over basement, ground and six stories in 1973 and refurbished in 2016 to provide modern Grade A specification accommodation throughout.	12,013.4 sq m (129,315 sq ft)	Freehold	€6,131,149
	The asset is fully let to the Governor and Company of Bank of Ireland with a weighted average unexpired lease term after allowing for break options of 19.01 years. The asset is let on standard institutional lease terms.			
40/42 Mespil Road, Dublin 4	Located on the southern side of Mespil Road in the Dublin 4 office district overlooking the Grand Canal. The property comprises an office	10,965.9 sq m (118,040 sq ft)	Freehold	€4,506,000

		6		
€4,868,221		£2,460,149		
Freehold and Long Leasehold — a number of long leasehold interests from 900 to 2,000 years unexpired at peppercorn rents not being demanded. Please note this is classified as a Freehold property for the purpose of the categorisation in the main report.		Freehold and Long Leasehold—250 years from May 2017 at a fixed rent of £50,000 per annum.		
12,890.3 sq m (138,755 sq ff)		17,082.7 sq m (183,883 sq ft)		
Situated on the corner of Harcourt Street and St Stephen's Green South occupying a prominent site in Dublin 2 and benefitting from its position adjacent to the LUAS Green Line. Comprises four second generation office buildings, constructed in 1982, and recently extensively refurbished internally. Arranged around a courtyard with car parking on site.	The asset is fully let to tenants including KPMG and the Government of Finland with a weighted average unexpired lease term after allowing for break options of 8.44 years. The asset is let on standard institutional lease terms.	Pioneer Point, a private rented sector (PRS) residential tower, is situated in the centre of Ilford, to the east of Central London. Ilford is east of Stratford and west of Romford within the London Borough of Redbridge.	The asset is located approximately 200 metres to the south of Ilford mainline rail station. Surrounding uses include a Sainsbury's supermarket, a Travelodge hotel, a cinema and a library with typical local high street shopping and amenities nearby.	The long leasehold interest comprises all of the recently completed residential development providing 294 residential units. It comprises two towers, north and south, that interconnect at the basement and lower commercial use floors only. The north tower rises to 32 storeys and the south tower to 24 storeys. The residential is a combination of one and two-bed apartments let mainly on Assured Shorthold Tenancies (AST's) with some corporate lettings
Russell Court, St Stephen's Green, Dublin 2		Pioneer Point, Ilford, London, IG1 2WS		

The asset is fully let to Governor and Company of Bank of Ireland with a weighted average unexpired term allowing for break options of 10.94 years. The asset is let on standard institutional lease terms.

building built in 2003 comprising seven stories over a double basement which has benefitted from a high specification fit-out.

	€5,512,306			55,391,624
	Freehold			Freehold
	14,498.3 sq m (156,063 sq ft)			26,846.7 sq m (288,985 sq ft)
for break options and are fully let on standard institutional terms.	Stillorgan is situated in the south suburbs of Dublin. Surrounding areas include Dundrum, Sandyford and Blackrock, with a large local residential catchment.	The asset comprises a first generation shopping centre facility constructed in 1966, arranged over two main open malls arranged in an "L" shape with anchor units at either end, and smaller units at ground and first floor levels at either side of both malls. The centre includes a Tesco which is undergoing refurbishment and extension, along with several other national and local occupiers. The centre also includes surplus land extending to 0.57 ha (1.40 acres) which is used as an additional car parking facility serving the centre.	There are 74 tenancies in total, with key tenants including Tesco, Dunnes Stores, Starbucks and McDonalds. The asset offers a weighted average unexpired lease term after allowing for break options of 4.57 years. The vacancy by ERV is 1.04%. All leases are on standard institutional terms.	Located in the South Manchester suburb of Didsbury, approximately 9.7 km (6.0 miles) from Manchester City Centre, Towers Business park is set across 6.07 ha (15.00 acres) and provides high quality office
	Stillorgan Shopping Centre, Stillorgan, Co Dublin 19			Towers Business Park, Manchester M20 2RY

The asset was constructed in 2007 but has been refurbished over the last year. Following completion of the works, the building was relaunched in June 2017 and as a result, is currently going through a

residential amenity space offering a cinema, games room, dining areas

and work spaces.

period of leasing. The property also includes a newly created communal

sq ft). The commercial uses include a children's' tuition provider, a buffet restaurant and a gym with 49 car parking spaces at basement

ground, first and second floors which extends to 3,697.4 sq m (39,800

The freehold interest comprises the commercial uses on the part

Explore Learning and Fore Fitness. The commercial uses offer a weighted average unexpired lease term of 12.93 years after allowing

level which are currently vacant. Tenants include Eastwood Investment,

		€5,634,030		€5,600,702	
		Freehold		Freehold	
		21,348.4 sq m (229,800 sq ft)		30,175 sq m (324,812.8 sq ft)	
The scheme attracts strong covenants and current occupiers include John Lewis, Syngenta Limited, British Airways, Honeywell Control Systems and Oracle Corporation.	The weighted average unexpired lease term after allowing for breaks options is 4.09 years. The vacancy by ERV of 3.87%. All leases are on standard institutional terms.	Block F&L Vantage is located within the Central Park development in the south Dublin suburb of Sandyford. The immediate area is characterised by offices and multi-family blocks, with retail / commercial uses at ground floor levels. The area is highly accessible with a LUAS light rail stop at Central Park, and with the M50 motorway accessible at Junction 16.	The property consists of two multi-family blocks built in 2008, comprising of 276 apartments, made up of 34 x 1 bed, 188 x 2 bed and 54 x 3 bed units. Current occupancy is at 96% and the units are let on standard 12 month residential tenancies. The commercial element is valued separately and excluded from this description.	The property comprises a shopping centre located in the north of Madrid, in an affluent neighbourhood with one of the highest income rates per capita. The scheme was first opened in 1995 and was extended in 2001.	The property is arranged over 2 buildings (South and North). The North building has 3 floors and South has 2 floors. Significant refurbishment works have commenced to improve the scheme including
		Block F&L Vantage, Central Park, Sandyford, Dublin 18		Moraleja Green, Spain	

self-contained accommodation of up to 5,016.6 sq m (54,000 sq fl), single floor plates of 1,718.7 sq m (18,500 sq fl) and suites from 92.9 sq m (1,000 sq fl) upwards. The park offers 981 car parking spaces providing a ratio 1:27 sq m (1:295 sq fl).

The buildings provide modern office space with the flexibility to provide

accommodation across ten individual buildings. Constructed between 1988 and 2003 with refurbishment since on an ad-hoc basis.

			52,619,421			
			Freehold			
			9,127.1 sq m (98,247 sq ft)			
modernisation of the internal finish, unit reconfiguration, relocation of the entrances in the north building, landscaping the garden adjacent to the south building and improvement of the leisure offer.	Tenants include Cinesa (Cinema), Sanchez Romero (Supermarket), H&M, Zara, Massimo Dutti and Michael Kors.	The weighted average unexpired lease term is 1.6 years to expiry after allowing for break options. All leases are on standard terms considered representative of the Spanish shopping centre market. There is currently 6,718 sq m (72,314.6 sq ft) vacant mainly on the top floor of the north building, representing a vacancy rate of 29% by ERV.	Friars Bridge Court constructed in 1991 is located on the south side of the River Thames in the London Borough of Southwark. The building is situated on Blackfriars Road within 50 metres of Southwark underground station and close to Blackfriars Bridge.	London Bridge and Waterloo mainline stations are within a short walk. The property is multi-let as offices with accommodation arranged over basement, ground and seven upper floors.	The property presents a potential redevelopment opportunity and leases are deliberately short term in order to offer flexibility, but there is good demand for the space as it stands. As a result it has a short term weighted unexpired average lease term after allowing for break options is 1.67 years.	Occupiers are dominated by rail operating companies associated with the nearby Waterloo station such as London and South Eastern Railway and Stagecoach South West Trains. All leases are on standard institutional terms.
			Friars Bridge Court, London SE1 8NZ			

Appendix B: Summary breakdown of Portfolio as valued by CBRE

For the avoidance of doubt this will include all properties including those described in Appendix A.

Y

SUB-PORTFOLIO	DESCRIPTION	FLOOR AREA	GROSS CONTRACTED RENT (PER ANNUM)	FAIR VALUE	ERV OF BUILT ACCOMMODATION (PER ANNUM)
UK Industrial	The portfolio comprises 25 assets predominantly located in England with one asset in Scotland. 22 of the assets are held freehold with three held leasehold. One of the leasehold assets has a short unexpired term of 47.5 years with a c.20% pay-away and the other two have over one hundred years unexpired with nominal ground rents payable. The majority by value of the assets are single let providing modern warehouse accommodation. There are four multi-let industrial estates where the accommodation is older but very functional.	256,781.2 sq m (2,764,060 sq ft)	11,320,161	2182,010,000	£13,499,075
	The assets are let in total to 46 tenancies with three units currently vacant representing 8.28% by ERV. Major tenants include Wincanton Group, TNT Express, Electrolux and Woodland Global. All of the leases are let on standard institutional terms.				
London & South East offices	The portfolio comprises 14 office buildings of which two are located within Central London, both described in Appendix A. The remaining 12 assets are spread across the South East in established office locations including Maidenhead, Warford and Farnborough.	106,016 sq m (1,141,184 sq ft)	528, 164,411	000'018'9153	169'856'923
	All of the buildings are held freehold apart from Buckingham Palace Road held long leasehold as described in Appendix A.				

		251,620,992				522,781,995
		5274,605,000				£346,420,000
		521,976,882				\$ 698'600'873
		133,870.1 sq m (1,441,013 sq ft)				172,037 sq m (1,851,852 sq ft)
ff) to on of	units nants Sable ional		ouse, pired	H) to on of	units nants nillips ional	
The buildings range from 2,348.5 sq m (25,280 sq ft) to 19,443.6 sq m (209,296 sq ft) and are a combination of single and multi-let assets.	The assets are let in total to 47 tenancies with ten units currently vacant representing 11.07% by ERV. Major tenants include British Telecomms, ASOS, Avaya, Canon and Cable & Wireless. All of the leases are let on standard institutional terms.	The portfolio is made up of a business park situated in Manchester as described in Appendix A and 13 individual offices located across the UK. Six of the assets are located in Scotland with three in Edinburgh and three in Aberdeen. The remainder are located in the south west of England, the Midlands and the north including Plymouth, Bristol, Edgbaston and Sheffield.	All of the buildings are held freehold apart from Apex House, Edgbaston held long leasehold with over 100 years unexpired at a rent geared to 7% of passing rents.	The buildings range from 975.7 sq m (10,503 sq ft) to 22,523.6 sq m (242,450 sq ft) and are a combination of single and multi-let assets.	The assets are let in total to 75 tenancies with 13 units currently vacant representing 14.56% by ERV. Major tenants include EDF Energy, HSBC, Big Lottery Fund, Conoco Phillips and Shell UK. All of the leases are let on standard institutional terms.	The portfolio comprises 88 assets located across the UK. The majority by number are located in London and the south-east with the next largest proportion being in the north. It is diverse in nature comprising high street shops, food convenience supermarkets, a department store, car showroom, retail warehouses and a tertiary shopping centre.
The 19,4 singl	The a curren includ. & Wir terms.	Rest of UK Offices Man office Scotl	All o Edgk at a	The 22,5 singl	The a currer includ includ and SI terms.	UK Retail The majc with with in n in n supe

All of the assets are held freehold apart from six assets. Three are virtual freehold (999 years at a nominal ground rent), two have over 85 years unexpired at a nominal ground rent and one has 75 years unexpired with a rent geared to 36.61% of ERV.

The assets are let in total to 268 tenancies with tenants

including Debenhams, Waitrose, Asda, B&Q, Peacocks and Matalan. There are 27 units currently vacant representing 3.35% by ERV. All of the leases are let on standard

ACCOMMODATION ERV OF BUILT (PER ANNUM) €35,808,173 €10,897,630 €14,171,434 €204,427,000 €215,430,000 €635,277,000 **FAIR VALUE** PASSING RENT (PER ANNUM) 38,764.6 sq m (417,272 sq ft) £12,250,678 35,565.4 sq m (382,835 sq ft) £11,541,389 €27,825,431 80,979.4 sq m (871,684 sq ft) **FLOOR AREA** Comprises a number of different properties including three Comprises 17 prime office buildings (including those listed in The retail portfolio in Ireland comprises a total of six retailing The portfolio is let on 155 tenancies, mostly accounted for by refurbished second generation buildings let to prime centre, as well as key suburban and provincial locations in individual multi-family assets, a licenced-premises, and a Assets are made up of a mixture of third generation and covenants and considered institutionally acceptable in terms A above) located in prime Dublin retailing pitches in the city Co Dublin, Co Cork and Co Louth. In addition, are two parcels of land at the Marshes Shopping Centre and at the Appendix A above) located in the Dublin 2 and Dublin 4 The portfolio is let to a total of 69 tenants and current vacancy and shopping centre assets (including those listed in Appendix central business district areas and south Dublin suburbs. is approximately 10.55% by ERV. All assets are held Freehold or on long leases with minimal ground rents payable and smaller national retailers making up the tenant mix. Tenants The assets are all held freehold and are largely fully let on the two shopping centres, with large multi-national and include Marks and Spencer, H&M, Boots, Bestseller, hotel and golf resort in North Dublin. considered virtual freehold interests. of location and specification. standard institutional terms. Leisureplex in Stillorgan. McDonalds and KFC. DESCRIPTION Ireland Leisure, Hotel **IRELAND & SPAIN SUB-PORTFOLIO** and Residential **Dublin Offices Ireland Retail**

Schedule of Properties

Ref	Property	Asset type
20440072	111,Buckingham Palace Road	Office
	Towers Business Park, Wilmslow Rd., Manchester, M20 2RY	Office
20440051	Friars Bridge Court, Blackfriars Road,London,SE1 8NZ	Office
20440273	Building 1,2,3 Leavesden Park, Watford, WD25 7GR	Office
20440274	Discovery Place, Summit Avenue, Farnborough, GU14 0NZ	Office
20440060	Rubislaw House,Aberdeen,AB15 6FZ	Office
20440052	Argyle House,Edinburgh,EH3 9DR	Office
20440062	Seafield House, Aberdeen, AB15 6FZ	Office
20440275	Form 1 & 2, Bartley Wood Business Park, Hook, RG27 9XA	Office
20440278	Eton House, Maidenhead Office Park, Maidenhead, SL6 3QH	Office
20440061	Marathon House,Aberdeen,AB15 6FZ	Office
20440277	Axis One, 10 Hurricane Way, Langley, SL3 8AG	Office
20440271	6 Roundwood Stockley Park, Heathrow, UB11 1JA	Office
20440017	Exchange Tower,19,Canning Street, Edinburgh	Office
20440054	Pennine Centre, Sheffield, S1 3RF	Office
20440016	Theta, yon Way,Camberley,GU16 7ER	Office
20440276	Thames House, Waterside Park, Bracknell, RG12 1XL	Office
20440018	Oracle House,6,Lochside Avenue, Edinburgh Park, Edinburgh	Office
20440279	Eldon Court, 75-77 London Road, Reading, RG1 5BS	Office
20440019	Douglas House,40,London Road,Reigate,RH2 9QP	Office
20440001	Apex House, Calthorpe Road,Edgbaston,B15 1TR	Office
20440015	25,Marsh Street,Bristol,BS1 4AQ	Office
20440229	334,Outland Road,Plymouth,PL3 5TU	Office
20440005	The Horizon Centre,32,Upper High Street,Epsom,KT17 4QJ	Office
20440241	4 Chalfont Park, Gerrards Cross, SL9 0BG	Office
20440166	Civic Court, Calverley Street,Leeds,LS1 3ED*	Office
20440004	Meriden Business Park, Copse Drive, Allesley, Coventry, CV 9RG	Office
20440204	30,32 & 34,Crofts Bank Rd,Urmston,M41 0UH	Office
27	TOTAL	OFFICE

Ref	Property	Asset type
20440200	Midpoint 18, Middlewich, CW10 0TN	Industrial
20440328	Duke Park Industrial Estate, Montrose Road, Chelmsford, CM2 6TE	Industrial
20440012	Units 1-3,Mcconnell Drive, Wolverton, Milton Keynes	Industrial
20440029	Colussus 1 & 2,Eurocentral,Glasgow,ML1 4XJ	Industrial
20440069	Universal Point, Walsall	Industrial
20440033	6, Wells Close, Gatton Park Business Centre, Merston, RH1 3DR	Industrial
20440213	Southam Road, Radford Semele, Leamington Spa,CV31 1TA	Industrial
20440059	Melton Enterprise Park, Hull	Industrial
20440026	Axis 40,Athlon Road,Alperton,HA0 1YJ	Industrial
20440031	Preservation House, Airport Way,Luton,LU2 9LF	Industrial
20440028	Beddington Lane, Croydon	Industrial
20440081	Units B1-C2 ,Ely Road, Angel Road Estate,Edmonton,London,N18 3BH	Industrial
20440011	Units 1&2,Lydiard Fields,Swindon,SN5 8UB	Industrial



20440245	Mill Lane Trading Est, Croydon, CRO 4AA	Industrial
20440329	Addison Way, Orion Business PK, Ipswich IP6 ORL	Industrial
20440032	Site A, Finlan Road, Stakehill Industrial Estate,Manchester,M24 2SJ	Industrial
20440250	Thorn Lighting, Merrington Lane Ind Est, Spennymoor	Industrial
20440199	Unit 1,Merlin Park,Trafford Park,Manchester,M32 OSZ	Industrial
20440030	Phase 2,Foxbridge Way,Normanton,Leeds,WF6 1TN	Industrial
20440119	Unit 6/6A & 12 ,Eurolink Industrial Estate,Sittingbourne,ME10 3RN	Industrial
20440034	8, Wells Close, Gatton Park Business Centre, Merston, RH1 3DR	Industrial
20440214	4,Crossgate Road, Park Farm Industrial Estate,Redditch,B98 7SN	Industrial
20440077	Blocks J & L, Stadium Way Trading Estate, Rayleigh	Industrial
20440156	TNT Park Farm Industrial ,Wellingborough,NN8 6TZ	Industrial
20440138	Plot 12&13 Robjohns R,Chelmsford,CM1 3BY	Industrial
27	TOTAL	INDUSTRIAL

D-f	Downste	Accellone
Ref 20440128	Property 30-31 High St,& 31The Square, Winchester, SO23 9BL	Asset type
20440128	2-14,2-14,Pinstone Street,Sheffield,S1 2HN	Retail
20440024	44-50,Bridge Street,Chester,CH1 1NQ	Retail
		Retail
20440088	9 - 19 Tooting High St,& 6 - 12 Totterdown Street ,London,SW17 OSN	Retail
20440162	103,New Street,Huddersfield,HD1 2UA	Retail
20440142	11 Gentleman's Walk,& 2-12 London Street,Norwich,NR2 1LH	Retail
20440141	2-10,Back of the Inns,Norwich,NR2 1PT	Retail
20440021	22-32,Commercial Road,Bournemouth,BH2 5LP	Retail
20440022	156-160,High Street,Bromley,BR1 1HE	Retail
20440074	50/51 Great Russell St,London,WC1B 3BA	Retail
20440108	134/142 High Street ,134/142 High Street ,Gillingham,ME7 1AS	Retail
20440122	107-125 Above Bar Street, MV with updated leases, Southampton*	Retail
20440184	115,Princes Street,Edinburgh,EH2 3AA	Retail
20440127	35-47 Howardsgate, Welwyn Garden City,AL8 6AP	Retail
20440154	1-7,Market Street ,Leicester,LE1 6DN	Retail
20440123	1-3,St Peter's Street, St Albans,AL1 3DH	Retail
20440091	24 A-C, Mitcham Road ,London,SW17	Retail
20440174	7/11,Blake Street,York,YO1 8QJ	Retail
20440185	95/96,Princes Street, Edinburgh	Retail
20440233	66/68,High Street,Weston-super-Mare,BS23 1HS	Retail
20440133	Suffolk Hotel, Buttermarket, Bury St Edmunds, IP33 1DW	Retail
20440124	61-65,St Peter's Street, St Albans,AL1 3EA	Retail
20440222	185-187,High Street,Cheltenham,GL50 1DF	Retail
20440160	Talbot Hotel,32 - 34,Bank Street,Bradford,BD1 1PR	Retail
20440096	112/116 High Street,Sutton,SM1 1LT	Retail
20440129	97 High Street, Winchester, SO23 9AP	Retail
20440120	1/1D Park Street & 92,1/1D Park Street& 92 Shirley High Street,	Retail
00446555	Southampton	
20440155	298/299 ,High Street,Lincoln,LN2 1AF	Retail
20440224	1 Ormond Place ,& 1-3 Ormond Terrace, Cheltenham, GL50 1JD	Retail
20440223	189 High Street & ,& 3 Pitsville Street, Cheltenham, GL50 1DF	Retail
20440112	7 Carfax & 6-7 Middle,Horsham,RH12 1DW	Retail



20440191	78/82,English Street,Carlisle,CA3 8HP		Retail
20440202	10/12 Eastbank Street,10/12,Eastbank Street,Southport,PR8 1[ΟJ	Retail
20440234	11-12,St Mary Street,Weymouth,DT4 8PB		Retail
20440131	105-109 Montague Street, Worthing, BN11 3BP		Retail
20440251	,231-240,High Street, Sunderland		Retail
20440189	205A Grange Road,4&6 Salisbury Street,Birkenhead,L41 2PF		Retail
20440079	72 North End,Croydon,CRO 1UJ		Retail
20440109	2-8 Sussex Road, Haywards Heath,RH16 4EA		Retail
20440076	145/151,High Street ,Beckenham,BR3 1AG		Retail
20440192	15 Eastgate Street, 15, Eastgate Street, Chester, CH1 1LG		Retail
20440236	58,Yorkshire St,Rochdale,OL16 1JP		Retail
20440196	Main Street,53/61,Main Street, Keswick		Retail
20440217	33,High Street,Worcester,WR1 2QL		Retail
20440232	1,St Nicholas Street,Truro,TR1 2RW		Retail
20440168	14-15,Westborough,Scarborough,YO11 1UH		Retail
20440186	135,High Street,Perth,PH1 5UF		Retail
20440187	53-55,George Street,Altrincham,WA14 1EH		Retail
20440193	Frodsham Street,25-29,Frodsham Street,Chester,CH1 3JJ		Retail
20440150	206/207,Station Street, Burton on Trent,PA1 2BE		Retail
20440218	40,The Shambles,Worcester,WR1 2RE		Retail
20440171	103-105,High Street,Scunthorpe,DN15 6LY		Retail
20440212	29/30,High Street, Dudley,DY1 1PJ		Retail
20440208	59, 61&63,Station Road ,Port Talbot		Retail
20440169	81,Newborough,Scarborough,YO11 1ET		Retail
20440198	46 Mill Street,46,Mill Street,Macclesfield,SK11 6LT		Retail
20440235	80,St Mary Street, Weymouth,DT4 8PJ		Retail
20440215	49/49A,Broad Street, Ross On Wye,HR9 7DY		Retail
20440099	6,High Street,Banbury,OX16 5DZ		Retail
68	TC	OTAL	RETAIL HIGH STREET

Ref	Property	Asset type		
20440058	Trident Retail & Leisure Park,Runcorn,WA7 2FQ	Retail		
20440183	Porthlethen Retail Park, Muirend Road, Aberdeen, AB12 4XP	Retail		
20440025	Staines Road Retail Park, Hounslow	Retail		
20440240	Southampton Road ,Southampton Road ,Fareham,PO14 4PH	Retail		
20440165	263,Selby Road,Halton,Leeds,LS15 7JR	Retail		
20440105	Magnet Ltd, Telford Place, Crawley	Retail		
20440170	Magnet Unit, Seamer Road, Scarborough, YO12 4DH	Retail		
7	TOTAL	RETAIL WAREHOUSE		

Ref	Property	Asset type
20440244	Tamar Industrial Estate,Saltash,PL12 6LD	Retail
20440111	Asda Superstore ,The Marlowes ,Hemel Hempstead,HP2 4AA	Retail
20440226	Plymco Supermarket, Mayors Avenue, Dartmouth, TQ6 9NF	Retail
20440114	Sainsbury's,19-27, High Street ,Littlehampton	Retail
20440118	Budgens, 49-53, High Street, Shepperton	Retail
20440116	Budgens Bepton Road, White City, Midhurst	Retail
20440110	Station Road, Heathfield	Retail
20440100	Budgens ,Market Square, High Street,Battle,TN33 0EY	Retail
20440239	Craig House, Uxbridge Road, West Ealing, West Ealing, W13 8RA	Retail



20440228	Somerfield Stores, Market Street ,Okehampton,EX20 1HN	Retail
20440102	Budgens Supermarket ,62-64,High Street ,Billingshurst,RH14 9NY	Retail
20440238	Budgens 324 Hook Road,324,Hook Road, Chessington	Retail
20440246	Budgens 57-67 High St, Cranfield	Retail
20440152	Co-Op, Gladstone Street ,Rushtone Road, Desborough	Retail
20440225	25,High St ,Coleford,GL16 8HD	Retail
16	TOTAL	FOOD CONVENIENCE
Ref	Property	Asset type
20440103	220-246, High Street, Chatham	Retail
20440195	Angel Shopping Centre, Units 1-4, Victoria Street, Crewe, CW1 2HF	Retail
2	TOTAL	
Ref	Property	Asset type
20440219	599-601,Christchurch Road,Boscombe,BH1 4AN	Retail
20440161	41-45,St Sepulchre Gate, Doncaster	Retail
20440173	Go Bowling Shipley, Market Street, Shipley, BD18 3QD	Retail
3	TOTAL	. RETAIL OTHER
Ref	Property	Asset type
20440066	Dixon, Priory Park,Hull,HU4 7DY	Retail
1	TOTAL	. RETAIL SHOWROOM
Ref	Property	Asset type
20440253	Fairmont St Andrews Bay Hotel	Leisure
20440073	Travelodge - Grays Inn, Greys Inn Road,London,WC1X 8BM	Leisure
20440078A	Norfolk House -Retail, Wellsey Road, Croydon, CRO 1LH	Leisure
20440078B	Norfolk House -Hotel/Office,Croydon,CR0 1LH	Leisure
20440164	Riva Hotel, Vicar Lane ,Leeds,LS1 6PJ	Leisure
5	TOTAL	LEISURE HOTEL
Ref	Property	Asset type
20440153	Nuffield Health, Blackbird Road, Byford Way,Leicester,LE4 0DZ	Leisure
20440070	Aspects Leisure Park, Bedford, MK41 9LN	Leisure
20440067	DP Leisure, Roebuck Way, Milton Keynes	Leisure
3	TOTAL	
Ref	Property	Asset type
20440000	Pioneer Point, Ilford	Residential*



* Mixed FH/LH tenure

PROPERTY ADDRESSES - SPAIN & IRELAND

Ref	IRELAND Property		Asset Type		
A15	The Marshes Shopping Centre, Dundalk (Lands)		Site		
C08	Leisureplex Site, Stillorgan, Co Dublin		Site		
A07	Stillorgan Shopping Centre, Stiillorgan, Co Dublin	Shopping Ctr			
A14	The Marshes Shopping Centre,Dundalk		Shopping Ctr		
A04	41 St Stephen's Green, Dublin 2		Retail		
B06	Block K, Commercial, Central Park, Dublin 18		Retail		
A02	40/42 Mespil Road, Dublin 4		Office		
A03	Charlemont House, Charlemont Place, Dublin 2		Office		
A09	Russell Court, St Stephen's Green, Dublin 2		Office		
A11	Baggot Plaza, 27/33 Baggot Street Upper, Dublin 4		Office		
A12	The Warehouse, 35 Barrow Street, Dublin,4		Office		
A13	South Bank House, Barrow Street, Dublin 4		Office		
A16	Gardner House, Wilton Place, Dublin 2		Office		
C01	14 - 15 Sir John Rogerson's Quay, Dublin 2		Office		
C02	16 Sir John Rogerson's Quay, Dublin 2	Office			
C03	Beaver House, Clonskeagh, Dublin 14		Office		
C04	Block C, Dundrum Business Park, Dublin 14		Office		
C05	Block 3 & 4, Blackrock Business Park, Co Dublin		Office		
C06	Block 5, Blackrock Business Park, Co Dublin		Office		
C07	5 Schoohouse Lane, Dublin 2		Office		
C09	The Chase, Sandyford, Dublin 18		Office		
B01	81 Apartments, Liffey Trust, Sherrif Street, Dublin 1		Multi Family		
B03	Block F & L Residential, Central Park, Dublin 18		Multi Family		
B05	Block K, Residential, Central Park, Dublin 18		Multi Family		
B04	Block F & L Commercial, Central Park, Dublin 18		Mixed Use		
B02	Portmarnock Hotel & Golf Links, Portmarnock, Co Dublin		Hotel		
A06	16 Westmoreland Street, Dublin 2		High Street		
A10	Marks & Spencer, St Patrick Street, Cork		High Street		
28		TOTAL	IRELAND		



Ref	SPAIN Property	Asset type		
	Moraleja Green	Retail	Dept Stores/SC	
20340268	Puerta del Sol 9	Retail	Retail	
20340270	Santisima Trinidad 5, 28010 Madrid	Retail	Parking	
20340282	Alcalá de Hénares,Alcalá de Hénares	Retail	Food/Covenience	
20340283	Local Almería,18,Paseo de Almería,Almería	Retail	Food/Covenience	
20340281	Aluche,93,Seseña,Aluche	Retail	Food/Covenience	
20340291	Local Rocafort,244,C Rocafort,Barcelona	Retail	Food/Covenience	
20340288	Local Berga,65,C Sant Fruitos,Berga	Retail	Food/Covenience	
20340287	Bilbao - Zabalburu,Nicolas Alkorta - Pza Zabalburu,Bilbao	Food/Covenience		
20340284	Local Cádiz,1,Plaza de la Libertad,Cádiz	Retail	Food/Covenience	
20340295	Local Esparraguera,s/n,Tres Rieres,Esparraguera	Retail	Food/Covenience	
20340293	Local Granollers,220,C Girona,Granollers	Retail	Food/Covenience	
20340290	Local Igualada,s/n,Av Paisos Catalans,Igualada	Retail	Food/Covenience	
20340295	Local Mora D'Ebre,21,C Antoni Asens,Mora D'Ebre	Retail	Food/Covenience	
20340280	Local Olérdola,41-49,Av Barcelona,Olédrola	Retail	Food/Covenience	
20340285	Salamanca,82,Calle Toro,Salamanca	Retail	Food/Covenience	
20340289	Local Viladecans,198 - 200,Av. de la Generalitat,Viladecans	Retail	Food/Covenience	
17		TOTAL SPA	AIN	





Olimpia Investment Fund

VALUATION REPORT

VALUATION DATE: 30TH JUNE 2017 REPORT DATE: 13TH SEPTEMBER 2017

PREPARED BY

Colliers Real Estate Services Italia Srl con socio unico

PREPARED FOR

Kennedy Wilson Europe Real Estate Plc

The Directors

47 Esplande

St Helier - Jersey

J.P. Morgan Limited (JP Morgan)

25 Bank Street

London E14 5JP

N M Rothschild & Sons Limited ("Rothschild")

New Court,

St Swithin's Lane

London

EC4N 8AL

Accelerating success.

272



COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO

Partita IVA 06 180000967

Via Durini 2 20122 Milano

Tel: +39 (0) 02 00640990 Fax: +39 (0) 02 00640999 www.colliers.com/it



TABLE OF CONTENTS

1	INTRODUCTION	4
1.1	OBJECT AND PURPOSE OF THE VALUATION	4
1.2	DEFINITION OF MARKET VALUE – BASIS OF VALUATION	4
1.3	COMPLIANCE WITH VALUATION STANDARDS – VALUATION	
	CRITERIA	5
1.4	GENERAL ASSUMPTION AND MARKET CONDITIONS	6
1.5	LIMITATIONS, RESTRICTIONS	6
1.6	CONFLICT OF INTEREST - DISCLOSURE	8
1.7	RESPONSIBILITY	8
1.8	RELIANCE	8
1.9	PUBLICATION	8
2	VALUATION	10
2.1	THEORY AND PRACTICE IN THE REAL ESTATE SECTOR	10
3	CONCLUSIONS	14



1 INTRODUCTION

1.1 OBJECT AND PURPOSE OF THE VALUATION

• The Valuation has been prepared for a Regulated Purpose as defined in the RICS Valuation – Professional Standards (January 2014) ("Red Book"). We understand that our valuation report and the Appendice to it (together the "Valuation Report") is required for inclusion in a scheme circular subject to the requirements of the UK City Code on Takeover and Mergers (the "circular") which is to be published by Kennedy Wilson Europe plc pursuant to a recommended merger of the Company and Kennedy-Wilson Holdings, Inc. (the "Merger").

The effective date of valuation is 30 June 2017.

In our opinion, having regard to the information provided to us by the Savills Investment Management, Fondo Olimpia, the aggregate value of the Properties which are still owned by the Company as at today's date, would not, in accordance with Rule 29.4 of the UK City Code on Takeovers and Mergers, be materially different from the valuation of those same Properties at 30 June 2017.

In accordance with the RICS Valuation – Professional Standards (January 2014) ("Red Book") we have made certain disclosures in connection with this valuation instruction and our relationship with the Company.

1.2 DEFINITION OF MARKET VALUE – BASIS OF VALUATION

The "Current Value" or "Market Value" of a property refers to the fair amount (price) at which a piece of real estate can reasonably be transferred, at a certain date, from a seller to a purchaser, with neither of the two parties being forced to sell or purchase and both of which are fully aware of all of the significant factors regarding the subject property, of its possible uses, of its characteristics and of the existing market conditions.

The sale price of the asset will therefore be determined based on normal sales conditions, which are present when:

- the seller truly intends to sell the asset and is not subject to circumstances of an economic/financial nature that influence its willingness to sell;



- there is a reasonable amount of time, considering the type of asset and the market situation, to market the asset, carry out sales negotiations and define the relative contractual conditions;
- the terms of the sales transaction reflect the true conditions of the real estate market of the zone in which the asset is located:
- offers to purchase the asset reflect real market conditions, and any offers that are not in line with the market (due to the subjective opinion of the purchaser, acting according to principles that do not significantly impact the market economically) are not taken into consideration.

For the avoidance of doubt, we have valued the Properties as real estate and the values reported above represent 100% of the market values of the assets.

Our opinion of Fair Value is based upon the Scope of Work and Valuation Assumptions attached, and has been primarily derived using comparable recent market transactions on arm's length terms.

We have valued the Properties individually and no account has been taken of any discount or premium that may be negotiated in the market if all or part of the portfolio was to be marketed simultaneously, either in lots or as a whole.

1.3 COMPLIANCE WITH VALUATION STANDARDS – VALUATION CRITERIA

The valuations have been prepared in accordance with the RICS Valuation – Professional Standards Global – January 2014 and the RICS Valuation Professional Standards January 2014 (and following revised version), ("the Red Book"). We confirm that we have sufficient current local and national knowledge of the particular property market involved, and have the skills and understanding to undertake the valuations competently. Where the knowledge and skill requirements of the Red Book have been met in aggregate by more than one valuer within Colliers Real Estate Services srl, we confirm that a list of those valuers has been retained within the working papers, together with confirmation that each named valuer complies with the requirements of the Red Book.

The Properties have been valued by a valuer who is qualified for the purpose of the valuation in accordance with the Red Book.

During the Valuation process of the Assets, Colliers RES will be compliant with the dispositions of the Provisions By Bank of Italy (Provvedimento di Banca d'Italia) of the 19th January 2015, Title V, Chapter IV, Section II, the Methodology for Independent Experts of SGR and, broadly, with the sector Best Practice.

COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO



Specifically, three reference methodologies will be adopted, Construction Cost Approach, Sales Comparison Approach and Income Approach (both the direct approach and the Discounted Cash Flow Analysis). In the Fund half year valuations, Colliers RES will be compliant with the valuations criteria of the RICS Red Book (Comparison approach, Income approach, Replacement cost approach).

1.4 GENERAL ASSUMPTION AND MARKET CONDITIONS

The property details on which each valuation is based are as set out in this report. We have made various assumptions as to tenure, letting, taxation, town planning, and the condition and repair of buildings and sites, as set below. If any of the information or assumptions on which the valuation is based are subsequently found to be incorrect, the valuation figures may also be incorrect and should be reconsidered. We confirm that no variation from standard assumption have been made for this valuation.

The values stated in this report represent our objective opinion of Fair Value in accordance with the definition set out above as of the valuation date. Amongst other things, this assumes that the properties had been properly marketed and that exchange of contracts took place on this date (as indicated in §1.2).

1.5 LIMITATIONS, RESTRICTIONS

We would like to draw attention to a number of fundamental assumptions to which we have referred in preparing this report:

- Sources of information: this report has been prepared on the basis of information provided to us by the Client. More specifically, the following were provided:
 - Floor plans of the asset
 - Lease agreement, Rent Roll and tenant situation updated
 - Cadastral details
 - Capex plan/tenant improvements (provided by the Client)
 - Costs: non-recoverable (Property tax, Property mgmt. tax, Insurance etc.)

For some asset, there might have been provided supplementary information/documents upon Colliers Request.

• Areas: these have been provided by the client and were verified random measurements during the course of the inspection;

COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO



- Title deeds: the valuation is based on information provided by the client. Unless
 otherwise specified, it is assumed that the client possesses or will possess
 regular titles of ownership and that there are no third-party claims, obligations,
 restrictions and/or encumbrances and/or pending litigation that may impact use
 of the asset;
- Leasehold situation: the property's leasehold situation was provided by the client
- Regulatory assessments: without prejudice to what has been determined from
 the information provided by the client, no environmental impact surveys have
 been carried out, and it is assumed that the property is not in abnormal condition
 and that neither archaeological remains nor dangerous or deleterious materials
 that could adversely affect occupation of the site, marketing of the property or
 current/future values of the asset are present;
- Expenses: no allowances have been made for any sale/purchase expenses, such as legal, tax or agency costs, except for those regarding letting of the areas. The property has been considered in its existing state, not encumbered by any mortgages and free from restrictions or easements of any type, except that which is specifically stated in the report;
- Taxes, duties and other transfer costs: no deductions have been made from the values expressed in this report for any taxes or duties, or any legal, agency or other costs.

Should this information prove to be incorrect or incomplete, the property valuation could alter and, therefore, we reserve the right, if necessary, to revise our conclusions.

We hereby confirm that as at the date of this report (13th September 2017):

- (i) We have not become aware (after having made enquiry of the Company) of any material change since 30th June 2017 in any matter relating to any property covered by our Valuation Report which in our opinion would have a material effect on the aggregate value as at today's date of the properties still owned by the Company, and
- (ii) In relation to market conditions and movements in the property markets in which the properties covered by our Valuation Report are located, based on observed transactions involving comparable properties which have occurred and independent data published, in each case,

COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO



since 30th June 2017, we do not consider that the movement in respect of the retained subject properties constitutes material change.

1.6 CONFLICT OF INTEREST - DISCLOSURE

In drawing up this report, we hereby confirm that we have accomplished the valuation independently and that any conflicts of interest have arisen during the valuation process, according to the best practice and Italian regulation.

We confirm that we have been appointed as Real Estate Indipendent Expert by Savills Investment Management SGR S.p.A. in order to run the half year valuation and, where required, the acquisition/sale opinion of value, as required by the Italian Legislative Decree no. 58 of 24 February 1998 and Italian Ministerial Decree no. 30 of 5 March 2015. Our activity will terminate with the half year valuation dated 30th June 2018, after 3 years, as required by the Italian regulation.

1.7 RESPONSIBILITY

For the purposes of the Takeover Code, we are responsible for this report and we will accept responsibility for the information contained in this report and confirm that to the best of our knowledge (having taken all reasonable care to ensure that such is the case), the information contained in this report is in accordance with the facts and contains no omissions likely to affect its import. This report complies with, and is prepared in accordance with, and on the basis of, the Takeover Code.

1.8 RELIANCE

This report is for the use only of the party to whom it is addressed for the specific purpose set out herein and no responsibility is accepted to any third party for the whole or any part of its contents save as set out in "Responsibility" above.

No reliance may be placed upon the contents of this Valuation Report by any party for any purpose other than in connection with the Purpose of Valuation.

1.9 PUBLICATION

We consent to the inclusion of the Valuation Report in the circular, and in any further announcement to be published or made available by the Company or by Kennedy-Wilson Holdings, Inc. in accordance with the Code, and to the publication and reproduction of this report as required by Rules 26 and 29 of the Code.

COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO



Save as set out above, neither the whole nor any part of our report nor any references thereto may be included in any published document, circular or statement nor published in any way without our prior written approval of the form and context in which it will appear. Before this Valuation Report, or any part thereof, is disclosed orally or otherwise to a third party, Colliers Real Estate Services Italia con Socio Unico's written approval of the form and context of such publication or disclosure must first be obtained. Such publication or disclosure will not be permitted unless where relevant it incorporates the Assumptions referred to herein. For the avoidance of doubt, such approval is required whether or not Colliers Real Estate Services Italia con Socio Unico is referred to by name and whether or not the contents of our Valuation Report are combined with others.

Such publication of, or reference to this report will not be permitted unless it contains a sufficient contemporaneous reference to any departure from the Royal Institution of Chartered Surveyors Valuation Standards or the incorporation of the special assumptions referred to herein.



2 VALUATION

2.1 THEORY AND PRACTICE IN THE REAL ESTATE SECTOR

In determining the valuation process, we adhered to the generally accepted principles and criteria, in accordance with the International Valuation Standards (IVS) and the professional best practices in the appraisal field.

As far as property valuation criteria, practices and valuation theory in the real estate sector are concerned, three different approaches are used:

- The construction cost approach, which expresses the value of an asset based on the costs necessary for its construction or replacement, appropriately depreciated according to age, general conditions, functional, economic or environmental obsolescence and all other age factors deemed relevant. The value of an asset can be calculated as the depreciated cost of reconstruction when there is no market for similar assets, and the value may be expressed as current cost of reconstruction. Determination of the construction cost has three fundamental components: appraisal of the land, estimate of the building construction cost and estimate of the appreciation/depreciation factors;
- The sales comparison approach, which expresses the value of an asset based on the average sales prices observed in the local markets, taking into account specific characteristics of the individual assets. The prices used are those obtained from transactions that are comparable in terms of type, location and use of the asset. Application of the sales comparison approach is carried out following the achievement of two objectives: identification of a sample of homogeneous assets and determination of the transaction prices. Furthermore, the appreciation/depreciation factors must also be estimated. Use of this method leads to the determination of average values per square metre inferred from the market transactions;
- The income approach, which determines the value of an asset according to the income or cash flows it will be able to generate. The income approach is based on the assumption that a rational buyer is not willing to pay a price greater than the present value of benefits that the asset will be able to produce. Application of this approach consists of three phases: determination of the expected economic benefits, definition of the type of



algebraic relationship linking the value to income and selection of a discount/capitalisation rate.

An Assumption is defined in the Red Book Glossary and Appendix 3 to be a "supposition taken to be true" (an "Assumption").

Assumptions are facts, conditions or situations affecting the subject of, or approach to, a valuation that it has been agreed need not be verified by the valuer as part of the valuation process. Assumptions are made when it is reasonable for the valuer to accept that something is true without the need for specific investigation.

The Company has confirmed and we confirm that our Assumptions are correct as far as the Company and we, respectively, are aware. In the event that any of these Assumptions prove to be incorrect then our valuations should be reviewed. The principal Assumptions which we have made are stated within this Valuation Report.

For the avoidance of doubt, the Assumptions made do not affect compliance with the approach to Market Value under the Red Book.

Our assumptions can be summarazied as follow:

- As stated in §1.5 no allowances have been made for any expenses of realisation nor for taxation which might arise or crystallise in the event of a disposal.
- All rents and capital values stated in this report are exclusive of VAT
- Passing rents quoted in this report are the rents which are currently payable under the terms of the leases, according to the information and the tenancy situation provided
- The rental value reflects the terms of the leases where the Properties, or parts thereof, are let at the date of valuation. Where the Properties, or parts thereof, are vacant at the date of valuation, the rental value reflects the rent we consider would be obtainable on an open market letting as at the date of valuation. Rental values indicated in our report are those which have been adopted by us as appropriate in assessing the capital value and are not necessarily appropriate for other purposes, nor do they necessarily accord with the definition of Market Rent.
- Fixed-term leases frequently incorporate either tenants' options to extend
 or tenants' break clauses; other leases are rolling to indeterminate, subject
 to stated notice periods. For the purposes of our valuations, we have made
 assumptions as to appropriate presumed expiry dates
- In the absence of any information to the contrary, we have assumed that:

COLLIERS REAL ESTATE SERVICES ITALIA SRL CON SOCIO UNICO



- (a) the Properties are not contaminated and are not adversely affected by any existing or proposed environmental law;
- (b) any processes which are carried out on the Properties which are regulated by environmental legislation are properly licensed by the appropriate authorities
- In the absence of any information to the contrary, we have assumed that:
 - (a) there are no abnormal ground conditions, nor archaeological remains, present which might adversely affect the current or future occupation, development or value of the Properties;
 - (b) the Properties are free from rot, infestation, structural or latent defect;
 - (c) no currently known deleterious or hazardous materials or suspect techniques have been used in the construction of, or subsequent alterations or additions to, the Properties; and
 - (d) the services, and any associated controls or software, are in working order and free from defect
- Unless stated otherwise within this report, and in the absence of any information to the contrary, we have assumed that:
 - the Properties possess a good and marketable title free from any onerous or hampering restrictions or conditions;
 - all buildings have been erected either prior to planning control, or in accordance with planning permissions, and have the benefit of permanent planning consents or existing use rights for their current use;
 - the Properties are not adversely affected by town planning or road proposals;
 - all buildings comply with all statutory and local authority requirements including building, fire and health and safety regulations
 - all rent reviews are upward only and are to be assessed by reference to full current market rents;
 - there are no tenant's improvements that will materially affect our opinion of the rent that would be obtained on review or renewal;



- tenants will meet their obligations under their leases, and are responsible for insurance, payment of business rates, and all repairs, whether directly or by means of a service charge;
- there are no user restrictions or other restrictive covenants in leases which would adversely affect value;



3 CONCLUSIONS

Since 30th June 2017 no acquisition or disposal have occurred.

The Market Value at the date 30th June 2017 was:

€ 199.150.000

(One hundred and ninety-nine million one hundred and fifty thousand /00 €)

Our opinion of Value at the date 13th September 2017 confirms the Value at 30th June 2017 equal to the total sum of:

€ 199.150.000

(One hundred and ninety-nine million one hundred and fifty thousand /00 €)

A Summary of the Property characteristics, as at 13th September 2017, are included within the Appendix A.

Colliers Real Estate Services Italia Srl con socio unico

Arch. Giulia Longo



APPENDIX A

Olimpia Investment Fund

Valuation Date: 30th June 2017



Change in Market Value since 30 June 2017	no changes	no changes	no changes	no changes	no changes	no changes	no changes	no changes	no changes	
Gross Passing Rent 2017	€ 3.872.413,17	€ 1.639.743,04	€ 717.028,67	€ 1.018.088,50	€ 1.309.958,76	€ 684.854,82	€ 2.941.656,11	€ 1.011.194,29	€ 725.072,24	€ 13.920.009,60
Current Occupier	Dogane e GDF	sdul	sdul	Entrate / MEF/ Monopoli	sdu	Territorio e MEF	Miur	sdul	Dogane e MEF	
Tenancy	sqm 26.326 Agenzia del Demanio	sqm 11.284 Agerzia del Demanio	sqm 8.245 Agenzia del Demanio	sqm 8.083 Agenzia del Demanio	sqm 11.157 Agenzia del Demanio	sqm 5.313 Agenzia del Demanio	sqm 16.158 Agerzia del Demanio	sqm 6.941 Agenzia del Demanio	sqm 5.700 Agenzia del Demanio	2
GLA(sqm)	sqm 26.32	sqm 11.28	sqm 8.24	sqm 8.08	sqm 11.15	sqm 5.31	sqm 16.15	sqm 6.94	sqm 5.70	sqm 99.207
Tenure	Freehold	reehold	Freehold	reehold	reehold	reehold	reehold	Freehold	Freehold	
Description	The subject property is in Via Valtellina, along the borders of the city centre, in a predominantly residential area. The property is partially subject to protection pursuant to fallant legislative Decree no. 402/2004. The complex comprises 12 buildings, partly for tentiary office use and partly for storage use, in addition to a number of abandoned buildings that were once used as warehouse space by Agenzia delie Dogane (Customs Agency).	The building has seven floors above ground and two basement levels. The sections of the building are. Freehold arranged to form a Y shape, and the layout of areas is identical on each of the floors, except for the ground floor, where the property is subdivided into two parallel blocks.	The property was constructed in the early 1960s and is situated in a central zone containing mixed. For office, residential and relate buildings, with excellent connections with et ofty is cased, system. It comprises an area of 2,160 sqm according to the land register (including covered and uncovered), on which a building with eight floors above ground, a roof and two basement levels is situated. The structural frame is in reinforced concrete, with a flat roof.	The subject property is a self-contained building designated for office use, it is in a central area of the Freehold city and has 6 floors above ground on three of the four sides, while the side along Via Giulio Razasco has only 3 floors above ground.	The subject complex comprises three self-contained buildings; the first two (A and B) are designated. Freehold for use as offices, while the last one contains the surveillance area. The complex is studend in a central part of the clip mer Santa Maria Novella station. It has 4 floors above ground plus one basement level for building A, two floors above ground for Building B and, lastly, 4 floors above ground and one basement level for Building C.	The subject property is a self-contained building for office use. The building has 4 floors above ground, Freehold with 1 basement level. It has a pentagonal floor shape, with structural frame in brickwork.	The subject building is a self-contained building for office use, it is situated within a central area of the Freehold city, siring a forces above ground, with 2 basener it leash. At the structural level, the building as a rectangular floor print with structural farme in mixed stone and brick masonry and floors in reinforced concrete. All floors have recently refurbished and are equipped as offices, including the top mansand floor.	The building was constructed in the second half of the 1980s, on an exclusive lot which according to the exacting mass an area of "Lists sym and witch contains building rising seven floors above ground, with a codificient comis level (roof with accessible terrace), in addition to three basement levels, currently used as archives but originally designed as appurtenant underground parking (floors -1 and -2).	The subject property is a self-contained building for office use, it is situated within a central area of the city, rising 2 flors above ground, with 1 semi-basement leel from the main fascele along Via Plutho. The rest flacade on Via Piria has 3 floors above ground, due to the difference in leels between the two parallel roads. The shape of the building is rectangular, and it contains an internal countyand that is at basement level with respect to the main façade.	
Main Use	Office	Office	Office	Office	Office	Office	Office	Office	Office	
Property Address	Milano, Via Vaitellina	Brescia, Via Croce	Trieste, Via Anastasio	La Spezia, Piazza Europa	Firenze, Viale Belflore	Arezzo, Via Petrarca	Roma, Via Carcani	Roma, Via Spegazzini	Reggio Calabra, Via del Plutino Office	TOTAL

LEASEHOLD SITUATION

The existing contract was stpulated between FIP (Fondo immobil pubblic), Closed-End Investment Fund on behalf of which Investire Immobiliare SGR SpA acts, and Agenzia del Demanio (State Property Office), signed in Rome on 28 December 2004 and registered on 18 January 2005. All of the amendments subsequently signed and any additional agreements are part of the lease agreement. Savills I.M., management company for Olimpia Investment Fund, which owns the subject property of this valuation, has replaced investire immobiliare SGR SpA as fund manager. The contract has a second expiry date at 28th December 2022.

Agenzia del Demanio sub-let the singol properties to the current occupiers.

NOTE: Rental fees are updated annually at 75% of the ISTAT index

Since 30th June 2017 no acquisition or disposal have occurred.

The Market Value at the date 30th June 2017 was:

€ 199.150.000

(One hundred and ninety-nine million one hundred and fifty thousand /00 €)

Our opinion of Value at the date 13th September 2017 confirms the Value at 30th June 2017 equal to the total sum of:

€ 199.150.000

(One hundred and ninety-nine million one hundred and fifty thousand /00 €)



CONTACT DETAILS

Colliers Real Estate Services Srl con socio unico Valuation & Advisory Via Durini 2, 20122 Milano





